CP/68

An M6800 Operating System by **Jack E. Hemenway** and

Robert D. Grappel

Edited by Edward R. Teja

Hemenway Associates, Inc.

101 Tremont St. Suite 208 Boston, MA 02108 (617) 426-1931 The authors of the programs provided with this book have carefully reviewed them to ensure their performance in accordance with the specifications described in the book. Neither the authors nor Hemenway Associates, Inc., however, makes any warranties whatever concerning the programs, nor assumes responsibility of any kind for errors in the programs or for the consequences of any such errors.

The programs provided with this book are protected by the Copyright Law of the United States, Title 15 of the United States Code. Lawful users of this book may use the programs themselves, but may not make copies or translations of them except to the extent necessary to so use the programs. Any other use of these programs, including copying or translating them for purposes of resale, license or lease to others, is prohibited, and, in addition to actual damages, can result in civil damages of up to \$50,000 and criminal penalties of up to one year imprisonment and/or a \$10,000 fine.

Copyright (c) 1979 by Hemenway Associates, Inc. All Rights Reserved. SoftwareSourceBook is a trademark of Hemenway Associates, Inc.

Library of Congress Catalog Card #79-89895

Printed in the United States of America

Table of Contents

Part	1General Information. Command Structure. System Commands System Device Errors. System Error Messages.	2 2 2 3 1 1 1 1
Part	Advanced User's Guide. Filename formatting. Directory handling routines. Disk-file sequential I/O. Initialization and warmstart Deleting a file. Program chaining. User entries. Format of binary files. Examples of CP/68 usage.	17 17 38 40 41 42 43 43 45
Part	Description of Routines. Resident BIOS. Logical device handlers. Command-line interpreter. Command Processing routines. Transient Commands. ASSIGN. BOOT. DELETE. INITIALIZE LINK. PIP. Character-oriented device handlers. SECURITY. SET. STATUS.	52 52 53 53 55 58 63 71 72 73 74 75 78 78
Part	FORMAT Utility	80 80

Table of Contents

Part	5	82
	Random files	82
	What are random-access files?	82
	Physical and logical records	82
	Entry points	83
	File-control block	83
	Data structures	85
	File routines	85
	Error codes	87
	Notes and warnings	89
	Example	90
	STRUBAL+ support	95
	Using files in STRUBAL+	97
	Deleting files	98
	0	,
Part	6	99
	Modifications	99
	Disk-handling software	101
	Modifications for monitor ROMs	101
Part	7	102
	Software listings	103

Introduction

CP/68 is a floppy-disc-based operating system that supports standard peripherals such as a line printer, CRT console, paper-tape reader and punch, and auxiliary consoles. The preliminary specification was described in EDN's Software Systems Design Course (Chapter 7), November 20, 1978. The current version of CP/68 is based on that specification and an improvement on it.

The operating system's modularity makes it easy to manage conceptually, and a pleasure to use. It is the most powerful system available for the 6800 family of microprocessors.

This book presents the entire operating system in seven distinct parts. Part I introduces you to the operation of the program; Part II adds the Advanced User's Guide; Part III covers the system's operation in detail; Part 4 explores the operation of the formatting utility; Part 5 introduces the random-access file support; Part 6 provides the information you will need to adapt the software to nonstandard hardware configurations; Part 7 gives complete source listings.

CP/68 GENERAL INFORMATION

COMMAND STRUCTURE

CP/68 commands consist of a command name and optional parameters. Some commands are memory resident and will execute immediately; others are transient (stored on disk) and must be loaded from disk before they are executed. System command names may be abbreviated to three characters; user-defined commands are invoked by entering their full names. These command files must be binary type with transfer addresses (type 01).

Where CP/68 requires numeric values, either decimal or hexadecimal notation may be used. Hex values must be preceded by a dollar sign (\$). The period (.) is used for an operator prompt.

FILENAMES

File names in CP/68 consist of two parts: a name and an extension. The name is a string of alphanumeric characters up to 8 characters in length. The extension consists of up to 3 alphanumeric characters. The first character of both the filename and extension MUST be an alpha character. The name is separated from its extension by a period. The following are valid filenames:

INPUT.TXT MYFILE.B H1.HEX JACKS.FIL

To specify a file, give the disk drive number, filename and extension. The drive number is given as a decimal digit followed by a colon. The following are examples of unique files:

O:INPUT.TXT 1:INPUT.TXT 1:INPUT.HEX 0:INPUT2.TXT

If the drive number is zero it may be omitted. The following identify the same file:

O:BOBS.BIN BOBS.BIN

Note that only alphanumeric characters may appear in filenames or extensions. The following are invalid filenames:

1:JACKSFILE.HEX (name more than 8 characters) 2:TEMP.FILE (extension more than 3 characters) O TEST.TMP (colon missing after drive number) STBL+.BIN (+ is a non-alphanumeric character) EDITOR (file extension missing)

WILDCARD FEATURES -----

CP/68 permits manipulation of classes of files. The mechanism for forming such classes is called wildcarding. Two wildcard characters perform unique identification tasks. The asterisk (*) matches an entire string of characters of arbitrary length. Since a complete filename consists of two strings, a name and an extension, the wildcard filename *.* is a short form of expressing all possible filenames. The wildcard filename *.HEX expresses all filenames with the extension HEX.

The second wildcard character is the question mark (?). This character substitutes for any single character, including a blank. Hence, the filename TEST?.HEX is equivalent to TEST.HEX or TESTP.HEX or TEST2.HEX. It is not equivalent to TESTING.HEX. The filename *.* is equivalent to ?????????.???.

CP/68 SYSTEM COMMANDS

ASSIGN (transient)

This command assigns logical device names to physical devices. CP/68 supports the following logical devices.

CON the console terminal I/O device

PTR paper-tape reader

PTP paper-tape punch

DSK disk drive

LPT line printer

MTA magnetic tape
TTY teletype (could be second console with

paper-tape facilities)

NUL null device (bit bucket)

ASSIGN manipulates the relationship of physical devices to logical device names. For instance, if it is desired to use the teletype as the console device, you need only enter

ASSIGN CON=TTY

CAUTION: Take care with assignments. It is possible to get into trouble. The console device should ALWAYS be a device capable of input.

Now, all console messages and input will use the teletype physical device. Suppose, however, that one wanted to test a routine which would simply output characters. The following command could be used to direct the paper-tape punch output to the Null device:

ASSIGN PTP=NUL

Devices can be repeatedly ASSIGNed. The STATUS command will give the present state of the device assignments.

ASSIGN CON=LPT

would lock up the system requiring a restart. One should not re-assign the DSK device, as that is where the system gets its transient commands.

The command can make as many assignments as desired at one time. After each command line, it will re-prompt for another command line. Enter the escape character followed by a CR (see SET under ES for a definition of this input), to leave ASSIGN and return to the command level.

BOOT (transient)

When a fresh copy of the system file is brought into the transient area from the disk, the system is said to have been booted. Any file which was LINKed on the disk in drive zero can be BOOTed. BOOT works as a specialized LOAD.

DELETE (transient)

This command is used to remove a file from disk. Wildcard characters in filenames can be used to remove categories of files. DELETE can process multiple command lines.

DELETE [drive:] filename.ext

where the drive number will default to drive zero. The filename and extension fields may contain wildcard characters. When the named file(s) are found the command issues a prompt that gives the user a chance to save the file.

DELETE MYFILE.TMP

DELETE-0:MYFILE.TMP ? .YES

The YES response assures the operating system that this is the file

to be deleted. The YES can be abbreviated to Y; any input other than Y is interpreted as a NO.

DELETE *.TMP

DELETE-0:MYFILE.TMP ? .NO

would be the correct response if MYFILE.TMP was not the one that was to be deleted. This strategy saves you from being wiped out by typos. If there are several matches—due to the use of a wildcard character in the filename—each will be prompted in turn, and any of the matches may be removed. Suppose, for example, that there are three TXT files on drive 1, named TEST1, TEST2, and TEST3. Then the following command:

DELETE 1:TEST?.TXT

DELETE-1:TEST1.TXT ? .NO

DELETE-1:TEST2.TXT ? .YES

DELETE-1:TEST3.TXT ? .YES

DELETE .

This removes files TEXT2.TXT and TEXT3.TXT while leaving TEXT1.TXT. Enter the escape character followed by a carriage return to leave DELETE and return to the command level.

DIRECTORY (resident)

The DIRECTORY command provides a list of the files on a specified disk. The listing prints on the console device unless directed to the line printer.

DIRECTORY (goes to console)

DIRECTORY /L (goes to lineprinter)

The directory listing has the following format:

NAME TYPE ACCESS FIRST-TRACK/SECTOR LAST-TRACK/SECTOR SECTOR COUNT

The type code, access code, track/sector, etc. are output in hexadecimal. The type codes defined in CP/68 are:

- 00 binary file
- 01 binary file with transfer address
- 02 random access
- 03 text file (hex file)

The access codes defined in CP/68 are:

- 00 can be renamed or deleted
- 01 can be renamed but not deleted
- 02 cannot be renamed or deleted

Filenames are listed as 8-character strings with 3-character extensions. Following the directory list, the total number of disk sectors used by the listed files is listed in decimal.

The DIRECTORY command allows several levels of file qualification for listing categories of files.

DIRECTORY [/L] [drive:] [filename.ext]

The drive will default to disk zero. The filename and extension may use wildcards. For example:

DIRECTORY 1:*.HEX

will list on the console all files from disk 1 which have the extension HEX . Another example:

DIRECTORY /L TEST?.*

will list on the line-printer all files from disk zero which have names beginning with TEST followed by a character (or blank).

EXIT (resident)

This command causes control to shift from CP/68 to the underlying hardware system monitor. To get back to CP/68 either jump to the cold-start location or re-boot the system.

INITIALIZE (transient)

The INITIALIZE command formats a specified disk. All disks must be initialized before they can be used with CP/68.

INITIALIZE drive number

The drive number must be present even if the drive is zero. The command echoes the drive number, allowing the user to save the disk's contents.

INITIALIZE 1

INIT. DISK IN DRIVE 1 ? .YES

will begin the initialization process on the disk in drive 1. The initialization process wipes out the entire contents of the disk.

INITIALIZE 2

INIT. DISK IN DRIVE 2 ? .NO

returns to the command level leaving the disk unchanged. Upon completion of the initialization process (which may take several minutes) CP/68 returns to the command level.

JUMP (resident)

This command allows the user to leave CP/68 and go to any arbitrary absolute address. If the program at that address does a subroutine return (RTS instruction), CP/68 will continue at the command level.

JUMP \$E113

will go to the address E113 (hexadecimal).

JUMP 256

will go to the address 256 (decimal).

LINK (transient)

This command sets the linkages for BOOT; it prompts the user for a file name. This file must be a binary file with transfer address (type 01). Once performed, the file named in LINK will be the file BOOTed when that disk is in drive zero.

LINK

ENTER SYSTEM FILE ? .CP68.SYS

links the file CP68.SYS as the file to be bootstrapped. The drive number defaults to zero; no wildcard characters are permitted in the filename or extension.

LOAD (resident)

This command puts programs into the transient area. They are not executed; control returns to CP/68 command level. LOAD requires that files be binary type (00 or 01 type).

LOAD [drive:] filename.ext

where the drive will default to zero. No wildcard characters are permitted in the filename or extension.

LOAD 1:PROG1.BIN

loads PROG1.BIN into the transient area.

PIP (transient)

The peripheral-interchange program (PIP) provides the operations for media conversion (eg, load, print, copy and combine disk files), referring to each peripheral device by name.

PIP destination=source[,source][,source]....

where destination and source are:

[drive:] filename.ext device

Device is one one of the logical devices (see ASSIGN).

In the case of a disk-to-disk copy, for example,

PIP newdrive:=source drive:

copies the contents of the source drive exactly (sector for sector) onto the disk in newdrive. PIP prompts the user, providing a chance to save the contents of the newdrive.

PIP 0:=1:

COPY FROM DRIVE 1 TO DRIVE 0 ? .YES

will make the disk in drive 0 an exact copy of the disk in drive 1. A selective disk-to-disk copy follows a different form.

PIP destination:=source:filename.ext

where the filename and extension may contain wildcards. This will cause copies of all files on the source disk which match the filename and extension to be reproduced on the destination disk. All files on the destination disk are untouched; only those new files which were copied from the "source" disk will be written on the destination disk. If files with the same filename.ext already exist on the destination disk, an error indication is printed and the file is not copied.

PIP 0:=1:*.REL

copies all files on drive 1 that have the extension REL onto drive 0. The following command will copy all files from disk 0 to disk 1.

PIP 1:=0:*.*

This is a different form of copy from PIP 1:=0: . Using the wildcard filename.ext will copy the files into as sequential as possible a format on the new disk. Only the data sectors are copied, not the entire disk. Also, this form prompts the user as each file is copied, allowing very selective copying.

PIP can also transfer data between devices. For example, the following command can be used to view the contents of a file containing ASCII text:

PIP CON=filename.ext

Similarly, the contents of the file could be printed using PIP.

PIP LPT=filename.ext

PIP can be used to create text files.

PIP filename.ext=CON

builds a new file with the data typed at the console device. The END-FILE character (control-D, hex 04) is used to end the file. PIP can be used to transfer data from device to device as follows:

PIP LPT=CON (print data from console)
PIP PTP=PTR (duplicate a paper-tape)
PIP TTY=CON (type from one device to another)

and many other combinations. PIP allows the user to combine several sources of input into one. This can be used to append several files into one, as in:

PIP bigfile=file 1, file 2, file 3,

Input from several devices can also be combined.

PIP newfile=oldfile.CON

lets you type new data after the oldfile is copied to the newfile.

PIP can also perform data translations. Internal storage of programs is in a binary format which cannot be listed, printed or copied to ASCII-character devices. PIP can convert the internal binary format to a hexadecimal format (MIKBUG) which can be used for listing, etc. Such data can also be converted into binary format. The following command converts a MIKBUG paper-tape file into an internal hexadecimal file:

PIP MYFILE.HEX/H=PTP

The following command can be used to convert the hex file into an internal binary file:

PIP MYFILE.BIN/B=MYFILE.HEX

PIP can be used to punch MIKBUG format tapes as follows:

PIP PTP/H=filename.BIN

so that, for example, one could punch a copy of the system with the

PIP PTP/H=CP68.SYS

where CP68.SYS is a binary file. PIP can also be used to list or view a program file as follows:

PIP LPT/H=INIT.CMD (transient INITIALIZE)

PIP CON/H=CP68.SYS

When copying a file from one disk to another, PIP maintains the filetype, and sets the access code to 00. It may be desirable at times to force the type of a file to TEXT (03). This can be done as follows:

PIP 1:TEMP.TXT/T=CON

The switch /T makes 1:TEMP.TXT a text file. Otherwise, a file produced by PIP will default to binary type. (00)

PIP can process multiple command lines. It will prompt the user after each command is completed. Enter an escape character to return to command level in CP/68.

RENAME (resident)

To change the name of a file without modifying its contents, use

RENAME [drive:] oldname.oldext, newname.newext

where the drive will default to zero. The file access code must be 00 or 01 to allow renaming. The newname must not exist already with that extension. The following command, for example, will rename the file BOBS.OLD to BOBS.NEW:

RENAME BOBS.OLD.BOBS.NEW

No wildcard characters are permitted in either the new or old names or extensions.

SAVE (resident)

This command saves an area of memory as a binary file.

SAVE [drive:] filename.ext, startad, endad [, transfer ad]

where the drive defaults to zero. The filetype of the save-file will be 00 if no transfer address is present, and 01 if a transfer address is supplied. For example, the following command will save the first 8k of memory as a system file to be entered at the address 07BC hexadecimal.

SAVE 1:CP68.SYS, \$0000, \$2000, \$07BC

Addresses can also be entered in decimal notation. To save the first 256 bytes of memory:

SAVE BASEPAGE.SAV, 0, 256

No wildcard characters are permitted in the filename or the extension.

SECURITY (transient)

The files's security is determined by its access code. (see DIRECTORY). The code permits protection of certain files from deletion or renaming. SECURITY loads into the transient area. Its syntax is

SECURITY [drive:] filename.ext,access-code

where the drive will default to zero. For example, to remove any protection from the file CP68.SYS on drive zero:

SECURITY CP68.SYS,0

or to protect the file INIT.CMD from deletion:

SECURITY INIT. CMD. 2

To allow INIT.CMD to be renamed but not deleted:

SECURITY INIT. CMD. 1

No wildcard characters are permitted in either the filename or extension.

SET (transient)

This command allows the user to control the characteristics of the console and lineprinter devices.

SET parameter=value

where the following parameters are defined for the console:

BS -- Backspace character. This character may be set to any ASCII character on the console device. Control-H (08H) is the default.

DL -- delete character; causes the entire line just entered to be deleted. Control-U (15H) is the default.

DP -- depth count. The console will be paged with DP lines per page. This can be used to avoid scrolling; defaults to zero which disables paging.

- WD -- Width. Sets the number of characters that will appear on a line. The default (zero) disables the line limit.
- NL -- null count. Sets the number of nonprinting null characters sent with each carriage return. Allows delays for mechanical terminals. The default is zero.
- TB -- tab character. Defines the character to be decoded as a tab. Default is Control-I (09H).
- DX -- duplex switch. Selects either full or half duplex operation for the console. Default is F (full); H is half duplex.
- EJ -- eject count. The number of lines skipped at the end of each page. If the pause switch is set the system waits for an escape character before continuing. Defaults to zero.
- ES -- escape character. Defines the escape character; default is the ASCII escape character (1BH).
- PS -- pause switch. Determines whether or not the system will wait at the end of a page. Valid values are Y (yes) and N (no); default is N.

Two parameters are exclusively for the line printer.

LD -- depth. Sets the number of lines per page; defaults to 60 decimal.

LW -- width. Sets the number of characters per line; defaults to 80 decimal.

With the exception of DX and PS, all parameters take a number which may be either decimal or hexadecimal. The following are some valid commands:

SET LD=50 (50 lines per page on LPT)
SET DX=H (half-duplex CON)
SET BS=\$08 (backspace CNTL-H)
SET EJ=0 (no formfeeds)
SET PS=Y (pause on)

SET allows multiple command lines. It will prompt after each command line. Enter an escape character followed by a carriage return to return to command level.

```
STATUS (transient)
```

A systems status is its list of the present state of device assignments—printed on the current console device. It returns directly to command level after listing the devices. See ASSIGN for a complete list of device names.

SUBMIT (resident)

This command allows the use of a file containing CP/68 command lines as a source of console commands. The text lines in the file are used as though they were typed at the console. The memory resident SUBMIT can invoke any other command under CP/68. The file must be a text file (type 03), built with either the editor or PIP. The syntax of the SUBMIT command is:

SUBMIT [drive:] filename.ext

where the drive defaults to zero. All commands from the file will be echoed as they are read. There is a special divert character used in SUBMIT files. This is the ampersand "&" symbol.

The use of the divert character allows a one-line console command to be inserted into a SUBMIT command string. When "&" is found in a SUBMIT file, the user is prompted for a command. This command is executed, and then the SUBMIT file is resumed. When the end of the file is encountered, the system returns to command level at the console. For example, suppose the file SUBMIT.TXT contains the following:

DIRECTORY 1
STATUS
ASSIGN PTR=TTY
(escape)
&
LOAD INIT.CMD
end of file

Then, the following command:

SUBMIT SUBMIT.TXT

would first list the directory of drive 1, give the device status of the system, assign the PTR device to the TTY, escape to command level, accept a user command from the console and execute it, load the file INIT.CMD, and return to command level.

No wildcard characters are permitted in the filename or extension.

SYSTEM DEVICE ERRORS

All device errors in CP/68 are reported in the following format:

device-name ERROR: number

where device-name is the three-character logical name and the error number is hex encoded. For example:

> LPT ERROR: OA DSK ERROR: 02

are system device error messages. The set of errors defined in CP/68 are:

- end of directory found in search 01-
- file already in use 02-
- 03file already exists
- no such file exists 04-
- 05read/write error
- directory overflow 06-
- disk full 07-
- end-file encountered 08-
- 09- bad disk sector. bad media
- OA- device not ready OD- illegal use of File Control Block
- 12- illegal operation (write a read file, etc.)
- 15- bad file name

CP/68 SYSTEM ERROR MESSAGES

FORMAT ERROR The command line does not conform to the syntax specified for the command.

NUMBER ERROR A bad numeric argument is present. The drive number is out of range or is not followed by a colon.

FILE NOT FOUND The requested file could not be found.

DISK ERROR: aa AT SECTOR bb, TRACK ce This error message comes from the INITIALIZE command. The error type (aa) is a device-error number.

SYNTAX ERROR INVALID SET PARM

> These error messages come from the SET command. They indicate a bad SET command line.

The following errors come from PIP:

BAD INPUT (OUTPUT)

A device error; usually accompanied by a device-error message.

ILLEGAL INPUT (OUTPUT) DEVICE

Refers to attempts to use a device in an invalid manner, such as reading from a lineprinter.

BUFFER OVER-RUN

An overly long input line was encountered. The input file is probably the wrong type for the operation desired.

ILLEGAL SWITCH

Indicates a syntax error in the switch portion of the command line.

READ (WRITE) ERROR

Encountered in disk-to-disk copying; accompanied by a device error message.

DIRECTORY ERROR

The directory on a disk could not be read properly. This message is usually accompanied by a device-error message.

CHECKSUM ERROR

The checksum of a hex-formatted file was not correctly read.

Additional errors are:

SUBMIT FILE ERROR The filename in the SUBMIT command line could not be found or was not a TEXT file.

ILLEGAL FILE TYPE The file specified for LOAD was not a binary file.

RENAMING ERROR DUPLICATE NAME SECURITY ERROR

These errors messages come from the RENAME command. RENAMING ERROR indicates some form of disk error in accessing the drive containing the old file. DUPLICATE NAME indicates that the new name already exists on the disk. SECURITY ERROR indicates that the old file is protected from renaming. (access code=01 or 02)

UNABLE TO CHAIN: filename.ext

This error message indicates that a CHAIN request was made to the CP/68 system with filename.ext but it cannot be done. (no such file, disk read error, file not the right type, etc.)

FILE DELETE-PROTECTED

This file is protected from deletion (access code 02). It cannot be deleted until its access code is reduced.

DELETE ERROR-OPEN OUTPUT FILES

As long as any output files are open ${\tt CP/68}$ cannot delete a file on that disk.

Advanced User's Guide

INTRODUCTION

CP/68 is fully relocatable, supports dynamic disk files on multiple drives, has a clean and logical command structure, provides device-independent I/O, and has features which facilitate complex system operations. It requires slightly less than 8K bytes of contiguous memory plus a section of base-page (0020H to 0046H). Transient files overlay some system commands and user files. User files can chain in new files. Files can be used as a source of system commands.

CP/68 provides an extensive set of "extended instructions" which greatly add to the power of the 6800 instruction set. These "extended instructions" were used frequently in CP/68 itself. This portion of the book describes the structures and algorithms used in CP/68 in sufficient detail to allow you to add functions to the system and to interface your own programs to CP/68.

CP/68 DATA STRUCTURES

CP/68 uses several data structures in memory to perform various functions. These data structures are involved in all I/O operations, and some of the other system operations. The data structures discussed in this section include: Base-page, Equipment table, Physical device table, Request-control block (RCB), File-control block (FCB), File information block (FIB), and stack.

BASE-PAGE

 ${\rm CP/68}$ uses an area of base-page memory from address 0020H to 0047H to store global variables and system parameters. Most of these locations deal with I/O, while others are involved with command parsing and other functions.

Command-parsing variables

DESCRA 0020H

This 2-byte location stores the address in memory of the beginning character of a token. (For a description of "tokens", see the CP/68 operation NXTOK)

DESCRC 0022H

This byte stores the number of characters in the current token.

CUCHAR 0023H

This 2-byte location stores the address of the next character in the command line to be processed. Typically, this means that CUCHAR=DESCRA+DESCRC+1. CUCHAR is initialized to the beginning of the command line when it is desired to parse that line. DESCRA is automatically set by the NXTOK operation. To back up a token, set CUCHAR=DESCRA.

RC 0025H

This byte returns the return-code of the extracted token. (See NXTOK for a description of token codes.)

CLASS 0026H

This byte returns the class of the extracted token. The class is a subclassification of the RC. (See NXTOK for a description of token classes.)

VALUE 0027H

This 2-byte location stores the binary value of a numeric token when one is encountered during parsing. It is an unsigned 16-bit number.

Conversion from hex or decimal bases is done automatically by CP/68.

Disk information locations

FCBCHN 0029H

This 2-byte location stores the address of the header of the linked list of open file-control blocks. If FCBCHN is zero, there are no open files. If FCBCHN is not zero, it contains the address in memory of the first FCB that is active. Each FCB contains a pointer to the next FCB. If the pointer is zero, the end of the chain has been reached.

FRETAB 002BH

This is a table consisting of four, 2-byte entries. Each entry corresponds to one of the four disk drives maintained by CP/68. The entry stores the track and sector numbers of the header of the free-space chain on that disk. When a disk is being used, CP/68 copies the header data into the FRETAB entry so that it does not have to continually read the data from the disk. The entries are cleared when CP/68 is re-started.

Unused locations

0033H to 0039H Reserved for future expansion.

Console parameters

BS 0039H

This byte is the character to be used as a backspace on the console device. The default value for BS is 08 hex.

DL 003AH

This byte is the character to be used as the line-delete on the console device. The default value for DL is 15 hex (control-U).

DP 003BH

This byte is the number of lines per page on the console device. The default value for DP is 00 hex. (no limit on page depth)

DPCNT 003CH

This byte is used as the counter for the lines on a page on the console device. When DPCNT=DP, the end of a page has been reached. DPCNT is initialized to 01 hex.

WD 003DH

This byte is the number of characters per line on the console device. The default value for WD is 00 hex. (no limit on line width)

NL 003EH

This byte is the number of nulls which will be output with each linefeed on the console device. This feature allows linefeed delays for consoles which need such delays. The default value for NL is 00 hex.

TB 003FH

This byte is the character to be recognized as a tab on the console device. The default value for TB is 09 hex. (control-I)

DX 0040H

This byte is a switch which determines if the console device is to echo input characters. (Full or half duplex) If DX=00, the console is full-duplex and will echo all input. If DX=FF, the console is half-duplex and will not echo. The default value for DX is 00 hex. (full duplex)

EJ 0041H

This byte is the number of linefeeds to be output at the end of a page on the console device. The default value for EJ is 00 hex.

PS 0042H

This byte is a switch which controls the "pause" feature on console output. If PS=00, the console will wait at the end of a page of output until an escape character is input. (See ES below) If PS is not zero, the console will not pause. The default value of PS is FF hex. (no pause)

ES 0043H

This byte is the character to be interpreted as an "escape" on console input. The default value for ES is 1B hex. (ASCII "ESC")

Lineprinter parameters

LDP 0044H

This byte sets the number of lines per page on the lineprinter device. The default value for LDP is 60 decimal.

LDPCNT 0045H

This byte stores the count of lines on a page of lineprinter output. When LDPCNT=LDP, a full page has been output. The value of LDPCNT is initialized to 00 hex.

LWD 0046H

This byte sets the number of characters on a line for the lineprinter device. The default value for LWD is 80 decimal.

EQUIPMENT TABLE (EQTAB)

The Equipment table, in conjunction with the Physical-device table, vectors I/O using the device name provided by the user in the RCB or FCB. Each table contains an entry for each physical device in CP/68. physical devices are: Console (CON), Papertape reader (PTR), Papertape punch (PTP), Disk (DSK), Lineprinter (LPT), Magnetic tape (MTA), Teletype or alternate Console (TTY), and Null device (NUL). The CON device is the command source. It must be capable of input and output of ASCII characters. The CON "SET" parameters control its behavior. The PTR device is input only. The X-ON (11 hex) and X-OFF (13 hex) characters are used to turn PTR on and off. Linefeeds (OA hex) and nulls (OO hex) are swallowed. The PTP device is output only. A linefeed (OA hex) is issued with each carriage return (OD hex) and 4 nulls (OO hex) are added. The DSK device is a floppy-disk drive. The details of its operation are handled in the system code. The LPT device is an output-only printer. The LPT "SET" parameters control its behavior. The formfeed (OC hex) character is used to control paging on the LPT device. Linefeeds (OA hex) are automatically provided with each carriage return. (OD hex) The MTA device is unsupported in the present CP/68. The NUL device is actually not a device at all but simply a "bit bucket" or "do nothing". This proves useful at times to check out programs. Each device is given a three-character name.

Each entry in the equipment table has three 2-byte fields. The first field is the address of an input routine for that device. This routine must handle a line or block of data; CP/68 does not use character or single-byte I/O. If the device does not support input (the LPT for example), then the NUL handler is used. The second field is the address of an output routine for that device. This routine must also handle a

line or block of data. If the device does not support output (the PTR for example), the the NUL handler is used. The third field is the address of the interface used by that device.

As supplied, CP/68 assumes the following:

CON ACIA at 8008H PTR ACIA at 8010H PTP ACIA at 8010H

DSK special case...the handlers for this device have interface addressing built in.

LPT PIA at 8002H

MTA not implemented in the current version.

TTY ACIA at 8010H NUL no device needed

Note that the PTR, PTP, and TTY devices are set up to share one interface. This allows using the papertape facilities of a teletype (ASR-33) as well as its keyboard/printer. Note that CP/68 initializes the CONsole ACIA device, the TTY ACIA device, and the LPT PIA device on cold start. Other devices will need to be initialized by the user. An example Equipment table entry is shown below.

CONSOL FDB INLIN input a line from the console FDB OTLIN output a line to the console

FDB \$8008 ACIA at address 8008H

PHYSICAL DEVICE TABLE (PDTAB)

This table vectors I/O calls to the proper entry of the equipment table. Each entry in this table consists of three fields. The first field is the three-character name of the device; the second field is the address of the entry in the equipment table which services the physical device; the third field is also the address of the equipment table entry. The use of both fields allows for reassignment of a physical device. Suppose, for example, that you wanted to use the TTY device as the console. (See the ASSIGN command) You would modify the second entry of the physical-device table CON entry to point to the TTY entry of the Equipment table. All I/O directed to CON would then be vectored to the TTY device using the TTY handlers. The third field of the physical-device table entry is used to maintain a pointer to the original address of the device. Thus, no matter how many times a device may have been reassigned, there is still a pointer to its original Equipment-table entry. This is needed by some CP/68 commands, such as STATUS. Hence, each entry in the physical-device table has seven bytes. As an example, here is the CON entry.

FCC 'CON' name is CON

FDB CONSOL Equipment table pointer

FDB CONSOL "same"

The physical-device table uses a zero entry as an end marker.

REQUEST-CONTROL BLOCK (RCB)

All requests for I/O through CP/68 require a data structure in memory called an RCB or FCB. An RCB consists of 9 bytes of memory. Disk I/O requires the extended block (FCB). All other I/O requests may use an RCB. There are five fields in an RCB; three must be filled in by the user and the system provides the other two. The structure of an RCB is as follows:

RCBEQT supplied by the system

This 2-byte space is the address of the EQTAB entry which applies to this request for I/O.

RCBGDT required from user

This three-byte space must contain the three-character name of the device from-or-to which I/O is desired. CP/68 looks up this name in PDTAB and uses the entry there to find the EQTAB entry which it stores in RCBEQT.

RCBSTA supplied by the system

This byte is the status of the I/O request. It should be cleared before a CP/68 I/O request is issued. It returns any error conditions. It is zero for successful I/O completion. If RCBSTA returns nonzero, an error has occured.

RCBDTT required from user

This byte is a switch to choose input or output. If RCBDTT=0, then input is being requested. If RCBDTT=FF, then output is being requested.

RCBDBA required from user

This 2-byte space should contain the address in memory of a buffer to be used for I/O. It is up to the user to provide sufficient space in the buffer.

Example of RCB setup for CONsole input

RMB 2 space for RCBEQT FCC 'CON' RCBGDT FCB 0 RCBSTA

FCB 0 RCBDTT input FDB BUFFER buffer address

Example of RCB setup for PTP output

RMB 2 space for RCBEQT FCC 'PTP' RCBGDT

FCC 'PTP' RCBGDT FCB O RCBSTA

FCB \$FF RCBDTT output FDB BUFFER buffer address

To access fields in the RCB, the following EQUates will be useful.

RCBEQT EQU 0 RCBGDT EQU 2 RCBSTA EQU 5 RCBDTT EQU 6 RCBDBA EQU 7

Now, if the index register points to the RCB address...

LDA A RCBSTA, X get the status LDX RCBDBA, X get the buffer address

and so on.

FILE-CONTROL BLOCK (FCB)

This data structure is an extended RCB with additional fields necessary for disk I/O. It consists of 42 bytes of memory. The first five fields are identical to the RCB fields.

FCBEQT=RCBEQT FCBGDT=RCBGDT FCBSTA=RCBSTA FCBDTT=RCBDTT FCBDBA=RCBDBA

There are 14 additional fields in an FCB.

FCBDRV required from user

This byte must contain the drive number of the disk containing the desired file. Drive numbers run from 0 upwards.

FCBTRK supplied by system

This byte must contain the track number of the desired sector on the disk in FCBDRV.

FCBSCT supplied by system

This byte must contain the sector number desired on FCBTRK.

FCBFWD supplied by system

This 2-byte space is filled in by CP/68 with the forward link (track and sector) of the requested sector in disk reads and writes.

FCBBAK supplied by system

This 2-byte space is filled in by CP/68 with the backward link (track and sector) of the requested sector in disk reads and writes.

FCBNAM required from user

This 13-byte field must contain the file name and extension of the desired file for use by the file-manager of CP/68. The file name must be exactly 8 characters; pad with blanks as necessary to fill 8 characters. The ninth character must be a period. "." The extension must be exactly three characters; pad with blanks as necessary to fill 3 characters. The 13th character should be an "end-string" character. (04 hex) A system function is provided to format a string of characters into this internal form...see FMTS.

FCBTYP user supplied for new file, system supplied for existing file

This byte gives the type of file. If a new file is being created, the user should set this byte as follows:

- 00 binary file
- 01 binary file with transfer address (runable)
- 02 random file
- 03 text or hex file

Other numbers may be used, but CP/68 type-checks files that are loaded into memory, copied, etc. If the file already exists, the file manager will fill this field with the file type.

FCBACS user supplied for new file, system supplied for existing file

This byte gives the access code of the file. If a new file is being created, the user should set the byte as follows:

- 00 no protection
- 01 file can be renamed but not deleted
- 02 file can neither be renamed or deleted

If the file already exists, the file manager will fill this byte with the access code of the file.

FCBFTS supplied by system

This 2-byte field is filled by the system with the first track and sector

of the named file.

FCBLTS supplied by system

This 2-byte field is filled by the system with the last track and sector of the named file

FCBNMS supplied by system

This 2-byte field is filled by the system with the number of sectors used by the named file.

FCBNFB supplied by system

This 2-byte field is filled by the system with a link to the next active FCB in the system. If this is the most recent FCB in the system, the link will be zero. (See FCBCHN in base-page)

FCBIND supplied by system

This 2-byte field is filled by the system with a pointer to the buffer supplied at FCBDBA. This pointer indicates the present data byte in the buffer.

FCBSCF required from user

This byte is a switch to control space-compression in text files. If FCBSCF=0 then no space-compression is performed. If FCBSCF is nonzero, then all spaces within a file (20 hex) will be compressed as follows:

Any data byte =20 hex will be compressed. Spaces are replaced by the negative (2's-complement) of the number of sequential spaces. Hence, if the file contained the following 5 bytes of data:

41 20 20 20 41 'A A'

it would be compressed to read

41 FD 41

where FD=-3 .

When a file is read back with FCBSCF nonzero, spaces are re-inserted where necessary. Only files of ASCII text should be compressed.

Example of FCB setup to read file MYFILE.TXT on disk 1

RMB 2 FCBEQT
FCC 'DSK' FCBGDT=DSK
FCB 0 FCBSTA
FCB 0 FCBDTT=input

FDB BUFFER sector buffer address

```
FCB 1
              FCBDRV=1
RMB 1
             FCBTRK
RMB 1
             FCBSCT
RMB 2
             FCBFWD
RMB 2
             FCBBAK
FCC 'MYFILE '
FCC '.'
FCC 'TXT'
             FCBNAM
FCB $04
RMB 1
             FCBTYP
RMB 1
             FCBACS
RMB 2
             FCBFTS
RMB 2
             FCBLTS
RMB 2
             FCBNMS
RMB 2
             FCBNFB
RMB 2
             FCBIND
FCB $FF
             FCBSCF (compression on)
```

Here is a set of EQUates which will ease access of FCB fields.

```
FCBEQT EQU 0
FCBGDT EQU 2
FCBSTA EQU 5
FCBDTT EQU 6
FCBDBA EQU 7
FCBDRV EQU 9
FCBTRK EQU 10
FCBSCT EQU 11
FCBFWD EQU 12
FCBBAK EQU 14
FCBNAM EQU 16
FCBTYP EQU 29
FCBACS EQU 30
FCBFTS EQU 31
FCBLTS EQU 33
FCBNMS EQU 35
FCBNFB EQU 37
FCBIND EQU 39
FCBSCF EQU 41
```

Thus, if the index register points to the FCB address

```
LDA A FCBFWD,X get forward link track
LDA B FCBFWD+1,X get forward link sector
STA A FCBTRK,X put into track
STA B FCBSCT,X put into sector
```

and so on.

FILE-INFORMATION BLOCK (FIB)

This data block contains the information in the file directory on disk. Each file has a FIB, consisting of 32 bytes. In the present CP/68, only the first 20 bytes are used. The FIB fields match the FCB fields starting with FCBNAM and ending with FCBNMS.

FIBNAM=FCBNAM FIBTYP=FCBTYP FIBACS=FCBACS FIBFTS=FCBFTS FIBLTS=FCBLTS FIBNMS=FCBNMS

The FIBNAM field is always maintained in the proper format. The following EQUates will ease the access of FIB fields.

FIBNAM EQU 0 FIBTYP EQU 13 FIBACS EQU 14 FIBFTS EQU 15 FIBLTS EQU 17 FIBNMS EQU 19

STACK

CP/68 contains its own stack in its RAM space. Cold or warm starts reset the stack pointer to the system stack location.

CP/68 provides a 256 byte stack which is quite ample. Since system calls are done via software interrupts, and the stack is used for parameter passage, a minimum of 100 bytes of stack is needed to run CP/68 successfully.

DO NOT UPSET THE CP/68 STACK POINTER!

CP/68 DISK FORMAT

A disk initialized for CP/68 (see INITIALIZE command) has some data structure written onto it which CP/68 uses to work with files. These data structures must be maintained or CP/68 may do unpredictable things to the disk. An uninitialized disk will not work with CP/68.

Track 0

The first track on the disk (track 0) is reserved for the system. The first sector (sector 1) is used for bootstrap space, system linkage, and the free-space header. If SECSIZ is the number of bytes per sector on the disk, then

SECSIZ-5	first track of system-linked file
SECSIZ-4	first sector of system-linked file
SECSIZ-3	last track of system-linked file
SECSIZ-2	last sector of system-linked file

(These values are written by the LINK command)

SECSIZ-1 track of first free sector SECSIZ sector of first free sector

(These values are initialized by INIT, updated by file manager.)

The beginning SECSIZ-6 bytes of the first sector of the first track provides space for a bootstrap program. The remainder of track 0 is space for the file directory information. Files are described by 32-byte FIB blocks that are stored sequentially as long as there is space. The directory space is initialized to all zero by INIT.

A directory search is terminated when a zero is found at the start of a FIB block. A FIB is removed from the directory by placing a blank in the first character of the file name field (first byte of FIB). This does not recover the file's sectors, however. The DELETE function is provided to both remove a FIB and replace the file's sectors on the free-space list of the disk. The next file to be created will use that space.

Tracks 1-n

The rest of the tracks on the disk are used as CP/68 file space. Every sector has forward and backward links in its first four bytes. These links are automatically maintained by the system. Hence, each sector has SECSIZ-4 usable bytes. An initialized disk has its sectors linked in a pattern found to optimize access times, not usually in a sequential manner. The free-space chain header on track 0 points to the start of this list; sectors are allocated to files from this list and links changed accordingly. Deleted files return their sectors to the head of the free-space list. A much-used disk will become "fragmented"--the links will be very far from sequential. This increases access times, but CP/68

29

will not lose data as long as the links are maintained. The PIP command provides a way to "compact" a disk that has become fragmented.

(Note: the backward links are not used in the present CP/68.)

ISSUING SUPERVISOR CALLS (SVC)

CP/68 was written to be relocatable. Each routine could not have an absolute address. Also, it was desired that routines have standardized calling sequences and that registers be saved in most cases. The mechanism of the 6800 software interrupt was used to solve the problem of calling CP/68 routines. CP/68 has only two entry points: the cold start at its first byte, and the software-interrupt handler (SWIHDR) three bytes later. All system calls are performed by a software interrupt (SWI) instruction followed by a routine number. These two bytes are collectively referred to as an SVC. CP/68 automatically vectors the call to the appropriate address. The SWI saves the registers on the stack and recovers them on return from the system. Those routines that use registers for parameters manipulate them on the stack. Once CP/68 has been called, the stack contains:

stack pointer:

SWIHDR return address condition code byte accumulator B accumulator A Register X Return address

Thus, the following code would recover the contents of the B accumulator.

TSX LDA B 3,X

The following would return the condition codes to the user.

TPA TSX STA A 2,X

Since each CP/68 routine call is done in the same way, SWI and a byte, they can be made macros and used like new instructions. For example, CP/68 has a routine to read a byte from an open file. It would be called as follows:

SWI call CP/68 FCB 24 file-read

A macro could be written:

READ MACRO SWI FCB 24 MEND

so that whenever a file read was desired, a READ instruction could be given. CP/68 was written with the express purpose of providing a list of useful "extended instructions".

Using the software-interrupt mechanism, up to 256 different system calls are possible. In fact, CP/68 uses only 54 of these. (numbered 0-53) An SWI followed by any number larger than 53 will be vectored to the usual SWI trap in the underlying monitor. (Check the SWIHDR routine for the location of this trap.) Thus, breakpointing can be done in CP/68 with a two-byte "SWI"

SWI call CP/68
FCB \$FF force call to monitor

which will operate exactly like the simple SWI did without it. Programs that use SWI instructions must be modified to add the second byte, or CP/68 routines will be called with unpredictable results.

SVC ROUTINES

General instructions

00 PSHAL

This routine pushes all the register contents onto the stack in the normal 6800 order.

01 PULAL

This routine is the reverse of PSHAL. It restores the register contents from the stack.

02 TXAB

This routine transfers the contents of the index register to the A and B accumulators. The high byte goes into A, the low byte into B. The index register is undisturbed.

03 TABX

This routine is the reverse of TXAB. The contents of the A and B accumulators are transfered into the index register. The contents of A and B are not disturbed.

O4 XABX

This routine exchanges the contents of the index register and the A and B accumulators. A and B become X, X becomes A and B.

05 PSHX

This routine pushes the contents of the index register onto the stack. The low byte is pushed first, followed by the high byte. No registers are disturbed.

06 PULX

This routine is the reverse of PSHX. The index register is loaded from the stack. Only the index register is changed.

07 ADXAB

This routine adds the 16-bit unsigned contents of the index register to the combined 16-bit value in the A and B accumulators. The result is left in A and B, X is unchanged. The condition codes are set to reflect the results of the addition.

08 ADABX

This routine works like ADXAB except that the result is left in X, A and B are unchanged. The condition codes reflect the results of the addition.

09 ADDAX

This routine adds the unsigned byte in the A accumulator to the 16-bit unsigned value in the X register. The result is in the X register, A is unchanged. The condition codes reflect the result of the addition.

10 ADDBX

This routine is like ADDAX except that the B accumulator is used. The condition codes reflect the results of the addition.

11 SBXAB

This routine subtracts the 16-bit unsigned value in the index register from the combined 16-bit value in the A and B accumulators. The result is left in A and B, X is unchanged. The condition codes are set to reflect the results of the subtraction.

12 SBABX

This routine is like SBXAB except that the result is left in X, A and B are unchanged. The condition codes reflect the results of the subtraction.

13 SUBAX

This routine subtracts the unsigned byte in the A accumulator from the 16-bit unsigned value in the index register. The result is left in the index register, A is unchanged. The condition codes are set to reflect the result of the subtraction.

14 SUBBX

This routine is like SUBAX except that the B accumulator is used. The condition codes reflect the results of the subtraction.

15 MUL8

This routine multiplies the unsigned bytes in A and B accumulators and puts the resulting 16-bit value high byte in A, low byte in B. The condition codes are set to reflect the product of the multiplication.

This routine multiplies the unsigned 16-bit value in the index register by the 16-bit value in the A and B accumulators. The 32-bit result is left in A,B,X. The condition codes are set to reflect the result of the multiplication.

17 MOVC

This routine moves up to 256 bytes from one place to another. The from-address and to-address are placed on the stack. (to-address first, followed by from-address.) The byte count is passed in the B accumulator. On return, B=0, the stacked addresses have been incremented B times, and A is undisturbed.

Example:

get to-address LDX TOADDR PSHX use CP/68 LDX FRMADDR get from-address PSHX use CP/68 LDA B #100 move 100 bytes MOVC move them INS INS clean stack INS

18 CMPC

This routine compares two strings. It can be used for comparing text strings or other data. It can compare strings of up to 256 bytes in length. If the "end-string" character (04 hex) is found in either string, comparison is terminated. The parameter setup is the same as MOVC--the addresses of the two strings are stacked and the byte count goes into accumulator B. The result of the comparison is returned in the condition codes.

Example of using CMPC

INS

LDX #STRNG2 point to second string **PSHX** LDX #STRNG1 point to first string PSHX LDA B #10 compare 10 characters CMPC compare INS INS clean stack INS INS was string 1 > string 2? BGT ----

so that if STRNG1='AAAAAAAAAAAAAAAAAAAAAA and if STRNG2='BAAAA ', then the branch would not be taken.

45 MOVS

This routine works like MOVC except that it does not use a byte count the B accumulator. The move continues until an "end-string" (04 hex) is found in the from-string.

46 INDEX

This routine adds the product of the unsigned bytes in the A and B accumulators to the 16-bit unsigned value in the index register. The result is left in the index register, A and B are unchanged. The condition codes are set to reflect the results of the operation.

50 DIV16

This routine divides the unsigned 16-bit value in the combination of the A and B accumulators by the 16-bit unsigned value in the index register. The quotient is returned in the A and B accumulators. The remainder is returned in the index register. The condition codes are set to reflect the quotient value.

Command-parsing routines

47 NXTOK

This routine breaks up a command line into "tokens". A token is a substring of the command line which is treated as a unit. CP/68 defines the following tokens:

NAME A name is a string of characters which begins with an alphabetic character and contains only alphanumeric characters. (no imbedded spaces)

NAME WITH WILD-CARD CHARACTERS

A name which may include the special characters "*" and **#?#**_

NUMBER A string of digits which may be decimal or hexadecimal. Hexadecimal numbers must begin with a dollar sign. (\$)

DELIMITER

One of the special characters defined by CP/68. This includes the period (.), comma (,), colon (:), dollar sign (\$), equals sign (=), semicolon (;), and the arithmetic routines +.-. and /

The ASCII carriage return character. (OD hex)

ERROR A token not falling into one of the above classes.

NXTOK uses base-page for its parameters. Scanning the command line begins at the character whose address is in CUCHAR. The address of the first character of the token is returned in DESCRA. Note that spaces are not part of any token. Spaces are skipped over by NXTOK unless they are imbedded in a token. The count of the number of characters in a token is returned in DESCRC. The base-page locations RC and CLASS return the classification of the token as follows:

NAME RC=01 CLASS=02 NAME (WCRD) RC=02 CLASS=02 NUMBER RC=03 CLASS=02

DELIMITER RC=ASCII code of character CLASS=04

CARRIAGE RET. RC=OD hex CLASS=OD hex

ERROR RC=00 CLASS=00

CUCHAR is returned pointing one character beyond the end of the present token. If the token is a number (RC=03), then its binary value is returned in the base-page location VALUE. NXTOK will automatically convert unsigned decimal or hexadecimal numbers into binary form. The hex numbers must have a leading dollar sign. (\$) NXTOK will trap numbers that are too large (>65535 or FFFF hex) as errors.

Example of use of NXTOK

command line='LOAD 1:MYFILE.EXT ' carriage return

first token='LOAD' RC=01, CLASS=02
second token='1' RC=03, CLASS=02, VALUE=0001
third token=':' RC=3A, CLASS=04
fourth token='MYFILE' RC=01, CLASS=02
fifth token='.' RC=2E, CLASS=04
sixth token='EXT' RC=01, CLASS=02
seventh token=c.r. RC=0D, CLASS=0D

19 IOHDR

This is the basic I/O routine in CP/68. It is called with the address of the RCB or FCB in the index register and it causes the system to perform the I/O operation. No registers are disturbed by this routine. IOHDR handles entire lines or blocks of data at once. All CP/68 devices are handled through IOHDR, although some additional routines are provided for disk I/O and special cases of system I/O. The status of the I/O request is returned in RCBSTA (or FCBSTA).

Example of use of IOHDR to write character string on terminal

LDX #RCB point to RCB IOHDR

where the RCB has been set up as follows:

RCB RMB 2 space for EQTAB
FCC 'CON' console device
FCB 0 status
FCB \$FF output

FDB DATA address of data characters

DATA FCC 'THIS STUFF WILL BE PRINTED'
FCB \$0D carriage return

Note that a carriage return was used to indicate the end of a line. CP/68 will add a linefeed automatically for CON, TTY, or LPT I/O. If a new line is not desired, use an "end-string" (04 hex) in place of the carriage return.

Reading or writing a disk sector is done through IOHDR by some additional setup in the FCB. The FCBGDT must be 'DSK'. The FCBSTA is cleared. The FCBDTT is set to 00 for reading or FF for writing. The FCBDBA is set to point to a sector buffer. The FCBDRV is set to the desired drive number. The FCBTRK is set to the desired track number. The FCBSCT is set to the desired sector number. IOHDR will perform the read or write to/from the indicated sector on the indicated disk. Any disk sector can be accessed in this manner. The only error checking performed is that the desired sector exists on the disk and that the desired operation can be performed by the drive. The user is warned that IOHDR does not preserve the links or other data structures on the disk. This is done by the routines READ, WRITE, etc.

This routine prints error messages for device I/O errors. It is called with the address of an RCB or FCB in the index register. If the status (RCB or FCBSTA) is zero (good), it does nothing. If the status is nonzero, it prints an error message on the console device. The error message is of the form:

AAA ERROR: BB

where AAA is the device name (RCB or FCBGDT) and BB is the status value (RCB or FCBSTA) in hexadecimal.

48 GTCMD

This routine accepts a command line from the console. The user is prompted and a new line may be typed in. GTCMD passes the line directly to NXTOK, so on return from GTCMD, the first token on the line has been parsed. If the user desires to back up to the start of the line, set CUCHAR=DESCRA in base page.

49 PRTMSG

This routine prints a string on the console device. The index register is pointed to the start of the string. If the string terminates with a carriage return, a new linefeed is issued. If the string terminates with 04 hex, no linefeed is issued.

Filename formatting

44 FMTFCB

This routine parses a complete file designation including drive number, filename, and extension, and places it properly formatted into an FCB. The format that FMTFCB expects is:

[drive:] filename.ext

where the drive number and colon are optional. If the drive number is omitted, drive 0 will be assumed. FMTFCB will allow no wild-card names; it works only with unambiguous file references. To use FMTFCB, place the address of the character string containing the file specification into CUCHAR in the base-page. Place the address of the FCB into the index register. FMTFCB will place the drive number into FCBDRV and the filename appropriately formatted into FCBNAM. Any error conditions are returned in FCBSTA. If FCBSTA=00, the file specification was correctly formatted. If there was some error, FMTFCB returns an error status=21.

This routine formats a filename from the input form which may vary in length to the fixed internal form. It also handles the expansion of wild-card characters. The calling sequence is like MOVC, with from and to addresses on the stack and a byte count in the B accumulator. The from-address is typically the start of a token in the command line. The to-address is typically the FCBNAM field of an FCB. The byte count is the total length of the name; the sum of the length of the three tokens (name, . , ext) which comprise it. FMTS expands the wild-card character "*" into a string of "?" of the proper length. FMTS returns a condition byte in the B accumulator as follows:

B=00 unambiguous name

B=01 ambiguous name (wild-cards found)

B=02 bad name (error)

Example of the use of FMTS

CMDLIN FCC 'ABC?.*' length=6 characters

LDX #FCB+FCBNAM point to FCB name field

PSHX

LDX #CMDLIN point to command line

PSHX

LDA B #6

FMTS format name

INS

INS clean stack

INS

INS

at this point, B=01 and the name field of the FCB contains

ABC?^^^.??? where "^" indicates a space

53 CMWC

This routine compares strings like CMPC, except that it skips over the wild-card character "?" which matches any character, including a space.

23 OPEND

This routine accesses the directory track on a particular disk and returns a pointer to the first FIB on the disk. It is called with the index register pointing to an FCB which has the drive number set up in FCBDRV and 'DSK' in FCBGDT. The FCBDBA must point to a buffer large enough for one disk sector. The status (FCBSTA) is returned as follows:

00=good 01=end of directory found >1=error condition value

OPEND

If the status is good, the buffer (FCBDBA) contains the first sector of the directory from the indicated disk and FCBIND is initialized to the start of the first FIB. It is up to the user to check that the FIB is not a deleted file. This is done by looking for a space (20 hex) in the first byte of FIBNAM. Hence, if the index register points to an FCB which has FCBGDT, FCBDRV, and FCBDBA properly set, the following code will check for a valid FIB entry.

open directory

TST FCBSTA,X good status?
BNE ERROR no, error!

LDX FCBIND,X point to FIB
LDA A 0,X check first byte
CMP A #\$20 space?
BEQ NOGOOD if so, not valid

Note that FCBSTA=01 indicates a totally empty disk.

26 GETDR

This routine gets subsequent directory entries from a disk after OPEND has been used. Each call to GETDR will move the pointer FCBIND to the next FIB in the sector buffer. GETDR automatically reads new directory sectors as necessary until the end of the track is encountered. The calling sequence for GETDR is the same as that for OPEND: address of FCB in the index register and status returned in FCBSTA.

27 PUTDR

This routine is used to put a new FIB into a disk directory. It assumes that OPEND and GETDR have been used to find a spot for the new FIB where it will overlay either a deleted FIB or the next unused FIB on the disk. It assumes that the necessary file specification has been placed into the FCB (FCBNAM, FCBTYP, FCBACS, FCBFTS, FCBLTS, and FCBNMS) The index register is pointed to the FCB. PUTDR will copy the FIB entries from the FCB to the disk directory location pointed to by FCBIND. Status information is returned in FCBSTA.

20 OPEN

This routine opens a disk file for an I/O operation. It is called with the index register pointing to an FCB which has been initialized with the appropriate information. To open an existing file, set the following:

FCBGDT='DSK'
FCBSTA=0
FCBDTT=0 input
FCBDBA=address of sector buffer
FCBDRV=desired drive
FCBNAM=filename, properly formatted
FCBSCF=00 or FF depending on type of file (space compression)

To create a new file, set all of the above plus the following:

FCBTTP=FF output
FCBTYP=desired file type
FCBACS=desired file access code

OPEN will check that a new file does not conflict with a file that already exists on the disk and check that a file opened for input actually exists. Error status is returned in FCBSTA. OPEN places the FCB on the active FCB chain (see FCBCHN on base-page). As many open files as desired may be kept in the system, as long as there is a unique FCB for each one.

21 CLOSE

This routine finishes the processing of an active file and removes its FCB from the active file chain. It is called with the address of the FCB in the index register. Error status is returned in FCBSTA. For new files, CLOSE pads the last incomplete sector with nulls (00) so that the file contains all the desired data. CLOSE updates the directory FIB of the file to include the last track/sector used (FIBLTS) and the number of sectors (FIBNMS). Once a file is closed, its FCB space and buffer may be reused.

22 REWD

This routine is actually a CLOSE followed by an OPEN on the same file and using the same FCB. It can only be performed on input files. The effect is to return the file pointers to the start of the file. REWD is called with the index register pointing to the FCB. Error status is returned in FCBSTA.

This routine gets a data byte from an file opened for input. Bytes are read sequentially from the file. READ is called with the FCB address in the index register. It returns the data byte in the A accumulator. Error conditions are returned in FCBSTA. If the end of the file is reached, the status will return 08. READ cannot go beyond the end of the file. If space compression is set (FCBSCF=FF), READ will expand the compressed spaces into real spaces. (20 hex)

25 WRITE

This routine places data bytes into a file opened for output. Bytes are written sequentially into the file. WRITE is called with the data byte in the A accumulator and the index register pointing to the FCB. Error conditions are returned in FCBSTA. If space compression is set (FCBSCF=FF (hex)), WRITE will convert spaces (20 hex) into compressed internal format.

Initialization and Warmstart

31 WARMST

This routine returns control to CP/68 from a running program. This is the proper way to terminate a program written to run under the CP/68 system. WARMST will reset the stack pointer to the system stack, close all open files on the FCB chain, clear the free-space entries in base-page, and prompt for a new command.

51 INTDK

This routine does all necessary initialization processes for the disk drives. CP/68 does this on cold-start. The user may use this routine if the drive initialization must be redone from outside CP/68.

Deleting a file

28 DELETE

This routine removes an existing file from a disk. It is called with the index register pointing to an FCB which has FCBGDT='DSK', FCBDRV=desired drive, FCBNAM= filename properly formatted. DELETE checks the access code of the file to be sure that the file may be deleted. If FIBACS>00, DELETE will issue an error message, set FCBSTA=18, and return. DELETE requires that all open output files on the disk be closed. If there are open output files on the disk, DELETE will issue an error message, set FCBSTA=18 and return. DELETE removes the FIB from the directory by putting a space in the first character of FIBNAM. It links the sectors of the file to the head of the free-space list on that disk. It updates the free-space header link as well. Error conditions are returned in FCBSTA.

Program chaining

29 CHAIN

This routine loads a new program file into memory and starts executing it. It uses LOADB to bring in the new file. CHAIN is called with the index register pointing to an FCB with the desired FCBDRV, FCBNAM, etc. CHAIN moves the data from the user FCB into a system space so that the new file may overlay the user FCB memory. If there was some error, CHAIN will issue an error message and return to the system for a new command. If the file to be CHAINed had no transfer address, this will be flagged as an error. If there was no error, the new file will begin execution at its transfer address.

37 LOADB

This routine loads a binary-format file into memory. The file type (FIBTYP) must be 00 or 01. If it is not, LOADB will issue an error message and return without changing memory. LOADB expects the index register to point to an FCB with FCBGDT='DSK', FCBDRV, and FCBNAM set to the desired file specification. If an error condition is encountered while reading in the new file, LOADB will close the file and return to the system. If the file had a transfer address, it will be stored in the location VALUE in base-page. If there was no transfer address, VALUE will be zero.

User entries

32-36, 38-43 USR1-USR11

These entries in the dispatch table of CP/68 (DSPTAB) are unassigned and are left for the user to add new routines.

FORMAT OF CP/68 BINARY FILES

Binary files under CP/68 (this class includes all transient commands, system utilities, SAVE files, etc.) are stored on disk in a binary format to conserve space. There are two types of data in a binary file: transfer address, and memory data. Each type of data is stored in a block of up to 256 bytes. The format of a transfer address is:

BYTE 1 transfer address mark (16 hex)
BYTE 2-3 transfer address

BYTE 1 memory data mark (02 hex)

BYTE 2-3 memory address

BYTE 4 count of data bytes

BYTE 5--- data bytes exactly as in memory

Memory data is loaded at the address specified with it. There may be more than one transfer address in a file. If so, the last one in the file will be used. The last sector of a binary file will be padded with nulls (00 hex) as necessary to complete the sector. This has no effect on memory loading.

Binary files cannot be transferred to an ASCII device like the PTP or LPT. Similarly, files read from ASCII devices like the PTR or CON are not in the binary format. The system command PIP provides format conversions for these two formats.

he following examples illustrate usage of CP/68 routines to perform seful operations. They are not intended to be optimal programs, but imply to show how easy the CP/68 "extended instructions" make the task f dealing with files, etc.

This example shows how to open, read, write, and close files for nput and output. It is assumed that the user will type filenames in at he console when prompted to do so. Six routines are presented here:

```
OPENI open an existing file for input OPENO open a new file for output GETB get a byte from existing file OUTB put a byte out to new file CLOSI close the file being read CLOSO close the new file
```

It is assumed that the disk system has been initialized by use of NTDK. Two FCBs are assumed, one for each file in use. In this example, t is assumed that SECSIZ=256 bytes.

```
INFCB RMB 2
                    define input FCB
      FCC 'DSK'
      FCB 0
      FCB 0
                    direction (input)
      FDB INBUF
      RMB 33
OUTFCB RMB 2
                    define output FCB
      FCC 'DSK'
      FCB 0
      FCB $FF
                    direction (output)
      FDB OUTBUF
      RMB 33
INBUF RMB 256
                    sector buffer for input
OUTBUF RMB 256
                   sector buffer for output
```

he examples assume that the EQUates for FCBs and base-page ocations have been set up.

```
OPENI LDX #INMSG
PRTMSG prompt for input filename
GTCMD get filename from CONSOLE
LDX DESCRA
STX CUCHAR back up to first token
LDX #INFCB point to FCB

*
OPEN2 CLR FCBSTA,X init. status
CLR FCBSCF,X no space compression
```

TST FCBSTA.X error? BNE FILERR yes, print error message OPEN open file TST FCBSTA, X error? BNE FILERR yes RTS done! OPENO LDX #OUTMSG prompt for file name PRTMSG GTCMD get user file name LDX DESCRA STX CUCHAR back up to first token LDX #OUTFCB point to FCB BRA OPEN2 finish like OPENI FILERR PRTERR print error message WARMST return to system INMSG FCC 'INPUT FILE?' FCB \$04 OUTMSG FCC 'OUTPUT FILE?' FCB \$04 CLOSI LDX #INFCB point to FCB CLOSE close file TST FCBSTA,X error? BNE FILERR yes RTS CLOSO LDX #OUTFCB point to FCB CLOSE close file TST FCBSTA, X error? BNE FILERR yes RTS GETB save B accumulator PSH B PSHX save index register LDX #INFCB point to FCB READ read a byte from file

* the A accumulator now contains the byte read in

LDA B FCBSTA, X check status

CMP B #8 status=08 is end-file

BEQ GETB2

BRA FILERR otherwise, error

¥

GETB1 PULX recover index register
PUL B recover B accumulator
RTS

¥

GETB2 set whatever EOF flag is desired BRA GETB1

* byte to be written in A accumulator

*

OUTB PSH B save B accumulator
PSHX save index register
LDX #OUTFCB point to FCB
WRITE write byte to file
TST FCBSTA X error?

TST FCBSTA,X error?
BNE FILERR yes

*

PULX recover index register
PUL B recover B accumulator
RTS

nd for good measure, here is how to rewind the input file.

REWIND LDX #INFCB point to FCB REWD rewind file TST FCBSTA,X error?

BNE FILERR yes

RTS

ere is another example of the power of CP/68 to do fairly complex tasks n a few simple lines. Suppose the user wishes to have one program load n another whose name is defined in the program. Assume that INFCB and NBUF exist from the previous example.

LOADER LDX #FNAME point to desired file spec. STX CUCHAR store in base page pointer

LDX #INFCB point to FCB

FMTFCB format file spec. into FCB

PRTERR take care of errors

TST FCBSTA,X error found?
BNE QUIT if so, quit

LOADB

load in new file

TST FCBSTA,X error found?
BNE QUIT if so, quit

LDX VALUE look at transfer address BEQ QUIT if zero, no transfer address

JMP 0,X go to transfer address

QUIT RTS

*

FNAME FCC 'O:MYFILE.BIN'

FCB \$0D carriage return

A somewhat more complex example is this piece of CP/68 which searches a disk directory for an empty FIB location. It assumes an FCB and sector buffer set up like INFCB, etc. The track and sector of the slot (if found) are returned in FCBTRK and FCBSCT. Error status is returned in FCBSTA as follows:

00=found slot 01=no space available >1=error

The value TRKSIZ is assumed to be EQUated to the number of sectors in a track of the disk. It is assumed that the A accumulator contains the desired drive to be searched.

SEMPTY LDX #INFCB point to FCB

STA A FCBDRV, X save drive number

TXAB

LDX #INBUF get buffer address
XABX now X=FCB, A,B=INBUF

STA A FCBDBA, X set buffer address into FCB

STA B FCBDBA+1.X

CLR FCBSTA, X init. status

OPEND open directory of drive

SEMPT2 LDA A FCBSTA, X check status BEQ SEMPT3 status O.K.

CMP A #1 end of directory?

BEQ SEMPT4 yes

JMP FILERR otherwise error

SEMPT3 LDX FCBIND,X point to FIB
LDA A 0,X check first byte

CMP A #\$20 space? BNE *+3 no

RTS yes, found an empty FIB

LDX #INFCB

RTS

point to FCB GETDR get next FIB from directory

BRA SEMPT2 keep looking

SEMPT4 LDA A FCBSCT, X get sector number CMP A #TRKSIZ at end of track 0? BNE *+3 no. found empty FIB

yes, no room

CLR FCBSTA,X return good status RTS

The next example shows a second way to chain a new program in from another using CP/68. Using the CHAIN SVC, the new program can overlay the one that called it in. The assumptions of an input FCB, etc. are used here.

> LDX #MSG get program name to chain in

PRTMSG GTCMD

LDX DESCRA back up to first token

STX CUCHAR

LDX #INFCB point to FCB

FMTFCB set name, drive into FCB

TST FCBSTA, X error?

BNE FILERR if so, quit

CHAIN

bring in new program

CHAIN never returns

* it will either start new program or give

* error message and return to system

This next example illustrates the active-FCB chain process. It will print on the console the filename of every active FCB in the system.

> LDX FCBCHN get chain header if=0, no active FCBs BEQ DONE

LOOP LDA A #\$OD

STA A FCBNAM+12,X put c.r. after name

PSHX save pointer

LDX FCBNAM, X point to name field

PRTMSG

PULX recover FCB pointer *

DONE RTS

This example is the actual code used by the FMTFCB SVC in CP/68. It illustrates the use of NXTOK in parsing a line of text. It also illustrates how register data is passed on the stack to CP/68 SVCs.

```
FMTFCB TSX
      LDX UXH,X
                     point to FCB
      CLR FCBSTA, X clear status
       CLR FCBDRV,X default drive=0
       NXTOK
                     get a token (assume CUCHAR init.)
       LDA B RC
                     check RC
       CMP B #3
                     number?
       BNE PARS2
                     no
       TST VALUE
                     valid drive no.?
       BNE PARS1
                     no
       LDA A VALUE+1 valid drive no.?
       CMP A #3
                     (0,1,2,3)
       BHI PARS1
                     not valid
       STA A FCBDRV, X init. drive number
       BRA PARS1A
PARS1
      TSX
       LDX UXH,X
                     point back to FCB
       LDA A #21
                    return error code
       STA A FCBSTA, X
       CLR VALUE
       CLR VALUE+1 return no value
       RTS
PARS1A NXTOK
                     get token from command line
       LDA B RC
                     check RC
       CMP B #1:
                     colon?
       BNE PARS1
                     if not, error
       NXTOK
                     get token
       LDA B RC
                     check RC
PARS2 CMP B #1
                     unambig. name?
       BEQ PARS4
                     yes, good
PARS3
       TSX
       LDX UXH,X
                     point to FCB
       LDA A #21
       STA A FCBSTA, X return error code
       RTS
```

```
PARS4 LDX DESCRA
                     point to name
       STX SAVEX
                     save it in temp. loc.
       LDA A DESCRC
                     get length of name
       STA A SAVEA
                     save it in temp. loc.
       NXTOK
                     get a token
       LDA B RC
                     check RC
       CMP B #1.
                     period?
       BNE PARS3
                     if not, error
       INC SAVEA
                     count the period in length
       NXTOK
                     get a token
       LDA B RC
                     check RC
       CMP B #1
                     unambig. name?
       BNE PARS3
                     if not, error
       LDA B DESCRC get ext. length
       ADD B SAVEA
                     get total length
       TSX
       LDX UXH,X
                     X points to FCB
       LDA A #FCBNAM
       ADDAX
                     X points to FCBNAM
       PSHX
                     set up for FMTS
       LDX SAVEX
                     point to name
       PSHX
       FMTS
                     format file name
       INS
       INS
                     clean stack
       INS
       INS
       TST B
                     error check
       BNE PARS3
       RTS
SAVEA RMB 1
                     temp. locations
SAVEX RMB 2
```

Description of Routines

INTRODUCTION

The CP/68 operating system consists of a memory-resident part and transient files which are loaded into memory when needed. The various transient files overlay each other, since only one is ever in use at a given time. The resident part occupies memory from 0100 hex to about 2000 hex. The transients load starting at 2000 hex and occupy no more than 4 K bytes each (up to 3000 hex). A part of base-page is also used (a description of these locations is given elsewhere in this book).

The resident portion of CP/68 consists of five parts:

BIOS- the Basic I/O System
CLI- Command Line Interpreter
DREAD- Directory Read
SFIO- Sequential File I/O
DRIVERS- Disk Drive handlers

There are nine transient commands:

ASSIGN- make device assignments

BOOT- bootstrap system

DELETE- delete a file (part of this command is resident)

INIT- initialize a new disk

LINK- link a system file for BOOT
PIP- Peripheral Interchange Program
SECURITY- manipulate access code of files

SET- manipulate parameters of CON and LPT devices

STATUS- display present device assignments

In addition to these commands, some disk systems require a formatter program.

FORMAT- format a soft-sectored disk

Also included in this book is the Random-Access file package. This transient package of subroutines provides the facilities for random-access file manipulation under CP/68.

Resident Routines

BIOS (Basic I/O System)

The BIOS package consists of the software-interrupt handler (SWIHDR) and a set of routines which are called from it. SWIHDR is the only entry point within CP/68; it vectors all requests for system services to their appropriate handler in the system. The system must vector SWI instructions to SWIHDR to enable CP/68 to function. SWIHDR accesses the byte following the SWI instruction to determine the desired system operation. Invalid bytes (CP/68 has 53 valid operations) are vectored to a monitor location trap. Valid bytes are used to index the dispatch table (DSPTAB) to find the 16-bit offset of the system handler. A subroutine jump is made to the handler, passing all the registers on the stack. (SWI placed them there) Upon return, the return address is incremented to skip the operation byte and an RTI instruction returns to the caller.

Extended Instructions in BIOS

BIOS contains a set of system operations which effectively extend the instruction set of the 6800 to include many useful capabilities from the 6809 set. These instructions are described elsewhere in this manual. They are simply listed here.

```
PSHALL
         push all registers
PULALL
         pull all registers
TXAB
         transfer X to A,B
TABX
         transfer A.B to X
XABX
         exchange X and A.B
PSHX
         push X
PULX
         pull X
ADDABX
         add A,B to X
ADDXAB
         add X to A,B
ADDAX
         add A to X
         add B to X
ADDBX
SUBABX
         subtract A,B from X
SUBXAB
         subtract X from A.B
SUBAX
         subtract A from X
SUBBX
         subtract B from X
INDEX
         X=X + A * B
MUL8
         A \cdot B = A * B
MUL16
         A,B,X = A,B * X
DIV16
         A,B = A,B / X (remainder in X)
MOVC
         move a character string of given length
CMPC
         compare character strings
CMWC
         compare character strings with wild-card matches
MOVS
         move a character string with 04 hex terminator
```

FMTS (Format a filename string)

This routine takes a filename string as might be input from the console and formats it into the required CP/68 format. CP/68 wants filenames in the form:

NAME: 8 characters

DOT: period

EXTENSION: 3 characters

FMTS is called with the addresses of the input and output strings on the stack and the length of the input string in the B accumulator. It fills the output string space with blanks (20 hex) and places the dot in the 9th character position. It then moves the name and extension from the input string to the output string. It checks the name and extension for validity as it goes, it also checks for wild-card characters. The B accumulator returns a status code as follows:

- 00 hex unambiguous, valid name
- 01 hex ambiguous, valid name
- 02 hex invalid name

DISPATCH TABLE (DSPTAB)

This table contains the 16-bit signed offsets of each of the CP/68 system routines relative to the SWIHDR handler. Note that \$FFFF is -1 in 16-bit binary. The somewhat strange-looking form of the table entries is required since the assembler does not allow unary operators or parentheses in address expressions. For example, *-@PSHAL*\$FFFF, could be re-written as -(*-@PSHAL). Note that DSPTAB is also defined as an offset from SWIHDR.

EQTAB (Equipment table) and PDTAB (Physical Device table)

These tables are described in detail elsewhere in this book. They are used by the I/O handler routines, the ASSIGN, and the STATUS transients. Together, they serve to vector I/O requests to the system to the required device handler.

IOHDR (I/O Handler)

This is the central handler for CP/68 I/O requests. It is called with the address of a control block in the index register. IOHDR calls PDSRCH to look through PDTAB for the handler address of the logical device named in the control block. It then calls the handler. Handlers are called with the address of the control block in A,B. If the device name is invalid, IOHDR returns a status of 80 hex which indicates that no such device exists.

PDSRCH

This subroutine is used by IOHDR to access the physical device table. It is called with the address of the control block in the index register. A linear search is performed through PDTAB. If the device name is found, PDSRCH uses the address in PDTAB to point to the EQTAB. There it loads either the input or output handler vector and stores it into the control block. A carry-clear on return indicates that the name was found. A carry-set is returned if the name was not found.

Logical Device Handlers

These routines handle the input and output operations for each of the CP/68 logical devices. Each handler is entered with the address of the control block in A,B. They return that address in the index register.

NULL

The null device simply moves the control block address to the index register and returns.

INLIN (line-oriented input)

This routine handles lines of data from console-type devices. It handles tasks such as fielding "line-delete" and "back-space". It handles echo based on the SET "DX" parameter. It provides the CP/68 input prompt. It also outputs a linefeed for each carriage return.

Calls: INCON, OUTCON

OTLIN (line-oriented output)

This routine handles output of lines to console-type devices. SET parameters such as the null count (NL), line width (WD), paging (DP), ejects (EJ), and pause (PS) are handled in this routine. Detection of a break (any key struck during output) is provided in this routine. This code assumes an ACIA-driven device. The address of the ACIA is derived from the Equipment table.

Calls: INCON, OUTCON

INCON

This routine performs the actual handling of the console ACIA for input. It is called with the index register holding a buffer address. This value is preserved in INCON. The address of the control block is passed on the stack. INCON uses this address to access the EQTAB to get the actual ACIA address. INCON strips the parity bit and returns the character in the A accumulator. INCON will wait for a character.

OUTCON

This routine performs the actual handling of the console ACIA for output. It preserves the index and B registers. It uses the address of the control block from the stack to access EQTAB which gives it the ACIA address. The A accumulator passes the character to be output.

INRDR (line input from papertape reader)

This routine handles input from the papertape reader (PTR) device. It issues the X-ON (11 hex) character to start the reader and uses the X-OFF (13 hex) to turn it off at the end of the line. Nulls (00 hex) are swallowed.

Calls: RDRIN, OUTPCH

OTPCH (line output to papertape punch)

This routine handles line output to the papertape punch (PTP) device. It appends a linefeed (OA hex) and 4 nulls to each line.

Calls: OUTPCH

RDRIN

This subroutine handles the actual input from the ACIA driving the papertape reader. It is identical to INCON except for the stripping of the parity bit.

OUTPCH

This subroutine handles the actual output to the ACIA driving the papertape punch. It is identical to OUTCON.

OTLPT (line output to lineprinter)

This routine outputs a line to the lineprinter device. It assumes a PIA-type interface. The SET parameters for page width (LWD) and page depth (LDP) are handled in this subroutine. OTLPT issues a formfeed (OC hex) to space pages. It automatically adds a linefeed for each line.

Calls: OUTLPT

OUTLPT

This subroutine actually handles output to a PIA port. It preserves the index and B registers. The address of the control block from the stack is used to access EQTAB to get the PIA address. The character to be output is passed in the A accumulator. An acknowledgement signal is expected from the device.

The rest of BIOS is a set of jumps to the other routines forming CP/68. These jumps are necessary for SWIHDR to vector to separately assembled modules. (CLI, Directory read, Sequential File I/O, and Disk Drivers)

Command Line Interpreter (CLI)

The CLI is the heart of CP/68. All command processing passes through it. It contains the routines that load transients and programs, that save memory onto disk, that parse command lines, etc.

Command Table (CMDTAB)

This table contains all the commands directly recognized by CP/68. Each table entry consists of the first three characters of the command name and the address of the command handler. Hence, all CP/68 command names can be abbreviated to their first three characters. A zero marks the end of the table.

Character Table (CHRTAB)

This table is used by the parsing routines (NXTOK) to evaluate a character for the type of token it could be in. Characters from the space (20 hex) to underline (5F hex) in the ASCII set have an entry in the table. Each entry is a byte where each bit has a significance as follows:

- Bit 7 Alphabetic
- Bit 6 Decimal digit
- Bit 5 unused
- Bit 4 unused
- Bit 3 delimiter
- Bit 2 Hexadecimal digit
- Bit 1 Wild-card character

A set bit indicates that the character is a member of the class. For example, the letter "A" has the entry 82 hex. This means that it is both an alphabetic character and a hex digit. Note that the wild-card characters are declared alphabetic (81 hex).

CLI Main loop

There are two entries to CLI, called COLDST and WARMST. There is a jump to COLDST at the beginning of BIOS (start of CP/68). This is the starting location of the system. WARMST is the return to the system, and it is reached through SWIHDR. COLDST performs the initialization steps for the system. The stack pointer is set to the internal stack space. The SUBMIT flag is cleared (no SUBMIT in process). The console and TTY ACIAs are initialized. The set-up for the CON device is:

Counter divide- 16
Word select- 8 bits + 1 stop, no parity

Interrupts- disabled

The set-up for the TTY device is:

Counter divide- 16

Word select- 7 bits + 2 stop, even parity

Interrupts- disabled

The lineprinter PIA is initialized as follows:

A side: undefined

B side: output, CB1 active low input, IRQ disabled

CB2 output

INITDK is called to initialize the disk hardware. The console control block CONRCB is initialized and the start-up banner is printed. The header of the active-file chain is initialized. Processing now begins the usual CLI loop.

WARMST also sets the stack pointer and clears the SUBMIT flag. It then looks through the active-file chain, closing all files that it finds. It then enters the usual CLI loop.

WARM3 marks the start of the command-processing loop. First, the four free-space headers are cleared. Now a command line is input using GTCMD. This line might come from the console or from a SUBMIT file. GTCMD automatically parses the first token from the line. If it is an ambiguous name (wild-cards), it is a format error. If it is a number, it is assumed to be the drive number of a filename. Otherwise, it is an unambiguous name which might be a command or else a filename on drive 0.

The command table is searched to determine if the name is a command. If the name is found, control jumps the the processor for that command which returns to WARM3 when it completes. If the name is not found, or if this is a filename not on drive 0, the system routine (LODCMD) brings the named file into memory. Since LODCMD does its own parsing of file names, the pointers are first returned to the start of the command line. If a transfer address was loaded, control jumps to that address. If no transfer address was found, or after the loaded process returns, control returns to WARM3 for a new command.

Calls: LODCMD

PRTMSG

This routine prints a message on the console and is used by all the CP/68 routines for printing error messages and prompts. It is called with the address of the text string in the index register. The string must be terminated with either a carriage return (OD hex) or a string terminator (O4 hex). The carriage-return causes an automatic linefeed, the string

terminator does not.

PRTERR

This routine prints a formatted error message on the console. It is called with the address of a control block in the index register. It tests the status byte in the control block for error conditions. If there was no error, it prints nothing. If the status byte is nonzero, it converts the byte to hex and stores it in the error message field DERNUM. The device name is taken from the control block and stored in DEVNAM. Finally, the error message is printed.

GTCMD

GTCMD is called to input a line of text from the user. Based on the SUBMIT flag SUBFLG, the line might come from the console or from an open SUBMIT file. If SUBFLG is cleared, GTCMD reads a line from the console. If SUBFLG is set, GTCMD reads a line from the open SUBMIT file, using the file-control block SUBFCB. If reading from a file, the special characters "&" and O4 hex (control-D) are processed. The control-D indicates the end of the SUBMIT file; the file is closed, SUBFLG is cleared, and a line is input from the console. The "&" indicates diversion in a SUBMIT file, one line is taken from the console without upsetting the file or SUBFLG. No matter where the line came from, GTCMD always goes into the parsing routine NXTOK to find the first token on the line.

Calls: NXTOK

Flags: SUBFLG

NXTOK (parsing tokens)

This routine performs the parsing function on a CP/68 command line. Each time it is called it determines the next lexical token of the command line. There are six types of tokens which are recognized:

Multi-character strings- Unambiguous name

Ambiguous name

Number

Single characters- Delimiter

Carriage return Error (undefined)

NXTOK uses the pointer CUCHAR to point to the starting point on the line to begin parsing. NXTOK moves CUCHAR to point just beyond the end of

the present token. NXTOK returns four values for each token. DESCRA is a pointer to the first character in the token. DESCRC is a count of the length of the token. RC is a code for the type of token. CLASS is a code for major classification of the token.

NXTOK first skips over any blanks up to the first non-blank character. If the character is less than 20 hex, it is either a carriage return or undefined. If it is greater than 5F hex, it is undefined. This means that lower-case characters are not recognized. Next, NXTOK calls GCHRTB which looks up the character in CHRTAB. If the character is alphabetic, NSCAN is called to parse the name. If the character is a decimal digit, DSCAN is called to parse the decimal number. If the character is neither, and it is not a delimiter, it is an error. If it is a delimiter, NXTOK checks for a "\$" character. If found, HSCAN is called to parse a hexadecimal number. Otherwise, the delimiter token is returned.

Calls: GCHRTB, NSCAN, DSCAN, HSCAN

DSCAN

This routine parses a decimal string. It looks at characters from the command line one at a time until a non-decimal digit is found. The pointers are decremented to the last decimal digit and it is checked for length (since CP/68 works with 16-bit numbers, it can accept nothing larger than 65535). CVDB is called to convert the decimal string into binary which is returned in VALUE.

Called by: NXTOK

Calls: GCHRTB, CVDB

NSCAN

This routine parses an alphanumeric string. It looks at characters from the command line one at a time until a non-alphanumeric character is found. The pointers are then decremented to point to the last alphanumeric character in the string. The B accumulator is used to indicate if a wild-card character was found in the name string.

Called by: NXTOK

Calls: GCHRTB

HSCAN

This routine parses a hexadecimal number as indicated by a leading dollar sign (\$). It looks at characters from the command line one at a time until a non-hexadecimal digit is found. The pointers are then decremented to point to the last hexadecimal digit in the string and the length is checked (since CP/68 can accept numbers up to \$FFFF). CVHB is called to convert the hex string into binary which is returned in VALUE.

Called by: NXTOK

Calls: GCHRTB, CVHB

GCHRTB

This routine accepts a character in the A accumulator and uses it to index the character table CHRTAB. The entry from the table is returned in the A accumulator.

Called by: NXTOK, NSCAN, DSCAN, HSCAN

Tables: CHRTAB

CVHB

This routine converts a hexadecimal string into binary. On entry, DESCRA points to the start of the string and DESCRC is the number of characters in the string. It returns the 16-bit unsigned binary value in the index register.

Called by: HSCAN

CVDB

This routine converts a decimal string into binary. Its calling sequence is identical to CVHB.

Called by: DSCAN

Command Processing routines

All command processing routines are called as subroutines from the CLI loop.

JMPCMD

This routine processes the JUMP command. It uses NXTOK to parse the jump address. It removes the return address (JMPCMD was called as a subroutine) from the stack and executes a jump to the address specified in the command line. If the routine jumped to executes an RTS, it will return to the CLI loop. A "safer" return would be to issue a WARMST call.

Transient Command Processor

The set of CP/68 commands processed by transients:

ASSIGN, BOOT, DELETE, LINK, PIP, SECURITY, SET, STATUS

must load the required file into the transient space. This is accomplished by using a "dummy command" which effectively forces the filename of the transient command to become the command line. LODCMD is called to bring the transient into memory. For the transients that require it, the address of PDTAB is passed in the A and B accumulators.

Calls: LODCMD

SUBCMD (SUBMIT command processor)

This routine processes the SUBMIT command. It uses FMTFCB to parse a filename from the command line into the SUBMIT FCB (SUBFCB). Blank expansion is turned on and the file is opened. The filetype is checked to insure that the file is a text file. The SUBMIT flag is set, indicating to GTCMD that lines should now come from the file, not the console.

This routine processes the SAVE command. It first initializes the control block SAVFCB as a type 0 file. FMTFCB is used to parse the filename into SAVFCB. The starting address is then parsed and saved in SAVEX. The ending address is parsed and saved in SAVEX1. If this is the end of line, then no transfer address is desired. If there is a delimiter, then a transfer address is parsed, the filetype is made 1, and a transfer-address block is written to the file. Next, data records consisting of 256 data bytes each are written out to the file. When the ending address is reached, the last data block is written out and the file is closed.

LODCMD (LOAD command processor)

This routine loads a file into memory. It processes the LOAD command and is used by the CLI loop and the transient command processor as well. It uses FMTFCB to parse the filename and then uses LOADB to actually load the file into memory.

Called by: CLI loop, Transient processing, INICMD

Calls: LOADB

LOADB

This routine actually loads a memory-image file (produced by SAVE) into memory. The file must be type 0 or 1 (memory-image). The load process opens the file and looks for either data blocks or transfer address blocks. Data blocks contain their load address, so the following data is stored into the indicated address. Transfer address blocks store their address into VALUE. Hence, the last transfer address found in the file will be used.

Called by: LODCMD, CHAIN

RENCMD (RENAME Command processor)

This routine processes the RENAME command. FMTFCB is first used to put the old filename into SAVFCB. SFILE is called to search the directory for this file. If found, the access code is checked to see whether this file is rename-able. If so, the second filename (the new one) is parsed. Note that the second filename can have no drive number, since the first drive number is assumed. Pointers to the directory entry of the old file are stored in SAVFCB. SFILE is called with the new filename to insure that it does not duplicate an existing name. If there is no duplication, the

directory entry for the old filename is re-accessed and the new name field is written into it.

Calls: SFILE

INICMD (INITIALIZE Command Processor)

This routine processes the INITIALIZE command. It parses the drive number and checks it for validity. LODCMD is used to bring the transient code for INIT. into memory. The drive number is passed in the A accumulator and control is given to the transient code. When it is complete, it returns to the CLI loop.

Calls: LODCMD

DIRCMD (DIRECTORY Command Processor)

This routine processes the DIRECTORY command. It begins by formatting ALLFIL into a temporary BUFLIN. ALLFIL is a wild-card specification which matches all filenames. The lineprinter flag LPTFLG is cleared to direct output to the console. A check is made for the lineprinter switch /L . If found, the lineprinter flag LPTFLG is set. Otherwise, DIRCMD looks for a drive number. If a number is found, it is checked for validity and if it is valid it is stored in SAVEA. Next, DIRCMD looks for a file specification. This file specification may contain wild-cards. If a file specification is found, it is formatted into BUFLIN. The number of sectors used (NSEC) is cleared. If LPTFLG is set, the output is redirected to the LPT device. The drive number is recovered from SAVEA and converted to ASCII. The header messages are printed. The directory of the desired drive is opened.

DIRCMD now loops through each directory block on the given disk. It compares each file on the disk with the name in BUFLIN. If they do not match (including wild-cards), DIRCMD looks at the next file in the directory. If a match is found, the data from the directory block is formatted into a string for output. The string is printed and DIRCMD looks at the next file. When the end of the directory is found, the number of sectors used (the sum of the number of sectors of each file which matched) is converted to ASCII and the finishing message is printed.

Imbedded in DIRCMD is a routine called CVBTD. This routine converts a 16-bit unsigned binary number to ASCII. The number is passed in the A and B accumulators. The address of the place to form the ASCII text is passed in the index register. CVBTD generates five characters.

CHAIN

This routine provides CP/68 the facility to load and run a transient file from an executing program. It works by moving the necessary information from the user's FCB to the system SAVFCB. The user's FCB address is passed in the index register. By moving to SAVFCB, the new program can overlay the user's FCB. CHAIN calls LODCMD to bring the new file into memory. If a transfer address is found in the new file, control jumps to it. Otherwise, control returns to the CLI loop.

Calls: LODCMD

SEMPTY

This routine is used to search a disk directory for an empty slot. It looks through the directory for either a directory block with a blank as its first character (indicates a deleted file) or the end of the directory. If a usable directory block is found, SEMPTY returns a status of 0. If no usable block is found, a status of 1 is returned. SEMPTY uses a system control block SYSFCB. It is called with the drive to search in the A accumulator. It returns the pointers to the directory block in SYSFCB. (FCBTRK, FCBSCT, and FCBIND) The status is returned in FCBSTA.

Called by: OPEN (sequential file I/O)

SFILE

This routine searches a disk directory for a given, non-ambiguous file. It is called with the address of a control block in the index register. This FCB contains the drive and filename of the file to be searched. SFILE returns status in the supplied FCB. A status of 0 indicates the file was found. A status of 1 indicates the file was not found. FCBIND in the supplied FCB points to the directory block. SFILE uses SYSFCB to manipulate the directory.

Called by: OPEN, CLOSE (sequential file I/O), RENCMD, DELETE

This routine handles the removal of a file from a disk. It is called with the address of an FCB in the index register. This FCB contains the filename and drive of the file to be deleted. First, SFILE is called to locate the file in the directory. The access code is checked to see if this file may be deleted. If so, all the active FCBs are checked to see if there are any open files on this disk. If there are, no file deletes may be performed on the disk, since this might corrupt the linkages of the sectors. If there are no active files on this disk, the directory entry of the file is read in. The first and last track/sector pointers are saved. A blank is inserted into the name field in the directory. The present header of the free-space list on this disk is saved. The first track/sector of the file becomes the head of the free-space list. The last track/sector of the file is linked to the old free-space header. This puts the sectors from the deleted file back onto the free-space list. The free-space sector is updated to match this.

Calls: SFILE

FMTFCB

This routine parses a file specification from the command line and places the result into a supplied FCB. The address of the FCB is passed in the index register. The pointer CUCHAR indicates the beginning of the file specification. FMTFCB first looks for a drive number. If none is found, drive 0 is assumed. If a number is found, it is checked for validity. FMTFCB expects an unambiguous name. (no wild-cards) If a syntax error is found while parsing, 21 hex is returned in the FCBSTA field of the FCB.

DIRECTORY-READ Routines

This set of routines provides the means to read and change a disk directory under CP/68. It consists of three entries: OPEND, GETDR, and PUTDR. A CP/68 directory is a sequence of 32-byte directory blocks stored on the first track of the disk. The end of the directory is marked by a directory block whose first character is a zero. If the first character is a blank (20 hex), this directory block is assumed to have been deleted and new files will over-write it.

OPEND

This entry opens a disk directory for use. It positions the drive to the first track (directory) and reads in the first sector of the directory. The first character of the directory sector is tested. If it is zero, the disk directory is empty and a status of 01 hex is returned, indicating

that the end of the directory was found. If it is not zero, a zero status is returned. OPEND is called with the address of a user FCB in the index register. The FCB must have the drive number set and the device-type must be set to DSK. It returns status information in the FCB.

GETDR

This entry reads directory blocks from an open directory. OPEND must be called prior to calling GETDR. GETDR moves the pointers to the directory 32 bytes forward each time it is called. This effectively accesses the directory block for the next file on the disk. GETDR will read a new sector when it finishes the previous one. It will return a status of 00 hex if it finds a good file block in the directory. It will return a status of 01 hex if it finds the end of the directory. Its calling sequence is the same as that of OPEND.

PUTDR

This entry updates a directory block that has been found with OPEND and GETDR. The changes to the file directory data are made to the copy in the sector buffer used with OPEND and GETDR. Calling PUTDR with the address of the FCB in the index register will re-write the directory sector into the directory, making the desired updates.

SEQUENTIAL-FILE I/O Routines

These routines handle sequential files under CP/68. They direct the directory-routines and the drivers to form a file-management system. There are five routines: OPEN, CLOSE, READ, WRITE, and REWD. Each is called with the index register pointing to an FCB. Those routines which pass characters (READ, WRITE) use the A accumulator. These routines also handle space-compression for text files.

OPEN

OPEN prepares files for use under CP/68. It first checks that the file is not already open, then it determines whether the file is to be opened for input or output. The in/out decision is based on the FCBDTT byte in the FCB.

Input files are checked against the disk directory to see if the file already exists. The system subroutine SFILE performs this check. Next, OPEN moves the file pointers, type, etc. from the directory to the FCB. The first sector of the file is read in; the forward and backward sector links are put into the FCB. Finally the FCB is added to the

Output files are processed differently. SFILE is called to check that the new filename does not duplicate an already existing file. Next, the system subroutine SEMPTY is called to find an available directory block for the new file. The FCBNMS (number of sectors), FCBLTS (last track/sector), and FCBBAK (back pointers) fields in the FCB are cleared. The free-space header for the desired disk is accessed. If it is nonzero, this is the track/sector of the next available sector. If it is zero, the free-space sector (link sector) is read and the header is updated. The free-sector is checked to see that it is not the end of the disk (0,0). The FCBFTS (first track/sector) field in the FCB is initialized to the free sector and the directory entry is written using PUTDR. The free sector is read in and the free-space header is updated to be the next available sector. Finally, the FCB is added to the active-FCB chain.

Calls: SFILE, SEMPTY

CLOSE

This routine finishes the processing of a file. First CLOSE checks that the FCB is open. If it is found in the active-FCB chain, it is removed from the chain. If it was an input file, CLOSE is finished. For output files, CLOSE must write out the last sector. It uses SFILE to find the directory entry for the file and updates the FCBLTS (last track/sector)

and FCBNMS (number of sectors) entries. The free-space record is updated. This completes the CLOSE process.

Calls: SFILE

sector buffer.

READ

This routine gets a byte from an open input file. It checks to see if the desired byte is in the sector buffer already. If it isn't, a new sector is read in and the forward and backward links are updated; the byte is accessed from the buffer. If no space-compression is required, the file pointer (FCBIND) is incremented and the data byte is returned. If space-compression is required, a test is made of the data byte. If the byte is positive (high-order bit is zero), the data byte is returned unchanged. If the byte is negative (high-order bit set), the byte is a compressed space. The data byte is actually the negative count of the number of spaces desired. The data byte is incremented and restored to the buffer while a space (20 hex) is returned. When the data byte reaches 00 hex.

the last space is returned and the file pointers are moved. Until then, spaces are returned while the file pointer stays in the same point in the

WRITE

This routine writes data bytes to an open disk file. It first checks that the file is open for output; next it checks to see if the end of the sector buffer has been reached. If it has, the present sector buffer is written to the disk. The number of sectors in the file (FCBNMS) is incremented; the free-space header is updated, as are the forward and backward file pointers (FCBFWD and FCBBAK). A new sector is read in from the free-space chain and linked to the file. In either case, the next step is to store the data byte into the sector buffer. If no space-compression is being done, WRITE is completed. If space-compression is being done, and if the data byte is a space (20 hex), the present value of the data byte in the file is checked. If it is negative (compressed space), the value is decremented (one more space) and restored. If it is not negative, a single compressed space (FF hex) is stored into the file. This completes WRITE.

REWD

This routine rewinds an input file to its starting point. Effectively, REWD is a CLOSE followed by an OPEN.

DRIVER Routines

These routines provide the interface between CP/68 and the disk hardware. Three entries are needed: INITDK, RDSEC, and WTSEC. The exact mechanism of these routines depends on the hardware being used.

INITDK

This routine performs all necessary initialization required by the disk system. This may include initializing peripheral interfaces, setting memory flags, calling ROM routines, etc. It is called with no parameters.

RDSEC

This routine reads a desired sector from the disk. It is called with the address of an FCB in the A and B accumulators. The FCB contains the drive, track, and sector pointers. It also contains a pointer to the buffer area. The status of the read must be returned in the FCB. It should also be returned in the A accumulator. Since these routines are called from software interrupts, they must change the stacked-value of the accumulator in order to return it. RDSEC must detect disk errors and return appropriate error status numbers.

This routine writes a desired sector to the disk. It is called with the address of an FCB in the A and B accumulators. The FCB holds the drive, track, sector, and sector-buffer pointers. The status should be returned in the same manner as RDSEC.

Transient Commands

ASSIGN Transient Command

This routine processes the ASSIGN command from CP/68. It re-directs a logical device by modifying the physical-device table entry (PDTAB) of a given device. PDTAB entries consist of 7 bytes. The first three bytes are the name of the device. The next two bytes are a pointer to the appropriate entry in the equipment table (EQTAB) where the device handler addresses are found. The last two bytes are also a pointer to the EQTAB. ASSIGN modifies the first pointer field, but the second pointer is left intact so that other routines (such as STATUS) can find the original device assignment.

When ASSIGN is called from the command-interpreter, the address of PDTAB is passed in the A and B accumulators. ASSIGN then proceeds to parse the command line, obtaining the names of the devices to be assigned. The device to be assigned is stored in DEV1, the device to which it is being assigned is stored in DEV2. The subroutine PDSRCH is used to check the names in DEV1 and DEV2 against the names in PDTAB to insure that both are valid device names.

If DEV1=DEV2, the second pointer field of the name is copied into the first pointer field of the name. If DEV1 is different from DEV2, then both names are checked with PDSRCH, and the second pointer field of DEV2 is copied into the first pointer field of DEV1. Note that even though DEV2 may have been re-assigned itself, the second pointer field retains the initial value.

Called by: CLI Calls: PDSRCH Tables: PDTAB

PDSRCH

ASSIGN uses this routine to check device names for validity. It searches the physical-device table (PDTAB) for a device name whose address is passed in the index register. The end of PDTAB is marked with a zero. PDSRCH returns with carry-set if the device was not found, and with

carry-clear if the name was found.

Called by: ASSIGN

Calls: none Tables: PDTAB

BOOT Transient Command

This routine bootstraps a system file from drive 0 using no system support. It assumes that the disk in drive 0 has had a bootable file linked on it (See LINK). It is written to be ROMable, with all necessary RAM locations in COMMON storage. It also uses its own stack space.

The first step BOOT performs is to initialize the disk drives. This process varies depending upon the hardware requirements. The next step is to read in the link sector. (track 0 sector 1) The last six bytes of this sector contain special information.

SECSIZ-6 First track of linked file

- -5 First sector of linked file
- -4 Last track of linked file
- -3 Last sector of linked file

SECSIZ-2,1 Free-space pointer

The track/sector pointers define the linked file.

BOOT loads the desired file into memory just like the system LOADB routine does. The marker 16 hex indicates a transfer-address block, the marker 02 hex indicates a data block. The loading process continues until the last sector of the file (as determined from the link sector) has been loaded. The program then jumps to the transfer address read from the booted file. Finding a null (00 hex) while searching for a data block will also indicate the end of the file and will cause a transfer to the start address read from the file.

Called by: CLI

Calls: GETBYT, RDSEC

GETBYT subroutine

This routine is used by BOOT to read in the desired file. It returns data bytes in the A accumulator. When necessary it calls RDSEC to get a new data sector from disk. When GETBYT finishes the last data byte of the last sector of the file, it jumps to the spot in BOOT which indicates an end-of-file condition.

Called by: BOOT Calls: RDSEC

RDSEC is the routine used to read individual sectors from the disk. It is called with the desired track in accumulator B, the desired sector in accumulator A, and the address of a buffer in the index register. RDSEC assumes drive 0. The actual mechanism of RDSEC depends on the hardware used to control the disks.

Called by: BOOT, GETBYT

DELETE Transient Command

This transient routine is used in conjunction with the resident DELETE code to handle the removal of files from the disk. The resident code actual performs the disk update, this transient handles set-up for it and also takes care of wild-card names, check-prompting, and other tasks.

DELETE first accepts a filename and tries to format the name into its internal SYSFCB. Since there may be wild-cards in the name, a temporary buffer called TEMP is used to hold the name. If the name parses as a good filename, the next step is to search the desired disk directory for a file whose name matches the given name in TEMP. If such a file is found, DELETE forms a prompt line with the file name and waits for a user response. If the response is "Y", the file is set up for the resident DELETE and is then erased from the disk. After the file is erased, or if the response was not "Y", the transient continues to search the disk directory for further matches. If more are found, they will each be prompted in turn. When the end of the directory is found, DELETE will prompt for a new filename. Entering an ESCAPE character returns the system to the command level.

Called by: CLI Calls: none

INITIALIZE Transient Command

This routine builds the necessary data structure for CP/68 on a blank disk. Soft-sectored disks must have been previously formatted before using this routine on them.

INITIALIZE first prompts the user that it is ready to initialize a disk in a given drive. The drive number is passed in the A accumulator from the CLI. If the user responds "Y", the initialization process begins. If the response is not "Y", the program returns to the CLI.

Initialization begins by writing the link sector. The last two bytes of this sector are set to point to track 1, sector 1 (the start of the

free-space). The remainder of track 0 (directory) is cleared. The rest of the sectors on disk (tracks 1 and above) are linked together into a free-space chain. The first two bytes of each sector point to the next sector. The third and fourth bytes point back to the previous sector. The remainder of the sector is cleared. The forward pointer of the last sector on the disk points to 0,0. The sectors need not be contiguous. A table called TBL is used to initialize the disk to an interleave pattern determined to provide the fastest access times for files. This table is entered with a logical sector number, it returns the physical sector number on the given track. The subroutine GETSC performs the lookup in TBL. The subroutine WRTBLK is used to write data sectors onto the disk. If a disk error occurs, the initialization process is aborted with an error message that indicates the sector and track of the bad spot on the disk.

Called by: CLI

Calls: GETSC, WRTBLK

GETSC

This subroutine converts a logical sector number into a physical sector number. using an interleave table TBL.

Called by: INITIALIZE

Tables: TBL

WRTBLK

This subroutine writes a data sector onto the disk. An internal control block FCBSPC is used to direct the writing. Errors are trapped to WRTERR which outputs the track, sector, and error numbers in hex.

Called by: INITIALIZE

LINK Transient Command

This routine is used to set the pointers in the link sector to point to a desired file. This is typically a CP/68 system file, but it can be any binary file which is to be bootstrapable.

The first step is to prompt the user for a file name. The name is parsed to be sure that it is a valid, non-ambiguous file name. LINK then looks up the file name in the disk directory. If found, the first and last tracks and sectors are recovered from the directory and placed in the internal SYSFCB. If the file is not found, or if it was not a valid filename, LINK gives an error message and returns to the CLI. If found, the link sector of the disk is read, the pointers updated to those from

the directory, and the link sector is re-written to the disk. It then returns to the CLI

Called by: CLI

PIP Transient Command

This routine handles all forms of data manipulation from one device or file to any other device or file. PIP (Peripheral Interchange Program) handles such diverse tasks as file concatenation, disk copy, binary-to-MIKBUG conversion, etc. It has several sections which perform different operations.

DEVTAB

This table lists the various devices supported by CP/68 and has the addresses of handlers for them. This differs from PDTAB and EQTAB in that PIP uses character-by-character I/O, not line-oriented I/O as used in the rest of CP/68. Each entry in DEVTAB consists of 11 bytes. The first three bytes are the device name. The rest of the entry is a set of four addresses, each two bytes. The first address is a handler for device "open". The second address is a handler for device "close". The third address is a handler for device character read. The last address is a handler for device character write. If one of these addresses is zero, it indicates that the device cannot perform the desired operation. (Read from line printer, etc.) The end of the table is marked with a zero.

CHARACTER-ORIENTED DEVICE HANDLERS

These short subroutines handle the various devices under CP/68 so that they can provide character-by-character I/O. The "open" routines check that the device is capable of the desired operation. The "open" for the lineprinter automatically emits a form-feed (OC hex). The "close" routines for devices like the paper-tape punch automatically add control-D (O4 hex) to indicate end-file. The "read" routines for devices like the paper-tape reader and teletype check for control-D and return end-file status when it is found. All the routines are called with the address of a control block (one of the internal FCBs) in INHND for input and OUTHND for output.

DLKUP

This subroutine performs the lookup of a device name in DEVTAB. The address of the device name is passed in the index register. Carry-set on return indicates that the name was not found. Carry-clear indicates that the name was found and the address of the table entry is in the index

PIP itself

The main body of PIP parses the command lines and determines the necessary processing. The first step is initialization of the input and output FCBs. The device is assumed to be disk 0 unless otherwise specified. A blank is placed in the first character of the filename field. PIP next processes the left side of the command line. If a number is found, it is checked for validity as a drive. If an error is found, PIP reprompts for another command. Otherwise, the program tries to complete the file name parsing. A valid filename is parsed into the input OUTFCB. If no number was found, the entry might be a file on drive 0 or a device name. DLKUP is used to check whether the entry is a device name. If not, the entry is formatted as a file name; if it is, the device name is placed in OUTFCB. The address of the device handler is placed in OUTHND.

PIP next looks for a slash (/) that indicates the presence of switches. If a slash is found, the switches are checked and appropriate flags set. Switches are separated with slashes. Parsing of the output portion of the command line ends with the equals sign in the line.

The output portion of the command line could also be a drive specification only (number followed by a colon). If this is the case, a flag is set to indicate that a form of disk-copy is requested (PIPFLG).

The input portion of the command line (right of the equals sign) is parsed much the same as the output side, except that no switches are allowed. Ambiguous filenames (with wild-cards) are allowed if in a file-copy (PIPFLG set).

Once the command line has been parsed, the transfer of data can begin. The character-oriented device handlers are used to move data from the input device to the output device. Upon completion of the transfer, PIP checks the command line for a comma or other delimiter on the right. If found, this indicates another input source is to be concatenated. The source specifier is parsed and if valid, its data is also transfered.

I/O errors during transfer are indicated, but the processing continues. Note also that since transfers are buffered by the handlers, there will be a one line lag between input and output.

Upon completion of data transfer, PIP reprompts for a new command line after issuing a "DONE" message. An ESCAPE character will allow return to command level.

DTDCPY

This routine is called when PIP determines that the form

drive: = drive:

has been commanded. This routine performs a direct sector-for-sector copy from one disk to another. A prompt is issued which indicates the direction of copy and gives the user a chance to correct mistakes in the command.

FILCPY

This routine is called when PIP determines that the form

drive:= drive: wildcard name

has been commanded. The wild-card filename is moved into temporary storage TMPBUF. The disk directory is searched for filenames which match the name. If a match is found, the name is echoed and the user is prompted for a response. If the response is "Y", the file is copied. After the copy, or if the response was not "Y", further matches are sought in the directory. Each match is prompted in turn until the directory is exhausted.

HEXFRM

This routine converts the internal binary-format of program files into MIKBUG or hexadecimal format. It is called when the H switch (HFLAG) is set by PIP.

BINFRM

This routine converts MIKBUG or hexadecimal-format data into the internal CP/68 binary format. It is called when the B switch (BFLAG) is set by PIP.

SECURITY Transient Command

This routine is used to change the access code of a given file. It first parses the filename passed to it by the CLI. This name is looked up in the disk directory. If not found, an error message is returned and the CLI is resumed. If the file is found, its directory information is retained in the internal SYSFCB. The command line is parsed for a comma followed by a number. If found, and if the number is less than 256, the number is placed into the directory access entry of the named file and the directory is updated. If an error was found, the program simply returns to CLI without changing the directory.

Called by: CLI

SET Transient Command

This routine processes the SET command. It manipulates the CONsole and LPT parameters in base-page. The set of legal parameter names is contained in the table SETAB. Each entry consists of 4 bytes. The first two bytes are the 2-character name of the parameter. The second two bytes are the address of this parameter. Two bytes are used because not all versions of CP/68 place the parameters in base-page. The subroutine SETSRC searches this table for the parameter whose name is contained in the index register. Carry-clear indicates that the parameter was found in the table and that its address is in the index register. Carry-set indicates that the name was not found.

The normal case of SET is PAR=number. In this case, the value of "number" is stored at the address recovered from SETAB based on "PAR". There are two special cases in SET. If PAR=DX, the appropriate values are not numbers but "F" or "H" (full or half-duplex). SET checks for these responses and stores FF hex into the DX parameter address for half-duplex and 00 hex for full-duplex. If PAR=PS, the appropriate values are "Y" or "N" (pause Yes or No). SET checks for these responses and stores FF hex into the PS parameter address for pause-off and 00 hex for pause-on.

Called by: CLI

STATUS Transient Command

This routine prints out the present state of logical/physical device assignments. It is called with the address of the physical device table (PDTAB) in the A and B accumulators. It works by taking the device name of an entry in the table and looking at its two address pointers. If they are the same, the device has not been re-assigned and so it can be printed as

DEV = DEV

If the pointers differ, it indicates that a re-assignment has been

done. PDTAB is searched for an entry whose second address pointer matches the first address pointer of our given entry. When found, its device name is the one to which the given device has been re-assigned. Therefore, if DEV1 is the given device name, and DEV2 is the name of the entry whose second address matched DEV1's first address pointer, STATUS prints

DEV1 = DEV2

STATUS performs this operation for all devices in PDTAB and then returns to CLI.

Called by: CLI

FORMAT Transient Utility

Those versions of CP/68 which utilize soft-sectored disks require a program which writes the necessary format data onto new diskettes. This information must be on the disk prior to initialization. It usually needs to be written only once.

The FORMAT program consists of three parts: the driver, the track-build subroutine, and the track-write subroutine. The driver and track-build sections are the same for all hardware (on 5-inch disks using 128-byte sectors). The track-write section varies for different hardware configurations.

DRIVER

This routine gets a drive number from the user. It checks this number for validity and issues another prompt to the user. The second prompt allows the user to change disks or to abort the formatting process. The rest of the driver is a loop which calls TRKBLD and then TRKWRT for each track on the disk.

TRKBLD

This routine builds an image of an entire formatted track in memory (TRKBUF). TRKBLD assumes 128-byte sectors, 18 sectors per track, and a Western Digital 1771 disk controller. The track format is:

```
GAP
       8 bytes of FF hex
GAP
       7 bytes of FF hex
                              sector starts here
SYNC.
       4 bytes of 00 hex
ID-MARK 1 byte
                of FE hex
TRACK # 1 byte (track number)
       1 byte
                of 00 hex
SECTOR
       1 byte (sector number)
       1 byte of 00 hex
                of 00 hex (128 bytes)
LENGTH
       1 byte
CRC
       1 byte
                of F7 hex
GAP
       11 bytes of FF hex
SYNC
       6 bytes of 00 hex
D-ADDR 1 byte
                of FB hex
DATA
       128 bytes (00 hex)
CRC
       1 bytes of F7 hex
PAD
       1 bvte
                of FF hex
                              end of sector
```

(repeat for 18 sectors)

GAP 400 bytes of FF hex

Track numbers are set by the driver in a location called TRACK. Sector numbers are set in a location called SECTOR. TRKBLD needs at least 3400 bytes for its track image.

TRKWRT

This subroutine is called by the driver to transfer the track image built by TRKBLD to the disk. TRKWRT must position the desired drive to the desired track. The drive number is found in the CP/68 location VALUE. The track number is found in TRACK. After positioning the drive, TRKWRT must do a track-write operation. The exact mechanism of this operation depends upon the hardware in use.

Random-access files

This section discusses the random-file support package provided with the CP/68 operating system. You can link it to STRUBAL+ or assembly programs which run under CP/68 and which will manipulate random-access files.

WHAT ARE RANDOM-ACCESS FILES?

Random-access files are a special type of file structure. There are two major differences between the normal CP/68 sequential file and the random-access file:

- 1. Random-access files can perform both input and output operations on an open file. Sequential files are opened for input or output but never both.
- 2. Random-access files can be arbitrarily positioned to locations within the file. Sequential files can be positioned to their origin via the REWD system call, but they cannot be positioned to other locations without reading or writing between the starting position and the desired position.

Random-access files are actually a special type of sequential file. The random-access file has a data structure written into it which facilitates positioning to arbitrary locations.

PHYSICAL AND LOGICAL RECORDS

There are two terms which must be differentiated in order to explain the functioning of random-access files. The first of these terms is physical record. A physical record is the block of data treated as a unit by the storage device being used.

In the case of floppy-disks, the physical record is also called sector because it is written (or read) out as a single unit. CP/68 allows the user to read and write arbitrary sectors with the IOHDR system call. Thus, random-access at the physical record level is provided in CP/68. The size of a physical record, however, is fixed by the hardware. This imposes severe restrictions on the user, whose data may not fit in the required record size. The user desires control over the size of record. It is desirable to vary the record size to fit the application. This variable-sized record is referred to as a logical record. The logical record does not depend on hardware; it is under program control. The

manipulation of logical records (hereafter simply called records) is done by the routines described in this manual. The routines in this package must convert the user's descriptions of logical records into internal descriptions in terms of physical records.

ENTRY POINTS IN THE RANDOM-ACCESS PACKAGE

There are seven entry points in this package.

1.	CREATE	build a new random-access file
2.	ROPEN	open an existing random-access file
3.	RCLOSE	close an open random-access file

4. RREAD read a byte from the current position of a random-access file

5. RWRITE put a byte into the current position of a random-access file

6. POSITION move the random-access file pointer to the start of a desired record.

7. EXPAND add new records to an open random-access file

User packages may link with these routines by using their names as EXTernals. Alternatively, a vector table is provided at the start of the random-access package which has jumps to each of the routines in the order given above. Each routine is called with the address of an FCB (File-control block) in the index register. The RREAD routine returns the byte just read in the A accumulator. The RWRITE routine is passed the byte to be written in the A accumulator.

THE RANDOM-ACCESS FILE-CONTROL BLOCK (FCB)

The file-control block (FCB) used with random files has five additional data fields appended to it, compared to the normal FCB as described in the CP/68 Advanced User's Guide. They are:

FCBRNM

This 2-byte field holds the number of records contained in the file. It must be set by the user when CREATE is called. It is set by the system on ROPEN. There is a maximum for this value. based on the sector size of the floppy disks in use, and given by the relation

MXRNUM = 20 * (SECSIZ-4)

where SECSIZ is the number of bytes in a disk sector. If SECSIZ=128, this value becomes 2480. For 256-byte sectors, the maximum is 5040 records.

FCBRSZ

This 2-byte field holds the number of bytes in each logical record. The user must set it when CREATE is called. It is set by the system on ROPEN. The record size can be as small as one byte or as large as 65535 (FFFF hex) bytes. It is recommended that record sizes be kept fairly large--there is a 3-byte overhead for each record in the file.

FCBRCD

This 2-byte field holds the record number representing the current file position. The system initializes it when the file is opened (the first record number is 1). The user must set this field before POSITION is called.

FCBPOS

This 2-byte field holds the present record pointer of the current file position. The system intializes it when the file is opened. It gives the location within the current record that data will be read from or written into. As data is read or written, FCBPOS is incremented until FCBRSZ is reached. At this point , FCBRCD is incremented and FCBPOS reinitialized. Thus, any byte in the file is addressed by its record pointer (FCBRCD) and its position within the record (FCBPOS).

FCBRTB

In order to rapidly address a record within a file, the randomccess package builds a table of addresses at the time that the file is opened. This table is built in the FCB of the file and occupies 120 bytes. The table consists of a 2-byte entry for each sector of the random-access file index. Hence, the table supports up to 60 index sectors per file. This leads to the limitation on FCBRNM.

The following EQUates will address the new FCB fields when used like the EQUates defined for the other FCB fields.

FCBRNM EQU 42 FCBRSZ EQU 44 FCBRCD EQU 46 FCBPOS EQU 48 FCBRTB EQU 50

Note that the FCB for a random-access file must be 170 bytes long. (The sequential-file FCB required only 42 bytes).

DATA STRUCTURES IN RANDOM-ACCESS FILES

Every random-access file built by CP/68 contains a data structure termed an index. This index is itself a sequential file containing pointers to the data records contained in the file. Thus, each random-access file is two sequential files: an index and the data record.

The file's first four bytes contain the values of FCBRNM and FCBRSZ--which describe the size of the file and data records. The index follows these two values. This index consists of a 3-byte entry; the first byte represents the track on which the data record begins, the second represents the sector on which the data record begins, and the third byte represents the position of the record's first data byte within the sector. The pointers are written sequentially as their data records are allocated during the CREATE processing. The end of the index is marked by a pointer containing all zero values. The index is padded with nulls (zero values) to fill out the last sector.

Data records begin on the next sector of the random-access file. They are simply a sequence of bytes FCBRSZ long and initialized to zero during the CREATE processing. There are no end-of-record marks; the end of one record is contiguous with the start of the next sequential record. Reading or writing past the end of a data record will automatically spill over onto the next data record. The RREAD and RWRITE routines will update the pointers FCBRCD and FCBPOS to indicate the current file position. The POSITION routine can be called at any time to move the file pointers to the start of a desired record.

RANDOM-ACCESS FILE ROUTINES

CREATE

This routine builds the structures for a new random-access file on disk. The user must provide a random-access FCB (170 bytes long) with the drive, filename, record size, and number of records set up. (FCBGDT=DSK, FCBDRV, FCBNAM, FCBRSZ, FCBRNM) A new file will be created with an index for each record. Each record will be cleared to zero. The filetype of the file will be set to 02. All random-access files disable space-compression. CREATE is called with the address of the FCB in the index register. It returns status information in FCBSTA of the user FCB. CREATE destroys the contents of the A and B accumulators and the condition codes. It leaves the index register intact. A CREATEd file is not open—it must be opened by a call to ROPEN before it may be accessed. CREATE may take a long time to build a large random-access file, since it must write the index as well as each data record in the file.

ROPEN

This routine prepares a previously CREATEd file for use. It is called with the address of a user FCB in the index register. The drive and filename must be set up by the user (FCBGDT=DSK, FCBDRV, and FCBNAM). It reads FCBRNM and FCBRSZ from the file and places them in the user FCB (which must be 170 bytes long). It also stores the filetype (must be 02), access code, first track and sector (T/S), last T/S (FCBTYP, FCBACS, FCBFTS, and FCBLTS) fields into the FCB. The file pointers (FCBRCD and FCBPOS) are initialized to point to the first record in the file. The ROPEN routine also reads the file index, building a table (FCBRTB) in the FCB containing the track and sector of each sector of the index. All unused table entries are cleared. The process of building this table may take many seconds for a file with many data records. ROPEN destroys the A and B accumulators and the condition codes; it returns the index register intact. Error status is returned in the FCBSTA field of the user FCB.

RCLOSE

This routine closes the file described by the user FCB whose address is passed in the index register. Any pending output is completed before the FCB is de-allocated. RCLOSE should only be used on random-access files. (type=02) It destroys the A and B accumulators and condition codes; the index register is returned intact. Error status is returned in the FCBSTA field of the user FCB.

RREAD

This routine reads a data byte from a random-access file. It is called with the address of the user FCB in the index register. The data byte read is returned in the A accumulator. RREAD reads sequentially from the current file position defined by FCBRCD and FCBPOS. If the last operation performed on the file was writing, RREAD will finish that operation before reading. Subsequent calls to RREAD will access sequential data bytes. RREAD destroys the B accumulator and the condition codes; the index register is returned intact. Error status is returned in the FCBSTA field of the user FCB. If a read error occurs, RREAD will return a null.

RWRITE

This routine writes a data byte into a random-access file. It is called with the address of the user FCB in the index register and the byte to be written in the A accumulator. The data byte will be written at the current file position defined by FCBRCD and FCBPOS. Subsequent calls to RWRITE will write sequential data bytes. RWRITE destroys the A and B accumulators and the condition codes; the index register is returned intact. Error status is returned in the FCBSTA field of the user FCB.

POSITION

This routine moves the current file position to the start of a desired record in the file. It is called with the address of a user FCB in the index register. The desired record is set in the FCBRCD field of the FCB. POSITION will initialize FCBPOS when the desired record is found. If the last operation performed on the file was writing, the last write will be finished before the file position is changed. POSITION destroys the A and B accumulators and the condition codes; the index register is returned intact. Error status is returned in the FCBSTA field of the user FCB.

EXPAND

This routine adds new records to an existing, open, random-access file. EXPAND is called with the address of a user FCB in the index register. The number of new records desired is set in the FCBRCD field of the user FCB. The new records will have the same size (FCBRSZ) as the others in the file. EXPAND will close the file after the new records have been appended. None of the old records will be affected by the EXPAND process. The new records are added after all the old ones. A file may be EXPANDed many times. EXPAND destroys the A and B accumulators and the condition codes. The index register is returned intact. Error status is returned in the FCBSTA field of the user FCB. Adding many records to a file may take a long time.

NEW ERROR CODES FOR RANDOM-ACCESS FILES

The random-access routines trap all the same file errors as the sequential routines do. In addition, they trap four new errors that are specific to random-access operations. They are:

- OB BAD RECORD SIZE PARAMETER
 The value specified for FCBRSZ was zero.
- OC BAD RECORD NUMBER PARAMETER
 The value specified for FCBRNM was zero or greater

than the MXRNUM for the system sector size.

- OE BAD FILE TYPE

 The file specified is not random-access type. (02)
- OF BAD POSITION PARAMETER
 The value specified for FCBRCD lies outside the file.
 (The last data byte of the last data record has been written or read.)

I.

Random-access files contain track/sector information in their indices. Hence, rearranging their sectors on the disk will corrupt the indexing and destroy the file. Disks which have random-access files on them should not be copied using the packing (drive:=drive:*.*) PIP command. Such disks should be copied exactly, sector-for-sector, using the nonpacking PIP copy command. (drive:=drive:) Using PIP to transfer a random-access file from disk to disk will corrupt the new file, making it worthless.

II.

The FCBDTT field of the FCB, which was used in sequential file handling to specify input or output, is under system control when working with random-access files. It should not be used by the programmer.

The following program illustrates the use of random-access file routines under CP/68. It allows exercise of all the CP/68 random-file operations.

```
NAM TESTRND
 EXERCISE PROGRAM FOR RANDOM-ACCESS FILES IN CP/68
# 101
      OPEN FILE (ONLY ONE FILE OPEN AT A TIME)
# 1C1
      CLOSE FILE
# IBI
      BUILD A NEW RANDOM-ACCESS FILE
 'R' READ FROM CURRENT POSITION IN FILE
       (END ON CARRIAGE-RETURN IN FILE)
# 'W'
      WRITE TO FILE AT CURRENT POSITION
       (END WITH CARRIAGE-RETURN)
* 'P' POSITION FILE TO DESIRED RECORD
 'E' EXPAND CURRENTLY-OPEN FILE
       JMP START
DEFINE RANDOM-FILE EXTERNALS
       EXT CREATE
       EXT ROPEN
       EXT RCLOSE
       EXT RREAD
       EXT RWRITE
       EXT POSITION
       EXT EXPAND
* DEFINE TEXT BUFFER FOR OUTPUT
                    80 CHARACTERS FOR LINE BUFFER
BUFFER RMB 80
BUFEND FCB $0D
                   FORCE C.R. ON LINE
BUFPNT RMB 2
                     BUFFER POINTER STORAGE
# DEFINE CP/68 EQUATES
       BASEQU
       FCBDEF
FCBRNM EQU 42
FCBRSZ EQU 44
FCBRCD EQU 46
FCBPOS EQU 48
# LOCAL RANDOM-FCB BLOCK
```

FCBLK RMB 2

```
FCC 'DSK'
       RMB 2
       FDB SECBUF
       RMB 162
SECBUF RMB 256
* SET OF PROGRAM PROMPT AND ERROR MESSAGES
M1
       FCC 'ENTER COMMAND: '
       FCB 4
M2
       FCC 'ENTER FILE SPECIFICATION: '
       FCB 4
       FCC 'ENTER RECORD SIZE : '
M3
       FCB 4
M4
       FCC 'ENTER NO. OF RECORDS: '
       FCB 4
M5
       FCC 'ENTER RECORD NUMBER: '
       FCB 4
M6
       FCC 'ENTER DATA: '
       FCB 4
M7
       FCC 'BAD NUMBER'
       FCB $0D
* BEGIN PROGRAM CODE HERE
START LDX #M1
                            PROMPT FOR COMMAND
       PRTMSG
       GTCMD
                            GET COMMAND
       LDX DESCRA
       LDA A O,X
                            "OPEN"?
       CMP A #'O
       BNE NEX1
                            NO
# PROCESS "OPEN" COMMAND
       LDX #M2
                            PROMPT FOR FILESPEC
       PRTMSG
       GTCMD
                            GET FILESPEC.
       LDX DESCRA
       STX CUCHAR
                            BACK UP A TOKEN
       LDX #FCBLK
       FMTFCB
                            PUT FILESPEC INTO FCB
       TST FCBSTA, X
                            ERROR?
       BEQ OPN2
                            NO
ERROR LDX #FCBLK
       PRTERR
                           PRINT ERROR MESSAGE (IF ANY)
       BRA START
                           GET NEW COMMAND
OPN2
       JSR ROPEN
                           OPEN FILE
       BRA ERROR
                            ERROR (IF ANY) AND LOOP
```

"CLOSE"? NEX 1 CMP A #'C BNE NEX2 NO # PROCESS "CLOSE" COMMAND LDX #FCBLK CLOSE FILE JSR RCLOSE BRA ERROR ERROR (IF ANY) AND LOOP "BUILD"? NEX2 CMP A #'B BNE NEX3 NO * PROCESS "BUILD" COMMAND LDX #M2 PROMPT FOR FILESPEC PRTMSG GTCMD GET FILESPEC. LDX DESCRA STX CUCHAR BACK UP A TOKEN LDX #FCBLK PUT FILESPEC INTO FCB **FMTFCB** TST FCBSTA, X ERROR? BNE ERROR YES PROMPT FOR RECORD SIZE LDX #M3 **PRTMSG** GTCMD GET VALUE LDA B RC NUMERIC? CMP B #3 BEQ BLD2 YES NUMERR LDX #M7 PRINT "BAD NUMBER" MESSAGE PRTMSG JMP START TRY AGAIN LDA A VALUE BLD2 LDA B VALUE+1 LDX #FCBLK PUT RECSIZ INTO FCB STA A FCBRSZ,X STA B FCBRSZ+1,X PROMPT FOR NO. OF RECORDS LDX #M4 PRTMSG GET VALUE GTCMD LDA B RC NUMERIC? CMP B #3 BNE NUMERR NO LDA A VALUE . LDA B VALUE+1 LDX #FCBLK PUT RECNUM INTO FCB STA A FCBRNM, X

STA B FCBRNM+1.X JSR CREATE BUILD NEW FILE JMP ERROR ERROR (IF ANY) AND LOOP CMP A #'R "READ"? NEX3 BNE NEX4 NO # PROCESS "READ" COMMAND RED1 LDX #BUFFER INIT. OUTPUT BUFFER POINTER STX BUFPNT RED2 LDX #FCBLK JSR RREAD READ BYTE FROM FILE TST FCBSTA, X ERROR? BEQ RED3 NO JMP ERROR YES LDX BUFPNT RED3 GET BUFFER POINTER STORE CHARACTER IN BUFFER STA A O.X CMP A #\$0D CARRIAGE-RETURN? BEQ RED4 IF SO, FINISH UP INX STX BUFPNT MOVE BUFFER POINTER CPX #BUFEND AT END OF BUFFER? BNE RED2 NO, LOOP LDX #BUFFER PRTMSG PRINT BUFFER CONTENTS BRA RED1 LOOP FOR NEW BUFFER RED4 LDX #BUFFER PRTMSG PRINT BUFFER CONTENTS JMP START NEX4 CMP A #'W "WRITE"? BNE NEX5 NO * PROCESS "WRITE" COMMAND PROMPT FOR DATA LDX #M6 PRTMSG GTCMD GET DATA LINE LDX DESCRA POINT TO IT WRIT1 LDA A O.X GET DATA BYTE LDX #FCBLK WRITE DATA BYTE JSR RWRITE TST FCBSTA, X ERROR? BEQ WRIT2 NO

WRIT1A JMP ERROR YES WRIT2 LDX DESCRA LDA A O,X GET BYTE AGAIN CMP A #\$OD C.R.? BEQ WRIT1A IF SO, DONE INX IF NOT, MOVE POINTER STX DESCRA BRA WRIT1 LOOP CMP A # P NEX5 "POSITION"? BNE NEX6 NO * PROCESS "POSITION" COMMAND LDX #M5 PROMPT FOR RECORD NUMBER PRTMSG GTCMD GET VALUE LDA B RC NUMERIC? CMP B #3 BEQ POS1 YES JMP NUMERR NO POS1 LDA A VALUE LDA B VALUE+1 LDX #FCBLK PUT RECNUM INTO FCB STA A FCBRCD, X STA B FCBRCD+1,X JSR POSITION POSITION FILE JMP ERROR ERROR (IF ANY) AND LOOP NEX6 "EXPAND"? CMP A #'E BNE NEX7 NO # PROCESS "EXPAND" COMMAND LDX #M4 PROMPT FOR NO. OF RECORDS PRTMSG GET VALUE GTCMD LDA B RC NUMERIC? CMP B #3 BEQ EXP1 YES JMP NUMERR NO EXP1 LDA A VALUE LDA B VALUE+1 LDX #FCBLK PUT RECNUM INTO FCB

STA A FCBRCD,X STA B FCBRCD+1,X JSR EXPAND JMP ERROR ENLARGE FILE

ERROR (IF ANY) AND LOOP

*

NEX7

JMP START

UNRECOGNIZED COMMAND

END

RANDOM-ACCESS FILE SUPPORT FOR STRUBAL+ PROGRAMS

All functions of the random-access file package are available to the STRUBAL+ programmer through procedures built into the random-access file-driver program supplied with the random-access package. This file-driver program includes all the support necessary for sequential file I/O plus all the additional random-file commands. Some of the random-file operations share the same keywords with the sequential operations. The shared keywords are:

OPEN open a file for use CLOSE close a file after use READ read data from a file WRITE write data into a file

The new set of keywords includes:

BUILD create a new random-access file

DELETE delete a file from the disk (random or sequential)

ENLARGE add records to a random-access file

LOCATE return the current file pointers of a random-access file

PLACE position a random-access file to a given record

This set of keywords provides support for all file manipulations under CP/68.

The shared keywords READ and WRITE work the same way for sequential and random-access files. Data is moved sequentially starting with the current file position. The .EOF. and .ERR. functions are used in the same way with random-access files as they were with sequential files. The shared keyword CLOSE also works the same for both types of files in CP/68. The shared keyword OPEN has the same syntax for both types of files. If a random-access file is to be opened, append ';R' to the file specification instead of ';I' or ';O' used with sequential files. This identifies the file to be opened for random-access.

Only files built for random access can be used as random-access files. Sequential files cannot be manipulated using random-access statements. A file with a filetype of 02 is a random-access file. Random-access files may be built under STRUBAL+ control using the BUILD procedure. They may be positioned to any desired record using the PLACE procedure. The current values of the file pointers may be obtained using the LOCATE procedure. Finally, records may be added to an existing random-access file through the use of the ENLARGE procedure.

BUILD procedure

This procedure is used to create a new random-access file. Such a file is defined by its file specification (drive, name, and extension), a record count, and a record size. If FNAME is a string of characters containing a valid CP/68 file designation (which does not already exist on the disk), RECNO is an integer which contains the desired number of records to be in the file, and RECSIZ is an integer which contains the desired number of characters to be in each record, then the following procedure call will build the desired file.

CALL BUILD(RFCB, FNAME, RECNO, RECSIZ)

RFCB is the name of the user-supplied file-control block (FCB). The FCB must contain 426 bytes for systems whose sector size is 256, and 298 bytes for systems whose sector size is 128 bytes. The BUILD procedure may take substantial time for a large file. The file is closed upon return from BUILD.

ENLARGE procedure

This procedure is used to add new records to an existing random-access file. The file must be already open before ENLARGE is called. ENLARGE requires the address of the file FCB and the desired number of records to be added as parameters.

CALL ENLARGE (RFCB, RECNO)

The file is closed upon return from ENLARGE. The ENLARGE procedure may take substantial time if many records are added to the file.

LOCATE procedure

This procedure returns the current file pointers of a random-access file. There are two pointers: the current record RECNO, and the current position within the record BYTNO. (These correspond to FCBRCD and FCBPOS.) LOCATE is called with the address of the file FCB as a parameter.

CALL LOCATE(RFCB, RECNO, BYTNO)

It returns two integer values containing the pointer contents.

PLACE procedure

This procedure moves the file pointers of a random-access file to a user-specified record. PLACE requires the address of the file FCB as a parameter, as well as an integer containing the desired record number.

PLACE always positions the file to the start of the desired record.

USING RANDOM-ACCESS FILES IN STRUBAL+

The following STRUBAL+ example program illustrates the use of the random-access procedures to exercise random-access files. The example is similar in function to the assembly-language example shown earlier.

- * ILLUSTRATE USE OF RANDOM-ACCESS FILES THROUGH STRUBAL+
- * ASSUME RANDOM-FILE PACKAGE AND DRIVERS LOADED

DSTRING DATA(80), RFCB(426), FNAME(30), CMD(10), TMP(1) INTEGER RECNO, RECSIZ, BYTNO

CALL INITIO

* AVAILABLE COMMANDS ARE:

* BUILD, CLOSE, ENLARGE, OPEN, POSITION, READ, WRITE

START INPUT /, 'ENTER COMMAND (B,C,E,O,P,R,W): ',%CMD XTRACT TMP=1,CMD STRING IF TMP .NE. 'O' THEN NEX1

* PROCESS "OPEN" COMMAND HERE

INPUT /, 'ENTER FILE SPECIFICATION: ', %FNAME STRING FNAME=FNAME,';R'
OPEN (RFCB) FNAME
GOTO START

NEX1 STRING IF TMP .NE. 'C' THEN NEX2

* PROCESS "CLOSE" COMMAND HERE

CLOSE (RFCB)
GOTO START

NEX2 STRING IF TMP .NE. 'B' THEN NEX3

* PROCESS "BUILD" COMMAND HERE

INPUT /, 'ENTER FILE SPECIFICATION: ', %FNAME INPUT /, 'ENTER NUMBER OF RECORDS: ', RECNO INPUT /, 'ENTER RECORD SIZE: ', RECSIZ CALL BUILD(RFCB, FNAME, RECNO, RECSIZ) GOTO START

STRING IF TMP .NE. 'R' THEN NEX4 NEX3

* PROCESS "READ" COMMAND HERE

CALL LOCATE(RFCB, RECNO, BYTNO)

PRINT /,[6], 'RECORD=', RECNO,' BYTE= ', BYTNO

* PRINT CURRENT POINTERS BEFORE READING READ (RFCB) %DATA PRINT /,[72], %DATA GOTO START

NEX4 STRING IF TMP .NE. 'W' THEN NEX5

* PROCESS "WRITE" COMMAND HERE

INPUT /, 'ENTER DATA: ',%DATA WRITE (RFCB) %DATA GOTO START

STRING IF TMP .NE. 'P' THEN NEX6 NEX5

* PROCESS "POSITION" COMMAND HERE

INPUT /, 'ENTER RECORD NUMBER: ', RECNO CALL PLACE(RFCB.RECNO) GOTO START

NEX6 STRING IF TMP .NE. 'E' THEN START

* PROCESS "ENLARGE" COMMAND HERE

INPUT /, 'ENTER NUMBER OF RECORDS: ', RECNO CALL ENLARGE(RFCB, RECNO) GOTO START

END

DELETING A FILE USING STRUBAL+

One additional procedure is contained in the new file driver program; this procedure allows STRUBAL+ programs to delete files from disk. Only unambiguous names can be used; no wildcards are allowed. The DELETE procedure requires an FCB in the user program. This FCB can be sized either for sequential files or random-access files. The file specification is passed as a string to the procedure.

CALL DELETE(RFCB, DNAME)

Care should be taken with this procedure, as once a file is deleted it is lost. There will be no prompting, unlike the DELETE command under CP/68.

MODIFICATIONS FOR DISK HARDWARE DIFFERENCES

CP/68 can be tailored for a wide variety of disk configurations. This section will describe the places which must be modified for most common hardware setups. There are three parameters which describe a disk to CP/68:

```
SECSIZ the number of bytes in a sector (128 assumed)
TRKSIZ the number of sectors in a track (18 assumed)
DSKSIZ the number of tracks on a disk (35 assumed)
```

In addition, CP/68 checks the number of drives. From 1 to 4 drives may be used. (CP/68 as described here assumes four drives.) More than four drives can be used if more space is allocated to the free-space pointer table (FRETAB) in the base-page. Two bytes are needed for each drive added.

SECSIZ

This parameter is the most important one, as it affects the buffer sizes for the sector buffers in the system. Sector buffers appear in:

```
CLI- SAVFCB, SYSFCB, SUBFCB
BOOT- BUFFER

DELETE- SYSFCB
INIT- FCBSPC
LINK- SYSFCB
PIP- INFCB, OUTFCB
SECURITY- SYSFCB
RNDFILE- RNDFCB
```

All sector buffers are sized for 128 bytes as shown. They could be enlarged to 256 bytes if necessary. Larger sectors would require extensive modification since byte counts are kept in 8-bit locations throughout CP/68.

SECSIZ also is used as a parameter in CP/68 to allow addressing of elements of a sector or to compute constants based on the sector size. Use of SECSIZ as a parameter appears in:

CLI, DREAD, Sequential File I/O, BOOT, INITIALIZE, LINK and RNDFILE

TRKSIZ

This parameter is used in the following routines:

CLI- in subroutine SEMPTY DREAD- in subroutine GETDR

INIT-

PIP- in subroutine DTDCPY

FORMAT-

The use of TRKSIZ in INITIALIZE includes the length of the sector-interleave table TBL. There must be a table entry for each sector on a disk track.

DSKSIZ

This parameter is used in the following routines:

INIT-

PIP- in subroutine DTDCPY

FORMAT-

CP/68 assumes that all disks have the same DSKSIZ.

Number of Drives in System

This parameter appears in the following routines:

CLI- at WARM3 (to initialize FRETAB)

in subroutine INICMD in subroutine DIRCMD in subroutine FMTFCB

SFIO- (mask off low 2 bits of drive number to access FRETAB)

DELETE-

INITIALIZE-

LINK-

PIP-

SECURITY-

FORMAT-

RNDFILE- (mask off low 2 bits of drive number to access FRETAB)

In all cases except SFIO and RNDFILE, the checks on drive number are used for error-detection only.

DISK HANDLING SOFTWARE

Any disk operating system like CP/68 must be modified for use on different hardware. The hardware-specific code is localized in the DRIVERS, BOOT, and FORMAT. The DRIVERS require initialize, sector-read, and sector-write capabilities for multiple drives. BOOT requires only initialize and sector-read from drive 0. FORMAT requires track-seek and track-write capability for multiple drives. Drivers for several common disk configurations are given here. They each perform the same functions-only one is needed for CP/68.

MODIFICATIONS FOR VARIOUS SYSTEM MONITOR ROMS

CP/68 makes no use of system monitor routines during its execution. As a result, any of the current "---BUG" monitors can be used with it. BIOS contains the addressing for the various I/O devices (EQTAB), which may need changing for different addressing of I/O devices. BIOS also contains an error trap for CP/68 calls (SWIs) that have an invalid function code. This trap should vector to the normal breakpoint entry in the monitor ROM. This vector is directed to E113 hex in SWIHDR. CLI also contains a vector to the monitor in its command table (CMDTAB). The EXIT command is vectored to the warm-start entry of the monitor ROM (the version shown goes to E0E3 hex). The BOOT transient contains an error trap which is jumped to in case of disk errors during boot. This vector is shown as E113 hex (like the one in BIOS).

One other modification will be necessary to use CP/68--point the SWI vector of the system to the SWIHDR entry. Some means must be found to force SWIs to be processed by SWIHDR. The BOOT process must set up the SWI vector, or else it must be set by code at the COLDST entry in CLI.

Part 7

Software Listings *******

Resident
BIOS103
CLI117
DIRECTORY140
SFI0142
ICOM driver150
Transients
ASSIGN153
BOOT156
DELETE158
INITIALIZE162
LINK165
PIP168
SECURITY184
SET187
STATUS189
RANDOM191
Hex dump of resident code
Hex dump of transients209
Southwest Technical Products drivers211
BOOT215
INITIALIZE218
FORMATTER221
Smoke Signal Broadcasting drivers224
BOOT227
INITIALIZE229
FORMATTER232
Percom Data Company
Single-sector read and write236
INITIALIZE238
BOOT

CLR A LDA B 0, X GET REQUEST CODE CMP B #53 VALID RC? BLS *+5 YES * JMP #E113 NO	ASL B *2 ROL A * * CALCULATE THE ADDRESS OF THE * APPROPRIATE PROCESSING ROUTINE	* BSR START * START TSX ADD B 1, X		LDX 0, X ADD B 1, X ADC A 0, X * TSX STA A 0, X STA B 1, X	* LDA A 2, X TAP LDX O, X INS INS	* JSR O, X CALL PROCESSING ROUTINE * TSX INC 6, X BNE *+4 * INC 5, X RTI RETURN TO USER	ISTER EGUAT 2 3 4 4 5 7 9 8 NG ROUTINES
0006 4F 0007 E6 00 0009 C1 35 000B 23 03 000D 7E E113	0010 58 0011 49	0012 8D 00 0014 30 0015 EB 01	CB 89 E7	0021 EE 00 0023 EB 01 0025 A9 00 0027 30 0028 A7 00 002A E7 01	002C A6 02 002E 06 002F EE 00 0031 31	0035 AD 00 0035 30 0036 AC 06 0038 26 02 003A AC 05	
0061 0063 0063 0065 0065	0068 0068 0070 0071	0075 0075 0076 0077 0078	0080 0081 0082 0083 0084	0086 0087 0089 0089 0090 0091	0093 0095 0097 0098 0098	0100 0102 0103 0104 0105 0106	0112 0113 0115 0115 0117 0118 0120
Z	* RCB EQU/ * RCBEAT EQU RCBSDT EQU RCBSTA EQU PCBNT'S EQU	RCBDBA EQU 7 * STATUS EQU/S* * STATUS EQU/S* BUSY EQU #80 RCB BUSY STATUMEN FOUR #7F AND RISY (ANDRE	N ENT EQTAB N ENT POTAB * BASEQU DESCRA EQU #20 DESCRIPTOR ADDRESS(2	DESCRC EQU \$22 CUCHAR EQU \$23 RC EQU \$25 CLASS EQU \$26 VALUE EQU \$27 FCBCHN EQU \$29 FRETAB EQU \$28	0033 BMEM EQU \$33 START OF TRANSIENT AREA(2) 0035 EMEM EQU \$35 END OF TRANSIENT AREA (2) 0037 CMEM EQU \$37 NEXT AVAIL TRANSIENT AREA (2) 0039 BS EQU \$38 BACKSPACE CHAR 0036 DL EQU \$38 DELETE LINE CHAR 0038 DPC EQU \$38 DEPTH; LINES/PAGE 0037 CM \$39 DECTH; LINES/PAGE	MU FRU \$3E NL FRU \$3E TB FRU \$3F DX FRU \$41 EU \$41 PS FRU \$42 ERU \$42 LDP FRU \$43 I DPCNT FRU \$45	# COLDSTART ENTRY * COLDSTART ENTRY * COLDSTART ENTRY * SWIHDR SWI INP * THIS IS THE ONL * SWIHDR TSX OS LDX 5, X
<u>o</u>	0000	700					
0000 0000	0000 0000 0000 0002 0000 0005	0000 0000	, ,,,,				0000 7E 0000 7E 0000 7E 0000 7E

UXH, X UXH+4, X UXL, X UXL+4, X

σσσσ

9H00

TSX LDA STA STA STA

A PSHXA

STA STA INX DEC BNE

009E

2, × 0, ×

* PSHXA

M M

86 8

DES DES TSX LDA

@PSHX

UXH+4, X UXH, X UXL+1, X UXL, X

~ < < <

000AB 000AB 000AF 000AF

TSX LDA STA LDA STA

PULX: PULL

ePULX

\$

Œ

A

* XABX: EXCHANGE A, B AND X

exABX

RTS

A UXH, X A B UXL, X @TABX+1 A STAB

TST LEGA X

EF

008C 008C ×

* PSHX: PUSH

×

* TABX: TRANSFER A, B TO

STA A UA, X STA B UB, X

STAB

9,00

A7 E7

007B A UA, X A UXH, X A UB, X A UXL, X

TSX LDA STA STA STA

007E 007F

@TABX

																									1	ŝ																												
*	PSHAI DES		DES	DES	DES		LUA B #V				:	DEC B			LDA B #5			LDA A UC, X	Œ	4 K L C			RTS	*		* FULALL: PULL ALL REGISTERS	**************************************	100 B #50	1	PULAC LDA A UC+7, X	Œ			* BNE PULAC	***********	1		A URL,	DEX		ב ב ב		SNI	SNI	INS		*	n -	* *	* TXAB: TRANSFER X TO A, B		TXAB TSX	LDA A UXH, X	B UXL,
							60		5	8) }		F8		05			25	Š.		α)						1 0	?	60	22		9	D	8		ဗ	8		٥	0												05	9
	96	5 6	5.6	9.6	34		90	Š		4		ξ 2 G			8	ဓ္တ		اع		2 6			39				ć	3 2				စ ဗိ	T d		3		8		8	- 5 6		31	31	31	31	31	Ş	'n						
	0031	1000	1000	000	0041	1	0042	100	0045	0047	0049	0040	004B		0040	004F		0020	0052	000	0000	3	0058			•		0000					0061	7900	0064				006A	900							0.00							
04400	0124	010	0125	0127	0128	0129	0130	0131	0133	0134	0135	0136	0137	0138	0139	0140	0141	0142	0143	44.0	0145	0147	0148	0149	0150	0151	0152	0150	0155	0156	0157	0158	0159	0160	0161	0163	0164	0165	0166	0100	0168	0170	0171	0172	0173	0174	0175	0178	01//	0179	0180	0181	0182	0183

କାର୍କ ର ଅଟି [.] ୪୪୪ ଅ	ee an iii a e .	* LDA B UXH, X SBC B #00 BRA STAUXH * * SUBBX: SUBTRACT B FROM X ESUBBY TSX BRA @SUB * * INDEX: X:=X+A*B *	# NUL8: A,B:=A*B # MUL8: A,B:=A*B # TSX # MUL8: A,B:=A*B # TSX # STA B UB,X STA B UB,X STA A UA,X TPA TST B # TST B # TST B
20 88D 20 80D 80D 80D 80D 80D 80D 80D 80D 80D 80	100 E6 20 20 100 E6 20 20 20 20 20 20 20 20 20 20 20 20 20	20 CC2 20 20 EE 6	0111 8D 38 0113 37 0115 32 0116 30 0117 20 B3 0117 20 B3 0110 E7 03 0110 E7 03 012 07 0121 5D
0306 0307 0308 0309 0311 0312 0313 0314 0315	0319 0320 0321 0323 0324 0326 0328 0330 0330	0335 0335 0337 0337 0337 0341 0342 0345 0346 0346	0349 0350 0351 0352 0354 0355 0357 0358 0360 0361 0362 0363 0363 0363
PULXA LDA B 8, X STA B 10, X DEX DEC A BNE PULXA INS INS * RTS * ADDXAB: ADD X TO A, B	# READAR TSX BSR EAABX+1 BSR EAABX+1 # RADDABX: ADD A, B TO X # CADDAB A UB, X LDA A UB, X LDA B UA, X ADDAB ADD A UXL, X CTA A UYL, X CTA A UYL, X	ADC B UXH, TAUXH TPA STA B UXH, TST UXL, X ESTZ BEQ TESTZA AND A ##FB ESTZA STA A UC, X RTS	* * ADDAX: ADD A TO X * GADDAX TSX LDA A UA, X ADDZ LDA B #00 * ADDBX: ADD B TO X * ADDBX: ADD B TO X * ADDBX TSX BRA ADDZ LDA A UB, X BRA ADDZ * SUBXAB: SUBTRACT X FORM A, B
00b5 E6 08 00b7 E7 0A 00b9 09 00bA 4A 00bB 26 F8 00bB 31 00bF 39	0000 30 0001 8D C6 0003 8D 03 0005 8D 02 0007 30 0008 A6 03 000C AB 06 000C AB 06	E9 07 6D 27 84 84 87 39	000E 30 000F 04 00E1 C6 00 00E3 20 E7 00E5 30 00E6 A6 03 00E8 20 F7
0245 0246 0247 0247 0249 0250 0251 0252 0253 0254 0256	0259 0259 0261 0261 0263 0265 0265 0268 0269 0269 0270	0273 0274 0275 0276 0278 0280 0281 0283 0283 0284 0285	0.288 0.289 0.291 0.292 0.294 0.296 0.300 0.300 0.300 0.300 0.300 0.300

GET ARGUMENTS STACK INPUTS		REPLACE STACK POSITION RECOVER POINTER POSITION STORE REMAINDER STORE QUOTIENT
* @DIV16 TSX LDA A UA, X LDA B UB, X LDA B UB, X LDX UXH, X PSH B PSH A PSH A PSH A PSH A FCB 5 DES TSX	LDA A #1 LDA A #1 LST 1, X BM1 D1V153 * ASL 2, X ROL 1, X LDA A #17 LDA A #17 LDA A 3, X CLR 4, X CLR 4, X BCC D1V165 * ADD B 2, X BCC D1V165 * CLC BRA D1V167 * CLC BRA D1V167 * CLC BRA D1V167 * CLC * ADD B 2, X ROD B 3, X ROD B 2, X ROD B 2, X ROD B 3, X ROD B 2, X ROD B 2, X ROD B 3, X ROD B 2, X ROD B 2, X ROD B 3, X ROD B 3, X ROD B 2, X ROD B 3, X ROD B 2, X ROD B 2, X ROD B 3, X ROD B 3, X ROD B 2, X ROD B 2, X ROD B 3, X ROD B 3, X ROD B 2, X ROD B 3, X ROD B 3, X ROD B 3, X ROD B 3, X ROD B 2, X ROD B 3, X ROD B 3, X ROD B 3, X ROD B 3, X ROD B 2, X ROD B 3, X ROD	* INS INS INS INS INS INS INX
015F 30 0160 A6 04 0162 E6 03 0164 EE 05 0165 37 0167 36 0167 36 0168 05		31 31 31 31 32 33 33 33 33 33 33 33 33 33 33 33 33
0428 0429 0430 0431 0433 0433 0434 0436 0436 0438	0440 0441 04443 04444 04443 04446 0446 0455 0456 0456 0466 0466 046	0474 0475 0477 0477 0477 0489 0481 0483 0483 0483 0485 0486
** * MUL16: A, B, X:=A, B*X * MUL16: LDA A #16 36 175 175 175 187 187 187 188 189 189 180 180 180 180 180	## MUL16L BCC MUL16S EB 04	07
0124 86 0126 36 0127 3 36 0129 5 4	2000 01334 01334 01338 01338 01334 01334 0134 0135	
0367 0368 0368 0370 0371 0372 0373 0374	0.378 0.338 0.3382 0.3882 0.3883 0.3884 0.3885 0.3885 0.3885 0.3886 0.4886 0.4886 0.4886 0.4886 0.4886	0414 0415 0415 0416 0417 0418 0420 0422 0423 0424 0425

TERMINATOR? YES	MATCH? NO	×	×	x, 1,	X ALL DONE? ND	SET CC SET B	PARE STRING	SAME PARMS AS CMPC		*	TERMINATOR?		TERMINATOR?	YES	WILD CHARACTER? Yes	MATCH? NO	× ::	×	< 2
LDA B O, X CMP B #\$04 PUL B BEQ CDONE		TSX INC PARM1+1, X BNE CMP1			INC PARMZ, DEC B BNE CMPO	TSX TPA STA A UC, X STA B UB, X	RTS COM	SAM	TSX LDA B UB, X		CMP A #4 BEQ CDONE	TSX LDX PARMZ, X		PUL B BEQ CDONE	CMP A #72 BEQ CMW1	CMP A 0, X BNE CDONE	TSX INC PARM1+1, X BNE *+4	INC PARM1, X INC PARM2+1, X	THE PERMIT
	* *	,		* CMP1	MP 2	CDONE	CMMC:	* * :	ecmuc *	○ # ○ # ○ *		*		*		*	π 1.	* *	
01E4 E6 00 01E6 C1 04 01E8 33 01E9 27 14	28 H	01EF 30 01F0 6C 0A 01F2 26 02	9	90 70 70	01FA 6C 0B 01FC 5A 01FD 26 D9	01FF 30 0200 07 0201 A7 02 0203 E7 03	0205 39		0206 30 0207 E6 03	E.	020B A6 00 020D 81 04 020F 27 EE	0211 30 0212 EE 0B	0214 37 0215 E6 00 0217 C1 04	0219 33 021A 27 E3	021C 81 3F 021E 27 04	0220 A1 00 0222 26 DB	0224 30 0225 6C 0A 0227 26 02	0229 6C 09 022B 6C 0C	3
0550 0551 0552 0553	0554 0555 0556 0556	0558 0559 0560	0562 0562 0563	0564 0565 0566	0567 0568 0569 0570	0571 0572 0573 0574 0575	0576 0577 0578	0579	0582 0583 0583	0584 0585	0587 0588	0589 0590 0591	0592 0593 0594	0595 0596 0596	626 0598 0598	0600 0601 0602	0603 0604 0605 0606	0607 0608 0609 0610	*****
TZ FIX UP ZERO FLAG FROM, TO ON STACK			GET COUNT	GET CHAR	MOVE CHAR	DONE? Yes						O STRINGS	PARM1, PARM2 ON STACK, B=COUNT ON MATCH PARMS=NEXT CHAR POSITION	PAKMS≕LASI CHAR PUSITION			GET A CHAR TERMINATOR? YES		
UMP TESTZ FROM	B=COUNT ON RETU	EQU 9 EQU 11	TSX LDA B UB, X	LDX FROM, X LDA A O, X	TSX LDX TO, X STA A O, X	TSX DEC B BEQ MOVC3				INC TO, X BRA MOVC1	RTS	COMPARE TWO STRINGS	PARM1, PARM2 ON STACK, ON MATCH PARMS=NEXT C	ON MISMAICH			LDX PARM1, X LDA A O, X CMP A #\$04 BEQ CDONE	TSX LDX PARM2, X PSH B	
* * * 0 0 CC		₽	# @MOVC	* *0vc1		* *	*		# # #		0VC3	* * *		* * 0		ω	0 ₩ 0 *		
01B3 7E 00D7		01B6 0009 01B6 000B	01B6 30 01B7 E6 03	01B9 EE 09 01BB A6 00	01BD 30 01BE EE 0B 01C0 A7 00		26 26 36	ပွဲ ပ	26 26	01D0 6C 0B 01D2 20 ES	0104 39			0105		30 E6	01D8 EE 09 01DA A6 00 01DC 81 04 01DE 27 1F	01E0 30 01E1 EE 0B 01E3 37	
0489 0490 0491 0492	0493 0494 0495 0495	0497	0501 0501 0502	0503 0504 0505	0506 0507 0508 0509	0510 0511 0513 0513	0515 0516 0517	0518 0519	0521 0521 0522	0523	0526	0529 0530	0531 0532 0533	0535 0535 0535	0537	0539 0540 0541	00042 00440 00440	0547 0548 0548)

SET RC	POINT TO "TO" BLANK	STORE BLANK	NO NO TO	•	STORE BLANK DONE? NO	SET DEFAULT RC	POINT TO FROM STRING CLEAR COUNT	GET CHARACTER FOUND	NAME TO LONG?	YES FORMAT ERROR	GET COUNT GET EXT COUNT SAVE NO EXT	TOO LONG? YES POINT TO FROM GET FIRST CHAR OF NAME
LDA B #2 STA B UB, X RTS * FORMAT "TO" STRING	FMTSA LDX TO, X LDA A #\$20 LDA B #8	FMISB STA A O, X INX DEC B	* BNE FMTSB * CLD A A * .	LDA B	FMTSC STA A 0, X INX DEC B BNE FMTSC	* FMTS1 TSX	* LDX FROM, X CLR B	* FMTSZ LDA A O, X CMP A #'. BEG FMTS3	* INX INC B CMP B #9 BNE FMTS2	BRA FMTSO * FOUND "." CHECK EXT * FMTS3 INC B	LDA A UA.X SBA STA A UA.X BEG FMTSO	CMP A #3 BHI FMTSO * EXT FIELD OK! * LDX FROM, X LDA A O, X
025E C6 02 0260 E7 03 0262 39	0263 EE 0B 0265 86 20 0267 C6 08	0269 A7 00 026B 08		02/1 H/ 00 02/3 08 02/4 C6 03 02/6 86 20	0278 A7 00 027A 08 027B 5A 027C 26 FA	027E 30 027F 6F 03	0281 EE 09 0283 5F	0284 A6 00 0286 81 2E 0288 27 08	0288 5C 028B 5C 028C C1 09 028E 26 F4	0290 20 CB	0294 A6 04 0296 10 0297 A7 04 0299 27 C2	029B 81 03 029D 22 BE 029F EE 09 02A1 A6 00
0673 0674 0675 0675 0677	0679 0680 0681	0663 0684 0684	0686 0687 0688	0687 0690 0691 0692 0693	0694 0695 0696 0697	0698 0699 0700 0701 0702	0703 0704 0705	0707	0710 0711 0712 0713 0714	0715 0717 0717 0718 0720	0722 0723 0724 0725	0722 0728 0728 0730 0731 0732
X Done?		M, TO ON STACK	ON RETURN TO=TO+COUNT FROM=FROM+COUNT	GET CHARACTER	MOVE CHARACTER	DONE? YES			NC TO.X RA MOVS1 TS REFORMAT A FILE NAME	STACK FROM ST MMBIG BIG	AD FILE NAM	C SAVE IN UA NAME TOO LONG? NO
BNE *+4 INC PARM2, X DEC B	BNE CMWO	FROM,	No.	TSX LDX FROM, X LDA A O, X	TSX LDX TO, X STA A O, X	TSX CMP A #04 BEQ MOVS3 INC FROM+1,X		INC TO+1, X BNE MOVS1	INC TO, X BRA MOVS1 RTS REFORMAT	FROM, TO ON B=COUNT OF B=RC= OO UN	05	LDA B UB, X STA B UA, X CMP B #12 BLS FMTSA MANY CHARACTERS
* *		:: \$00 * * *	* * * *	MOVS OVS1	* *		*	* MOVS2 I *	OVS3 FMTS:		* * * * ®	* * TOO * * FMTSO 'L
	0232 26 D5 0234 20 C9			0236 30 0237 EE 09 0239 A6 00	023B 30 023C EE 0B 023E A7 00	0240 30 0241 81 04 0243 27 0E		024B 6C 0C 024D 26 E8	024F 6C 0B 0251 20 E4 0253 39		9	0255 E6 03 0257 E7 04 0259 C3 06 0258 23 06 0250 30
0611 0612 0613 0613 0614	0616 0617 0618 0619	0 6 20 06 21 0622	0623 0624 0625 0625	0627 0628 0629 0630	0631 0632 0633 0634	0638 0638 0638	0641 0642 0643	0644 0645 0646 0647	0648 0649 0650 0651 0653	0654 0655 0656 0657 0658 0658	0660 0661 0662 0663 0663	0665 0666 0667 0668 0669 0670 0671

STORE CHAR WC? NO	SET AMBIG RC FIX "TO" POINTER	WILD CARD? NO SKIP OVER "."	STORE "?" ALL DONE FIX TO POINTER
STA A 0, X TSX DEC B INC TO+1, X BNE *+4 INC TO, X * CMP A * ? BNE FMTSSA	LDK STR BR4K BR4K BR4D BR4D BR4D STR STR STR STR		* LDA A #'? * LDA A #'? * FMTS7 STA A 0.X BNE FMTS7 * SET AMBIG RC * TSX LDA A #1 STA A UB. X * MOVE "FKOM" EXT TO "TO" EXT * FMTS8 TSX INC TO+1, X F FROME LDA A FF
02EB A7 00 02ED 30 02EF 56 0C 02F1 26 02 02F3 6C 0B 02F5 81 3F 02F7 26 E0	02F8 86 01 02F8 30 02FC A7 03 02FE 20 D9 0300 30 0301 EB 0C 0303 E7 0C 0305 E6 0B 0307 C9 00		0318 C.6 03 0316 A7 00 031E 08 0320 26 FA 0322 30 0323 86 01 0325 A7 03 0325 A7 03
0795 0796 0797 0799 0800 0801 0803 0803	0.807 0.807 0.807 0.808 0.808 0.811 0.811 0.815 0.815 0.815 0.815	0823 0823 0823 0824 0825 0826 0827 0828 0831 0833 0833	0836 0837 0838 0840 0841 0844 0845 0845 0846 0846 0846 0846 0851 0851 0851 0851 0851
NO NAME? YES WILD CARD? NO POINT TO "TO" STRING	STORE "?" DONE? NO RC STORE		GET CHARACTER DONE? YES
CMP A #'. BEG FMTSO * CMP A #'* BNE FMTS5 * WILD CARD FILL WITH "?" TSX LDX TO, X LDX D #O	** SET AMBIG RC ** SET AMBIG RC ** TSX LDA A #1 ** SET AMBIG RC ** TSX LDA A #1 ** FIX POINTERS	LDA A FKOM+1, X ADD A #2 STA A FROM+1, X LDA A FROM, X ADC A #00 STA A FROM, X ADD A #8 STA A T0+1, X ADD A #8 STA A T0+1, X ADD A #8 STA A T0+1, X ADC A #00 STA A T0, X ADC A #00 STA A T0, X ADC A #00 STA A T0, X	** BRA FMTS6 ** MOVE NAME FROM -> TO ** MISS LDA B #8 ** FMTS5A TSX LDA FKOM, X LDA A O, X TSX INC FKOM+1, X BNE *+4 ** INC FKOM+1, X BNE *+4 ** CMP A #' CMP A #' BEG FMTSSB **
02A3 81 2E 02A5 27 B6 02A7 81 2A 02A9 26 2C 02AB 30 02AC E 08	26 4 6 8 6 8 6 8 6 8 6 8 6 8 6 8 6 8 6 8	02BB A6 0A 02BF 8B 02 02C1 A7 0A 02C5 89 00 02C7 A7 09 02C9 A6 0C 02CB 8B 08 02CB A7 0C 02CB A8 0C 02CB A7 0C	02D5 20 34 02D7 C6 08 02D6 A6 00 02DF 60 04 02DF 60 04 02E1 26 02 02E5 EE 08 02E7 81 2E 02E7 81 2E
0734 0735 0735 0737 0739 0739 0740 0741	0745 0746 0747 0750 0750 0753 0753 0753 0755 0755	0760 0761 0761 0762 0764 0765 0768 0768 0770 0771	0776 0777 0778 0778 0778 0781 0782 0783 0785 0785 0786 0787 0787 0787 0787 0787

77.27.27.27.27.27.27.27.27.27.27.27.27.2	
-60PENDFFFF **-60PEND***FFFF **-60P	0.0
	. RDSEC . WTSEC 0 NULL OTLPT \$8002 . MTIN
* * * * * * * * * * * * * * * * * * *	* LPTK * MTAPE
	~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~
0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
0.38	0302 0304 0306 0306 0306 0306 0360
09197 09237 09237 09237 09237 09237 09337 09337 0934 0934 0934 0934 0935 0935 0935 0935 0935 0935 0935 0935	0967 0968 0969 0970 0971 0972 0973 0975 0975
a iii	
UTINE	
ET AMBIG ROONE ING ROUTING	
GET EXT COUNT WC? NO VES SET AMBIG ROCCESSING ROUTINE TO THE TO T	20 11 12 11 10 0 8 1 1 10 10 10 10 10 10 10 10 10 10 10 10
MC? NO VES SET AMBIG PROCESSING ROUT: 2 2 3 4 4 5 6 6 7	
X 1, X	
FROM, X A O, X A O, X FROM+1, X ++4 FROM, X TO+1, X ++4 TO, X A O, X A O, X A O, X FROM, X FRO	
INC TO.X LDA B UA.X TSX LDA A O.X LDA A O.X TSX INC FROM.X LDX TO.X STA A O.X TSX INC TO.X STA A O.X TSX INC TO.X CMP A #7? BNE FMTS10 LDA A #1 STA A UB.X DEC B BNE FMTS10 LDA A #1 STA A UB.X DEC B BNE FMTS10 LDA A #1 STA A UB.X DEC B BNE FMTS10 LDA A #1 STA A UB.X DEC B BNE FMTS10 LDA A #1 STA A UB.X DEC B BNE FMTS10 LDA A #1 START FMB = -@POLIAL#FFF FMB = -@POLIAL#FFF FMB = -@POSHAL#FFF FMB = -@POLIAL#FFF F	***EADDAX**FFFF ***EADDAX**FFFF ****EADDAX**FFFF ****ESUBAX**FFFF ****ESUBAX**FFFF *****EMUL.8**FFFF *****EMUL.8**FFFF *******************************
INC TO.X LDA B UA.X TSX LDA A O.X LDA A O.X TSX INC FROM.X LDX TO.X STA A O.X TSX INC TO.X STA A O.X TSX INC TO.X CMP A #7? BNE FMTS10 LDA A #1 STA A UB.X DEC B BNE FMTS10 LDA A #1 STA A UB.X DEC B BNE FMTS10 LDA A #1 STA A UB.X DEC B BNE FMTS10 LDA A #1 STA A UB.X DEC B BNE FMTS10 LDA A #1 STA A UB.X DEC B BNE FMTS10 LDA A #1 START FMB = -@POLIAL#FFF FMB = -@POLIAL#FFF FMB = -@POSHAL#FFF FMB = -@POLIAL#FFF F	***EADDAX**FFFF ***EADDAX**FFFF ****EADDAX**FFFF ****ESDABX**FFFF ****ESDABX**FFFF ****ESDABX**FFFF ****EMUL.8**FFFF ****EMUL.8**FFFF ****EMUL.16**FFFF ****EMUL.16**FFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFF
# INC TO, X # LDA B UA, X FMTS9 TSX LDX FROM, X LDA A O, X TSX INC FROM+1, X B NE *+4 B NE *+4 B NE *+4 INC TO+1, X BNE FHTS10 CMP A #1 TSX INC TO+1, X BNE FMTS10 * TSX INC TO+1, X BNE FMTS1 * TSX INC TO+1, X BNE FMTS10 * TSX INC T	FUB *-@ADABX**FFFF FUB *-@ADDAX**FFFF FUB *-@ADDAX**FFFF FUB *-@SBABX**FFFF FUB *-@SBABX**FFFF FUB *-@SUBAX**FFFF FUB *-@YULL8**FFFF FUB *-@YULL8**FFFF FUB *-@YULL8**FFFF FUB *-@YULL8**FFFF FUB *-@YULL8**FFFF FUB *-@YULL8**FFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFF
E6 04	FD63 FD8 **eADABX**FFFF FD7 FD8 **eADAX**FFFF FD80 FD8 **eADAX**FFFF FD80 FD8 **eSSXAB**FFFF FD90 FD9 **eSUBAX**FFFF FD90 FD9 **eSUBAX**FFFF FD90 FD9 **eSUBAX**FFFF FD90 FD8 **eSUBAX**FFFF FD80 FD80 FD80 FD80 FD80 FD80 FD80
EC 0B	10.20

TURN OFF BUSY BIT	SAVE RCBADR PT TO RCBGDT SAVE PTR TO RCBGDT	POINT TO PDTAB SAVE ON STACK CHARS TO MATCH MATCH!	GET PD PTR POINT TO NEXT ENTRY	SKIP RCBGDT GET RCBADR SAVE
* ON RETURN X=RCBADR * IOHURA LDA A #NTBUSY AND A RCBSTA, X STA A RCBSTA, X * RTS * SEARCH PDTAB FOR DEVICE * X=RCBADR		* FCB 5 * LDX #PDTAB * PDSRCA PSHX SW1 FCB 5 LDA B #3 CMPC SW1 FCB 18 FCB 18 FCB 18 ** NOMATCH	* PULX SMI SMI FCB 6 ADDBX SMI FCB 10 INX INX INX INX	NAAB SWI PULX SWI FCB 6 SWI SWI SWI NX INX INX SWI SWI
0439 86 7F 043B 64 05 043D 67 05 043F 39		0445 GE 03F0 0446 GE 03F0 0448 GE 044B GE 03 044B 12 044F 12	0451 3F 0452 06 0453 3F 0454 0A 0455 08 0457 08 0457 08	0.459 3F 0.456 02 0.45E 3F 0.45E 06 0.45F 3F 0.460 05 0.462 08 0.462 08
1043 1044 1045 1047 1049 1050 1053 1053		1064 + 1065 1065 1065 1066 1067 1068 1068 1071 1071 1072 1073 + 1077 1077 1077 1077 1077 1077 1077	1078 1080 1081 1082 1083 1084 1085 1087	1089 1091 1092 1092 1093 1096 1098 1099 1100 1100 1101 1103
		* * * * * * * * * * * * * * * * * * *	TTYIO TTYIO TNUL NULLIO NULLIO O THANDLER	LDX UXH, X
049C 04E7 0489 0499 0000 43	0.3F5 03C0 F 0.3F7 50 0.3FC 03C6 F 0.3FE 50 0.401 03CC F 0.403 03CC F	444 03D2 03D2 03D8 03D8 03DE 03DE	03E4 03E4 4E 03EA 03EA 30	042A EE 05 042E 87 05 042E A7 05 0433 25 04 0435 3F 0435 04
0.980 0.981 0.983 0.984 0.986 0.988 0.989 0.990 0.990	0992 0993 0994 0995 0997 0999	1002 1003 1004 1006 1006 1007 1011 1012	1015 1016 1017 1018 1019 1020 1022 1022 1022	1027 1028 1028 1030 1031 1034 1035 1035 1037 1038 1037 1038

1.64 0.497 0.49 0.49 0.49 0.40
D-PTR 1166 + 0496 06 1167 0497 0C 1168 0498 39 1169 0498 39 1173 0498 39 1173 0498 39 1173 0498 39 1173 0498 39 1173 0498 39 1173 0498 39 1175 0498 39 1175 0498 39 1175 0498 39 1175 0498 39 1180 0498
F RCBGDT R TO EQT S OF EQT B EQT ADR UTPUT? NT TO OUTPUT DRIVER ADDRESS B
END OF TABLL NO SKIP RCBGDT SKIP RCBGDT SET RETURN GET RCBADR GET ADDRESS OF EQT SAVE IN A, B SKIP RCBGDT GET RCBADR SKIP RCBGDT SAVE ON STACK SAVE ON STACK INPUT OR OUTPUT? INPUT OR OUTPUT? INPUT OR OUTPUT? GET DRIVER ADDRESS SAVE IN A, B GET RCBADR
FCB 5 TABX SWI FCB 3 TST 0,3 X BNE PDSRCA ** NOT IN TABLE PULX SWI FCB 6 PULX SWI FCB 6 PULX SWI FCB 6 END A #5 LDA A #5 LDA A #5 LDX 0,0 X GE SWI FCB 6 END SWI FCB 6 SWI FCB 7 SWI FCB 5 SWI FCB 7 SWI FCB 3 SWI FCB 5 SWI FCB 6 SWI FCB 5 SWI FCB 3 SWI FCB 3 FCB 6 SWI FCB 5 SWI FCB 3 FCB 6 SWI FCB 5 SWI FCB 3 FCB 6 SWI FCB 7 SWI FCB 3 FCB 3 FCB 3 FCB 6 SWI FCB 3 FCB 6 SWI FCB 3 FCB 6 SWI FCB 7 FCB 6 SWI FCB 7 FCB 6 FCB 6 SWI FCB 7 FC
1105 + 0464 05 1106 + 0465 3F 1108 + 0466 03 1109 + 0466 03 1111 0467 26 DE 1111 0467 26 DE 1111 0468 3F 1115 + 046B 3F 1116 + 046E 06 1117 046F 0D 1118 + 046E 06 1119 + 046E 06 1119 + 046E 06 1120 0472 8F 1121 0472 8F 1122 0474 87 05 1123 0476 06 1131 0470 3F 1132 0478 06 1134 047E 3F 1140 + 047E 06 1141 047F 3F 1142 0483 8F 1143 + 0483 05 1144 0485 E7 01 1150 0487 6D 06 1151 0487 6D 06 1151 0487 6D 06 1152 0488 B 02 1153 0488 B 02 1154 0488 B 02 1155 0489 3F 1156 0490 03 1166 0490 8 1167 0490 8 1168 0490 8 1169 0490 8 1160 0490 8 1160

RESET POINTER	CR? NO	4 7	NULLS? NULLS? NO		PAGING ON? NO	PAGE? NO	RESET DPCNT	PAUSE? NO	GET A CHAR ESCAPE? NO	GET EJECT COUNT NO EJECTS	. F	DONE? NO	GET RCBADR	RESTACK	GET EQT ADDRESS GET PHYSICAL ADDRESS	GET STATUS NO BREAK	READ DATA WAIT FOR ANY INPUT	GET RCBADR
XNI	CMP A #\$0D BNE OTLIN1	LDA A ##0A R JSR OUTCON *	LDA A #00 LDA B NL BEQ OTLIN3		OTLING TST DP BEQ OTLIN7 *	DEC DPCNT BNE OTLIN7	LDA A DP STA A DPCNT *	TST PS BNE OTLIN5 *		OTLINS LDA B EJ BEQ OTLIN7 *	OTLIN6 LDA	DEC B BNE OTLIN6	TLIN7 PULX SWI	PSHX 6	FCB 5 LDX RCBEQT, X LDX 4, X	LDA A O, X ASR A BCC OTLIN8	* LDA A 1, X USR INCON	OTLINB PULX SWI FCB 6 * RTS
0517 08	0518 81 0D 051A 26 D3	051C 86 0A 051E BD 057D F	0521 86 00 0523 D6 3E 0525 27 06	0527 BD 057D R 052A 5A 052B 26 FA	052D 7D 003B 0530 27 21	0532 7A 003C 0535 26 1C	0537 96 3B 0539 97 3C	053B 7D 0042 053E 26 07	0540 BD 0568 R 0543 91 43 0545 26 F9	0547 D6 41 0549 27 08	054B 86 0A 054D BD 057D R	0551 26 F8	. 0553 3F	0555	co	055B A6 00 055D 47 055E 24 05	0560 A6 01 0562 BD 0568 R	0565 3F 0566 06 0567 39
1291	1293 1294 1295	1296 1297 1298	1299 1300 1301	1303 1304 1305	1307 1308 1309	1310	1313 1314 1315	1316 1317 1318	1319 1320 1321 1322	1323 1324 1325	1326	1329	1331			1341 1341 1342	1343 1344 1345 1346	1348 + 1348 + 1349 + 1350
	SAVE RCBADR	X POINT TO BUFFER	STORE CHAR BUMP BUFFER POINTER	CR? NO LF	GET RCBADR		O CONSOLE	X:=RCBADR	SAVE	GET BUFFER ADDRESS GET CHARS/LINE	GET A CHAR	YES, DONE	SEND IT		S. L.	NULLS? NO	NULL	GET LINE WIDTH POINT TO LAST CHAR
SBC A #00	NL INS	TABX SWI FCB 3	BRA INLINZ * INLIN4 STA A O, X INLIN4 INX	CMP A #\$OD BNE INLINZ * INLIN6 LDA A #\$OA	* JSR PULX	SWI FCB 6 RTS	* OTLIN: OUTPUT A LINE TO CONSOLE * A, B=RCBADR	* OTLIN TABX SWI	FCB 3 PSHI SMI FCB 5	* LDX RCBDBA, X LDA B WD	* OTLIN1 LDA A O, X CMP A **O4	. 5	S JSR OUTCON INX DEC B		LDA A ##OD CONTROL LDA A ##OA CONTROL CONTRO	LDA BEG		* OTLINB LDA B WD OTLINC DEX LDA A O.X
04b0 82 00	+ 0402 3F + 0403 05	+ 0404 3F + 0405 03	04D6 20 CF 04D8 A7 00 04DA 08	84 24 86	•	+ 04E5 06 + 04E5 06 04E6 39		04E7	- 04E9 3F - 04E9 3F - 04EA 05	04EB EE 07 04ED D6 3D	04EF A6 00	27	04F5 BD 057D R 04F8 08 04F9 5A		04FC 86 0D 04FE BD 057D R 0501 86 0A 0503 BD 057D R	0506 86 00 0508 D6 3E 050A 27 06	050C BD 057D R 050F 5A 0510 26 FA	0512 D6 3D 0514 09 0515 A6 00
1229			1237 1238 1239 1240	1241 1242 1243 1244			1252 1253 1254	1255 1256 1257 +	1258 + 1259 1260 + 1261 +	1262 1263 1264	1265 1266 1267	1268 1269	1270 1271 1272	1273 1274	1275 1276 1277 1277	1279 1280 1281	1282 1283 1285	1286 1287 1288 1289 1290

٦	٦	4	

	* ************************************	A=CHAR PARITY STRIPPED	1418 1419 1420	BD 81
0568 3F 0569 05 0564 30			1421 1422 1422 1423	27
056B EE 04 056D EE 00 056F EE 04	LDX 4, X LDX RCBEQT, X LDX 4, X	GET RCBADR GET EQT ADDRESS GET PHYSICAL ADDR	1424 1425 1425	
0571 A6 00		GET	1427	05AB 08 05AC 81 0D
		NOT READY	1430	OSBO 84 13
0576 A6 01 0578 84 7F	* LDA A 1, X AND A #\$7F	GET CHAR STRIP PARITY	1433	05B2 BD 05ED
057A 3F	*	GET BUFFER ADDR	1434 1435 1436	+ 05B5 3F + 05B6 06
90 9700	*		1438	0587 39
97.50	* OUTPUT A CHAR TO CONSOLE * X:=BUFFER ADDRESS; CHAR IN A	CONSOLE CHAR IN A	1441	
	* OUTCON PSHX	SAVE BUFFER ADDRESS	1444	
057D 3F 057E 05	SWI FCB 5		1440	+ 0588 3F + 0589 05
	TSX LDX 4, X	GET RCBADR	144/	8 8
0582 EE 00 0584 EE 04		GET GET	1449 1450	OSBD EE OO
	* * * * * * * * * * * * * * * * * * *	GET STATUS	1452	05C1 A6 00 05C3 47
26.12	ASR		1454	05C4 24 FB
058B 24 FA	BCC OUTCO1	NOT READY	1456 1457	05C6 A6 01
058D A7 01 058F 33	STA A 1, X	SEND CHAR	1458 1459	+ 05C8 3F + 05C9 06
	PULX	GET BUFFER ADDR	1460	0504 39
0591 06	FCB 6		1463	
0592 39	RTS * READ A LINE FROM PAPER TAPE READER	PAPER TAPE READER	1465	
	* * A, B=RCBADR			+ 05CB 3F + 05CC 03
	INRDR TABX	X: =RCBADR		+ 05cn 35
0593 3F 0594 03	PSHX 3	STACK		+ 05CE 05 05CF EE 07
0595 3F 0596 05	SWI FCB 5	ATT DIFFERD ANDRESS	1474	
03%/ EE 0/	* CON VCBUBEL V		244	

	GET A CHAR LF? YES	NULL? Yes	CR? NO	X-OFF	GET RCBADR	APER TAPE READER JRN IN A	SAVE BUFFER ADDR	GET RCBADR GET EQT ADDR GET PHYSICAL ADDRESS	NOT READY	GET BVTE GET BUFFER PTR	OF PAPER TAPE BETWEEN LINES	X: =RCBADR STACK RCBADR	GET BUFFER ADDRESS	GET A CHAR CR?
JSR OUTFCH	INKD1 JSR RDRIN CMP A #\$OA BEQ INRD1	CMP A #\$00 BEG INRD1	STA A O, X INX CMP A #\$OD BNE INRD1	* INRD2 LDA A #\$13 JSR OUTPCH *		* READ A CHAR FROM PAPER * X:=BUFFER PTR RETURN IN	* RDRIN PSHX SWI FCB 5		* RDRIN1 LDA A O, X ASR A BCC RDRIN1	* LDA A 1, X PULX SWI FCB 6	RTS PUNCH A LINE WITH NULLS A, B=RCBADR	* OTFCH TABX SWI FCB 3 PSHX	SWI FCB-5 LDX RCBDBA, X	* OTFCH1 LDA A O, X INX USR OUTPCH CMP A ##0D
059B BD 05ED R	059E BD 05B8 R 05A1 81 0A 05A3 27 F9	05A5 81 00 05A7 27 F5	05A9 A7 00 05AB 08 05AC 81 0D 05AE 26 EE	05B0 86 13 05B2 BD 05ED R	+ 0585 3F + 0586 06		+ 05B8 3F + 05B9 05	05BB EE 04 05BB EE 00 05BF EE 00	05C1 A6 00 05C3 47 05C4 24 FB	05C6 A6 01 + 05C8 3F + 05C9 06	05CA 39	+ 05CB 3F + 05CC 03	+ 05CD 3F + 05CE 05 05CF EE 07	05D1 A6 00 05D3 08 05D4 BD 05ED R 05D7 81 0D
15	35822	2237	3878	3333	388888	544	6 4 4 4 4 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	4 4 8 A S	223	38283	649	258862	327	25278

SET CHARS / INE	OEI CHANS/LINE		ر TT م	!	RESET LINES/PAGE	-		CK.		r.	PAGE?	ON	INIT LDPCNT				GET RCBADR			INE PRINTER			SAVE BUFFER PTR		GET RCBADR	GET EGT ADDR	GE! PHYSICAL ADDR		CLEAR ACK	OUIFUI ACK?	NO	о ш)			TOR		
JSR OUTLPT			CMP A #50C BNE OTLP3		100	STA B LDPCNT PUL B	*	DILFG CRF A #500		LDA A #\$0A	DEC	BNE OTLPT4	LDA		* LDA A #\$0C	JSR OUTLPT	TLPT4 P	SWI FCB 6	*	* PRINT A CHAR ON LINE PRINTER	* * X=BUFFER PTR	*	OUTLPT PSHX	FCB 5	ISX LDX 4, X		PSH B			LDA B 1. X	. 4	a ≅ ā		NAT L	RTS	* READ A SINGLE SECTOR * A. REPERADE	i	EXT . RDSEC *
061B BD 0648 R 061E D6 46	8	8 8	0624 81 0C 0626 26 06	37				0630 26 D9	} ;	0632 86 0A	7A 0045	063A 26 09	063C D6 44	D7	98	0642 BD 0648 R		+ 0645 3F + 0646 06	07.47	0047 54			+ 0648 3F	•	0648 EE 04		37	ì	0652 E6 00	Н	0658 2A FC	0650 33			0			065E 7E 0000 X
1542 1543	1544 1545	1546	1548 1549	1550	1552	1553	1555	1557	1558	1559	1561	1562	1564	1565	1567	1568	1570	1571 + 1572 +	1573	1576	1578	1579	1581 +		1584	1585	1587	1588	7 00 00 10 0	1591	1592	1593 1594		1596 +		1600	1602	1603 1604
1 NO	۳. Ali		NULLS	T	3 NOT DONE	RESTORE RCBADR			TER ON PUNCH	¥i	SAVE BUFFER PTR			GET RCBADR	GE		GET STATUS		NOT READY	SEND BYTE	GET BUFFER PTR			alintaa 6		aqoaya x		70710	0.110		GET	GET CHARS/LINE	GET A CHAR	PRINT			CR	<u>"</u>
BNE OTPCH1	LDA A #\$0A JSR OUTPCH	LDA B #4	* ОТРСНЗ LDA	USR OUTPCH DEC B		* PULX	SWI FICE A	RTS	* PUNCH A CHARACTER ON PUNCH		UTPCH P	10 E	1	LDX 4, X I DX BCBEGT, Y		PSH B	UTPC1		BCC OUTPC1	STA A 1, X	× 1	SET SET	*	RTS * OHIDERT A LINE TO BEINIES		* OTIPT TARY	SWI	FCB 3	₹ 3 00		LDX RCBDBA, X	LUA B LWU	OTLPT1 LDA A O, X	JSR OUTLPT		* BNE UILFIZ		JSR OUTLPT LDA A #\$0A
05 D9 26 F6	05DB 86 0A 05DD BD 05ED R	05E0 C6 04	00 98	05E4 BD 05ED R 05E7 5A	05EB 26 FB		+ 05EA 3F + 05EB 0A	OSEC				+ OSEE OS	ဓ္	05F0 EE 04		05F6 37			05FB 24 FA	05FD A7 01		+ 0600 3+		0602 39			+ 0603 3F	+ 0604 03	+ 0605 3F	09090	出;	0609 D6 46	060B A6 00		5. S.	0012 20 UC	86 OD	0616 BD 0648 R 0619 86 0A
47.6	1481	1484	1485	1487 1488	1489	1490	1492		1496	498			502	ည် မ 4	302	507	508	510	$511 \\ 512$	513		. 716 . 717		519	522	523		526			689 689 689 689 689 689 689 689 689 689	532	533	ţΩ	536	538	539	540

v	06B5 7E 0000 X 06B8 7E 0000 X 06B8 7E 0000 X 06B8 7E 0000 X	EXT @USR3 EXT @USR4 EXT @USR5 EXT @FMTFCB END	USER DEFINED 3 USER DEFINED 4 USER DEFINED 5
OPEN SEQUENTIAL FILE CLOSE SEQUENTIAL FILE REWIND SEQUENTIAL FILE OPEN DIRECTORY READ A SEQUENTIAL FILE WRITE A SEQUENTIAL FILE GET A DIRECTORY ENTRY	0688 7E 0000 X 0688 7E 0000 X 068E 7E 0000 X	EXT @USR4 EXT @USR5 EXT @FMTFCB END	
OPEN SEQUENTIAL FILE CLOSE SEQUENTIAL FILE REWIND SEQUENTIAL FILE OPEN DIRECTORY READ A SEQUENTIAL FILE WRITE A SEQUENTIAL FILE GET A DIRECTORY ENTRY	06BB 7E 0000 X	EXT @USR5 EXT @FMTFCB END	
OPEN SEQUENTIAL FILE CLOSE SEQUENTIAL FILE REWIND SEQUENTIAL FILE OPEN DIRECTORY READ A SEQUENTIAL FILE WRITE A SEQUENTIAL FILE GET A DIRECTORY ENTRY	06BE 7E 0000 X	EXT @FMTFCB	
SEQUENTIAL FILE E SEQUENTIAL FILE UD SEQUENTIAL FILE DIRECTORY A SEQUENTIAL FILE E A SEQUENTIAL FILE A DIRECTORY ENTRY		END TO TO T	
шч н			
CL in cr i			
WRITE A DIRECTORY ENTRY			
DELETE A FILE			
CHAIN A PROGRAM FILE			
PRINT AN ERROR MESSAGE			
WARM START ENTRY			
USER DEFINED 6			
USER DEFINED 7			
USER DEFINED 8			
USER DEFINED 9			
USER DEFINED 10			
LOAD A BINARY FILE			
LOAD A RELOCATABLE FILE			
PARSE A COMMAND LINE			
GET A COMMAND LINE (CALLS @NXTOK)			
PRINT AN ERROR MESSAGE			
INITIALIZE DISK SYSTEM			
GET PRI	4	4	4

N NAM CLI COMMAND LINE INTERPRETER * COPYRIGHT 1978 BY HEMENWAY ASSOCIATES INC * BOSTON MASS. 02111 * ALL RIGHTS RESERVED * COLD START ENTRY POINT * RECLUST JMP COLDST * WARM START ENTRY POINT	K
0001 0002 0003 0003 0004 0005 0006 0007 0008 0009 0010	0015 0006 7E 0000 0016 0009 7E 0000 0017 0000 7E 0000 0018 0000 0071 0020 0000 0071 0021 0000 0374 0023 0000 0374 0024 0000 0376 0025 0000 0376 0027 0000 0376 0028 0000 0588 0029 0000 0588 0029 0000 0588 0030 0000 0588 0030 0000 0588 0031 0000 0588 0032 0000 0588 0033 0000 0588 0034 0000 0025 0041 0000 0025 0042 0000 0025 0044 0000 0025 0045 0000 0038 0047 0000 0038 0047 0000 0038 0048 0000 0038 0049 0000 0038 0050 1 0000 0038
0565 R 1ESTZA 00DB R 050C R 10 000B 0512 R 1TYID 03E4 R 0514 R 1TYAB 2183 M 062E R UA 0003 0600 R UB 0003 0620 R URH 0007 0640 R UKH 0007 0640 R UKH 0005 0550 R UKH 0005	057D R WRITE 23D2 M 00548 R XABX 21B5 M 00009 8 0449 R 0440 R 0440 R 0550 R 0550 R 0550 R 00050 R 2151 M 00050 R 2151 M 00050 R 00050 R
HMISS 0331 R 01LINB HMISA 0263 F 01LINB HMISC 0263 F 01LINB HMISC 0278 R 01LINB HMISC 0278 R 01LINB HMITOR 0009 01LP7 GETUR 23EC M 01LP71 INCON 0568 R 01LP71 INCON 0558 R 01LP71 INCON 0571 R 01PCH INCON 0571 R 01PCH INCON 0571 R 01PCH	LINI
ALDAR OOCC R ALDARX 2219 M ALDBX 2232 M ALDBX 2232 M ALDBX 2248 M ADUXAB 2200 M ADUXAB 2200 M BNSEQU 2A2A M B10S 0000 RN BMEM 0033 BS 0039 BUSY 0080	CLASS 0026 CLUSE 2369 M CMPO 01108 R CMPO 01108 R CMPO 01108 R CMWO 0203 R CMWO 0204 R CUCHAR 0022 DESCRA 0020 DESCRA 0030 DES
0664 RX 0654 RX 0654 RX 0651 RX 0001 R 0005 R 0000 R	0.0522 RX 0.0524 RX 0.0524 RX 0.0672 RX 0.0672 RX 0.0429 RX 0.0526 RX 0.0533 RX 0.0648 RX 0.0658 RX

	DIRECTORY	PERIPHERAL INTERCHANGE PROGRAM	RENAME	DELETE	SAVE	ASSIGN	LOAD	P	INITIALIZE	EXĮT MONITOR	SECURITY	SETCON	STATUS	BOOTSTRAP	LINK	SUBMIT	END OF TABLE	SPACE	
* RMB 256 STACK RMB 1 * COMMAND TABLE	* CMDTAB FCC 'DIR' R FDB DIRCMD	* FCC 'PIP' R FDB PIPCMD	FCC 'REN' R FDB RENCMD	* FCC 'DEL' R FUB DELCMD	* FCC 'SAV' R FDB SAVCMD	* FCC 'ASS' R FUB ASNCMD	FCC /LOA/ R FDB LODCMD	R FDB JMPCMD	R FDB INICMD	FUB \$E0E3	* FCC 'SEC' R FDB SECCMD	* FCC SET' R FDB SETCMD	FCC STA' R FDB STACMD	FCC 'BOO' R FUB BOOTCD	FCC 'LIN' R FDB LNKCMD	FCC /SUB/ R FDB SUBCMD	* FCB 00 * CHARACTER TABLE *		FCB \$00 #
0122 0123 00B6 0100 0124 01B6 0001 0126	01B7 44 01BA 0BES	01BC 50 01BF 0620	01C1 52 01C4 09F1	01C6 44 01C9 061B	01CB 53 01CE 07B4	0142 0143 01D0 41 0144 01D3 062F	01D5 4C 01D8 0956	01DA 4A 01DD 0607	0152 01DF 49 0153 01E2 0AF9	0155 01E4 45	01E9 53 01EC 0625	01EE 53 01F1 062A	0165 0164 01F3 53 0165 01F6 0634	01F8 42 01FB 0639	01FD 4C 0200 063E	0202 53 0205 06C6	0175 0176 0207 00 0178 0179	0180 0208 00 0181 0209 00 0182 020A 00	020B 020C
DEPTH TEMP WIDTH CHARS/LINE IG DEFINITIONS	EQUIPMENT TABLE ADDRESS	GENERALC DEVICE TYPE STATUS DATA BRANSFER TYPE DATA BREGGE ANDESS	EQUIPMENT TABLE ADDRESS	STATUS STATUS DATA TRANSFER TYPE	DAIA BUTTEK ADDRESS DRIVE NUMBER TRACK NUMBER	SECTOR NOMBER FWD LINK TRACK/SECTOR BACK LINK TRACK/SECTOR	FILE TYPE FILE TYPE FILE TYPE FILE TYPE FILE TYPE FILE TYPE		NEXT FUB IN MULTOR CHAIN INDEX INTO DATA BUFFER SPACE COMPRESSION FLAG	FILE NAME (8.3 + EOT=13) FILE TYPE	FILE ACCESS CODE FIRST TRACK/SECTOR LABOR OF COLOR	NOR /BIOS' CALLS			128 BYTES/SECTOR	'ROL BLOCK		DATA SPACE	SYSTEM
LDPCNT EQU \$45 LWD EQU \$46 BLOCK ADDRESSIN	RCBDE EQU	RCBGDI EWU 2 RCBSTA EQU 5 RCBDTT EQU 6	FCBEGT EGU O						FCBIND EQU 39 FCBSCF EQU 41			* * REGISTER OFFSETS ON		UXL EQU 6 * nick Attermittee	* DISK FILLIDOLES * SECSIZ EQU 128	SYSTEM F.	STSFLE KMB 2 FCC / DSK/ RMB 2 FMB 2 FMB 2	RMB * BUFFER RMB	* * STACK SPACE FOR SYSTEM
0061 + 000C 0045 0062 + 000C 0046 0063 0064	3000 +	0068 + 000C 0005 0069 + 000C 0005 0070 + 000C 0005	7000 + +	2000 + + +	2000 + + +	2000 + + +	+ + +	2000 + + +	0090 + 000C 0027 0091 + 000C 0027	2000 +	+ + +	7000 +	2000	0104 000C 0006 0105 000C 0006	0108 0109 000C 0080	0	0113 000C 0002 0114 000E 44 0115 0011 0002 0115 0013 0034	0015	

-	-	-

INITIALIZE STACK POINTER INIT. 'SUBHIT' FLAG POINT TO EGTAB POINT TO CONSOLE ENTRY	GET ACIA ADDRESS RESET ACIA	INIT ACIA POINT TO EGIOB	POINT TO TTY ENTRY GET ACIA ADDRESS	RESET ACIA INIT ACIA POINT TO EQTAB	POINT TO LPT ENTRY GET PIA ADDRESS DDRB: =OUTPUT	INIT PIA INITIALIZE DISK SYSTEM	X +1, X +2, X ISSUE A CR, LF	PRINT HEADLINE INIT. ACTIVE FCB CHAIN NOW PROCESS COMMANDS
ST LDS #STACK CLR SUBFLG LDX EQTAB+1 LDA B #4 ADDBX CLU	्री∂दद	crot μ	LDX EWIND-1 LDA B #40 ADDBX SWI FCB 10 LDX 0, X	LDA A #3 STA A 0, X LDA A #1 STA A 0, X LDX EQTAB+1	LDA B #28 ADDBX SW1 FCB 10 LDX 0, X LDA A #\$FF STA A 0, X	LUM A #\$ZC STA A 1,X INITUK SWI FCB 51	LDX #CONRCB LDA A # C STA A RCBGDT, X LDA A # * O STA A RCBGDT+1, X LDA A # * N STA A RCBGDT+2, X LDA A # * N STA A RCBGDT+2, X LDA # CRLF IS	PRIMSG SWI FCB 49 LDX #BANNER PRIMSG SWI FCB 49 LDX #0 STX FCBCHN BRA WARM3
* 02A3 8E 01B6 R COLDST 02A6 7F 0707 R 02A9 FE 0007 R 02AC C6 04	98E	86 A7 EF	C6 28 C6 28 C9 3F C 0A EE 00	02C3 86 03 02C5 87 00 02C7 86 01 02C9 A7 00 *	88 EE 0	02D8 86 2C 02D6 A7 01 * 02DC 3F * 02DD 33 *	02DE CE 0298 R 02E1 86 43 02E3 A7 02 02E5 A7 03 02E7 A7 03 02E9 A6 4E 02EB A7 04 02EB A7 04	02F0 3F 02F1 31 02F2 CE 0300 R 02F5 3F 02F6 31 02F7 CE 0000 02F7 CE 0000
	0253 + 02554 + 02555 0255	0257 0258 0259	0260 0261 0262 0263 + 0264 +	0266 0267 0268 0269 0270 0271	0272 0273 0274 + 0275 + 0276 0277	0279 0280 0281 0282 0283 + 0284 0285	0288 0289 0291 0291 0293 0294	0297 0298 + 0299 + 0301 0302 + 0303 + 0304 0305
								,
× × · · · *	+ - 1 . \	0 = 0 = 0	m ← N √ ► 00	о ∨ 1: ∧	¢∙Ф∢¤С⊡ШI	r0I∺JYJΣZ	:○ ₢₢₢ ₢०⊢⊐>з	X Y Z I CAROT CONSOLE BUFFER CONSOLE RCB
	FCB #04 FCB #04 FCB #04			CB #42 CB #04 CB #04 CB #06 CB #00			2000 000 000 000 000 000 000 000 000 00	∞ ⊷
<u> </u>	. L. L. L. L	. L. L. L. L	<u> </u>	<u> </u>	u u u u u u u u u i	և և և և և և և և և և		CONBUF RECONDER
	0213 04 0214 04 0215 04 0216 04		021B 42 021C 42 021U 42 021E 42 021F 42	0221 42 0222 04 0223 04 0224 00 0225 04		022E 82 022F 80 0230 80 0231 80 0232 80 0233 80		
0185 0186 0187 0188 0189	0191 0192 0193 0194	0196 0197 0198	0199 0200 0201 0202 0203	0205 0206 0207 0208 0209	0211 0212 0213 0214 0215 0216	0218 0219 0220 0221 0223 0223 0224 0224	0228 0228 0229 0230 0231 0233 0234	0236 0237 0238 0238 0240 0241 0242 0243 0244

۷	U	

POINT TO NEXT ENTRY	" SAVE PTR TO CMDTAB	STACK PARM	RESTORE CMDTAB PTR	END OF TABLE? NO	SKIP DESCRA PARM	* * PROCESS AS A TRANSIENT COMMAND FILE-NAME		GOOD LOAD? IF NOT, QUIT HAVE TRANSFER ADDRESS?	IF SO, CALL IT	GET PTR TO ROUTINE	GET ADDRESS OF ROUTINE TRANSFER TO A, B	SKIP DESCRA PARM RESTORE ROUTINE ADDRESS CALL ROUTINE
SWI FCB & ADDBX SWI SWI	INX INX TXAB SWI FCB 2	CULA SWI FCB 6 LDX DESCRA PSHX	SWI SCB 5 TABX	T B NOT IN	PULX SWI FCB 6	PROCESS AS A TRAN			JSR BRA	FOUND		PUCB 2 SWLX 5 SWLX 6 FCB 6 SWLI 5 FCB 3
	0361 08 0362 08 0363 3F 0364 02	0365 3F 0366 06 0367 DE 20		0365 26 0360 60 0367 26 E4	0371 3F 0372 06	į	DF 23 BD 0956 R CE 08AC R	26 26 26 26 26 26	0385 AD 00 0387 20 A7	0389 3F	•	038E 0390 0391 0391
0368 + 0369 + 0370 0371 + 0372 +	0373 0374 0375 0376 +	0378 + 0380 + 0381		0.387 0.387 0.388 0.390 0.391	0394 + 0395 + 0395 +	0397	0400 0401 0402	0403 0404 0405 0406	0407 0408 0409 0411	0412 0413 0414 0415 +		0422 0422 0423 0424 + 0425 0425 0427 0427 0427
LF, CR AY ASSOCIATES CP/68-1. 0'	REINIT STACK POINTER INIT. 'SUBMIT' FLAG ANY ACTIVE FCBS? NO	(OPEN) FCBS	X POINT TO NEXT FCB	INIT. FREE-SPACE TABLE 2 4 DRIVES 4	GET COMMAND	CHECK RC UNAMBIG. NAME? YES		NO PRINT ERROR		STACK PARM	STACK PARM COUNT OF THREE	MATCH
RLF FDB \$0A0D LF,CR ANNER FCB \$0A FCC 'HEMENWAY ASSOCIATES CP/6 FDB \$0A0D	* MARMST LDS #STACK REINIT STACK P CLR SUBFLG INIT. 'SUBMIT' WARM1 LDX FCBCHN ANY ACTIVE FCB BEQ WARM3 NO		FCB 21 LDX FCBNFB, X POINT TO NEXT F BNE WARM2	ARM3 LDX #0 STX FKETAB INIT, FREE-SPAC STX FRETAB+4 STX FKETAB+4 STX FKETAB+6		CMP B #1 UNAMBIG. BEQ CLI2 YES	* CMP B #3 NUMBER? BEQ TFILE IF SO, *	CLI1 LDX #FORMAT NO PRIMS SMI FCB 49	OMMAND DESCRA	PSHX SWI FCB 5 LDX #CMDTAB	8 2 # 3	CMPC SWI SWI FCB 18 BEG CMDSRB MATCH * NO MATCH *
FDB \$0A0D LF,CR ER FCB \$0A FCC 'HEMENWAY ASSOCIATES CP/6 FDB \$0A0D	LDS #STACK REINIT STACK PCLR SUBFLG INIT. 'SUBMIT' LDX FCBCHN ANY ACTIVE FCB	* CLOSE ALL ACTIVE (OPEN) * MARM2 CLOSE CALT	3F 5W 21 15 FCB 21 EE 25 LDX FCBNFB, X POINT TO NEXT F 26 FA BNE WARM2	LDX #0 STX FRETAB INIT. FREE-SPAC STX FRETAB+2 4 DRIVES STX FRETAB+4 STX FRETAB+6	CLIO GTCMD SWI FCB 48	CMP B #1 UNAMBIG. BEQ CLI2 YES	C1 03 * CMP B #3 NUMBER? 27 2C * BEQ TFILE IF SO, *	AA R CLII LDX #FORMAT NO PRIMS SWI FCB 49	DK UP COMMAND LDX DESCRA	PSHX SWI FCB 5 LDX #CMDTAB	PSHX SWI FCB 5 LDA B #3	CMDSRB

GET A COMMAND FROM THE CONSOLE PARSE THE FIRST TOKEN FROM THE COMMAND LINE		SET F	CON' SOLE DEVICE	C	T+1, X		T+2, X			POINT TO BUFFER				A-1. X ANTI DHIH BOPPEN HEDNESS	IN 'SUBMIT'?	ON	7.1188112.			SAVE BUFFER PUINIER	READ A CHARACTER			×÷	END-FILE?	YES	ERROR STATUS?				RESEL CONSOLF COMMAND			CLUSE FILE		RESET 'SUBMIT' FLAG	GET CONSULE CUMMAND	POINT TO BUFFER			6		!	ECHO COMMAND LINE		NOTAN GOS GISTATOS TINI	TOTAL TOTALER TON MAION
A COMMAND FR SE THE FIRST	Ě		CIM A #/C	(Œ	STA A RCBGDT+2, X	A P. B.	FCB 2		XABX	SWI	STO A BORDON X	STA B RCBDB		BEG GTCD2	COMMAND FROM		ŠĚ	IN ACHBELE	E E	SWI	FCB 24	Ω		BEW SUBEUF	TST B	BEQ SUBMT2		JSR SUBERR	BRA eGTCMD		LDX #SUBFCB	CEUSE	FCB 21		BRH CUICHU	LDX SUBTMP	STA A O, X		STX SUBTMP		:	LDX #CONBUF PRTMSG	SWI	FCB 49 STX CUCHAR	(1)
* * PAR		0401 6F 06			0409 A7 03	88	040D A7 04		040F 3F 0410 02	0411 CE 0248 R		0414 3F	0414 67 67	E S	70	041D 27 4B	GET	*	041F CE 0248 R	F (2		0428 3F	0429 18	E6	042C C1 08	1	0430 5D	0431 27 12		0433 BD 06E7 R	20 03		043B CE O70A R SUBEOF	043E 3F		0440 7F 0707 R	**************************************	FE 0708 R		08	044B FF 0/08 K 044E 81 OD	8	6	0452 CE 0248 R		0456 31 0457 DF 23) i
0492 0493 0494	0495	0496	0497	0499	0200	0501	0502		0202 +	9020		0208		0511	0512	0513	0515	0516	0517	0519	0520	0521 +	0522 +	0523	0524	0526	0527	0528	0529	0230	0532	0533	0034 0898	0536 +		0538	0540	0541	0542	0543	0545		0547	0548		0551 +	
X=A(MESSAGE)			HUUNESS	POINT TO CONSOLE RCB			TURECT TON: HOLLTRIT		OUTPUT MESSAGE							Ť		π,								AN ERROR MESSAGE ON RCB OR FCB STATUS	OL BLOCK	0.144	SIRIUS		POINT TO CONTROL BLOCK	r DEVIC	ור לבאט, שטון	CONVERT LEFT DIGIT TO ASCII		CONVERT RIGHT DIGIT		GET DEVICE NAME					PRINT FREDR / INF				
* BRA WARM3 * PRINT ERROR MESSAGE;	*		B UXL, X	LDX #CONRCB	Œ		STA A RCRUTT. X	:	IOHDR		FCB 19	**	*	* MESSAGES	6	FURTH FCC 'FURTH ERRUR'S			* FCB *OD		m		DEFINIOR KMB 2	* FDB \$CHOD	* *	* PRINT AN ERROR MESSAG	INDEX	* CONTRACT TO TO THE CO. *		PRIEKR ISX	LDX UXH, X	LDA A FCBSTA, X		JSR	STA A	LDA H FCBSIA, X	STA	LDA A	STAP	STA A DEVNAM+1	LDA	STA A	LDX #DEVERR PRIMSG	IMS	FCB 49	PRIERZ RIS	
0395 20 99		0397 30	E 2	빙	6		0365 67 06	È		0387	03A8 13	0349 39				03B6 OD		03B7 4E	03C3 0D				0300 0002							03D4 30	H	03D7 A6 05	ì	BD	B7 03D0	OSES BD OESD R	B7 03D1	A6 02	USEB B/ USUS K	B 19	A6 04	03F5 B7 03C7 R	CE 03C4		031-031	03FD 39	
0429 0430 0432	0433	0434	0436	0437	0438	0439	0441	0442	0443		0445	0447	0448	0449	0450	0452	0453	0454	0456	0457	0458	0459	0460	0462	0463	0464	0465	0466	0460	0469	0470	0471	0473	0474	0475	0477	0478	0479	0480	0482	0483	0484	0486 0486		0488	0450	

GET BYTE FROM CHARTAB	NAME? No	WILDCARD CHAR?	NO VES	YES SCAN NAME STRING		DECIMAL? No	YES, SCAN DECIMAL STRING	DELIMITERS?	NO, UNRECOG. CHAR	GET CHAR	92 (HEX)	SWIGHT AND OUR				TROUBLE, SET RC, CLASS=00		STRING STOP AT	CHAR	POINT TO NEXT CHAR	BUMP COUNT		GET BYTE IN CHARTAB	DECIMAL?	YES CONTINUE SCAN	CHECK LENGTH OF TOKEN	OVE OUT COLOR THE PARTY	WENT UNE CHAR. TOO THE	LENGTH OK?	NO, ERROR	CONVERT TO BINARY	SAVE VALUE
GCHRTB		## ##01			NXT7	#\$40 NXT6	DSCAN NXT7				B #\$24 NYT7	, , ,	NOCH.	CLASS Br		CLASS	Ę	٠, ١	* FIRST NON-DECIMAL C		A O, X DESCRC		CUCHAR	A #\$40	DSCAN	B DESCRC		B DESCRC		NUMERR	CVDB	VALUE B #3
SS.	BIT A BEQ	LDA B			BRA	BIT A BEQ	JSR BRA	BIT A		ž Č		į į	ž,	STA A			RTS	SCA			N C	X	X LS		BNE BNE	LDA		STA			SS	STX
* *XX	* NXT4	*	*	*	*	NXT5	ķ.	* ***				*	*	NXT7	*	NXTER		* DSCAN	* FIRS	DSCAN					;	•			,		* R DSCANZ	
* 0493 BD 0550 R N		C6 01 85 01	27 02	Co 02		04A7 85 40 04A9 27 05	04AB BD 04CB R	: K	27 10	04B4 DE 20	ដីជំ	50 07	04BC BD 0513 K	04BF 97 26) } } }	0026	¥ %	`		DE 23	& C	8	0403 DF 23	5 5	7 9	04DC DA 22	8		04E3 2D 03	04E5 7E 0546 R	BD OSBB	
0615	0617 0618 0619	0620 0621 0621	0623	0626	0628	0630	0633	0635	0637	6690	0641	0642	0644 0645	0646	0648	0650	0651	0654	0655	0657	0658	0990	0661	7990	0664	0665	0667	8990	0670	0671	0673	0675
	/DIVERSION/?	SB GET A LINE FROM CONSOLE	CUTL-D? (EOF IN TEXT FILE)	I GET NEW CHARACTER	GET A LINE FROM THE CONSOLE		S INIT PARMS FOR NATOK		NEXT TOKEN RUUTINE * SCANS A LINE OF SOURCE CODE AND RETURNS	THE NEXT TOKEN CLASS AND RC THE ADDRESS OF THE TOKEN IS RETURNED IN	# OF BYTES IN THE TOKEN IS SCRC.	THE RC AND CLASS ARE:	CLASS [A]	02 SUBSTRINGS	02	I) 04 DELIMITERS	ş	on Eur	00 ERRORS							LESS THAN 20	520		ER NO, UNRECOG. CHAR		200	~
@NXTOK	A #'& SUBMT4	#CONRCB GTCD2	A #\$04 SUBEOF	SUBMT1	æ	SWI FCB 19 LDX #CONBUF	CUCHA	XX	TOKEN KOUTINE	TOKEN ESS OF	DESCRA AND THE # 0 RETURNED IN DESCRO	ĕ □ □	RC [B]	10	88	(ASCII)	ş	<u> </u>	8		DESCRC	CUCHAR		ν, (Ο (Ψ	CUCHAR	A ##20	NXT1		NXTER	NXT7	4	C .
BRA	S M	LDX	4 CMP BEG	BRA	I	SE ZG			NS A L	ADDRE	CRA A	£ 22 €		ш	MCARU	DEL IMS			ZO.		동 당 당	Ž	STX	¥ ž	STX	<u>a</u> (H			TAB BRA	Ş	BR BLS
,	SUBMT3		* SUBMT4	*	* GTCD2						* DES	뿣 * *	* TYPE:	* NAME	* NUMBER	* *	* *	5 * *	* ERROR	* *	ENXTOK	OTXN						*			* 4	<u> </u>
		œ				οğ															0022	73 23 23	50	8	23	25	52					. 61
0459 20 16	045B 81 26 045D 26 05	045F CE 0298 0462 20 06	0464 81 04 0466 27 D3	0468 20 BB		+ 046A 3F + 046B 13 046C CE 0248	046F DF 23														7		H:	047B A6 0	8 H	8 2	0482 27 F	;	0486 81 0D 0488 26 3A	048A 16 048B 20 32	č	0491 20 31 0491 20 31

PRINT 'NUMBER ERROR' MESSAGE

PRINT 'NUMBER ERROR' MESS		CLEAN STACK	RETURN ERROR TOKEN	IN CHRTAB INDEXED BY VALUE OF		C 001 10		VALID CHAR ?	NO, > 5F	,		ADD IN CHARACTER OFFSET			GET BYTE					BINARY			# OF BYTES IN STRING			TEMP STORAGE	SET ADDRESS OF STRIKE	5		COLECT COLECT	DELI COOMI	DECKETEN FIR ID SIKING		BYIE UF THE CTOTES	SIKING	GET COUNT	COARCET	CONVERI	SAVE	DECKEMENT COUNT	(1 FEX DIGIT)	FOINT TO NEXT LEFT BYTE	CONVERI	SMIFT TO LEFT MIBBLE				CONVERT TO BYTE	SAVE	DECREMENT COUNT	(2 HEX DIGITS)
LDX #NUMBER PRTMSG SWI	FCB 49	INS	JMP NXTER	BYTE IN CHRTAB	CHAN IN KEU A	CMP A ##20		⋖	BHI GCHRTR		LDX #CHRTAB-\$20	ADDAX		'n,	LUM M O, X	2		CLK A	2	CVHB CONVERT HEX TO BINARY		E	li	RETURN [X] = VALUE		RMB 2	various Au			Ω	Q	¥ 2.	1 X X	20	INE CVHB1	I DA B DESCRE	3	<		Ŋ	BEG CVHBD				HSL A			€ <	SIA A HVAL+1	20	BEW CVHBU
* R NUMERR I					* *	CHOTE		. •	_	*		•		•				מראאוא ני	*		*	* ON EN		8		HVAL,	an		, .	<i>,</i> –	7 6	7		36	1 21 ≉		1 -	20	ne	- 1	20 C	-	7 6	a	Σ <	τ «	Œ (-	io i	Δ α	9.0
0546 CE 03B7 0549 3F	0546	054B 31 054C 31	054D 7E 04C4 R			ğ	0552 25 00	8	0556 22 08		OSS CE OIES R		055B 3F		OCCUPAGE OCC	U33F 3Y		0360 4F								2000 2950	06 20 06 20	75 0542	4	n4 22	8				o V	0573 D6 22	RD OSAR	2 6	50C) (2003	1 5	05/L 2/ 29		50 CONTR 48				6730	1 1	5/ UD63	0380 38	3 8
0739 0740 0741 0742 +		0744	0746	0748	0750	0751	0752	0753	0754	0755	97.70		+ 86/0		0,760	0761	70/0	0763	0765	0767	0768	69/0	07.70	0771	0772	07/3	4//0	07.76	0777	0778	0770	00.00	0707	0707	0783	0784	0785	2070	00/0	0700	0700	0780	0790	0/91	2670	0707	40/0	0790	9//0	76/0	0770
			0	0 0	o c	Ċ	0	٥	•																																										
TRING STOP AT MERIC CHAR		POINT TO NEXT CHAR GET CHAR	BUMP COUNT	GALLO TYPE OF FUTOG	GET BYTE IN CHARTAB	WILDCARD?		0	YES			YES CONTINUE SCAN		NOTENIC:	OCHIN COLUMN			MENT ONE CHAR TOO EAR	FINISH UP				œ		DESCRE: #0	_	POINT TO NEXT CHAR		BUMP COUNT		POINT TO NEXT CHAR		•	VES CONTINUE SCAN		CHECK TOKEN LENGTH		WENT ONE CHAR. TOO FAR	}	LENGTH OKO		NO, ERROR	The state of the s	CONVERT HEX TO BINGRY	SAVE BINARY VALUE					CUCHAR: = CORRECT VALUE	LOAD CLASS RC
		CUCHAR POINT TO NEXT CHAR A O, X GET CHAR	DESCRC BUMP COUNT	CALC TYPE OF THICK	GCHRTB GET BYTE IN CHARTAB	A #\$01 MILDCARD?	*++		B #2 YES	(NSCAN YES CONTINUE SCAN	V ## 40	MOCON VEG CONTINUE	NOCHIN TEO CONTINUE OCHIN	A DESCRE		A DESCRE WENT ONE CHAR TOO	ENDSCN FINISH UP				IN-HEX CHAR		DESCRO DESCROT =0	THESCRE INTO PERCRE	CUCHAR POINT TO NEXT	A O, X GET CHAR	DESCRC		CUCHAR POINT TO NEXT	GCHRIB GET BYTE IN CH	4 ##C/2 JEX/2	HSCAN1 VES CONTINUE		B DESCRC	B	B DESCRC WENT ONE CHAR TOO		HSCAN2	700	NUMERK		CVHB CONVERT HEX TO	VALUE SAVE BINARY VAL	100 mm m m m m m m m m m m m m m m m m m	9	CUCHAR		CUCHAR	A #2
NSCAN SCAN NAME STRING STOP FIRST NON-ALPHANUMERIC CHAR		CUCHAR POINT TO NEXT CHAR A O, X GET CHAR	DESCRC BUMP COUNT	GALLO TYPE OF FUTOG	GCHRTB GET BYTE IN CHARTAB	A #\$01 MILDCARD?	*++		LDA B #2 YES	(BNE NSCAN YES CONTINUE SCAN	***	MOCON VEG CONTINUE	NOCHIN TEO CONTINUE OCHIN	9		A DESCRE WENT ONE CHAR TOO	ENDSCN FINISH UP		*	HSCAN SCAN HEX STRING STOP AT	* FIRST NON-HEX CHAR		DESCRE: #0	THESCREA INIT DESCREA	CUCHAR POINT TO NEXT	LDA A O. X GET CHAR	DESCRC		CUCHAR POINT TO NEXT	GCHRIB GET BYTE IN CH	4 ##C/2 JEX/2	HSCAN1 VES CONTINUE		LDA B DESCRC	20	B DESCRC WENT ONE CHAR TOO		HSCAN2	700	UMP NUMERK		SCAN2 JSR CVHR CONVERT HEX TO	STX VALUE SAVE BINARY VAL	D #W	9	NDSCN LDX	Ě	CUCHAR	A #2

B:=10 POINT TO POWER OF TEN (X):=TENVL*10 RESTORE POINTER TO STRING	POINT NEXT LEFT DIGIT TOONE? OGET FINAL VALUE RETURN GET ADDRESS CHECK RC VALID NUMBER? YES NO, ERROR CLEAN STACK LOAD ADDRESS GO THERE	'STATUS' 'SETCON' 'STATUS' 'LINK' 'BOOT' LLIN NS NS CLIN NS TLIN NS NLIN NS ALIN NS
STA A DVAL-1 STA B DVAL-1 CLR A #\$0A LDA B #\$0A LDA TENVL MUL16 SWI FCB 16 STA TENVL PULX	PROCE MPC2	* 'DELLETE' * 'ASSIGN' DELCMD LDX #DE * PIPCMD LDX #PI * SECCMD LDX #SE SECCMD LDX #SE * SETCMD LDX #SE * BRA TRA SETCMD LDX #SE * BRA TRA * BRA TRA
OSEA B7 OSB6 R OSED F7 OSB7 R OSF0 4F OSF1 C6 OA OSF3 FE OSB9 R + OSF6 3F + OSF7 10 OSF8 FF OSB9 R	+ 05FB 3F + 05FC 06 05FD 09 05FD 09 0601 26 D3 0603 FE 05B6 R 0608 2F 0608 2F	061B CE 0666 R 061E 20 21 0620 CE 0673 R 0623 20 1C 0628 CO 17 0624 CE 069C R 062D 20 12 062F CE 0696 R 0637 20 0D
0863 0864 0865 0865 0866 0869 0870 0870		0905 0906 0908 0908 0911 0911 0911 0911 0923 0923 0923
SAVE DECREMENT COUNT (3 HEX DIGITS) POINT TO NEXT LEFT BYTE CONVERT SHIFT TO LEFT NIBBLE	OND HAND CONVERT TO BYTE STA A HUAL SAVE HBD LDX HUAL GET FINAL VALUE RTS H # # # FTURN ROUTINE TO CONVERT ASCII TO BINARY HBS LDA A 0, X GET BYTE CMP A # # 50 CONVERT CMP A # 50 CONVE	INII DECUDNI POINT TO LEAST SIGNIFICANT DIGIT SAVE POINTER GET DIGIT CONVERT TO BCD CLEAR ACCUMULATOR POINT TO POWER OF TEN (X):=TENVL*DIGIT (A, B)=(X) DVAL:=DVAL+TENVL*DIGIT
STA A HVAL DEC B BEQ CVHBD * DEX JSR CVHBS ASL A	CVHBD LDX	CVDB1 INX BEC B BNE CVDB1 * CVDB2 PSHX SWI FCB 5 LDA TENVL MUL16 SWI FCB 16 TENVL MUL16 SWI FCB 16 TAND FCB 16 TAND FCB 2 ADD B DVAL+1 ADD B DVAL+1 ADD B DVAL+1
0593 B7 0562 R 0596 5A 0597 27 0E 0599 09 0590 48 059E 48 059E 48	948 0562 R BT 0562 R 39 0562 R 80 30 81 09 80 30 81 09 80 07 39 07 39 7F 0596 R 7F 0598 R 7F 0598 R 7F 0598 R 7F 0598 R	05CF F7 05B8 R 05D2 08 05D3 5A 05D4 26 FC 05D6 3F 05D7 05 05D 40 F 05D0 FE 05B9 R 05E0 3F 05E1 10 05E2 3F 05E3 02 05E3 02
0801 0802 0803 0804 0805 0806 0808 0808		0843 0845 0845 0846 0846 0847 0848 0850 0851 0857 0857 0857 0858 0857 0857 0858 0857 0858 0857 0858

* * SYNTAX: SUBMIT [DRIVE:] FILENAME.EXT * FILE MUST BE 'TEXT' TYPE	* SUBCMD LDX #SUBFCB	CLK FUBDIL,X MAKE INPUT FILE	A FCBSCF, X	FRITCH FURTH LUKIVE: J FILENAME, EXT	4	PRIERR PRINT ERROR MESSAGES		TCB 30 IST FCBSIA, X ERROR?	SUBERR	# # # # # # # # # # # # # # # # # # #		FCB 20		A SUBSTAN	LDA A FCBTYP, X 'TEXT' FILE?	A #3	BNE SUBERK IT NO!, EKKUK	R INC SUBFLG	RTS	* * * * * * * * * * * * * * * * * * *	PRIMSG	IMS		I K LUX #SUBLUB PRIERR PRINI ERROR MESSAGE		30	CLUSE TRY TU CLUSE FILE	FCB 21	STX.	* SUBLIN FCC / SUBMIT FILE ERROR/	FCB		SUBTMP RMB 2	* SUBFCB RMB 2		RMB 2	RMB	* SUBBUF RMB SECSIZ	* PROCESS 'SAVE' COMMAND	* STORE MEMORY CONTENTS ON DISK IN BINARY FORMAT	* * SYNTAX: SAVE [DRIVE:] FILENAME.EXT, STARTAD, ENDAD [, TRANSA]
	Ä,	06CB 86 FF	06CD A7 29	+ 060F 3F	+ 06D0 2C		+ 06D1 3F + 04n2 1E	06D3 (+ 0607 3F		0609 60 05	HO 07 9780	06DD A6 1D		UOE1 20 U4		06E6 39	V370 30 2370		06EA	+ 06EB 31	VOEL LE U/UH R	06EF	+ 06F0 1E	+ 06F1 3F	06F2	06F3 39	06F4 20	00 9020	0707 0001	0708 0002	070A 0002		070F 0002		0734 0080			
0660 8860 8860	0991	0994	3660	0997			1000	1002	1003	1004			1008	1010	1011	1012	1014	1015	1016	1017	1019		1021	1023			1026		1029	1030	1032	1033	1035	1036	1038	1039	1041	1042 1043	1045	1047	1049
					SAVE OLD CUCHAR					TES SELECTION OF THE SE	ISSUE / LOAD / COMMAND			RESTORE OLD CUCHAR		GOOD LOAD?		TRANSFER ADDRESS?	NO, QUIT	PASS "POTTOR" ADDRESS IN (A.B.)	,		YES, GU IMERE							<u>.</u>											
* 06B0 R BOOTCD LDX #BOOTLN 03 BRA TRANS	LNKCMD LDX #LNKLIN		TRANS TXAB	FCB 2	LDX CUCHAR	PSHX				FCB 3	JSR LODCMD	PULX	SMI FOR A)		TST FCBSTA, X		LDX VALUE	BEG TRANS2	* FOR ASSIGN' TRANSIENT.	LDA A PDTAB+1		* C TEC	THANSZ RTS		DELLIN FCC 'O: DELETE, CMD'		PIPLIN FCC 10: PIP. CMD1	* FUB WOLL	ECLIN FCC	FCB #OD	* SETLIN FCC '0: SET. CMD'	FCB \$OD	SNLIN FCC	FCB \$OD	STALIN FCC 10: STATUS, CMD1	FCB	BOOTLN FCC (0: BOOT, CMD)	FCB #OD	LNKLIN FCC 'O: LINK, CMD'	* PROCESS 'SUBMIT' COMMAND
0639 CE 06B0 R 063C 20 03	063E CE 06BB R		0441 35	0642		,	0645 3F		0647	0648 03 0649 DF 23			064E 3F	-	띵	0655 6D 05 0657 26 00	2	띰	80		B6 000A R	F6 000B	0003 05 00	0665 39		0666 30 0677 00	ā	08.0		30	00 8890	30		30	06A2 0D	06A3 30	ac ac	30	OGBA OD	06BB 30 1	
0926 0927 0928	0830	0932	0933		9860		+ 6860			0942	0944		0946 +	8460	0949	0.950 0.950	0952	0953	9554	0929	0957	0958	0860	0961	0965	0963	0962	9960	0968	6960	0970	0972	0973	0975	0976	0978	6/60	0981	0983 0983	0985	2860

		126		
YES (NO TRANSFER ADDRESS) DELIMITER? NO, ERROR GET A TOKEN FROM CLI CHECK RC	CHECK STATUS SE RECORD HERE POINT TO FCB COMMMAND TYPE OPEN SAVE FILE CHECK STATUS GOOD?	HEADER BYTE HIGH-BYTE OF ADDRESS LOW-BYTE OF ADDRESS POINT TO FCB	FSAVE-FILE OPEN GOOD? BAD RECORDS GET END-ADDRESS ARE THERE > 256 BYTES LEFT?	E FULL R IN TO FCE BYTE OF BYTE OF IT TO ME!
BEQ SAV6 CMP A #4 BNE SAV5 NXTOK SWI FCB 47	LDA B TC. CMB B #3 NUMBER? BME SAV5 NO. BEROR OUTPUT TRANSFER-ADDRESS RECORD HERE LDX #SAVFCB INC FCBTYP, X OPEN SMI SMI FCB 20 TST FCBSTA, X CHECK STA BEQ *+5 UMP SAVERR BAD	LDA A #\$16 JSR PUTBYT LDA A VALUE JSR PUTBYT LDA A VALUE+1 JSR PUTBYT BRA SAV7 LDX #SAVFCB	SWI FCB 20 TST FCBSTA, X FS BNE SAVERR BA MRITE OUT BINARY RECORDS LDA A SAVEX1 LDA B SAVEX1 SBC A SAVEX BEG SAVEX	
0800 27 2B ** 0802 81 04 0804 26 CC ** 0806 3F 0806 3F 0808 D6 25	0808 US 25 0806 C1 03 0806 C2 C4 ** ** 080E CE 08AC R 0813 3F 0814 14 0815 6D 05 0817 27 03 **	081C 84 14 081E BD 0841 R 0821 94 27 0823 BD 0841 R 0826 96 28 0828 BD 0841 R 082B 20 09 **	0830 3F 0831 14 0832 &D 05 0834 & & & & ** NOW ** NOW ** NOW 0836 &B OBDF R SAV7 0837 F6 OBE R 0837 F OBE R 0837 F OBE R	64 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1
1111 1112 1113 1115 1116 1116 1118 1118 1118	1117 1120 1123 1124 1125 1127 1128 1130 1131	1136 1138 1138 1138 1140 1140 1144 1144 1144 1144	+ + + + + + + + + + + + + + + + + + +	1159 1160 1161 1163 1164 1165 1167 1168 1169 1170
POINT TO FCB BINARY TYPE NO COMPRESSION ACCESS-TYPE=0 OUTPUT FORMAT EDRIVE: 1 FILE. EXT	PRINT ERROR MESSAGES ERROR? IF SO, RETURN GET TOKEN FROM CLI	YES NO, FORMAT ERROR GET TOKEN FROM CLI CHECK RC NIMBER?	NO, ERROR SAVE STARTING ADDRESS GET TOKEN FROM CLI DELIMITER? NO, ERROR	CHECK RC NUMBER? NO, ERROR SAVE ENDING ADDRESS GET TOKEN FROM CLI
D LDX #SAVFCB CLR FCBTVP, X CLR FCBSCF, X CLR FCBACS, X LDA A #\$FF STA A FCBDTT, X FMTFCB	SMI FCB 44 PRTERR SWI FCB 30 TST FCBSTA, X BEQ *+3 RTS NXTOK SWI FCB 47 LDA A CLASS	BEG SAV5B LDX #FORMAT PRTMSG SWI FCB 49 RTS RTS NXTOK SWI FCB 47 CMP R #3		SWI SWI FCB 47 LDA B RC CMP B #3 BNE SAV5 LDX VALUE STX SAVEX1 NTTOK SWI FCB 47 LDA A CLASS CMP A #\$0D
000000	07C1 07C3 07C4 07C5 6 07C7 2 07CA 07C6 07C6	0700 27 0702 CE 0705 39 0707 39 0708 38 0709 28	070E 26 070E 26 07E2 FF 07E5 3F 07E6 27 07E7 96 07E9 81	07EE 2F 07EE 2F 07EE 2F 07E 06 25 07F 1 C1 03 07F3 26 DD 07F5 DE 27 07FA 3F 07FB 2F 07FB 2F 07FB 2F 07FB 2F 07FB 2F 07FB 2F
	1058 + 1058 + 1059 + 1060 + 1061 + 1065 1066 1066 1069 + 1067 1069 + 1070 + 107	1073 1074 1075 1075 1076 1079 1080 1081 1082 1083 1083 1083 1083 1083 1083	1086 1087 1088 1099 1099 1095 1095	1008 + 1009 + 1009 1009 1009 1009 1009 1009

SAVBUF RMB SECSIZ * PROCESS / LOAD / COMMAND	* LOAD MEMORY FROM BINARY OR COMMAND DISK FILE * IF TRANSFER ADDRESS IN FILE (COMMAND TYPE), SAVE IT	IN "VALUE" BASE-PAGE LOCATION OTHERWISE, "VALUE"=0	* * * * * * * * * * * * * * * * * * *	FMTFCB	4	PRTERR PRINT ERROR MESSAGES SWI	30	IST FCBSTA, X ENROR? BNE LODS IF SO, QUIT	PERFORM	1	PRINT ERROR MESSAGE	SWI FCB 30	ODS RTS	* *	**	* LOAD-BINARY PROCESSING *		VALUETA	LDX UXH, X POINT TO FCB	CLR FCBSCF, X NO COMPRESSION	CLR FCBSTA, X GOOD STATUS OPEN OPEN SAVE-FILE		FCB 20 IST FCBSTA.X CHECK STATUS	BEG LOADB2 GOOD	* RTS BAD, QUIT	* LOADB2 LDA A FCBTYP, X CHECK FILE TYPE	BEG LOADB3 (0) BINARY	CMP	BEG LOADB3	R LDX #TYPMSG FILE-TYPE ERROR PRIMSG	SMI		LDX UXH, X POINT TO FCB CLOSE CLOSE FILE SWI
08B5 0021 08D6 0080			OSSA CF ORAC		+ 0959 3F + 095A 2C	+ 095B 3F	0820	095D 6D 05 095F 26 04		+ 0961 3F	7040	+ 0963 3F + 0964 1E	0965				0966 7F 0027	- e	096D EE 05	P 4			+ 0976 14 0977 6D 05	27	097B 39	097C 66 1D	27	8	0982 27 1E	0984 CE 098F	+ 0987 3F	0889 30	098A EE 05 . 098C 3F
1233 1234 1235 1237	1238 1239 1240	1241	1243			1248		1251	1253		1257			1261 1262	1263	1264	1266	1268	1269	1271	1272		1275	1277	1279	1280	1282	1284	1285 1286	1287			1292 1293 1294 +
POINT TO FCB	DONE?	KEEP GOING	POINT TO FCB	HEADER BYTE	HIGH-BYTE OF ADDRESS	LOW-BYTE OF ADDRESS	i	COUNT	POINT TO MEMORY ADDRESS GFT BYTE	1	POINT TO FCB	DONES		CLOSE SAVE-FILE		CHECK STATUS		RETURN TO CLI		PRINT ERROR MESSAGE		MUST CLOSE FILE			WRITE BYTE TO FILE		CHECK STATUS	2000	REMOVE SUB-RETURN	ERROR AND QUIT	GOOD RETURN		
R STX XXX XXX XXX	DEC B BNE SAV8 *	BRA SAV7	AC R SAV9 LDX #SAVECB * LETTE OUT LAST RECORD HERE		BSR R LDA	BSR PUTBYT LDA A SAVEX+1	BSR	TBA BSR PUTBYT	R SAV10 LDX	X Z I	צנב	BSR PUTBYT		* CLOSE	IMS	FCB 21 TST FCBSTA, X		* RTS	* * * * * * * * * * * * * * * * * * *		SWI FCB 30	CLOSE	SWI FOR 21	RTS	PUTBYT WRITE	SWI FCB 25	TST FCBSTA, X		oni Sni	BRA SAVERR *	PUTB2 RTS	AVFCB RMB	FCC JDSK RMB 2 RMB 2 FDB SAVBUF
085F 0860 0863 0866	0868 5A 0869 26 EF	086B 20 C9	OBED CE OBAC		0872 8D 0874 B6	0877	087C 8D	087E 17 087E 8D 20	0881 FE	80 9880	OSSA CE OSAC		0880		+ 0892	+ 0893 15 0894 6D 05	0896 26 01	0898 39	00000	USYY CE USHC K SHVEKK	+ 089C 3F + 089D 1F		+ 089E 3F + 089E 15	0840		+ 0861 3F + 0862 19	8080 8080	,	08A7 31 08A8 31	08A9 20 EE	08AB 39		08AE 44 08B1 0002 08B3 08D6
1172 1173 1174 1175	1176	1179	1181	1183	1184 1185	1186	1188	1189	1191	1193	1155	1196	1198	1199 1200	1201	1202 1203	1204	1206	1207	1208	1210	1212	1213	1215	1216	1218	1220	1222	1223	1225	1227	1229	1230 1231 1232

128

NET SECON 1373 + OFF B 3F SHITE SHITE 1375 + OFF B 1E FUE 1375 OFF C 26 28 SHITE 1377 OFF C 26 28 SHITE 1377 OFF C 26 28 SHITE 1378 OFF C 26 28 SHITE 1380 OAO1 CE OOOC R LDA	A CALLEL-1 CET ADDRESS-LOW 1374 OPE 3F PRITE	POINT TO FCB INPUT FORMAT LDRIVE:		PRINT ERROR MESSI ERROR? YES, QUIT	SEARCH DIRECTORY PO- GET DIRECTORY PO- TRACK SECTOR X BLOCK POINTER	GET DIRECTORY BLC CHECK ACCESS CODE RENAME-ABLE? YES NO, ERROR ERROR CHECK NO ERROR	RENAME ERROR		NO, FORMAT ERROR
NET COUNT 1372 + 09F7 2C 1372 + 09F7 1E 1373 + 09F8 3F 1374 + 09F8 3F 1375 + 09F8 3F 1376 09F2 26 28 1378 0960 6 2 8 1378 0960 6 1 8 1378	STA A NOTE	* RENCMD LDX #SAVFCB CLR FCBDT1, X FMTFCB SWI		FCB 44 PRIERR SMI FCB 30 TST FCBSTA, X BNE RENAMS	N # σωσυσσασα	irαα*			
1368 1368 1368 1368 1368 1371 1371 1372 1373	LE TRANSFER ADDRESS HERE 1348	1	09F1 CE 08AC R 09F4 6F 06 09F6 3F	09F7 2C 09F8 3F 09F9 1E 09FC 26 28	BD OFFE R CE 000C R AS 0A 1F F CE 20 CE 00 CE AS 0A CE CE CE 20 CE	EE 27 A6 0E 81 02 26 03 7E 0AE3 R CE 08AC R 6D 05 27 16	0A26 CE 0A2C R 0A29 3F 0A2B 39 0A2C 20 0A3B 0D	0A3C 3F 0A3D 2F 0A3E 96 26 0A4C 81 04 0A42 27 06	03AA
TO FCE BYTE F DDRESS- DRESS- D	LE TRANSFER ADDRESS HERE BSR GETBYT GET A BYTE F CMP A ##16 NO LE TRANSFER ADDRESS HERE BSR GETBYT STA A VALUE BSR GETBYT STA A VALUE BSR GETBYT STA A VALUE BSR GETBYT CMP A ##02 FRAME HEADER BSR GETBYT GET ADDRESS- SWI FCB 21 DONE WITH LC LCSE SWI RTS CLOSE SWI RTS COUNT DOWN F BRE LOADB6 BRA LOADB6 BRA LOADB6 CMP B #8 END-FILE? BWE GETB3 CLR FCBSTA, X CHECK STATU BWE GETB3 CLR FCBSTA, X END-FILE?	1364 1365 1366	1367 1368 1369 1370 1370	1372 1373 1374 1374 1375 1377	1375 1380 1381 1382 1383 1384 1385 1386 1386	1390 1390 1391 1392 1394 1395 1396 1396 1396	1400 1401 1402 1403 1405 1405 1406 1406	1409 1410 1411 1412 1414 1415	1416 1417
	LE TRANSFER BSR GETBYT STA A VALUE BSR GETBYT STA A SAVEX SWI FCB 21 RTS CLOSE SWI FCB 21 RTS STA A SAVEX BSR GETBYT STA A SAVEX STA A SAVEX BSR GETBYT STA BSR GETBYT STA BSR GETBYT STA BSR GETBYT STA B FCBST BNE GETBS CMP B #8 BNE GETBS CLOADB6 BNE GETBS CLOADB6 CMP B #8 BNE GETBS	INT TO FC T A BYTE ER-ADDR.	NO ADDRESS HERE GET ADDRESS	GET ADDRESS GET NEW FRA FRAME HEADE VES	FILE ITH L DRESS	GET GET GET GET STOR	I DOWN FRAME HEW FRAME BYTE FROM	CHECK STATU 600D END-FILE? 40	END-FILE IS

14.00 1.00	NO, ERROR	POINT TO DIRECTORY FCB	SET UP TO SAVED T/S			READ DIRECTORY SECTOR		ERROR?	YES	SOLUTION OF THE SECTION	FOINT TO OLD NAME IN SECTOR			POINT TO NEW NAME				MOVE NAME INTO DIRECTORY					CLEAN STACK	POINT TO SYSTEM FCB			WRITE NEW NAME INTO DIRECTORY		MAKE 'INPUT' AGAIN		DIED TOTE NAME ERROR					£.,			SECURITY ERROR					Σ.	GND	TO PROCESS THIS COMMAND	GET NEXT TOKEN FROM COMMAND LINE	
1477 1478	Ð.		FCBSTA, X A FCBFTS,	LDA B FCBFTS+1, X	STA B FCBSCT, X	IOHDR	SWI FCB 19	TST A	BNE RENMS	I DX FCBI TS. X	PSHX	IMS	FCB 5	LDX #SAVFCB+FCBNAM	PSHX	INS	FCB 5	n	IMS	m	INS	SNI	SNI			αç	LOHOR Sult	FCB 19	CLR FCBDTT, X	RTS			IMS	œ	0 K		FCB #OD		Š	SHI	FCB 49			\ ¥	INITIALIZE	CALL TRANSIENT FILE TO PR		IMS
Hone Hone	0A9B 7E 0A26 R	OASE CE DOOC R	OAM1 6F OAM3 A6	0AA5 E6	OAA9 E7		- + +	OAAD	OAAE 26 EB	OABO FF 21	OHDV EE	+ 0AB2	+ 0AB3	OAB4 CE OBBC		+ 0AB7	+		+ OABB	+ OABC	OABD	OABE	Page 1	OAC1 CE OOOC	OAC4 86 FF	0AC6 A7	+	9000	OACA 6F	0ACC 39	OACH CF OARS R		+ OADO	+ OAD1	20 7040	0AD3 20	OMEZ		OAE3 CE OAE9 R	+ 0056	+ OAE7	OAE8 3		0AE9 20	*	* *		+
+ OA47 3F			GET TOKEN FROM CLI		CHECK RC	UNAMBIG, NAME?	אט, באאטא	GET NAME POINTER	SAVE IT	GET LENGTH SAVE TT	GET TOKEN FROM CLI			CHECK RC	PERIOD?	NO, ERROR	dotata Filipo	GET TOKEN FROM CLI			CHECK RC	UNAMBIG. NAME?	NO, ERROR	GET LENGTH OF EXT		FIELD									CLEAN STACK			ERROR?	YES		SEARCH DIRECTORY		CHECK STATUS	NOT FOUND?		NOT FOUND (GOOD)?	YES	
+ 0A44 3F + 0A48 31 0A46 35 + 0A48 2F 0A46 10 25 0A50 26 F2 0A50 26 F2 0A57 96 22 0A57 96 22 0A57 96 22 0A59 B7 0BE4 R + 0A50 2F 0A69 12 E 0A60 12 E 0A67 15 B 0BE4 R + 0A67 3F + 0A76 3F + 0A77 3F + 0A76 3F + 0A76 3F + 0A76 3F + 0A76 3F + 0A76 3F + 0A76 3F 0A89 31 0A81 31 0A82 26 BD	SWI FCB 49	RTS	RENAMB NXTOK	SWI FCB 47	LDA B RC	CMP B #1	NG N	LDX DESCRA				IMS	FCB 47	LDA B RC	CMP B #			NXTOK	IMS		B RC	B #1	i i		ADD B SAVEA	LDX #SAVFCB+FCBNAM	ING	FCB 5	LDX SAVEX	XIX IX				20	SNI	INS									JMP RENAM9			
1418 1419 1420 1420 1422 1423 1425 1425 1427 1427 1428 1427 1428 1429 1429 1429 1429 1429 1429 1429 1429	0A47 0A48	39	÷ 1£	0A4A 3F	18	ដ	74 97	DE 20	FF OBDD	96 22 B7 0BE4		OASC 3F	0A50 2F	90	5	Z6 E0	۵			0A68 2F	8		20 DJ		4	ç	0677 3F	0478 05		OA7C 3F	0A7U 05	I •	OAZE 3F	. "				3	26 BD	CE OBAC R	BD OEFE	CE 08AC	8	26 03	7E OACD R	81 01	27 03	K

STORAGE FOR FORMATIED FILE NAME DEFAULT FILE NAME	SPACE FOR COUNT OF SECTORS SECTORS USED^	POWERS-OF-10 TABLE	TEMP. STORAGE	DEFAULT DRIVE=0 DEFAULT TO CONSOLE DEFAULT NAME HAS 12 CHARS. SAVE IT	DEFAULT FILE NAME=* *	FORMAT *. * INTO BUFLIN CLEAN STACK	GET TOKEN FROM CL.I	CHECK RC END OF LINE? NO	YES, USE DEFHULIS SWITCH INDICATOR? NO	GET SWITCH LINE-PRINTER ('L')?
BUFLIN RMB 13 * ALLFIL FCC '*. */ * DIRLNZ FCB \$0A	RMB 50	KIOK FDB 10000 FDB 1000 FDB 100 FDB 10	SAVEX RMB 2 SAVEX RMB 2 TMPX RMB 2 LPTFLG RMB 1 CPTFLG RMB 1 **	* * DIRCMD CLR SAVEA CLR LPTFLG LDA A #12 STA A SAVEX LDX #BUFLIN PSHX		LDA B #3 FMTS FMTS SWI SWI FCB 52 INS INS	DIRCDO NXTOK SWI FCB 47	LDA B RC CMP B #\$0D BNE *+5	* OMF DIRCUS * CMP B #// BNE DIRCUI	NXTOK SWI FCB 47 LDA DESCRA LDA D O.X CMP A #'L
OBBD OCOD	04 0005 2E 040D	0BD3 2710 0BD5 03E8 0BD7 0064 0BD9 0006 0BD8 0001	OBDD 0002 OBDF 0002 OBE1 0002 OBE3 0001 OBE4 0001	OBES 7F OBE4 R OBE8 7F OBE3 R OBE B 86 OC OBED B7 OBDD R OBFO C OBFO CE OBAD R	+ OBF3 3F + OBF4 O5 OBF5 CE OBBA R + OBF8 3F + OBF9 O5	OBFA C6 03 + OBFC 3F + OBFD 34 OBFE 31	31 31 32F	3 3 3 3 3	0C0A 7E 0C94 R 0C0D C1 2F 0C0F 26 0F	+ 0C11 3F + 0C12 2F 0C13 DE 20 0C15 A6 00 0C17 81 4C
		0 4 4 4 4 4 4	0000000	33333333	333333	334444	2 2 2 3	0000	653 654 655 655 656	658 659 660 661 663
1603 1604 1605 1605 1606	त च च च र		1618 1619 1619 1620 1623 1623	1624 1625 1627 1627 1627 1628 1629 1631	SIENT	1638 1640 1641 1641 1642	1645 1645 1646 1647 1648	1649 1650 1651 1651	1653 1654 1655 LT-LS NS' 1655 1654	ਰ ਜ ਜ ਜ ਜ ਜ
CHECK RC 14 NUMBER? 14 YES 15	CHECK CLASS OF TOKEN DELIMITER? IF SO, PARSE AGAIN	IF NOT, PRINT ERROR MESSAGE	CHECK NUMBER BAD (>255) CHECK NUMBER (4 DRIVES) BAD	SAVE NUMBER SET UP DUMMY CLI ISSUE 'LOAD' COMMAND FOR TRANSIENT RESTORE NUMBER X CHECK FOR EKRORS	YES, QUIT IS THERE XFER-ADDRESS? NO, QUIT PASS DRIVE NO. TO TRANSIENT	START IT UP DONE: PRINT ERROR MESSAGE	TURN TO CLI	· · · · · · · · · · · · · · · · · · ·	T A FT-FS LT-LS NS'	DIRECTORY LINE BUFFER 16 16 16 16 17 18 18 18 19 19 19 19 19 19
FCB 47 LDA B RC CHECK RC 1 CMP B #3 NUMBER? 1 BEQ INICD2 YES	LDA A CLASS CHECK CLASS OF TOKEN CMP A #4 DELIMITER? BEQ INICMD IF SO, PARSE AGAIN	LDX #FORMAT IF NOT, PRINT ERROR MESSAGE PRIMSG SWI FCB 49 RTS RETURN TO CLI		MAND FOR TRANSIENT	INICD3 YES, QUIT VALUE IS THERE XFER-ADDRESS? INICD3 NO, QUIT A VALUE+1 PASS DRIVE NO. TO TRANSIENT	O, X START IT UP DONE! #NUMBER PRINT ERROR MESSAGE		#OD DIRECTORY COMMAND DIRECTORY OF DRIVE (A FT-FS LT-LS NS'	#OHOD 40 DIRECTORY LINE BUFFER 1 40D 1 2 NUMBER OF SECTORS USED 1

	GET NEXT TOKEN FROM CLI	CHECK RC NAME? VFC	WILD-CARD NAME?		CRC EX GET TOTAL LENGTH IN			FORMAT NAME INTO BUFLIN	CLEAN STACK	BAD NAME? FORMAT ERROR	1 NO USED SECTORS YET 3 WANT LINE-PRINTER? NO	CB MAKE CONSOLE INTO 'LPT'	NCBGD1, X # YP RCBGD7+1, X # X-1	T+2) MAKE ASCII JRV JR PRINI HEADER LINE 1	PRINT HEADER LINE	B POINT TO SYSTEM FCB
TNC SAVEX	NXTOK SWI FCR 47	LDA B RC	CMP B #2		LDA B DESCRC ADD B SAVEX LDX #BUFLIN	SWI FCB 5 LDX SAVEX1	PSHX SWI		FCB 52 INS INS INS	INS CMP B #2 BEQ DIR3	3 CLR NSEC CLR NSEC+1 TST LPTFLG BEQ DIR7	# ∢		a		PRTMSG SWI FCB 49 LDX #DIRFLD PRTMSG	FCB 49 LDX #SYSFCB TXAB SWI
* * OCKA 7C ORDO R DIRS	<u>:</u>	0C6F I	127	20 D7	0C7B D6 22 DIR6 0C7D FB 0BDD R 0C80 CE 0BAD R	+ 0C83 3F + 0C84 05 0C85 FE 0BDF R	0C88 3F	OCSA	0080 0080 0080 008E	0C8F 31 0C90 C1 02 0C92 27 BE	0C94 7F OBAB R DIRCD3 OC97 7F OBAC R OC9A 7D OBE3 R OC9D 27 OF	S 6	0CA4 B4 02 0CA6 B6 50 0CA8 A7 03 0CA8 A5 54	A7	OCB1 8B 30 OCB3 B7 OB56 R OCB6 CE OB43 R	OCB9 OCBA OCBB C	0CBF 31 0CC0 CE 000C R 0CC3 3F
1725	1727		1733 1734 1735	1736 1737 1738	1739 1740 1741		1746 1747 + 1748 +		1751 + 1752 1753 1754	1755 1756 1757	1759 1760 1761 1761 1762	1764	1766 1767 1768 1768	1770	1773		1782 + 1783 1784 1785 +
NO, FORMAT ERROR	YES, SET FLAG GET NEW TOKEN	TOKEN≐NUMBER? NO	CHECK FOR VALID DRIVE NO. BAD?	+1 CHECK DRIVE NO. DRIVE NO. 0, 1, 2, 3 BAD?	SAVE DRIVE NO.	NUMBER EKROR	RETURN TO CLI	GET NEXT TOKEN FROM CL.I	CHECK RC END OF LINE? YES, USE DEFAULT FILE NAME	COLON? IF NOT, BAD DRIVE NO.	GET NEXT TOKEN FROM CLI CHECK RC	NAME? YES	WILD-CARD NAME? YES	OTHERWISE FORMAT ERROR	RETURN TO CLI	POINT TO NAME SAVE POINTER : GET LENGTH SAVE II GET NEXT TOKEN FROM CLI	CHECK RC PERIOD? NO, EKROR
BNE DIR3	INC LPTFLG BRA DIRCDO	CMP B #3 BNE DIR2	TST VALUE BNE DIR1	LDA A VALUE+1 CMP A #3 BHI DIR1	STA A SAVEA BRA DIRIA	LDX #NUMBER PRTMSG SWI	FCB 49 RTS	NXTOK SWI	FCB 47 LDA B RC CMP B #\$OD BEQ DIRCD3	CMP B #7: BNE DIR1	NXTOK SWI FCB 47 LDA B RC	CMP B #1 BEQ DIR4	CMP B #2 BEQ DIR4	LDX #FORMAT PRTMSG SWI	FCB 49 RTS	LDX DESCRA STX SAVEX1 LDA A DESCRC STA A SAVEX NXTOK SWI	FCB 47 LDA B RC CMP B #'. BNE DIR3
0019 26 37	0C1B 7C 0BE3 R 0C1E 20 E2 *	0C20 C1 03 DIRCD1 0C22 26 26 **	0C24 7D 0027 0C27 26 0B **	0C29 96 28 0C2B 81 03 0C2D 22 05	* 0C2F B7 0BE4 R 0C32 20 06 *	0C34 CE 03B7 R 0C37 3F	+ 0C38 31 0C39 39 *	OC3A 3F	+ 0C3B 2F 0C3C D6 25 0C3E C1 0D 0C40 27 52	0C42 C1 3A 0C44 26 EE	+ 0C46 3F + 0C47 2F 0C48 D6 25	0C4A C1 01 DIR2 0C4C 27 0A	0C4E C1 02 0C50 27 06 *	0C52 CE 03AA R 0C55 3F	+ 0057 39 *	0000	+ 0C63 2F 0C64 D6 25 0C66 C1 2E 0C68 26 E8

1786 + 1787 1788	. 0CC4 02 0CC5 CE 0036 R	FCB 2 LDX #BUFFER XABX	PUT BUFFER ADDRESS IN	1847 + 1848 + 1849	0021 3F 0022 1E 0023 39	SWI FCB 30 RTS	RETURN TO CLI
1789 + 1790 +		4		1850 1851		* * OUTPUT FORMATTED DIRECTORY RECORD HERE	ORY RECORD HERE
1791 1792	OCCA A7 07 OCCC E7 08	STA A FCBDBA, X STA B FCBDBA+1, X	×	1852 1853		IRLST LDX	GET FIRST CHAR. OF BLOCK
1793	OCCE B6 OBE4 R	LDA A SAVEA	NI ON BUILD X	1854	0D26 A6 00	CMP 6 ##20	FIRST CHAR=BLANK?
	i ian	9.	OPEN DIRECTORY	1856	26	BNE *	NO
1796 + 1797 +	. OCD4 17	SWI FCB 23		1858	OD2C 7E 0E31 R	JMP DIRNXT	GET NEXT ENTRY
1798	* OCD5 A6 O5 DIRCD4	LDA		1859 1860	*	PSHX	
1800	27 4B		VAL. ID				
1801	* ocno 81 01	CMP A #1	END-OF-DIRECTORY?	1862 + 1863	OD3O OS OD31 CE OBAD R	FCB 5 LDX #BUFLIN	
1803	26 44		IF NOT, ERROR		1000 1000 1000	PSHX	
1804 1805	OBAB	LDA A NSEC	OUTPUT NUMBER OF SECTORS USED	1865 +	0035 05		
1806	F6			1867	OD36 C6 OC	LDA B #12	COMPARE NAMES (WITH WILD-CARDS)
1808	OCES CE OBBE K	LUA #UIRLNATI		1869 +	OD38 3F		
1809	* CON	VERT BINARY (16	CONVERT BINARY (16 BITS) TO 5 DECIMAL CHARS.	1870 +	0039 35	FCB 53	
1810	[(()	B)=BINARY VALUE - ADDRESS TO PLA	(A, B)=BINARY VALUE (X)= ADDESS TO PLACE CHARS IN ASCII	1871	003A 31 003B 31	SNI SNI	
1812				1873		INS	
1813	FF OBLD	X E	SAVE DATA POINTER	1874	33	SVI	
1814	OCEY DE OBUS R OCEC ZE OBEA R CVIECT	LDX #KIOK	FOIN TO CONSTANT THATE	1876	*	3	
1816	E0 01	SUB		1877	OD40 7E 0E31 R	JMP DIRNXT	NO MATCH, GET NEXT NAME
1817	OCF1 A2 00	SBC A O. X		1878	* OD43 CE OB69 R	LDX #DIRLIN+39	
1819	3			1880	02 C 9 C 9		BLANK OUT DIRECTORY LINE
1820 1821	OCFS 7C OBE4 R OCF8 20 F5	INC SAVEA BRA CVDEC2	NO, BUMP CHAR.	1881 1882	86 27 E7 00	LDA A #39 DIRCD5 STA B 0, X	
1822				1883			
1823 1824	OCFA EB 01 CVDECS OCFC A9 00	S ADD B 1, X ADC A 0, X	RESTORE PARTIAL RESULT	1884 1885	OD4D 4A OD4E 26 FA	DEC A BNE DIRCDS	
1825	36			1886	*		CTACK /TO/ ADDRESS
1826	OCFF FF OBDF R	LDX SAVEXI	SAVE REGISTERS GET POINTER	1887	0D50 3F	SWI	STHEN TO HERICAS
1828	B6 OBE4	Œ	GET CHAR.			FCB 5	
1829	ODO8 8B 30	ADD A ##30	MAKE ASCII	1890	OD52 CE 000C R	LDX #SYSFCB	
1831	33	ľŒ	CONT. CORN.	1892	17 17 17 17 17 17 17 17 17 17 17 17 17 1	×	STACK 'FROM' ADDRESS
1832	08		מיידוויסט זיירסט	1893 +	0057 3F	I ZSC U	
1833	ODOE FF OBDU K	LDX SAVEX	SHVE POINTER RECOVER REGISTER				SET UP FOR MOVE (FROM DATA BLOC
1835	80		MOVE TO NEXT TABLE ENTRY		. !	MOVC	TO DIRECTORY LINE, 12 CHARACTER
1836	0D15 08	INX CPX #KIOK+10	DONE?	1897 +	ODSB 3F ODSC 11	SWI FCB 17	
1838	01	BNE CVDEC1	NO				
1839 1840	ODIB CE OBBD R	LDX #DIRLN2	PRINT END LINE	1900 + 1901 +	0050 3F 005E 06	SWI FCB 6	
1841	, onte se	PRTMSG		1902 1903 +	005F 3F	PULX	RECOVER LINE PUINIER
	ODIF	FCB 49			0060 06 0061 08	FCB 6	
1845)		1906	0062 08	XXI	(St.) Little Cd. Aireach
1846	DIRER	DIRERR PRTERR	PRINT ERROR MESSAGE	1907	0D63 08	INX	PUINI IU LINE (15)

GET FIRST SECTOR	CONVERT HIGH NIBBLE	PUT INTO LINE POINT TO LINE (27)	CONVERT LOW NIBBLE PUT INTO LINE	POINT TO LINE (30)			GET LAST TRACK	CONVERT HIGH NIBBLE		PUT INTO LINE POINT TO LINE (31)	CONVERT LOW NIBBLE		FUINI IU LINE (33)			GET LAST SECTOR	CONVERT HIGH NIBBLE		PUT INTO LINE	1	CONVERT LOW NIBBLE PUT INTO LINE		POINT TO LINE (37)			GET SECTOR COUNT	
LDA A FIBFTS+1, X TAB	OSK DOTHL PULX SWI	STA A O, X INX IDX	JSR OUTHR STA A O, X	X X X	PSHX SWI FCB 5	LDX #SYSFCB LDX FCBIND, X	LDA A FIBLTS, X TAB	BSR OUTHL PULX	SWI FCB 6	Œ	TBA BSR OUTHR		PSHX		LDX #SYSFCB LDX FCBIND, X	LDA A FIBLTS+1, X TAB	BSR OUTHL		STA A O, X	TBA	BSR OUTHR STA A O, X	XXI	INX	PSHX SWI		LDX FCBIND, X LDX FIBNMS, X	TXAB SWI FCB 2 LDX NSEC
A6 10 16 10 0000	OUBE BU OE39 K + ODC1 3F	5000		ODCE OS	h 10	병끱	98 16	ODD9 8D 5E	+ ODDE 3F + ODDC 06		ODEO 17 ODE1 8D 5A	ODES H7 CO	>	! !	병ᇤ	ODEE A6 12 ODFO 16		+ ODF3 3F	ODF5 6	12		ODFD 08		+ 0E00 3F	0E02 (0E05 EE 27 0E07 EE 13	+ 0E09 3F + 0E0A 02 0E0B FE 0BAB R
1969	1972 1972 1973	1975	1978	1981	1983 1984 1985	1986 1987	1988 1989	1990	1992	1994 1995	1996 1997	1999	2001	2002	2004	2006	2008		2012	2014	2015 2016	2017	2019	2020 2021 2021	2023	202	2026 2027 2028 2028
	B TA BLOCK	IISK				(19)			ODE							_											
	POINT TO FCB POINT TO DATA		- C	LOI INIO LINE		POINT TO LINE (GET FILE ACCESS CODE	CONVERT TO ASCII		PUT INTO LINE				POINT TO LINE (23)			GET FIRST TRACK	CONVERT HIGH NIBBLE		PUT INTO I INF	POINT TO LINE (24)	CONVERT LOW NIBBLE	POT INTO LINE	FUINI (U LINE (26)	
PSHX SMI STOP	SYSFCB FCBIND, X	OUTHL	, 40 (X :	JSK GUIHK STA A O. X INX	POINT TO LINE		FCB 5 LDX #SYSFCB	FCBIND, X A FIBACS, X	OUTHL CONVERT	PULX SWI FOR 4	X '0 V		STA A O, X	I PX	POINT TO LINE	IMS		A FIBFTS, X	OUTHL CONVERT HIGH	PULX SWI	FCB 6 STA A 0, X PUT INTO I INF	POINT TO LINE	JEH JSR OUTHR CONVERT LOW NIBBLE STA A D.Y PHI INFO	•	LOINE OF LINE	FCB 5 LDX #SYSFCB LDX FCBIND, X
	+ 0065 05 FCB 5 0066 CE 000C R LDX #SYSFCB 0069 EE 27 LDX FCBIND, X	ODGE BD OE39 R JOSE OUTHL	SWI	10	BD 0E3D R JSK A7 00 STA 08 INX	POINT TO LINE	OD7F 3F SWI		EE 27 LDX FCBIND, X A6 OE LDA A FIBACS, X	OUTHL CONVERT		0.086 A7 00 STA A 0, X	17 TBA	8D 0E3D K JSK A7 00 STA	OS INX	INX POINT TO LINE PSHX		ODSC CE OOOC R LDX	LDA A FIBFTS, X	BD 0E39 R JSR OUTHL CONVERT HIGH	ODA7 3F	3.6 A.0,x	OS INX. POINT TO LINE	OE3D R JSR OUTHR	XNI XNI	CUBS OS TINA PULNE IU LINE	+ ODBS OS COOC R LDX #SYSFCB ODB9 EE 27 LDX FCBIND, X

CLEAN STACK USE SYSTEM BUFFER		IF NOT, ERROR CLEAN STACK (9 BYTES) 7 BYTES FROM SWI 2 BYTES FROM JSR GO TO TRANS. ADDRESS	CHAIN ERROR NAME OF FILE 12 CHARACTERS	CLEAN STACK CLEAN STACK (SWI+JSR)
SWI FCB 17 INS INS INS INS INS INS ELEX #SAVFCB SWI FCB 2 LDX #SAVBUF XAB SWI FCB 4	SIA A FUBBALA, X SIA A FUBBALA, X CLR FCBSTA, X LOADB SWI FCB 37 TST FCBSTA, X BNE CHANER	BEQ CHANER INS	CHANER LDX #CHANME PSHX SH1 FCB 1 LDX #SAVFCB+FCBNAM PSHX SH1 FCB 5 LDA B #12 MOVC SH1 FCB 17	* (0
+ 0E54 3F + 0E55 11 0E56 31 0E57 31 0E59 31 0E59 3F + 0E5D 3F + 0E5E 02 0E5F CE + 0E62 3F + 0E63 3F	0664 A7 0666 E7 0668 E7 0668 SF + 0668 25 066 C6 066 26	0E72 27 0E74 31 0E75 31 0E75 31 0E78 31 0E78 31 0E78 31 0E78 31	0E7F CE 0EB7 R + 0E83 05 - 0E84 CE 08BC R + 0E87 3F + 0E88 05 0E89 C6 0C + 0E88 3F + 0E88 3F + 0E88 3F + 0E88 3F	0E8E 3 0E8F 3 0E8F 3 0E90 3 0E91 C 0E95 4 0E95 3 0E97 3 0E98 3 0E98 3
2092 2093 2094 2094 2097 2097 2100 2101 2103 2103 2103 2103 2103	⊌ BY1E) BYTE)	ASCII	ij	
SAVE 'A' CONVERT HIGH NIBBLE PUT INTO LINE POINT TO LINE (38) RESTORE 'A' CONVERT LOW NIBBLE PUT INTO LINE		POINT TO FCB GET NEW DIRECTORY BLOCK CONTINUE CONVERT LEFT NIBBLE TO	CONVERT RIGHT NIBBLE TO	MOVE 30 CHARACTERS
ADDABX SWI FCB 8 STX NSEC PULX SWI FCB 6 PSH A BSR OUTHL STA A O, X INX PUL A BSR OUTHR STA A O, X	INX TBA BSR OUTHL STA A O, X INX TBA BSR OUTHR STA A O, X LDX *BIRLIN PRTMSG	SWI SWI FCB 49 * FCB 49 * OFTDR SWI FCB 26 R JMP DIRCD4 * OUTHL LSR A LSR A LSR A LSR A	LSK JTHR AND CMP CMP BLS BLS RDD RTS LOAD AND	* * INDEX VORTINGED TOTALS AND THE SHALL BENT SHALL BY SHALL BY SHALL SH
0EOE 3F 0EOF 08 0E10 FF 0BAB R 0E13 3F 0E14 06 0E15 36 0E16 8D 21 0E18 A7 00 0E1B 32 0E1B 32	0620 08 0621 17 0622 80 15 0624 67 00 0627 17 0628 80 13 0626 68 80	0E2F 3F 0E30 31 * 0E31 CE 000C R D 0E34 3F 0E35 1A 0E35 7E 0CD5 R * 0E39 44 0E3B 44 0E3B 44	88 07	# 0E48 CE 08AC R @(0E4B 3F 0E4C 05 0E4D 30 0E4E EE 07 0E50 3F 0E51 05 0E52 C6 1E
2030 2031 + 2032 + 2033 + 2035 + 2035 + 2037 2038 2040 2040 2041	2044 2045 2046 2047 2048 2050 2050 2052 2053	++ ++	2005 2005 2006 2007 2077 2073 2074 2076	++ ++

SWI FCB 26 BRA SEMPT2 KEEP LOOKING FOR EMPTY ** CMP A #TRKSIZ END OF TRACK? BNE *+3 NO, FOUND EMPTY ** RTS YES, OUT OF SPACE ** CLR FCBSTA, X RETURN GOOD STATUS PTS	* SEARCH DIRECTORY FOR UNAMBIGUOUS FILE REFERENCE * SEARCH DIRECTORY FOR UNAMBIGUOUS FILE REFERENCE * CONTAINING DESIRED FILE NAME AND DRIVE NO. * RETURNS ADDRESS OF DIRECTORY BLOCK IN FCBIND * RETURNS STATUS IN FCBSTA * I=FILE NOT FOUND * OFFOUND FILE * OTHERWISE-FRROR CODE	ш		LDX #BUFFER PROVIDE A BUFFER ADDRESS XABX SWI FCB 4 STA A FCBDBA, X STA B FCBDBA+1, X CLR FCBSTA, X INIT. STATUS	OPEND OPEN SWI FCB 23 FILEZ LDA A FCBSTA, X CHEC BEQ SFILE3 STAT	CMP A #1 END OF DIRECTORY? BEG SFILES YES PULX SMI FCB 6 STA A FCBSTA,X NO, EKROR STATUS UMP DIRERR ISSUE ERROR MESSAGE	LDX FCBIND, X LDA A 0, X CMF A #\$20 BNE SFILE4	FNEXT LDX #SYSFCB POINT TO SYSTEM FCB GETDR GET NEXT DIRECTORY BLOCK
2215 + 0EF0 3F 2216 + 0EF1 1A 2217 0EF2 20 E5 2218 0EF4 A6 0B S 2220 0EF6 81 1A 2221 0EF8 26 01 2222 0EF8 39 ** 2224 0EFB 6F 05 2224 0EFB 6F 05) 1 1	+ OEFE 3F + OEFF 05 0E00 04 09	+ +	2247 0F09 CE 0036 R 2248 2249 + 0F0C 3F 2250 + 0F0D 04 2251 0F0E A7 07 2252 0F10 E7 08 2253 0F12 6F 05	+ 0F14 3F + 0F15 17 0F16 A6 05 0F18 27 08	2261 OF 1A 81 01 2262 OF 1C 27 3B ** 2264 OF 1E 3F 2265 + OF 1E 3F 2266 + OF 1F 06 2267 OF 20 A7 05 2268 OF 22 7E 0021 R	0F25 EE 0F27 A6 0F29 81 0F2B 26	2275 OFZD CE OOOC R SFNEXT 2276
#SAVFCB CLOSE FILE: 21 QUITWILL GO TO CLI	ANME RMB 12 SEARCH DIRECTORY FILE FOR EMPTY SLOT USES SYSTEM FCB AND BUFFER PASS DIVE NO. IN 'A' REGISTER RETURNS TRACK, SECTOR OF SLOT IN FCBTRK, FCBSCT RETURNS ERROR STATUS IN FCBTA OFFORIND SID FCBTA	ALUE	26 26 SECTORS/TRACK #SYSFCB POINT TO SYSTEM FCB A FCBDRV, X SET DRIVE NO.	SET BUFFER ADDRESS	OPEN D	STATUS O. K. END-DIRECTORY? YES OTHERWISE ERROR	CHECK FIRST CHAR. OF SLOT BLANK? NO YES, FOUND EMPTY SLOT	POINT TO FCB GET NEXT DIR. BLOCK
INS INS INS INS INS INS ILLS CLOSE SWI FCB 21 RTS CLOAN N ECC / INORY	CHANNE RMB 12 CHANNE FDB \$0A0D * SEARCH DIRECTORY * USES SYSTEM FCB A * PASS DRIVE NO. IN * RETURNS TRACK, SE * RETURNS ADDRESS O * RETURNS ENERGY STA * O-FOLIND SIDT	OTHER	TRKSIZ EQU 26 * SEMPTY LDX #SYSFCB STA A FCBDR' TXAB	SWI FCB 2 LDX #BUFFER XABX SWI SWI STA A FCBDBA,X			LDA A O, X CMP A #\$20 BNE *+3 * RTS * A ** RTS	LDX #SYSFCB GETUR
+ +	0EC3	2175 2176 2177 2178	21177 0EC5 001A 2181 0EC5 001A 2182 0EC5 CE 000C R 2183 0EC8 A7 09 2184	2185 + OECA 3F 2186 + OECB 02 2187 OECC CE 0036 R 2188 2189 + OECF 3F 2190 + OED0 04	+ +	OEDB 27 OEDD 81 OEDF 27 OEE1 7E OEE4 EE		2213 OEED CE 000C R 2214

POINT TO DIR. ENTRY ACCESS CODE=0? 0. K.	NO, CANNOT DELETE	POINT TO FCB ERROR CODE	PROTECTED' SEARCH OPEN FCBS	FCB OUTPUT? NO, KEEP LOOKING GET DRIVE NO. SAVE FCB POINTER	POINT TO THIS FCB SAME DRIVE? YES	RESTORE POINTER GET NEXT FCB KEEP LOOKING DONE:	CLEAN STACK ERROR MESSAGE ERROR CODE	UXH, X A FCBSTA, X RETURN ERROR CODE DELETE ERROR-OPEN OUTPUT FILES \$0D	POINT TO FCB POINT TO DIRECTORY ENTRY GET FIRST T/S OF FILE
LDX FCBIND, X TST FIBACS, X BEQ DEL3	LDX #DELERR PRTMSG SWI FCB 49	TSX LDX UXH, X LDA A #18 STA A FCBSTA, X RTS	R FCC FCB LDX BEQ	⊢ w 1 r	SWI FCB 5 TSX LDX UXH+2, X CMP A FCBDRV, X BEQ DEL32	PULX SWI FCB 6 I LDX FCBNFB, X BNE DEL30 BRA DEL33	2 INS INS LDX #FOPERR PRTMSG SWI FCB 49 I DA A #18	TSX LDX STA RTS FCC FCC	S TSX LDX UXH, X LDX FCBIND, X LDA A FIBFTS, X LDA B FIBFTS+1, X TSX
	OF7E R						DEL32 OFBF R	* F.OPERR	* DEL33
OF 6B EE 27 OF 6D 6D 0E OF 6F 27 24	OF71 CE OF74 3F OF75 31	0F76 30 0F77 EE 05 0F79 86 12 0F78 A7 05 0F7B 39	20 00 27	0F99 6D 06 0F9B 27 0D 0F9D A6 09	0F9F 3F 0F40 05 0F41 30 0F62 EE 07 0F64 A1 09 QF66 27 08	OFA8 3F OFA9 06 OFAA EE 25 OFAC 26 EB OFAE 20 2F	OFBO 31 OFBI 31 OFB2 CE OFB5 35 OFB6 31	0FB9 30 55 55 55 55 55 55 55 55 55 55 55 55 55	OFUF 30 OFEO EE 05 OFE2 EE 27 OFE4 A6 OF OFE6 E6 10
2339 2340 2341	2342 2343 2344 2345 2346 +	2348 2348 2349 2350 2351	2352 2353 2354 2355 2355 2355 2357	2359 2360 2361 2362 2362 2363	2364 + 2365 + 2368 2369	2370 2371 2372 + 2373 + 2375 2375 2377	2378 2379 2380 2381 2383 2383 2384 +	2386 2386 2388 2389 2391 2391	2393 2394 2395 2396 2397 2398
KEEP SEARCHING	STACK DIRECTORY NAME ADDRESS	POINT TO SAVED FCB AD POINT TO NAME FIELD I	STACK SEARCH NAME ADDRESS 12 CHARACTER COMPARISON COMPARE NAMES	CLEAN STACK NO MATCH, KEEP LOOKING	RECOVER DIR. BLOCK ADDRESS		FULX FULX FULX FULX FULX SITA A FCBSTA, X RETURN STATUS RTS E A FILE FROM DISK CSTACKED, POINTS TO FOR WITH FILENAME AND DRIVE ND	RETURN ERROR STATUS IN FCBSTA CHECK ACCESS CODE FOR PROTECTION ERRORS MUST HAVE ALL OUTPUT FILES ON THIS DISK CLOSED SELETE TSX LDX UXH, X BSR SFILE SEARCH DIRECTORY	FOUND FILE? YES NO, QUIT
SWI FCB 26 BRA SFILE2	SFILE4 PSHX SWI FCB 5	LDX 2, X LIM A #FCBNAM ADDAX SWI FCB 9	PSHX SWI FCB 5 LDA B #12 CMPC SWI FCB 18	INS INS INS INS BNE SFNEXT	* SFOUND	PULX SWI SWI FCB 6 STA A FCBIND, X STA B FCBIND+1, X CLR FCBSTA, X GO RTS **	SFILES FOLX SFILES FOLX STA A FCBSTA, X RETI RTS * DELETE A FILE FROM DISK * INDEX: (STACKEN) POINTS:	* CHECK ACCESS CODE * CHECK ACCESS CODE * MUST HAVE ALL OUTP *	TSX UXH, X LDX UXH, X TST FCBSTA, X BEQ DEL2 * RTS
OF30 3F OF31 1A OF32 20 E2	0F34 3F 0F35 05 0F36 30	0F37 EE 02 0F39 86 10 0F3B 3F 0F3C 09	0F3D 3F 0F3E 05 0F3F C6 0C 0F41 3F 0F42 12	m m m m	0F49 CE 000C 0F4C EE 27 0F4E 3F 0F4F 02	0F50 3F 0F51 0A 0F52 A7 27 0F54 E7 28 0F56 6F 05 0F58 39	0F59 3F 0F5A 06 0F5B A7 05 0F5D 39		
2277 + 2278 + 2279	2280 2281 2282 + 2283 + 2284	2285 2286 2287 2288 + 2289 +	2290 2291 + 2292 + 2293 2294 2294 2295 +		2302 2303 2304 2305 2305 +	2308 2309 + 2311 2312 2313 2314 2315 2315	2316 2317 + 2318 + 2320 2321 2323	2325 2325 2327 2328 2328 2330 2331	2332 2333 2334 2335 2336 2337 2337

1	3	7

ş		WRITE UPDATED FREE-SPACE SECTOR	TI.X BAKE FOR TINDITA	×	FF YES		UXH, X POINT TO USER FCB A FCBLTS, X GET LAST T/S OF FILE	S+1, X	FCB USE SYSTEM FCB	BSCT, X	READ THAT SECTOR		BSTA, X ERRORS?	A YES		WEX GET T/S OF OLD FREE-SPACE	SAVEX+1 0, X UPDATE LINKAGES		USE SYSTEM FCB	TI,X MAKE FCB 'OUTPUT' MRITE UPDATED SECTOR			ILLA MAKE FUB 'INPUL'		ROUTINE TO PARSE AN UNAMBIGUOUS FILE NAME	NAME. EXT		FOR ADDRESS PASSED IN INDEX REGISTER	ERRORS RETURN '21' IN STATUS			TA,X NO ERRORS YET RU.X DEFAHLT DRIVE≡O	-		CHECK BY		2 NO		1 NO	.UE+1	(4 DRIVES)
STA	œ.	SWI	FCB 19 CLR FCBDT1, X		* BNE DELSH	TSX		LDA	R LDX #SAVFCB	STA A FUBIRK, X STA B FUBSCT, X	IOHDR	IMS	LDA A FCBSTA, X	BNE DEL3A	rnx	A AG	K STABSA			COM FCBDTT, X TOHDR	IMS	FCB 19	CLR FUBU		* ROUTINE TO PA *	* [DRIVE:] FILENAME. EXT		* FCB ADDRESS P		* @FMTFCB TSX	רים חאר א	CLR FCBSTA, X	NXTOK	SWI	100 B PC		BNE PARS2	TST	BNE PARS1	LDA	CMP A #3
1052 A7 1054 E7	1056 CE 08AC 1059 63 06	+ 1058	+ 105C 13 105D 6F 06		1001	1063 30	1064 EE 1066 A6	1068 E6 22	106A CE 08AC	106F E7 0B		+ 1071 3F + 1072 12	1073	1075 26 96	1077 EE	1079 B6 OBDD	107E A7 00	1081 E7	1083 CE 08AC	1086 63 06	+ 1088	+ 1089 13								1080 30	108E EE	1090 6F 05 1092 6F 09		+ 1094 3F	1096.1	1098 C1	109A 26 29	109C 7D	109F 26 0A	8	1083 81 03
2461 2462	2463 RY 2464 2464	2465	2467	2469			FIELD 24/3 2474	2475	2476		2479	2480	2482	2483 2484	2485	2486	2487	2489	2490	2492	2493	2494	2496	2497	2459	2501	2502	2504	2505	2507	2508	2509	2511	2512	2514	2515	2516		2519	2521	2522
SAVE IN FCB	POINT TO DIRECTORY ENTRY	GET LAST T/S OF FILE		STORE IN FCB	POINT TO DIRECTORY ENTRY		PUL BLANK INIU NAME FIE		MAKE 'OUTPUT'	WALLE OF DHIED DIRECTOR		RESTORE 'INPUT' STATE	GOOD STATUS?	IF NOT, GUIT		GET DRIVE NO.		GET FREE-SPACE SECTOR	\$ 1.25 4.1	IRACK#O SECTOR#3	INPUT	READ SECTOR		ERRORS?	YES	POINT TO SECTOR BUFFER	GET 1/8 OF FREE-SECTOR	SAVE THEM			GET FIRST T/S OF FILE	USE SYSTEM FCB	POINT TO SECTOR BUFFER	UPDATE LINKAGE	SAVE 'A'	USE SYSTEM FCB	UEL DRIVE NUMBER TIMES 2	ACCESS FREE-SPACE TABLE)	
LDX UXH, X STA A FCBFTS, X STA B FCBFTS+4			TSX LDX UXH, X		LDX FCBIND, X	ď		Œ.	STA A FCBDTT, X	SWI	3 19	CLR FCBDTT, X	DEL4	RTS		LDA A FCBDRV, X			LDA B #3	STA B FCBSCT, X	CLR FCBDTT, X	IOHDR		LDA A FCBSTA, X	BNC DELSH	FCBDBA, X	LUM A SECSI/-Z, X	A SAVEX			A FCBF1S, X		FCBDBA, X	STA A SECSIZ-2, X	A	¥ <	LUH H FLBURV, X ASL A	LDX #FRETAB	ADDAX SWI	FCB 9	
0FE9 EE 05 0FEB A7 1F	P E E		OFF5 30 OFF6 EE 05	A7		98	음	86 FF	1007 A7 06	1009 3F		100B 6F 06 100D 6D 05 DFL34	27 01	* 1011 39	:	1012 A6 09 DEL4 1014 CE 086C B	A7 09	88	101B C6 03			1023 3F	1024 13	1025 A6 05	9	EE	1028 A6 /E 1020 F6 7F	B7	1032 F7 OBDE R 1035 30	H	1038 A6 1F	e H	Ë	1041 A7 7E 1043 E7 7E	8	ᆼ			104F 3F	1050 09	
	3 5 5	5 5	cc	¢č	5 ਣ	ਟ :	¥	Ξ.	Ä		+	~ -		_					_ •		_	+			٦.			1		-	~~ ~	~	Π.		* ***		4 =	Ä	+	+	•

POINT TO NAME IN CLI	FORMAT NAME INTO FCB	CLEAN STACK	ERRORS? YES					
SWI FCB 5 LDX SAVEX PSHX	SWI FCB 5	FCB 52 INS INS INS	1NT B BNE PARS3 RTS END					
2584 + 10FA 3F 2585 + 10FB 05 2586 10FC FE 0BDD R	+ 10FF + 1100	+ + 1102 1103 1104 1105	2596 1106 31 2597 1107 5D 2598 1108 26 BF * 2500 110A 39 * 2601 2602					
NOT VALID INIT. DRIVE	POINT TO FCB	RETURN ERROR CODE 21 RETURN NO VALUE	GET A TOKEN FROM CLI CHECK RC COLON? NO, ERROR	GET A TOKEN FROM CLI CHECK RC UNAMBIG. NAME? YES	POINT TO FCB RETURN ERROR STATUS 21	POINT TO NAME SAVE POINTER SAVE POINTER SAVE IT GET A TOKEN FROM CLI CHECK RC PERIOD? NO, ERROR	COUNT PERIOD GET A TOKEN FROM CLI CHECK RC UNAMBIG. NAME?	GET LENGTH OF EXT TOTAL LENGTH POINT TO FCB POINT TO NAME FIELD IN FCB
BHI PARS1 N	PARS1A UXH, X	LDA A #21 STA A FCBSTA, X CLR VALUE CLR VALUE+1 R RTS	NXTOK SWI FCB 47 FCB 8 RC CMP B # 11 BNE PARSI	NXTOK SWI FCB 47 LDA B RC CMP B #1 BEG PARS4	TSX LDX UXH, X LDA A #21 STA A FCBSTA, X RTS	LDX DESCRA STX SAVEX LDA A DESCRC STA A SAVEA NXTOK SWI FCB 47 LDA B RC CMP B #*.	INC SAVEA NXTOK SWI FCB 47 LDA B RC CMP B #1 BNE PARS3	LDA B DESCRC ADD B SAVEA TSX LDX UXH, X LDA A #FCBNAM ADDAX SWI FCB 9
*	* PARS1		* PARS1A	* PARS2	PARS3		* °	* cc
10A5 22 04 10A7 A7 09	30 EE:	85 7F 7F 39	1089 1086 1088 1080 1080	10C1 3F 10C2 ZF 10C3 D6 25 10C5 C1 01 10C7 Z7 08	10C9 30 10CA EE 05 10CC 86 15 10CE A7 05 10D0 39	1001 DE 20 1003 FF 0800 1006 96 22 1008 B7 08E4 100C 2F 100D 06 25 100D CZ 100D CZ 100D CZ 100D CZ 100D CZ 100D CZ 100D CZ	10E3 7C 0BE4 10E6 3F 10E8 D6 25 10E6 C1 01 10EC 26 DB	10FE D6 22 10FO FB 0BE4 10F3 30 10F4 EE 05 10F6 86 10 10F8 3F 10F9 09
2523 2524 2525	2526 2527 2528 2528	2530 2531 2532 2533 2534	2535 2536 2537 + 2538 + 2539 2540 2540	2552 2552 2552 2554 2555 4 2554 4 4	2549 2550 2551 2552 2553 2554	2555 2556 2557 2558 2560 2561 2563 2564 2564	2564 2564 2564 2570 2571 2572 2573	2574 2575 2576 2578 2578 2580 2581 + 2582 +

```
αααα ΣΣ
             00005
00006
00027
00326
0328
0320
00320
2302
2302
2185
           UXH
UXL
VALUE
WARM1
WARM3
WARMST
WD
WD
           SAVV6 0836 R
SAVV7 0836 R
SAVV8 0836 R
SAVVBUF 0865 R
SAVVEND 0784 R
SAVVEND 0784 R
SAVVENT 0809 R
SECURE 0A69 R
SUBMIT 0A65 R
SUBMIT 0A65 R
SUBMIT 0A65 R
SUBMIT 0A65 R
SUBMIT 0A66 R
SUBMIT 0A66
                                                                                                                                                                                                                                                                                 {\tt xxxx}_{{\tt X}}{\tt xxx}_{{\tt X}}{\tt xxx}_{{\tt X}}{\tt xxx}{\tt xxx}{\tt xxx}{\tt xxx}{\tt xx}
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       ααεεαεαααααααααας <u>κ</u>ε
                                                                                                    2301
2267
2267
2267
2267
2067
0003F
0003F
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
000005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
00005
           MUL 16 MUL 18 MUL 19 MU
                                                                                                                                                                                                                                                                                                                                                                                                                                        αααααΣαΣααααΣααααΣαΣααα
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            ~~ × ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~
        0000C
00027
00027
00029
00029
00029
0000R
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00000
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00000
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
00010
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           06BB
246E
097C
09A2
09B5
09BC
      FCBFWD
FCBGDT
FCBLTS
FCBLTS
FCBNAM
FCBNAM
FCBNAM
FCBSCT
FCBSCT
FCBSCT
FCBSCT
FCBSCT
FCBSCT
FCBSCT
FCBSCT
FCBTR
FCB
                                                                                                                                                                                                                                                                                                                                                               FMTS
FOPERR
FORMAT
FORMAT
FORMAT
FORMAT
GETBS
GETBS
GETBS
GETBR
GCHKTR
GETBR
GCHKTR
GETBR
GCHKTR
GETBR
GTICDS
GTCDS
GTCD
                                                                                                                                                                                                                                                                                                                                                  FMIFCB
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   LODICMD
                                                                                                                                                                                                                                                      αααααααααααααααααααααααα
   ακακεκακ
Σ
                                                                                                                                                                                                                                                                                                                                                                                                            \alpha\alpha\Sigma\alpha\alpha\alpha\alpha
0E48
0000
0F5E
108D
03FE
0966
0471
0397
0003
2232
2232
2230
068A
                                                                                                                                                                                                                                                                                 0696
0300
262A
0633
0680
0680
                                                                                                                                                                                                                                                                                                                                                                                                                                                 0036
0BAD
               eCLDST
eDELET
eDELET
eSTEMP
eLOADB
eLOADB
eRNYTOK
ePRYTER
ePRYTER
ADDARX
ADDAX
ADDAX
ADDAX
ADDAX
ADDAX
BANCMD
ASNCMD
ASNCMD
ASNCMD
ASNCMD
BANNER
BASERU
```

FCB 19 TST A BNE: OPEND2 YES * OPEND1 LDX FCBIND, X POINT TO DATA BLOCK TST O.X FIRST CHAR=0? BEG OPEND3 YES, EOF * TSX LDX UXH, X POINT TO FCB CLR FCBSTA, X RETURN NO ERRORS RTS RETURN	* OPEND2 STA A FCBSTA, X RETURN ERROR CODE * OPEND3 TSX LDA A #1 STA A FCBSTA, X RETURN STATUS=1 ** ** ** ** ** ** ** ** **	** ** ** ** ** ** ** ** ** **	SUBJECT STATE STAT
+ 0018 3F + 0019 13 001A 4D 001B 26 0C 001D EE 27 001F 27 09 0021 27 09 0023 30 0026 6F 05 0026 6F 05	0022 47 05 002B 39 002C 30 002D EE 05 002F 86 01 0031 47 05	0034 30 0035 EE 05 0037 A6 27 0038 CB 20 0038 CB 20 0037 A7 27 0043 EE 07 0045 36 0049 37 0044 32	004B 3F 004D 27 08 004B 27 08 005B EE 05 0053 47 05 0053 47 05 0055 20 C6 005B EE 05 005B EE 05 005B EE 05 005B EE 05
0061 4 0062 4 0064 0064 0065 0066 0066 0067 0068 0068 0068 0068 0068	0075 0075 0076 0077 0078 0078 0080	0083 0085 0085 0087 0089 0092 0093 0095 0095 0096 0096 0096 0098	0100 0100 0100 0100 0100 0100 0110 011
50 500 W"	S IN FCBSTA: O=BLOCK FOUND 1=END OF DIRECTORY ELSE ERROR ATA BLOCK IN FCBIND EQUATES STATUS FLAGS	DIRECTION BUFFER ADDRESS DRIVE NO. TRACK NO. SECTOR NO. FILE-NAME FIELD BUFFER INDEX NTERS 128 BYTES/SECTOR 26 SECTORS/TRACK 32 BYTES/DIRECTORY BLOCK	SYSTEM FCB LOCATION POINT TO FCB INPUT TRACK=0 SECTOR=4 (START OF DIRECTORY) X NO ERRORS X 1, X POINTER TO BUFFER START X INIT. DIR. BLOCK POINTER 1, X READ SECTOR
OPEN, CP-68 eOPEND eGETDR ePUTDR ADDRES MUST S	IRN STATU ESS OF D ADDRESS FROU 5	FCBDTT EQU 6. DIRECTION FCBDBA EQU 7 BUFFER ADDRE FCBDRA EQU 7 BUFFER ADDRE FCBDRA EQU 10 TRACK NO. FCBRAM EQU 10 TRACK NO. FCBRAM EQU 11 SECTOR NO. FCBRAM EQU 12 FILE—NAME FIFCBIND EQU 39 BUFFER INDEX * * REGISTER POINTERS * UXH EQU 5 UXH EQU 5 UXH EQU 5 UXH EQU 5 UXH EQU 128 * SECSIT EQU 128	N ENT GOPEND N ENT GETUR N ENT GETUR X * EXT SYSFCB SY X EXT SYSFCB SY CLR FCBDTT, X TR CLR FCBDTT, X TR CLR FCBDTT, X TR CLR FCBTRK, X TR LDA A #4 SEI LDA A FCBSTA, X NO CLR F
0000 0000		0000 0007 0000 0007 0000 0008 0000 0008 0000 00027 0000 0000 0000 0000 0000 0010	0000 0003 0000 0034 0000 7E 0000 0003 30 0004 6F 06 0008 6F 07
0001 0003 0003 0005 0005 0007 0007 0009 0011	0015 0015 0017 0018 0018 0020 0021	0023 0024 0025 0026 0028 0039 0033 0035 0035 0036	0042 0043 0044 0045 0046 0048 0050 0052 0053 0053 0055 0055

PULLAL PULX PUTDR KCBDEF KEAD

REWIND SECSIZ SUBABX SUBAX

```
ΣžΣ
                                                                          Σ
                                                                                                                                       ΣΣ
SUBXAB 2265 P
SYSFCB 0000 F
TABX 219C P
TRKSIX 0010
TXAB 2183 P
UXH 0005
UXH 0005
WILL 23D2 P
XABX 2185 P
 X X X Z E E E E E E E E E E
                                                                                                                                                                                                                                                                         ž
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    60ETDR 0034
60PEND 0003
60DRAX 2232
60DBX 2233
60DBX 22
                                                                                                                                                                                                                                                                                                                                                                                                           21 BYTES TO MOVE
MOVE FROM FCB TO DIR. BLOCK
                                                                                                                                                                                           POINT TO NAME FIELD IN FCB
                             POINT TO FCB
GET ADDRESS OF DIR. BLOCK
STACK IT
                                                                                                                                                                                                                                                                                                                                                                                                                                                                  POINT TO SYSTEM FCB
MAKE OUTPUT
ISSUE I/O REQUEST
                                                                                                                                                      GET ADDRESS OF FCB
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               TSX
LDX UXH, X POINT TO FCB
STA A FCBSTA, X RETURN STATUS
RTS
                                                                                                                                                                                                                                                                                                                                                                                                             CLEAN STACK
                                                                                                                                                                                                                                                      STACK IT
              EPUTDR TSX
LDX UXH, X F
LDX FCBIND, X C
                                                                                            SWI
FCB 5
TSX
LDX UXH+2, X G
LDA A *FCBNAM
ADDAX
                                                                                                                                                                                                          0076 3F
0077 11
0078 31
0079 31
0070 31
0076 51
0077 62 0001 R
                                                                                              0069 3F
006A 05
006B 30
006C EE 07
006E 86 10
                  0064 30
0065 EL 05
0067 EE 27
                                                                                                                                                                                                                                                                     0072 3F
0073 05
0074 C6 15
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     ន
ខ
                                                                                                                                                                                                               0070 3F
0071 09
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            0081 3F
0082 13
0083 30
0084 EE 0
0086 A7 0
0123
00123
00124
00125
00127
00127
00128
00133
00135
00145
00145
00145
00145
00145
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
00153
0
```

142

	142		DPEN
NUMBER OF SECTORS NEXT FCB IN ACTIVE CHAIN INDEX INTO DATA BUFFER SPACE COMPRESSION FLAG FILE NAME (8.3 + EOT=13) FILE TYPE FILE ACCESS CODE FIRST TRACK/SECTOR LAST TRACK/SECTOR NUMBER OF SECTORS 128 BYTES/SECTOR	S ACTIVE-FCB-CHAIN HEAD LINK FREE-SPACE POINTERS (4 DRIVES) POINT TO FCB	START OF ACTIVE-FCB CHAIN EMPTY CHAIN? FOUND FCB?	YES, ERROR GET NEXT CHAINED FCB IF NOT END OF CHAIN, LOOP IF END, OK POINT TO FCB ERROR STATUS (FILE ALREADY OPEN
	* EXT SYSFCB * BASE PAGE POINTERS * FCBCHN EQU \$29 FRETAB EQU \$28 GOPEN TSX TXAR	SMI SMI FCB 2 LDX FCBCHN BEG OPEN3 * OPEN1 PSHX SWI FCB 5 SWI SWI SWI SWI FCB 12	FCB 6 BEG OPEN2 * LDX FCBNFB.X BNE OPEN1 * BRA OPEN3 * SWI OPEN2 TABX SWI FCB 3 LDA A #13 STA A FCBSTA.X
0000 0023 0000 0025 0000 0027 0000 0027 0000 0006 0000 0006 0000 0011 0000 0080 0000 0080 0000 0080 0000 0080 0000 0080 0000 0080	0008 7E 0000 0008 0029 0008 0028 0008 30	cc	0018 3F 0019 06 0010 EE 25 001E 26 F4 0020 20 07 0022 3F 0023 03 0024 86 0D 0024 86 0D
0061 + 00062 + 00063 + 00063 + 00063 + 00064 + 00065 + 00065 + 00067 + 00067 + 00073 + 00072 + 00072 + 00072 + 00072 + 00083 + 00083 + 00083 + 00084 +	0087 0087 0090 0092 0094 0095		01109 01110 01111 011113 01114 01115 01117 01118 0120 0121
* SEQUENTIAL FILE I/O PACKAGE * SEQUENTIAL FILE I/O PACKAGE * GOPEN * GOPEN * GOPEN * GOPEN * GOPEN * CLOSE SEQUENTIAL FILE * CLOSE SEQUENTIAL FILE * CLOSE SEQUENTIAL FILE * CLOSE SEQUENTIAL FILE * EWRITE * WRITE A BYTE INTO A SEQUENTIAL FILE * EWRITE * REMIND A SEQUENTIAL FILE * INUEX REGISTER (STACKED) POINTS TO FCB * CHARACTEKS PASSED IN 'A' REGISTER * STATUS CODES: (IN FCBSTA) * CHARACTEKS PASSED IN 'A' REGISTER * STATUS CODES: (IN FCBSTA) * O=GOOD * O=GOOD * = FILE IN USE * S=FILE IN USE * S=FILE IN USE * S=FILE IN USE * S=FILE FOUND * S=TOO MANY FILES FOR DIRECTORY * S=TOO MANY FILES FOR DIRECTORY * S=TOO MANY FILES FOR DIRECTORY * S=DISK FULL * S=END-FILE FOUND * S=BND-FILE FOUND	13=1L 18=1L 21=BA EGISTER EGU EGU	ADDRESSING POIL RCRDEF EQU 2 EQU 5 EQU 5 EQU 6 EQU 6 EQU 6 EQU 6 EQU 6 EQU 6 EQU 7	FCBSTA EQU 5 STATUS FCBDTT EQU 6 DATA TRANSFER TYPE FCBDRA EQU 7 DATA BUFFER ADDRESS FCBDRA EQU 7 DRIVE NUMBER FCBRCT EQU 11 SECTOR NUMBER FCBFWD EQU 12 FWD LINK TRACK/SECTOR FCBRAK EQU 14 BACK LINK TRACK/SECTOR FCBRAM EQU 14 FILE NAME (8. 3+EOT=13)) FCBTYP EQU 29 FILE TYPE FCBRCS EQU 30 FILE ACCESS CODE FCBFTS EQU 31 FIRST TRACK/SECTOR
z ç	ळ के एं∙	0000 0000 0005 0000 0000 0000	0005 0006 0007 0009 0006 0000 0010 0011 0011 0011
0000 0000		C	
			+ + + + + + + + + + + +

٦	L	7
_		4

GET FORWARD LINKS	POINT TO FCB PUT IN LINKS	POINT TO SECTOR BUFFER GET BACKWARD LINKS		POINT TO FCB PUT IN BACKWARD LINKS		INIT. BUFFER INDEX		CHAIN		MAKE FCB END OF CHAIN	SEARCH CHAIN FOR END LINK	EMPTY CHAIN?	GET FCB ADDRESS INTO (A, B)	NEGOTI	, Chair	RESTORE FCB ADDRESS		SOUD SIMIOS	AT END OF CHAIN?	ON	AT END OF CHAIN?	CONTRACTOR OF CHAIN ADDRESS	CONTROL WIND TO THE TABLE		GET FCB ADDRESS INTO (A, B)			PATCH CHAIN	RESTORE FCB ADDRESS		GOOD STATUS
LDA B 0, X LDA B 1, X TSY	ع د	a <u>r</u> ∢		LDX UXH, X STA A FCBBAK, X		щα	STA A STA B	* * PUT FCB ONTO ACTIVE-FCB CHAIN * COMMON TO READ AND WRITE		OPEN4 CLR FCBNFB, X		BNE OPENS *	TSX FIDA 6 UXH. X	ф <	ĽΩ	TABX	FCB 3	CLR FCBSIA, X RTS		BNE OPEN6	TST FCBNFB+1, X BNF OPFNA			FCB 5 Tex	LDA A UXH+2, X	LDA B UXL+Z, X	SWI	FCB 6 STA A FCBNFB, X		SWI FCB 3	
006C A6 00 006E E6 01	P E	0075 E7 0D 0077 EE 07 0079 A6 02			E7	800				9 E	유범	0096 26 OE	0098 30	E 6		+ 0001 3F	00A2 0	0043 6F 05 0045 39		00A8 26 16	000A 6D 26	ì	OOME	+ 00AF 05	00B1 A6 07	00B3 E6 08	00B5	ξŞ	00B9 E7 26	+ 00BB 3F + 00BC 03	OOBD 6
0184	0187	0189	0192	0194	0196	0199	0201	0203 0204 0204	0206	0208	0209	0211	0213	0215	0217	0218		0221	0223	0225	0227	0229		0232	0234	0235		0238	0240		
aca et tarted		READ OR WRITE? READ	WRITE	R INPUT	SEARCH DIRECTORY CHECK STATUS	COODS (AID SILE)		POINT IN TYPE FIELD IN FCB		STACK ADDRESS		POINT TO FCB	2	POINT TO TYPE FIELD IN DIR. BLOCK		STACK ADDRESS		8 BYTES TO MOVE FROM DIR. TO FUB.			CLEAN SIACK		PUINT TO FUB	001000/7000t +181	INIT: THEN SECTOR	READ FIRST SECTOR OF FILE		ERROR? No	RETHEN FRENE STATUS		POINT TO SECTOR BUFFER
RTS RTS	1 38 E	TST FCBDTT, X BEQ OPENR *	WHO OPENW	* OPEN SEQUENTIAL FILE FOR *	OPENR JSR SFILE TST FCBSTA, X		STA A FCBSTA, X RTS	* OPENR1 LDA A #FCBTYP Annox	IMS		SWI FCB 5	TSX I DX UXH+2.X	LDX FCBIND, X		SET	PSHX MHX	. 10	& #	SWI FCB 17		INS INS		LUX UXH, X LDA A FCBFTS, X	æ	STA B FCBSCT, X	答.	SWI FCB 19	TST A BEG OPENR2			* OPENR2 LDX FCBDBA, X
0123 0028 39 0124	+ +	0128 002B 6D 06 0129 002D 27 03	0130 0131 002F 7E 00C4 R 0132	0133 0134	0032 BD 0005 R 0035 6D 05	/2 /500	0137 0037 88 04 0140 003B A7 05 0141 003B 39	0142 0143 003E 86 1D	0145 + 0040 3F	7500 +	0148 + 0042 3F 0149 + 0043 05	0150 0044 30 0151 0045 FF 07	0047	× 1	0155 + 0048 3F 0156 + 004C 09	0157 0150 ± 0040 35	٠+	C	0162 + 0051 3F 0163 + 0052 11	0023	0165 0054 31 0166 0055 31	0057 30	0169 0058 EE 05 0170 005A A6 1F	005C E6	0172 005E A/ 0A 0173 0060 E7 0B		0175 + 0062 3F 0176 + 0063 13	0065 4	2007 07	68 6900	0183 006A EE 07

	FREE-SPACE TABLE			PACE RECORD)								L	l			FFER	SPACE		CE TABLE PUINIE	•	144	4	0 ig 0 i	HDLE								œ						ú	n						
TABLE INIT. YET? YES	R AND INIT.	and of thing	MAKE 'INPUT'	TRACK=0 SECTOR=3 (ERFF-SPACE RECORD)			ISSUE READ COMMAND		PUT 'OUPUT' BACK	ERROR?	OZ.	RETURN FRROR CODE	CLEAN STACK		QUIT	POINT TO DATA BUFFER	GET T/S OF FREE SPACE		RECOVER FREE-SPACE TABLE PUINIE		PUT BACK ON STACK			INI. TREE-STACE			AT END OF DICK?	YES	i i	AT END OF DISK?		END OF DISK ERROR					INIT, FIRST T/S	SAT TANT OF THIS	LOTUS TO LANGUA	UPDATE DIRECTORY			GOOD		
TST B BNE OPENW6	O IN FREE-SPACE	OPNW4 TSX		LDA A #O	4		IOHDR	SWI FCB 19	COM FCBDTT, X	TST A	BEG OPENWS	* STA A ECBSTA. X		INS	RTS	OPENWS LDX FCBDBA, X	LDA A SECSI7-2, X	LDA B SECSIZ-1, X	PULX	ECB 6	PSHX		ທີ່	STA B 1, X	; 1	OPENW6 TSX		BEG OPNWAS			BNE OPNWEB	PNW6A LDA A		NS INS	ST8		Œ	m	SIA A FUBIRK, X	PUTER	IMS	FCB 27	BEO OPENETA		SNI
0116 5D 0117 26 27	i i	8	1 1 1 1	011E 86 00	A	E		0126 3F	_	012A 4D	012B 27 05	20.70.05				0132 EE 07	8	0136 E6 7F	0	. 0138 3F		0130	013B 05	013C A7 00	ì	င္တ		0143 40	ì		0147 26 07	0149 86 07		0140 31			A7	E7	0154 A7 0A	ì		0159 15	015A 6D 05	ì	015E 31
0306	0308	0310	0312	0314	9317	0317	0318	0319 +		0322	0323	0324	0326	0327	0328	0330	0331	0332		0335			+ 8880	0339	0341	0342	0343	0344	0346	0347	0348	0320	0351	0352	0354	0355	0329	0357	0358	0360		0362 +	8980	0360	0366
	GET NEXT LINK IN CHAIN		FILE FOR COIPUI	SEARCH DIRECTORY	FILE FOLIND	NO		ERRUR SIRIUS (FILE EXISIS)				SEARCH FUR DIK. SPACE	CHECK STATUS			NO KOUM IN DIRECTURY?				KEIUKN EKKUK SIRIUS		ERROR STATUS (NO ROOM)		SET DIE BLOCK ANDRESS FROM SVS	The state of the s		POINT TO FCB	SAVE ADDRESS	INIT. NO. SECTORS=0		INIT. LAST T/S=0,0	INIT, BACKWARD POINTERS		GET DRIVE NO.	CIMIL KANGE (U-G)	ACCESS FREE-SPACE TABLE				SHVE INDLE FULKIEN		GET FREE T/S		NO	Q.
* RTS	OPEN6 LDX FCBNFB, X BRA OPEN5		* UPEN SEGUENTIAL FILE F	OPENW JSR SFILE		Ö	•	LUA A #3 STA A FCRSTA.X			OPENW1 LDA	USR SEMPTY	4	OPENW4		CMP 6 #1				STA A FUBSIA, X	*	OPENW3 LDA A #6	BRA	* COECHIES TO A COECHIND X	1 P P			STA A FCBIND, X				CLR FCBBAK, X		A FCBDRV,	ANII A #\$U3		a.	IMS	FCB 9	בהת. השני	FCB 3	LDA A O, X	m	Name of the	*******
00BF 39	00C0 EE 25 00C2 20 E2			00C4 BD 0005 R	C T	27		00CD 86 03	è é		A6 09	0004 BD 0002 R	A 050 7	00DC 27 0E		00DE 81 01	ì			00E5 A7 05			20	00EC 04 33	5 H	ဓ	H !	00F3 A7 27) 19	4 9	1 9 i	OOFF 6F 0E	Å.	8	0105 84 03		ļ	010B	· 010C 09	0100 35		010F A6 00	0111 E6 01	0113 40	0119 27 03
0245	0247	0250	0251	0253	000	0256	0257	0258	0260	0261	0262	0263	026	0266	0267	8970	0220	0271	0272	0273	0275	0276	0277	0278	0280	0281	0282	0283	0284	0286	0287	0288	0520	0291	0292	0223	0295		0297 +	8670		0301	0302	0303	0304

_	4	

SWI FCB 6 PME NOTEND NO	rB, X	d d	ರ	NOTHND CMP A FCBNFB, X AT DESIRED FCB? BNF NXTFCR NO		CMP B FCBNFB+1,X AT DESIRED FCB? BNE NXTFCB NO			(A, B) POINT	>H000 ★	IMS	ın.	TABX POINT TO THIS FCB	1357 T T T T T T T T T T T T T T T T T T T		B FCBNFB+1, X	PULX POINT TO PREVIOUS FCB	LEG A	STA A FCBNFB, X UPDATE ITS LINKAGE	B FCBNFE+1	* BKA CLUSEZ FINISH PROCESSING	XTFCB LDX	BNE NOTFND IF NOT END OF		NOCHN TABX POINT TO THIS FCB	IMS	(**	CLDM M #13 KETUKN ERKUK CUDE (CAN'T FIND FCB) STA A FERSTA, Y		*	×≱r:		CLUSEZ 15X 1 DY 11YH: Y DOINT TO THIS FOD	V. X. BEAD OR WRITE?	CLOSEW		RTS READ IS DONE		CLUSEW IST FUBIRK, X AT END UF DISK? BEQ CLSW1 YES		TST FCBSCT, X AT END OF DISK?	CLOWA	CLSW1 LDA A FCBBAK, X
+ 01A9 3F + 01AA 06 010B 24 00	01AD A6 25	26	20	01B7 A1 25 01B9 26 14	ì	01BB E1 26 01BD 26 10					OIBF	. 0100 05	Č			01C5 E6 26	2007		01C9 A7 25		01CD 20 08	01CF EE 25	01D1 26 E4		,	0103 3F	0110	4	ဗို				01DR FF 05	9	56		01E1 39	4	01E4 27 04	!	01E6 6D OB	ì	01EA A6 OE
0429	0432 0432 0433	0435	0438	0439	0441	0442 0443	0444	0445	0447	0440	0450 +	0451 +	0452	0454 +	0455	0456	7040	0459	0460	0461	0462	0464	0465	0466		0468 +		0471	0472	0473	0474	0,47	0470	0478	0479	0480	0481	0482	0483	0485	0486	0488	0489
CLEAN STACK QUIT	MAKE 'INPUT' ISSUE READ COMMAND	RESTORE 'OUTPUT'	CHECK FOR ERROR GOOD	X RETURN EKROR CODE		CLEAN STACK QUIT		GET FORWARD POINTERS	COTATO DIGAT DOVOCA DIGATED	NECOVER TREE STACE INDLE FOIRIER		OUT OF SPACE?		OUT OF SPACE?	Q		NEIGRN ERROR CODE (UC) OF SPHCE)	POINT TO FCB	~		UPDATE FREE-SPACE TABLE				POINT TO DATA BUFFER	-4 C. FAR ALT BUELER					don of Fished	FULN: 10 FUE			GET FCB ADDRESS	GET HEAD OF FCB CHAIN	NO ACTIVE FCBS?	S TOPS			AT DESIRED FCB?		RESTORE X
INS RTS	OPENW7 CLR FCBDTT, X IOHDR SWI	FCB 19 COM FCBDTT, X	TST A BEQ OPENWS	* STA A FCBSTA, X		RTS	2	UPENWS LUX PLBUBH, X LDA A O, X	LDA B 1, X	IMS		TST A		TST	BNE OPENW9	*	/# H H07		STA A FCBSTA, X	* RTS	PENW9 STA A	STA			LDX FCBDBA, X	CPNW9A CLR 4.X	X	DEC B	BNE OPNW9A	*	X		ectose TSX	LDA A UXH, X	B UXL, X	FCBCHN	BEG NOCHN	× ×			SUBABX	FCB 12	
015F 31 0160 39	0161 6F 06 + 0163 3F	Ç	0167 4D 0168 27 05		0160 31	016E 39	100	1 8 1	0173 E6 01	F 0175 3F		0177 40	gn 97 9/10	0174 50	017B 26 08	70 70 UZ 10	3 8	H		0184 39	0185 A7 00		င္က မ	# E	018C EE 07	ر 14	80		0194 26 FA		0197 50	7F 0084	30		Εę	01A1 DE 29			0105	0146 05	0197		
0367 9369 0369			0375 0376	0377	0379	0381	0387	0384	0385	0387 +		0386	0391	0392	0393	0394	0396	0397	0398	0399	0401	0402	0403	0404	0400	040	0408	040	0410	0411	0412	0414	0416	0417	0418	0419	0420	0421		0424 +	0425		0428

LDA A O, X GET TRACK/SECTOR LDA B 1, X TSX		A SECSIZ-2, X PUT NEW T/S B SECSIZ-1, X		IOHDR WRITE OUT UPDATED SECTOR	IMS	FUB 19 STA A FORSTA, X SAVE STATUS	RTS	UXH, X	FCBDTT, X	BEG READ2 UK	LDA A #18 ERROR CODE	H LCDSIH	3	EAD2 LDA A	LDA B FCBIND+1, X	4	CMP A		CMP	* BEG READS YES	EADZA LDX FCBIND, X	LDA A O, X OEI BYIE	CHECK FOR SPACE COMPRESSION	* IN SPACE-COMPRESSION MUDE, BYIE™ NECHTIVE STACE COOK! *	BPL READ2C NOT A COMPRESSED SPACE	**		FCBSCF, X		LDX FCBIND, X POINT TO BUFFER	NOTI ST		LDA A #\$20 IF SO, REPLACE WITH SPACE.		NOTLST STA A O, X PUT NEW CHAR. IN BUFFER LDA A #\$20 OUTPUT A SPACE	TSX STA A LACK	UXH, X	CLR FCBSIA, X GUUD SIMIUS RTS DONE		
0551 023C A6 00 0552 023E E6 01	0241	0245 A7	0249 30	0559 024A EE 05 0560	+	0562 + 024B 13	0250 39	0566 0251 30 0547 0252 FF 05	0254 6D	0256	0258	0236 A/	0573 0250 37	0250	0576 025F E6 28	0263 A2	0265 B1	0580 0268 26 05	026A F1	0583 026D 27 36	026F EE	0586 0271 A6 00	0588	0589	0591 0273 2A 1E	0592	0276 EE	0278 6D	7		0599 027E 4C	27 1772	0602 0281 86 20	X7 C07A	0605 0285 A7 00 0606 0287 86 20	0289	028C EE	0610 028E 6F 05		
	ERROR FIX-UP FOR END-OF-DISK	WRITE OUT LAST SECTOR OF FILE			ONE MORE SECTOR IN COUNT			2	SK RMATION		MAKE 'INPUT' FIND DIRECTORY SLOT		POINT TO FCB BESTOBE (GITPHT)		0000	TE NO GOOD, PINT		GET LAST TRACK WRITTEN	SECTUR PUT INTO FCB POSITION		WRITE DATA INTO DIRECTORY	COLLEGE GOOD	SOUD WRITE? YES		NO 600D, 4011::	MAKE / INPUT/	INHUNEO SECTORE3 (FREE-SPACE RECORD)		READ FREE-SPACE RECORD			KENIUKE TUULFUL	NO	BAD READ, QUIT!!	GET DRIVE NO	LIMIT RANGE (0-3)	2 BYTES/ENTRY ACCESS FREE-SPACE TABLE			
	SIA B PUBSULY BRA CLSW3	CLSW2 IOHDR WF	SWI FCB 19	LDA A FCBNMS, X	B #1	4	STA B FCBNMS+1, X	F	* LASI SECTOR NOW ON DISK * UPDATE DIRECTORY INFORMATION		CLSW3 CLR FCBDTT, X JSR SFILE			FCBSTA, X	BEG CLOSE3	* *		∢			PUTDR	3 27	IST FUBSIA, X BEG CLOSE4		* RTS	LOSE4 CLR FO	DA B #3	Œ	STA B FCBSCT, X IOHNR	SWI	FCB 19	COM FCBDILL X	BEO	*	* * * * * * * * * * * * *		ASL A		SWI FCB 9	
OIEC E6	0492 01F0 E7 08 0493 01F2 20 0E		0496 + Olf4 3f 0497 + Olf5 13	00	OIFA CB	01FC 89	0502 01FE A7 23 0503 0200 E7 24	0504	0505 0506		0202		0511 0208 EL 05	0200 60	020E 27	0515		0211 A6	8 24 24	0217	0522 0523 + 0219 3F	+	0525 021B 6D 05 0526 021D 27 01		0528 021F 39 0529	0220 6F	0531 0222 86 00 0532 0224 CA 03	0226 A7	0534 0228 E7 0B	+ 022A	+ 022F	0538 022C 63 06	022F	0541	0233	0234 84	0546 0236 48 0547 0237 FF 002B		0549 + 023A 3F 0550 + 023B 09	

: : : : :	@WRITE TSX LDX UXH, X POINT TO FCB TST FCBDTT, X CHECK FOR OUTPUT BNE WRITE2 OK *	LDA A #18 STA A FCBSTA, X RETURN ERROR CODE RTS QUIT	WRITE2 LDA A FCBIND, X CHECK F LDA B FCBIND+1, X WRITZO SUB B FCBDBA+1, X SBC A FCBDBA, X	* *	MRITZA TSX LDA A UA, X GET CHARACTER TO BE WRITTEN LDA UXH, X POINT TO FCB TST FCBSCF, X IN SPACE-COMPRESSION MODE? BEQ WRITZB NO * * HANDLE COMPRESSION OF SPACES HERE	AND A #\$7F CMP A #\$20 BNE WRIT2B LBX FCBIND, X LDA B O, X	* DEC B ONE MORE SPACE ADDED BPL SPC128 SPACE COUNT>128? * STRSPC STA B O, X PUT COUNT BACK TSX LDX UXH, X POINT TO FCB CLR FCBSTA, X GOOD STATUS RTS DONE	B ##FF STRSPC STRSPC	SWI FCB 2 TSX LDX UXH, X POINT TO FCB STA A FCBIND, X SAVE NEW POINTER STA B FCBIND+1, X
02E5 02E7 02E9	0679 OZEC 30 0680 OZED EE 05 0681 OZEF 6D 06 0682 OZF1 26 05		02F8 A6 27 02FA E6 28 02FC E0 08 02FE A2 07	0303 0303 0305	0698 030A 30 0699 030B A6 04 0701 030B EE 05 0702 0311 27 27 0703	0707 0708 0313 84 7F 0709 0315 81 20 0710 0317 26 21 0711 0319 EE 27 0713 0318 E6 00	0315 24 0316 56 0320 26 0322 E7 0324 30 0325 EE	032A C 032C 2	0730 + 032F 3F 0731 + 0330 02 0732 0331 30 0733 0332 EE 05 0734 0334 A7 27 0735 0336 E7 28
			6666	00000		0 6 6 6 6 6 6 6		363633	6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6
POINT TO BUFFER	MUVE BUFFEK FUINIEK STACK IT	RETURN BYTE GET POINTER FROM STACK	POINT TO FCB SAVE NEW POINTER X X SOOD STATUS	DONE AT END OF FILE?	NO X AT END OF FILE? NO RETURN END-FILE STATUS	GET FORWARD LINK T/S X PUT LINK INTO T/S READ LINKED SECTOR	FEROR CHECK RETURN ERROR CODE X X RE-INIT. INDEX SKIP FOUR BYTES OF LINK X	POINT TO BUFFER GET NEW FORWARD LINKS POINT TO FCB STORE FORWARD LINKS	POINT TO BUFFER GET NEW BACKWARD POINT TO FCB
FCBIND, X POINT	JNX MUVE PSHX STACK SWI FCB 5	A UA+2, X RETURN BYTE A GET POINTER FROM S B		DONE A FCBTRK, X A FCBLTS, X AT	AT END OF FILE? NO RETURN END-FILE	A FCBFWD, X B FCBFWD+1, X A FCBTRK, X B FCBSCT, X OR	EAD3A RETURN ERROR CODE FCBDBA, X FCBDBA+1, X #4 RE-INIT. INDEX #0 SKIP FOUR BYTES OF FCBIDA1, X FCBIDA1, X	FCBDBA, X POINT TO BUFFER A 0, X GET NEW FORWARD B 1, X CUXH, X POINT TO FCB A FCBFWD, X STORE FORWARD LI	

GET DRIVE NO. LIMIT RANGE (0-3) ACCESS FREE-SPACE TABLE	Z BYIES/ENIKY GET NEXT TRACK	GET NEXT SECTOR SAVE INDEX TO FREE-SPACE TABL	POINT TO FCB	NEW TRACK TO GET	NEW SECTOR	READ IN SECTOR		REPLACE 'OUTPUT'		O.		RETURN ERROR CODE	CLEHIN STREN			GET NEW LINK TRACK	GET NEW SECTOR	RECOVER PREE SPACE INDEX	1	PUT LINK INTO TABLE		POINT TO FCB	SET FORWARD LINKS	GET BUFFER ADDRESS		RE-INIT. BUFFER INDEX			GET BACKWAKU LINK	POINT TO BUFFER	PUT IN BACKWARD LINKS		ZERO OUT REST OF BUFFER				CONTINUE WITH NEW SECTOR	DISK FULL ERROR	Disk Folk Envior
	ASL A ADDAX SWI SWI FCB 9 LDA A 0.X		FCB 3 TSX LDX UXH+2, X		STA B FCBSCT, X	IOHDR		FCB 19 COM ECRDIT. X	TST A	BEQ WRITES		STA A FCBSTA, X	0 2	RTS			LDA B 1, X	PULX		STA A 0, X	STA B 1, X			SIA B FUBLWD+11.X	m	m	ADC A #0 STA A FCBIND, X		a		Œ	STA B 3, X	4, X	X		BNE WRITE6	* JMP WRITZA	* WDITE7 DA A: #7	WKITE/ LIM N #/
	0.393 48 + 0.394 3F + 0.395 09 0.396 A6 00	0398 E6	+ 0398 05 0390 30 0390 FF 07	A7	03A1 E7 0B			+ 03A6 13	3 5			03AC A7 05	03AE 31			H	03B5 E6 01	+ 0387 3F	0388	4	03BB E7 01	3 #	A7	03C2 E7 0D	2	8	03CA 89 00	E S	03D0 A6 OE		A7	03D8 E7 03	3 12		å å	03E0 26 FA	03E2 7E 030A R	20 70 1300	03E5 86 07
0797	0800 0801 0803 0803	0805 0806 0807	0808	0811	0812	0813	0815	0816	700	0819	0820	0821	0822	0824	0825	0826	0828	0829	0831	0832	0833	0835	9836	0837	083	0840	0841	0843	0844	0846	0847	0848	0820	0851	0852	0853	0854 0855	0856	0857
ONTINUE WITH SPACE	OINT TO BUFFER HAR. ALREADY THEKE? ES	TORE CHARACTER IN BUFFER OVE POINTER		OINT TO FCB	PUT NEW INDEX IN FCB	X OOD STATUS	ONE		UEL UKIVE NU.	ACCESS FREE-SPACE TABLE				GET FREE-TRACK			END OF DISK?		POINT TO FCB	POINT TO DATA BUFFER	NEW FORWARD LINK TRACK	NEW FURWHALL LINA SECTOR	POINT TO FCB	WRITE OUT SECTOR		ERROR?	ON	RETURN ERROR STATUS		MAKE ERROR RETURN		GET SECTOR COUNT	INCREMENT IT				GET TRACK JUST WRITTEN	15	×
BRA WRITZO CONTINUE WITH SPACE *	* WRITZB LDX FCBIND, X POINT TO BUFFER TST 0, X CHAR. ALREADY THEKE? BNE SPC128 YES	STA A O, X STORE CHARACTER IN BUFF INX MOVE POINTER TXAB	SMI SMI SECOND			STA B FCBIND+1, X CLR FCBSTA, X GOOD STATUS	DONE	2	FCBURY, X	#FRETAB ACCESS FREE-SPACE	A TWO BYTES/ENTRY	ADDAX			WRIT3A END	2	BEQ WRIT3A END OF DISK?	*		FCBDBA, X	A 0, X		UXH, X	œ	1 M.O.		BEG WRITE4	* STA A FCBSTA, X RETURN ERROR STATUS		* WRIT3A .MP WRITE?	*	OET	B FCBNMS+1, X	0# 0#	∢	B FCBNMS+1, X	FCBTRK, X GET TRACK JUST FCBSCT. X GET SECTOR	A FCBBAK, X PUT	C
BRA WRIT20	LDX FCBIND, X POINT TST 0, X CHAR. BNE SPC128 YES	00 STA A O, X STORE CHARACTER IN BUFF INX MOVE POINTER TXAB	⊢ 00	EE 05	A7 27 STA	B FCBIND+1, X FCBSTA, X GOOD	39 RTS DONE	*	A6 09 WRITES LIPS A FUBLIKY, X	#FRETAB ACCESS FREE-SPACE	48 ASL A TWO BYTES/ENTRY		0357 3F SWI	3 Y GFT	27 18 BEQ WRIT3A END	*	WRIT3A END	*	UXH, X	EE 07 LDX FCBDBA, X	A7 00 STA A 0, X	OI SIABI,X TSY	EE 05 LDX UXH, X	IOHDR	+ 0.35b) 3F 581	036F 4D TST A	06 BEG WRITE4	STA A FCBSTA, X	39 RTS	WAP MRITEZ	*	A6 23 WRITE4 LDA A FCBNMS, X GET	E6 24 LDA B FCBNTS+1, X	OO ADC A #O	A7 23 STA A	E7 24 STA B FCBNMS+1, X	A6 OA LDA A FCBTRK, X GET TRACK JUST	A7 OF STA A FCBBAK, X PUT	E7 OF STA B

```
Σ
                                                                                                                                                                                                                                                                                                        ααααπαααααα
                                        E C C C C C C C C C C C
       2335
2465
2465
2267
2267
2267
0326
0187
0187
00187
00028
00028
00028
00028
00028
00038
00038
00068
00068
00068
00068
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   0100
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    0149
      PULLAL
PULX
PULX
PULX
PULX
PULX
PULX
PULX
RUBBA
RCBDBA
RCBDTT
RCBDTT
RCBDTT
RCBDTT
RCBDT
R
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 ΣΣ
    019C
000B
0251
03ED
02EC
2219
2222
2248
2220
242A
0000
243A
021D
0220
                                                                                                                                                                                                                                                                  eccose
erend
                                                                                                                                               ERROR CODE (REWIND OUTPUT FILE)
                                                                                                                                                                                                                                                                                                                              RETURN ERROR STATUS
                                                                               POINT TO FCB
CHECK FOR INPUT
OK?
                  POINT TO FCB
                                                                                                                                                                                                                                                                             CHECK STATUS
OK?
                                                                                                                                                                                                                                                                                                                                                              RE-OPEN FILE
                                                                                                                                                                                                                              CLOSE FILE
                                                                                                                                             LDA A #18
STA A FCBSTA, X
RTS
                 UXH, X
A FCBSTA, X
 TSX UXH, X
STA A FCBSTA,
RTS
TSX
TSX
LDX UXH, X
TST FCBDTT, X
BEG REWD2
                                                                                                                                                                                                                                                            FCB 21
TST FCBSTA, X
BEQ REWD3
                                                                                                                                                                                                             CLR FCBSTA, X
CLOSE
                                                                                                                                                                                                                                                                                                                                                                                               ŝ
                                                                                                                                                                                                                                                                                                                                                                                         FCB
RTS
                                                                                                                                                                                                                                                                                                                                                             OPEN
                                                                                                                                                                                                                                                is
                                                                                                                                                                                                                                                                                                                            RTS
                                                                                                                                                                                                                                                                                                                                                                                                                                            S
                                                                                                                                                                                                  *
REWD2
                                                                                                                                                                                                                                                                                                                                                *
REWD3
                                                                 PREMD
                                                                                                                                               03F4 86 12
03F6 A7 05
03F8 39
                                                                                                                                                                                                                                           03FB 3F
03FC 15
03FD 6D 05
03FF 27 01
                                                                               S S S
                 ន
ខ
                                                                                                                                                                                                               03F9 6F 05
                                                                                                                                                                                                                                                                                                                                                                           0402 3F
0403 14
0404 39
 33.47 EE 33.
                                                                                                                                                                                                                                                                                                                            0401 39
 03E7
03E8
03EA
03EC
03EU
03EE
03F2
```

23EC 24F0 24BC 253E

150	
-----	--

INIT. CREG ISSUE CLEAR COMMAND SEEK TRACK 0	FOR SINGLE-SECTOR READ TA FROM FCB NESS IN (A, B) NAI CB 3 A FCBDRV, X CB 7 A MOVE UNIT BITS	~~~	SEND UNIT/SECTOR DRIVE OK? YES NO. DRIVE BAD A=TRACK SEEK TRACK 5 RETRIES	ISSUE READ COMMAND GET STATUS CRC ERROR? NO RESET ERROR FLAGS	TRIED ENOUGH? NO, KEEP TRYING RETURN ERROR CODE=5 DDAM?
LDA A #\$2C STA A CMDCTL LDA A #CLEAR STA A CMDDAT LDA A #SEEKTO JSR OUTCMD		202	F USR XMITUS. USR DRIVCK BCC GETBFO BRA QUIT O TBA USR SEEKTK LDA B #5	LDA LDA BIT BEO	DEC B BNE GETBF1 LDA A #5 BRA GUIT 2 BIT A #\$80 BEQ GETBF3
2C * * 80 * 0C * 00F3 * *		7 D B B B B B B B B B B B B B B B B B B	1.67 R 0.0 R	02 GETBF1 00F3 R ECOO 08 0A *	65 * * * * * * * * * * * * * * * * * * *
001C 86 2C 001E B7 EC 0021 86 80 0023 B7 EC 0026 86 0C 0028 BD 001	+ 002C 3F + 002D 03 002E 66 09 0031 46		003A BD 000 003D BD 01 0040 24 02 0042 20 49 0044 17 0045 BD 01	86 85 27 85 85	
0061 0063 0063 0065 0065 0068 0068 0069		0082 0083 0085 0086 0087 0087 0090 0091 0091	0095 0096 0097 0098 0100 0101 0102 0103	0105 0106 0108 0108 0110 0111	0113 0115 0115 0115 0118 0119 0120
ICOM D/WRITE P-68 SYSTEM INITIALIZE INTERFACE READ A SECTOR WRITE A SECTOR	DATA/STATUS INPUT DATA/STATUS CONTROL COMMAND OUTPUT COMMAND CONTROL DATA OUTPUT CONTROL EFINTIONS	READ WITE READ CRC SEEK SEEK SEEK TRACK O LOAD UNIT/SECTOR LOAD WRITE BUFFER CLEAR DEFINITIONS	US BUFFER ADI NUMBER K NUMBER OR NUMBER RN 'A' REGI X-REG (REGI	.WTSEC CALLED AS SUBROUTINES :NTERFACE CLEAR CONTROL REGISTER DDR=INPUT	
* DISK DRIVEKS FOR ICOM * SINGLE-SECTOR READ/WRITE * TO BE USED WITH CP-68 SYSTEM * ENT @INTDK INITIALIZ N ENT . WISEC WRITE A SE * PIA DEFINITONS	EGU EGU L EGU L EGU TROL C	READX EQU \$02 WRITEX EQU \$04 RDCRC EQU \$04 SEEK EQU \$06 CLREKF EQU \$06 LDTRAD EQU \$10 LDUS EQU \$30 SHFTKB EQU \$30 CLEAR EQU \$80 * FCR ADDRESS DEFIN	CBSTA EQU 5 CBDBA EQU 7 CBDRV EQU 7 CBTRK EQU 10 CBSCT EQU 11	* NOTE: . RDSEC AND . WTSEC C * INITIALIZE DISK INTERFACE * EINTDK CLR INCTL CLEAR CLR CMDCTL CLR OUTCTL CLR INDAT DDR=IN	
0000 0000 0000 0000 0000 0000 0000 0000		0000 0004 0000 0004 0000 0006 0000 0006 0000 0010 0000 0030 0000 0040	0000 0005 0000 0007 0000 0009 0000 0009 0000 0008	0000 7F EC01 0003 7F EC03 0006 7F EC07	88 B7 B B
0001 0002 0003 0004 0005 0007 0008 0009	0012 0014 0015 0015 0017 0019 0020	0022 0023 0024 0025 0026 0029 0031 0032	0035 0035 0036 0038 0038 0040 0041 0042	0045 0046 0047 0048 0049 0050	0059 0059 0059 0059

GET A BYTE	OUTPUT BYTE	LOAD WRITE BUFFER	LOOP UNTIL DONE	RESTORE TRACK RESTORE U/S	SEND U/S DRIVE OK? VFS	NO, DRIVE BAD	A=TRACK SEEK TRACK	5 RETRIES	SEND WRITE COMMAND	SEND CHECK CRC COMMAND	GET STATUS OK? VES	RESET ERROR FLAGS	RETRIED 5 TIMES YET?	YES, ERROR CODE=5	SET RC	UNIT/SECTOR FROM 'A'	CLEAR ERROR FLAGS	SEND LOAD U/S COMMAND		JA 'A'	SAVE COMMAND CLEAR BUSY FLAG	RESTORE COMMAND OUTPUT COMMAND	DONE? WAIT FOR DONE	CLEAR BUSY
∢ .	Q Q	STA A CMDDAT DEC B		PUL B	* JSR XMITUS R JSR DRIVCK BCC WRIBE1	BRA	* WRTBF1 TBA R JSR SEEKTK	LDA B #5	WRTBF2 LDA A JSR OL *	LDA ASR	LDA A INDAT BIT A #\$08 BEO HOTBES	*	DEC B BNE WRTBF2	LDA A #5 BRA QUIT	* WRTBF3 CLR A BRA QUIT *	TRANSMIT				* UDIFUT CUMMAND FROM 'A'		PUL A STA A CMDDAT	OUTCM1 LDA A INCTL BPL OUTCM1	* LDA A INDAT RTS
86 80	B7 86	00B1 B7 EC02		00B7 33 00B8 32	00B9 BD 00E7 F 00BC BD 011A F 00BF 24 02	70	00C3 17 00C4 BD 010C F	0007 06 05		38 30 30	00D3 B6 EC00 00D6 85 08	. Bi	OODE 26 E9	00E0 86 05 00E2 20 A9	00E4 4F 00E5 20 A6		OOE7 BD 0104 R	86 B7	00F2 39			00F7 32 00F8 B7 EC02	OOFB B6 ECO1 OOFE 2A FB	0100 B6 EC00 0103 39
0183	0185	0187	0189	0191	0193 0194 0195 0195	0197	0199 0200 0201	.0203	0205 0206 0206	0208	0210 0211 0211	0213	0215 0216 0216	0218 0218 0219	0221 0221 0222 0223	0224	0270 0270 0270	0228 0229	0230	0232	023 4 0235	0236	0239 0239 0240	0241 0242 0243
ō.	œ		ROL REGISTER			ROL.					v	1												œ
RETURN ERROR CODE=9	8 128 BYTES IN SECTOR	Ü	CTL INIT. COMMAND CONTROL REGISTER FTRR	T REAL	AI GETA BYTE SAVE IT C		FIRB STRUBE DAT READ T BUFFER	GET DATA BYTE MOVE TO BUFFER		YES, SET RC	Y DETLIEN ANY CONTENTS	GET RCBADR	OIA, A KELUKN SIALUS	GLE SECTOR WRITE PASSED IN (A, B)	POINT X TO FCB	DRV, X		MOVE UNIT BITS SCT, X	TRK, X DBA, X	TO DISK		ADDRESS	SAVE U/S SAVE TRACK	B 128 BYTES IN BUFFER
LDA A #9 RETURN EKROR BRA QUIT	BYTES IN SEC	**************************************	STA A CMDCTL INIT.	* STA A CMDDAT	LDA A INDAT GET PSH A SAVE * LDA A **2C	* STA A CMDCTL	LDA A #SHFTRB STA A CMDDAT CLR CMDDAT *	PUL A STA A O. X	B DONE GETBF4 NO	CLR A YES, SET		UXH, X GET RCBADR A FCBSTA, X		SET UP FOR SINGLE SECTOR ADDRESS OF FCB PASSED IN	WTSEC TABX POINT X SWI FCB 3			A FCBSCT, X	ěě		A=U/S	* B=TKACK * X=BUFFER ADDRESS *	RTBUF PSH A SAVE	
0064 86 09 LDA A #9 RETURN EKROR 0066 20 25 BRA QUIT	* 0068 C6 80 GETBF3 LDA B #128 128 BYTES IN SEC	LDA A	B7 ECO3 STA A CMDCTL INIT.	0071 B7 EC02 STA A CMDDAT	ECOO LDA A INDAT GET	007A B7 EC03 * STA A CMDCTL	LDA A #SHFTKB STA A CMDDAT CLR CMDDAT	0085 32 PUL A 0086 47 00 STA A 0.X	0088 08 INX 0089 54 DEC B DONE 008A 26 DE BNE GETBF4 NO	* CLR A YES, SET	* QUIT TSX QUIT TSX CTA A 110, Y BET146N AA	EE 07 LDX UXH, X GET RCBADR AA 05 ORA A FCBSTA, X	* * * *	SET UP FOR SINGLE SECTOR ADDRESS OF FCB PASSED IN	#TSEC TABX POINT X + 0.097 3F SWI FCB 3	0099 A6 09 LDA	0095 46 ROR 0090 46 ROR	46 ROR A FCBSCT, X	E PA	WRITE A SECTOR TO	A=U/S		PSH A SAVE	LDA B #128 128 BYTES IN BUFF

36 250A -L 2151	a ~ L	C 0006 2.23R8 M 2.32R8 M 2.38.4 M (7.0008 (7.0000C TK 010C R FRB 0040		2302 0004 0003	:225 233
PRTMS6 PSHALL	PSHX PULLAL PULTOR PUTOR QUIT	KDCRC HADD READY REWIND SEEK SEEKTO SEEKTO SHFTTRB	SUBABX SUBBX SUBBX SUBBXAB TABX TABX TABX UA	WRITE WRITEX WRIBFO WRIBFO WRIBFO	WRTBF2 WRTBF3 WRTBF3 XABX XABX XAITUS
002C 0097	×× 00 5		₩ .Χ₩₩₩	000B 0005 000A 2940	
¯, ¯, «	ध ददद ळः		3033Wili	. ii. ii. ii. ii. ii	. COOOOXIII ILLLIAARAAQOOOOOO
	SAVE U/S ISSUE RESET-FLAGS COMMAND RESTORE U/S	OUTPUT TRACK SEND LOAD-TRACK-ADDRESS COMMAND SEND SEEK COMMAND		DRIVE OK	RETURN ERROR=10
* * CLEAR ERROR FLAGS	* ERFRST PSH A LDA A #CLRERF STA A CMDDAT PUL A RTS	* SEEK TRACK IN 'A' * SEEKTK STA A OUTDAT LDA A #LDTRAD STA A CMDDAT LDA A #SEEK LDA A #SEEK USR OUTCMD	। धुरदन		PRVCK1 SEC LDA A #10 * RTS * END
	0104 36 0105 86 0A 0107 87 EC02 010A 32 010B 39		0119 39 0110 B6 EC00 011D 84 20 011F 26 03	0121	0127 99 0A 0127 39
0244	0246 0247 0248 0249 0250 0251	0253 0253 0254 0255 0255 0257 0258	0260 0261 0263 0263 0264 0265 0266	0268 0269 0270 0271	0277 0273 0274 0275 0275

		"T00"			A "FROM"			13100	1000			FIX STACK				GET "="				OK?	ON.		GE I DEVZ NAME			NOME	9			VALID?	Q.			"100"	2			A "FKOM"			COLINT								72					
* SAVE DEVI NAME		LDX #DEV1	Y NHX	FCB 5	LDX DESCRA	PSHX	T MAN			IMS	FCB 17	INS	INS	SNI	SNI	NXTOK		FCB 47	œ		BNE ASSNI		NATOR	ECB 47	THA B BC	CMP B #1			Ω		BNE ASSN1	100	SHVE LIEVZ NAME	I DY #DEV2		IMS	FCB 5	LDX DESCRA	PSHX	¥ 0.00				FCB 17	SNI	SNI	o Car		SEE IF DEV1=DEV2		LDX #DEV1	PSHX	1 20 1	
		0025 CE 011D R	+	+	002A DE 20		+ 0020 35	002F		+	+ 0031 11	0032	0033		0032				90	ن د	003C Z6 U3	*	+ 003€ 3€	. +	0040	0042 C1 01	26		90	ដ	004A 26 C5	* *	* *	004C CE 0120 R		+ 004F 3F		0051 DE 20	+ 0053 35	0054	0055 (0057			005A 31	00.00		* *	*	005D CE 011D R	16 0700 1	+ 00041 05	1000
0061	0062	0063	0065	9900	2900	8900	0020	0071	0072	0073	0074	0075	9200	0077	0078	0079	0800	0081	2800	0083	1000	0900	0000	0088	6800	0600	0091	0092	6600	0094	2600	0000	0098	6600	0100			0103	0104			0108			0111	0112	0113	0115	0116	0117	0118	0119		
Ne		DEVICE HOSIGNMEN! TRANSLEN! MODIINE SYNIAX: ASSIGN DEVI-DEV?	TERMINATE WITH AN ESCAPE, CR		DESCRIPTION APPROPRIATION		CURRENT CHAR (2)	TOKEN RETURN CODE	TOKEN CLASS	-	TOP OF FCB CHAIN (2)	14	_	ENE OF TRANSIENT AREA (2)	NEXT AVAIL TRANSIENT AREA (2)		DELETE LINE CHAR		_	WIDIN: CHRIS/LINE	TAB CHAR	DE IDI FX: FF=H. OO=F	EJECT COUNT		ш		•	WIDTH CHARS/LINE		FUIND SHVE FUINIER IU FUINB	THOM	GET DEVICE NAME			₫.		ESCAPE?		YES, DONE		"SYNTAX ERROR"						NAME?	ON	į		WO WO			
NAM ASSIGN	************			*	BASEUU DECCO EGI 420		Eau		EGU			AB EQU	9		EM EM	EUC	DL EQU #3A				Carlo H	FOL	EQU	EGU	EQU	EGU	INT EQU \$4	L.W.D EQU \$46	<	STA R PDT	Q.	ASSNO NXTOK		FCB 47				BNE ASSNZ	RTS			PRIMSG	IN SEC	TCB 49	ANDONA *	ASSNZ LDA B RC	<u>a</u>	Œ		LUM B DESCRU	ď			
N 0000 0000					0000 0000	0000		+ 0000 0025		0000		0000	0000		0000	0000	+ 0000 0038				0000	+ 0000 0040		+ 0000 0042				+ 0000 0000	0 0110	0003 F7 011C R	N OTTO /L			+ 0007 2F	0008 DE 20		5 3	000E Z6 09	0010 39		OO11 CE OOF1 R ASSN1		+ 0014 94 42 4100 4	7 7 7 7	/E 00/E R	0019 D6 25	01	26	ć	0011 06 22				
0001	2000	0000	0002	9000			0010	0011									0000					0026					0031	7 2000	0000	0035	0036	0037		6800	0040	0041	200	0043	0045	0046	0047		000 000 000 000 000		0052	0023	0054	0055	0056	\ 000 000 000 000	0000	0900		

Q	DEV1 OK INVALID NAME		x=A(ADDR1) OF DEV1 2, X 0, X 3, X 1, X sinxT EV2 B FOR DEV2	FOUND EKROR GET DEV2 ADDR2 SAVE FOUND
LDX #DEV2 PSHX SWI FCB 5 LDA B #3 CMPC SWI FCB 18 INS INS INS INS INS BNE ASSNA	SEARCH	* R ASSNXT LDX #MSGC PRTMSG SWI FCB 49 GTCMD SWI FCB 48 LDX DESCRA STX CUCHAR R N	DEV1=DEV2 SSN3 LDA A LDA A LDA A STA A BRA AS DEV1 . NE. E SEARCH PDTA	ASSN4 LDX *DEV2 JSR PDSRCH BCC ASSN5 * BRA PMSGB * ASSN5 LDX 2, X PSHX SM1 FCB 5 * LDX *DEV1 JSR PDSRCH BCC ASSN6
0062 CE 0120 R + 0065 3F + 0066 05 0067 C6 03 + 0069 3F + 0068 31 0068 31 0065 31	0071 CE 011D R 0074 BD 00BB R 0077 24 13 0079 CE 00FE R	+ 007b 31 007E CE 0112 + 0081 3F + 0083 3F + 0083 3F 0085 DE 20 0087 DF 23 0087 7E 0006	0.08C A6 02 A8 0008E A6 02 A9 009E A6 03 0092 A7 01 0094 20 E8 **	0096 CE 0120 R 0097 BD 00BB R 009C 24 02 009E 20 D9 0040 EE 02 + 0042 3F + 0043 05 0044 CE 011D R 0047 BD 00BB R 0048 24 02
0122 0123 0123 0124 0127 0128 0132 0133	0135 0137 0138 0138 0140 0141 0143	0145 0146 0147 0152 0152 0153 0155	0159 0158 0159 0161 0162 0164 0165	0168 0169 0170 0171 0172 0174 0175 0176 0177 0178 0178 0178

EKBOB			SAVE IN A.B		GET DEU2. ADDRO	DE 4.27		SWITCH		NO 95516N					STACK DEV ADDRESS		PUINI 10 PUINB	STACK PDTAB PTR		!	COON			FOUND			GET PDTAB PTR		FIX POINTER						STACK PDIAB POINTER			GET DEVICE ADDR POINTER			RESET		SAVE			
* BRA PMSGB	*	* X= A(ADDR1) DEV1	* ASSN6 TXAB		FCB 2	SEL	FCB 6	XABX			STAB 1.	2	*	* SEARCH PUTAB	PDSRCH PSHX	. F.C.		* PDSRCA PSHX		in.	LDA B #3	J.E.S.		BEG POSRCB	* NO MATCH		PULX	TMS	×		FCB 10	XX	×	XVI	PSHX		FCB 5	LDX 2, X		SBA	SUBAX	FCB 13			FCB Z	STA A 2, X
87 06 JA00	į			+ OOME 3F	+ 00AF 02	+ 00BO 3F			+ 00B2 3F	֓֞֞֞֞֜֞֞֞֞֓֓֓֓֓֓֞֟֜֓֓֓֓֓֓֓֓֓֓֓֓֓֓֓֓֓֓֓֡֓֡֓	0084 H/ 00	Æ				C 05	00BD FE 011B R		+ 0000 3F	S .	00C2 C6 03	+ 0004 35	0005 12	00C6 27 23				+ 0008 3F	200			80 2200	00CF 08				+ 0001 05		8	0007 10		+ 0008 + + 0008 01		OODA	+ 000B 02	00DD A7 02
0183	0185	0186	0187	0189	0190	0191	0193	0194		0196	0197	0199	0200	0201	2020		0206	0207	0209		0211	0212	0214	0215	0216	0218	0219	0220	0222	0223	0224	0225	0220	0228	0229	0230	0231	2520	0234	0235	0236	0237	0239	0240	0241	0243

```
ΣΣ
                                                                                                                                                                                                                                                                                                                                          Σ
                                                                                                                                                                                                                                                                                                                                                                                      2 C E
  AUDDA 2219 A
AUDDA 2222 A
AUDDA 2222 A
AUDDA 2222 A
AUDDA 2222 A
ASSNO 0006 R
ASSNA 0009 R
ASSNA 0009 R
ASSNA 0009 R
ASSNA 0006 R
ASSNA 0009 R
BME 0003 CCHEN 0004 CCHEN 0004 CCHEN 0004 CCHEN 0004 CCHEN 0004 CCHEN 0004 CCHEN 0003 CCHEN
               SET PDTAB POINTER
                                                                                                                                                                                                        GET POTAB POINTER
                                                           OF TABLE?
                                                                                                                                                                                                                                                                                                                                                                                 FCC /INVALID DEVICE NAME/
FCB #OD
                                                                                                                                 FIX STACK
                                                                                                                                                                                                                                                 FIX STACK
                                                                                                                                                              2
                                                                                                                                                                                                                                                                               SET RC
                                                           N ON
                                                                                                                                                                                                                                                                                                                                      FCC 'SYNTAX ERROR'
FCB #OD
                                                                                                                                                              SET
                                                                                                                                                                                                                                                                                                                                                                                                                           FCC 'ASSIGN- '
FCB 4
STA B 3, X
PUL.X
SWI
FCB 6
TST 0, X
BNE PDSRCA
                                                                                          * YES NOT IN TABLE
*
                                                                                                                                                                                                                                      •
                                                                                                                                                                                          *
PDSRCB PULX
SWI
FCB 6
INS
INS
CLC
RTS
                                                                                                                                 INS
INS
SEC
RTS
                                                                                                                                                                                                                                                                                                                                                                                                                                                                      RATE STATE
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              EN3
                                                                                                                                                                                                                                                                                                                                                                                                                                                                   PDTAB
DEV1
DEV2
                                                                                                                                                                                                                                                                                                                                        MSGA
                                                                                                                                                                                                                                                                                                                                                                                 MSGB
                                                                                                                                                                                                                                                                                                                                                                                                                           MSGC
                             00E1 3F
00E2 06
00E3 6D 00
00E5 26 D9
  OODF E7 03
                                                                                                                                                                                                                                                                                                                                                                                                                                                                     0002
0003
0003
                                                                                                                                                                                                                    00EB 3F
00EC 06
00ED 31
00EE 31
00EF 0C
                                                                                                                               00E7 31
00E8 31
00E9 0D
00EA 39
                                                                                                                                                                                                                                                                                                                                     00F1 53
00FD 0D
                                                                                                                                                                                                                                                                                                                                                                                 OOFE 49
0111 OD
                                                                                                                                                                                                                                                                                                                                                                                                                          0112 41
011A 04
                                                                                                                                                                                                                                                                                                                                                                                                                                                                   011B
011D
0120
```

OPEN 234F
OPEND 239E
PUSRCA 0000
PUSRCA 0000
PUSRCH 00BB
PUSRCB 0079
PWTFRS 2454
PWTFRS 2454
PWTFRS 2105
PWLX 2105
P

CRC ERROR? NO	RESET ERROR FLAGS	RETRY READ	FATAL ERROR	DDAM? NO	VES. EATAL FURNE		128 BYTES IN SECTOR	'L INIT. CMD. CNTL. REG.		1	SAVE DATA BYTE	L RESET CMD. CNTL. REG.	danalia desa sacata, ti		RECOVER DATA BYTE		COUNT DOWN		DONE!		ROM /A/ REGISTER			CLEAR BUSY	IT ISSUE COMMAND		WAIT FOR DONE	CLEAR VBISV										
BIT A #\$08 BEG RDSEC2	LDA A #\$0A STA A CMDDAT	BNE	* JMP ERROR	RDSEC2 BIT A #\$80 BEQ RDSEC3			LDA B	RDSEC4 LDA A #\$3C STA A CMDCTL	LDA A #\$40 STA A CMNDAT	< <	PSH A	Iα	LDA A #\$40	ເປົ້	PUL A			BNE RDSEC4	RTS	x‡: x	* * OUTPUT COMMOND FROM 'A' REGISTER	*	OUTCMD PSH A			₫	BPL OUTCD1	TAUNT A AUT		*	END							
010C 85 08 010E 27 0B	0110 86 0A 0112 B7 EC02		0118 7E E113	011B 85 80 011D 27 03	1		ర్ట		0129 86 40 0128 87 FC02	8 1	0131 36	B2	0137 86 40	<u>1</u>	013F 32	8	e d	0144 26 DE	0146 39				36	0148 86 5000	B7	B6	0152 2A FB		0157 39									
0185	0188 0188 0189	0191	0193	0195	0197	0199	0200 0201	0202	0204	9020	0207	0209	0210	0212	0213	0215	0216	0217	0219	0220	0227	0223	0224	0225	0227	0228	0228	0230	0232	0233	0234							
28 NEED NEW SECTOR? YES	3YTE	MOVE POINTER	GDT-73 1961 77907	j		NOT LAST	EOF-GO TO TRANSFER ADDRESS		GET FORWARD 1/S LINK	UPDATE PRESENT T/S	BEAD MEM CECTOR	NEHD NEW	GET DATA BYTE	RE-INIT. INDEX		ROUTINE					SAVE SECTOR	RESET ERROR FLAGS		AUTORIT SECTOR	LOAD U/S COMMAND		CHECK DRIVE STATUS	VES VES		ON		T TRACK	ISSUE LICHE INHICK ADDRESS CONTINUE	ISSUE 'SEEK' COMMAND	FIVE RETRIES		ISSUE 'READ' COMMAND CHECK STATUS OF DRIVE	
CPX #BUFFER+128 BEQ GETSEC		SIX INDEX RTS	* GETSEC LDA B PTS	(∢ @		BNE GETS2	BRA BOOT4	LDX #E	LDA B O, X	C 004	STA A PTS+1		LDA A O, X	STX INDEX	RTS	SINGLE-SECTOR READ ROUTINE	0-1,00	UKIVE=0 TRACK='B'	SECTOR='A'	BUFFER='X'	EC PSH A	LDA A		STA A DITTIAT	Œ	₫	LDA A INDAT	: ‡		JMP ERROR	TRA		CTA A CMUNAT	T C	ESK COICMU		BSR OUTCMD LDA A INDAT	
0046 8C 0090 C 0069 27 07	06 00 08 00	00AE FF 0096 C 00B1 39	F6 0094 C	0093	i	26 02	* 00C2 20 DA	* 00C4 CE 0010 C GE1S2		F 7	0095 000	당당	00D6 A6 00		00DC 39		* 3	* *	*	* *	00DD 36 RDSEC	86 0A		COES 32	8	B7	OOEC B6 ECOO	5 6	ì	00F3 7E E113	*	B 2	COFF BO IO	3 8 3 8	0101 8D 44 0103 C6 05	86 02	0107 8D 3E 0109 B6 EC00	
0123	0125	0128	0130	0133	0135	0137	0138 0139	0140	0142	0144	0145	0147	0148	0150	0151	0154	0155	0156	0158	0159	0160	0162	0163	0164	0166	0167	0168	0170	0171	0172	01/3	0175	0178	0178	01/9	0181 0182	0183	

	COMMAND PROCESSOR	G SYSTEM	EFINITIONS		EQUIPMENT TABLE ADDRESS		S	DATA IKANSFEK LYPE	DATA BUFFER ADDRESS DRIVE NUMBER	TRACK NUMBER	SECTOR NUMBER	¥	BACK LINK IRACK/SECTOR	TYPE	₫	FIRST TRACK/SECTOR	LAST TRACK/SECTOR		INDEX INTO DATA BUFFER	SPACE COMPRESSION FLAG	0 0	١		_	Š	NUMBER OF SECTORS											128 BYTES/SECTOR									STUKAGE FUR FILE NAME	
N NAM DELFILE	TRANSIENT	* FOR CP/68 OPERATING	* BLOCK ADDRESSING DEFINITIONS	*	FCBEGT EQU O	EGU	EQU	-	FCBUBA EUU /	EGU	EQU 1	EGU		FUBINATION TO FORT 29	Egu	EQU	0 0 0 0 0 0 0	FCBNFB FQU 37	EQU	FCBSCF EQU 41	FIBDE	FIBNAM EGO O		EQU 1	EGU	FIBNMS EWO 19	* BASE-PAGE EQUATES		Eg	DESCRC EQU \$22	-	EgU		ESCAPE EQU \$43	* DISK ATTRIBUTES		SECSIZ EQU 128	* 500 GOD *	Ę			RMB 2				* TEMPORARY STURAGE P	•
0000 0000					0000 0000				0000 0000					0000 0010				0000 0025		0000 0029		0000 0000				0000 0013				0000 0022				0000 0043			0800 0000			0000 0000		0005 0002		002A 0080			
0001	0003	0004	9000	0007	+ 0008 0000	0010	+ 1100	0012 +	0013 +	0015 +	0016 +	0017 +	0018	0019	0021 +	0022 +	0023 +	0025 +	0026 +	0027 +	0028	4 4	0031 +	0032 +	0033 +	0034 +	9800	0037	0038	600	0040	0042	0043	0044	0046	0047	0048	0049	0000	0052	0023	0054	0000	0057	0028	0059	

AUDRES 009A C.
BUOTT 0000 RN
BUOTT 0067 R
BUOTT 0067 R
BUOTT 0068 C.
CMDCTL EC02
CMDDAT EC02
CMDDAT EC02
CHDAT 009C C.
FIS 0090 C.
GETRYT 0063 R
GETSZ 00C4 R
GETSZ 00C4 R
GETSZ 00C4 R
GETSZ 00C7 R
INUAT EC00
INUAT EC00
INUAT EC07
OUTCTL EC07
COUTCTL EC07
COUTCTL EC07
COUTCTL EC07
COUTCTL EC07
COUTCTL EC07
COUTCTL EC07
COUTCTCL EC07
COUTCTL EC07
COUTCTL EC07
COUTCTL EC07
COUTCTL EC07
COUTCTL EC07
COUTCTCL EC07
C

GET LENGTH OF EXT TOTAL LENGTH

AMBIG. NAME? IF NOT, EKROR POINT TO BUFFER

CHECK RC UNAMBIG. NAME? YES

SAVE LENGTH OF NAME GET A TOKEN

GET NEW CLI

CHECK RC PERIOD? IF NOT, ERROR

COUNT PERIOD GET A TOKEN FORMAT NAME INTO TEMP BUFFER

CLEAN STACK

ERROR? YES

POINT TO FILE-NAME

FILE FOUND DURING SEARCH? NO, ERROR

END OF DIRECTORY? NO

CHECK STATUS 6000?

CLEAR 'FILE-FOUND' MARK OPEN DIRECTORY

F (55	LDX DESCRA STX BUFFER	STA A BUFFER+2	SWI	FCB 47	CMP B #7	BNE DEL2A	INC BUFFER+2	SWI	FCB 47			CMP B #2	BNE DELZA	LDA B DESCRC	ADD B BUFFER+2	LDX #TEMP PSHX	SMI	FCB 5		SWI			FCB 52	INS	INS	CMP B #2		I nx #SYSECB	CLR FCBSCF, X	OPEND	SWI	LDA A FCBSTA, X	BEQ DEL5	CMP A #1	TET ECBECE V	BEQ *+5
7E 0221 R	0114 20 0121 0D	0122 DE 0124 FF	0129 0127 96 22 0130 0129 B7 002C R		+	0134 012E U6 23 0135 0130 C1 2E	0136 0132 26 D8 **	0134 7C 002C R	+	+ 0138 2F	0142 0139 D6 25	0130 27	013F C1 02	0147 0141 26 C9	0143 D6 22	0145 FB 002C R	CE OOAC	+ 014B	+ 014C 05	3	0157 + 0150 3F		+ 0152	•	0155	0164 0156 31	0158	015A 27 B0	0168 **			0172 + 0161 3F	0163 A		0177 0167 81 01	07 2010	0181 016D 27 03 *
	DEFAULT DRIVE=0	INPUT GET TOKEN FROM CLI		CHECK FOR 'ESCAPE'		NOT ESCAPE	'ESCAPE' DONE	CHECK RC OF TOKEN	NUMBER? NO		VALID DRIVE NO. ?	3	VALID DRIVE NO. ?	Q		SET DRIVE NO. IN FCB		NUMBER ERROR			GET NEW CLI			NOYOT V HUO		7000	COLON?	IF NOT, ERROR	NEW O TOKEN			CHECK RC	UNAMBIG, NAME: YES	AMBIG NAME?		FORMAT EKROR	
RMB 2 RMB 12	LDX #SYSFCB CLR FCBDRV, X	CLR FCBDTT, X NXTOK	SWI FCB 47	LDX DESCRA	CMP A ESCAPE	BNF. DEL1	RTS	LDA B RC	CMP B #3 BNE DEL2		TST VALUE		CMP A #3	ä	LDX #SYSFCB	STA A FCBDRV, X	BRA DEL 1B	LDX #NUMBER	PRTMSG	SWI FCB 49	UMP DELNXT	FOC / NUMBER FREDRY	FCB	2	SWI	FCB 47	יא מישר פאר	BNE DEL 1A	70177	LINS.	FCB 47	LDA B RC	CMP B #1 BEQ DEL3		BEQ DEL3	LDX #FORMAT	FCB 49
OOAA OOOZ SAVEX OOAC OOOC TEMP	00B8 CE 0000 R DELO 00B8 6F 09	00BD 6F 06	+ 00BF 3F + 00C0 2F	00C1 DE 20 00C3 A6 00	0005 91 43	00C7 26 01 **	\$ 6E 6300	OOCA D6 25 DEL1	C1		00D0 7D 0027	9	0005 96 28 0007 81 03	22 07	OODB CE OOOO R	A7 09		OOE2 CE OOEA R DEL1A		+ 00E5 3F + 00E6 31)221 R	* * * * * * * * * * * * * * * * * * *	1 2	: *:	00F8 3F	+ 00F9 2F	00FF 04 30	OOFE 26 EZ	*	+ 0100 3F		06 25	0104 C1 01 DELZ 0106 27 14	5	010A 27 16	010C CE 0114 R DELZA	+ 010F 3F + 0110 31
0061	0063 0064 0065	0066		0070	0072	0073	0075	0078	0078	0800	0081	0083	0084 2084	9800	/800 0088	6800	0600	0092		0094	9600	0097	6600	0100			2010	0106	0107			0111	0112	01114	0116	0118	

PUT IN TERMINATOR OUTPUT 'FILENAME. EXT' OUTPUT '?'	GET USER RESPONSE 'YES'? NO, DO NOT DELETE FILE	POINT TO FCB NAME POINT TO DIRECTORY NAME	12-CHARACTEK MOVE MOVE DIR. NAME TO FCB CLEAN STACK	CALL FILE-DELETE RESTORE INDEX	CHECK STATUS GOOD DELETE? YES OUTPUT /FILE DELETED/	OUTPIJT 'DELETE-'
STA A 12, X PRTMSG SWI FCB 49 LDX #@MKK PRTMSG SWI	FCB 49 GTCML/SWI SWI FCB 48 LDX DESCRA LDA A 0, X CMP A # \ Y BNE DELSA	LDX #SYSFCB+FCBNAM PSHX SWI FCB 5 LDX #SYSFCB LDX FCBIND, X STX SAVEX	SWI FCB 5 LDA B #12 MOVC SWI FCB 17 INS INS	INS LDX #SYSFCB DELETE SWI FCB 28 LDA A SAVEX LDA B SAVEX STA A FCBIND, X	Z = 0	FCC / FILE DELETED/ FCB \$0D T LDX #DPRMPT PRTMSG SWI FCB 49 GTCMD SWI FCB 48
		++ .	0268 + 01Er 3+ 0268 + 01Er 3+ 0270	25.78 E. 2. 87.41	0208 0208 0200 0200 + 0200 + 0210 0211	0213 20 0220 0D 0221 CE 022F R + 0224 3F + 0225 31 + 0226 3F + 0227 30
YES, GET NEW CLI LINE FILE-NOT-FOUND ERROR	PRINT ERROR MESSAGE	POINT TO DIRECTORY ENTRY CHECK FIRST CHARACTER BLANK? (ALREADY DELETED) YES	POINT TO FILE NAME 12 CHARACTER COMPARE WILD-CARD COMPARISON	CLEAN STACK FOUND FILE? GET A NEW DIRECTORY ENTRY	MARK 'FILE FOUND' OUTPUT 'DELETE-'	GET DRIVE NUMBER MAKE ASCII OUTPUT 'DRIVE:' POINT TO FILE NAME IN DIRECTORY
JMP DELNXT LDX #FNFND PRTMSG SWI FCB 49 JMP DELNXT	FCC / FILE NOT FOUND/ FCB \$0D PRTERR SWI FCB 30	LDX FCBIND, X LDA A O, X CMP A #\$20 BEQ DEL.5A PSHX SWI FCB 5	LUX #lenr PSHX SWI SWI FCB 5 LDA B #12 CMWC SWI SWI FCB 53	INS INS INS BEG DEL6 LDX #SYSFCB GETDR SY 1 FOR 34	FCB AS BRA DEL4A LDX #SYSFCB INC FCBSCF, X LDX #DPRMPT PRTMSG SWI FCB 49 LDX #SYSFCB	LDA A FCBDRV, X ADD A #\$30 STA A DRIVE LDX #DRIVE PRTMSG SWI FCB 49 LDX #SYSFCB LDX FCBIND, X LDA A #\$04
016F 7E 0221 R * 0172 CE 017A R 0175 3F 0176 3I 0177 7E 0221 R		018F EE 27 BEL5 0191 A6 00 0193 81 20 0195 27 11 *	0197 LE UONL K 0190 3F 0190 05 019E C6 0C 01A0 3F 01A1 35	0143 31 0144 31 0145 31 0146 27 07 * 0148 CE 0000 R DELSA	OIAC IN OIAD 20 B4 * OIAF CE 0000 R DEL6 OIBS 6C 29 OIBS 31 OIBS 31	

```
25588 22588 22588 22588 22588 22588 22588 22588 22586 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 22587 
 ΣΣΣ
 BACK UP ONE TOKEN LOOP AGAIN
                               / DELETE-
DESCRA
CUCHAR
DELO
                                      4
STX
                               55 6
                                                      RAB
FCC
FCB
                                                                                    FCB
                                                                                                           END
                         *
DPRMPT
                                                *
DRIVE
                                                                               GMEK
CMEK
                 œ
DE 20
DF 23
7E 00B8
                                                     0001
3A
04
                               20
04
                                                                                     200
                               022F
0238
                                                      0239
023A
023B
                                                                                    023C
023F
```

EEGGETTEETTEETTEETTEETTEEG EEEG

* * R INITR2	*** * * I I I I I I	### ### ### ### ######################
0008 DE 20 0006 A6 00 000E 81 59 000E 27 01 00E0 39 00E1 CE 0000 00E4 AF 0A 00E6 86 01		0100 0101 0108 0108 0108 0108 0108 0110 0110 01115 01118
0061 0063 0063 0064 0065 0066 0067 0068 0070	0072 0073 + 0075 + 0076 0076 0077 0082 0082 0083 0085 0085 0086 0087 0086 0087 0087 0088	
E A DISK FOR CP-68 OPERATING SYSTEM 8 INCH FLOPPY DISKS SECTORS 1,2 BOOTSTRAP SECTOR 3 HEADER OF FREE-SPACE LIST SECTORS 4-26 DIRECTORY SPACE 777	128 BYTES PER SECTOR 26 SECTORS PER TRACK 76 TRACKS ON DISK (LESS TRACK O) BLOCK ADDRESSES ERROR STATUS FLAG DATA BUFFER ADBRESS DRIVE NUMBER TRACK NUMBER SECTOR NUMBER SECTOR LINK POINTER	## SECSIZ SECTOR BUFFER ## COMMAND-LINE INTERPRETER BASE-PAGE LOCATIONS ## COMMAND-LINE INTERPRETER BASE-PAGE LOCATIONS ## ADDRESS OF TOKEN ## ENJ #27 VALUE OF NUMERIC TOKEN ## FROMPT FCC / INIT. DISK IN DRIVE / DRVNO RMB 1 ## ENT . INITR ENTRY POINT FROM CLI ## ENT . INITR ENTRY POINT FROM CLI ## ENT . INITR ENTRY POINT FROM CLI ## ENT . INITR ANGE (ICOM PERMITS 4 DRIVES) LDX #FCBSPC POINT TO FCB LDX #FCBSPC POINT TO FCB STA A BROWNT ADD A ##30 MAKE DRIVE NUMBER ASCII STA A BROWNT FCB 49 GTCMD GET USER RESPONSE SWI FCB 49 GTCMD SWI FCB 48
** INITIALIZE A DISK ** FOR ICOM 8 INCH FL ** TRACK 0, SECTORS 3 ** TRACK 0, SECTORS 3 ** TRACK 0, SECTOR 3 ** TRACK 1-77	DISK ATTRIBUN ECSIZ EQU 128 RKSIZ EQU 128 RKSIZ EQU 128 RKSIZ EQU 126 SKSIZ EQU 16 CBSTA EQU 9 CBDBA EQU 17 CBDRK EQU 10 CBSTK EQU 11 CBSTK EQU 11 CBSTK EQU 13 CBSTK EQU 13 CBSTK EQU 13 CBST EQU 13	** COMMAND-LINE INTER ** COMMAND-LINE INTER ** COMMAND-LINE INTER ** CALUE EQU \$20 ** PROMPT FCC 'INIT. DI PROMPT FCC 'INIT. DI PROMPT FCC '?' ?' FCB \$0.4 ** ENT . INITR ** ENT . INITR ** ENT . INITR ** A DRVNO LDX #FCBSPC STA A FCBDRV, AND A #\$30 LDX #FCBPRV STA A FCBDRV, AND A #\$30 STA A FCBDRV, AND A #\$30 STA A FCBBRV STA A FCBBRV STA A FCBBRV STA A FCBRV STA A FCBR
z	* * <u>Q = Q</u> * * * r r r r r r r r r r r r r r r r	**************************************
0000 0000	0000 0010 11 11 11 11 11 11 11 11 11 11	002A 0080 00AA 0027 00AA 49 00BE 0001 00BE 20 00C2 20 00C2 84 03 00C5 84 03 00C5 84 03 00C5 84 03 00C6 87 09 00C6 87 09 00C6 87 09 00C6 87 00 00C6 87 00 00C7 88 00 0

B #1 YES, SECTOR=1 A NEXT TRACK A #DSKSIZ+1 END OF DISK? INITR7 NO	LAST SECTOR POINTS TO 0,0	TRACK LINK SAVE LSEC GET PSEC	1 SECTOR LINK	RESTORE LSEC WRITE SECTOR	DONE? (=0)	NO.	DONE? (=0)	ON O	ñ L	X SAUF ISEC	GET PSEC	A GET LSEC	KEEP WRITING	lu!	OUTPUT ERROR MESSAGE		RETURN TO CLI	'INITIALIZATION FAILED'		23		SAVE X-KEGISIEK	Place (Appendix Applied to TMICO	ADD LOGICAL OFFSET		SECTOR STARTS AT 1	GET PSEC	KESTUKE X-KEG		ONLACTION ACCOUNT	SAUF AA
LDA B #1 INC A CMP A #DSKSIZ BNE INITK7	CLR A CLR B			PUL B BSR WRTBLK		BNE INITR8	TST B BNF TNITES		2	INITRE STA A FCBTRK, X PSH B		SIM B PUBSULA PUL B	BRA INITR6	# * FATAL ERROR MESSAGE	R INITO LDX #QMSG	PRTMSG	FCB 49 RTS	* QMSG FCC /INITIALI)	FCB #0D	* CONVERT LISEC TO PSEC * LISEC IN R-REG	*		FCB 5	ADDBX	SWI FCB 10		LDA B 0, X		FCB 6 RTS	* WRITE A SECTOR WITH ERROR CHECKING	* TELE PSH A
0183 0166 C6 01 0184 0168 4C 0185 0169 81 4D 0186 0168 26 02 0187	0188 016D 4F 0189 016E 5F 0190	0191 016F B7 002A R 0192 0172 37 0193 0173 8B 33	0175 F7	0178	017B 4D	0198 017C 26 04 0199	0200 017E 5D 0201 017F 26 01	00 1010	2010	0205 0182 A7 0A 0206 0184 37	0185	0189 33	0210 018A 20 D5	0212 0213	018C CE 0192		+ ~	0220 0192 49	0107	0224 0225	0226	+ 0148	0229 + 01A9 05 0230 01AA CF 0229 R	·	0232 + 01AD 3F 0233 + 01AE 0A	OIAF	0235 01B0 E6 00	0237 + 01B2 3F	01B3	0240 0241	0185 36
5 BUFFER EXCEPT FOR LAST 2 BYTES	#BUFFER B #SECSI?-2 A	x '0	INITK3	#1 TRACK, SECTOR=1	×× ċ·	<.		WRTBLK WRITE BLOCK 3 FCRSTA, X CHFCK FOR DISK FRROR	ð	INITG FATAL DISK ERROR, QUIT	AD ASA BONGE	R=4	BUFFER+SECS17-2 BUFFER+SECS17-1	INITIALIZE DIRECTORY TO ZERO	WRITE	FCBSIH, A CHECK FOR DISK ERRUR *+4 OK	INITO FATAL DISK ERROR, QUIT		ZIS	INITKS YES	A FCBSCT, X INTEG NO CONTINUE DESTINO	Ì	#1 FCBSCT, X SECTOR=1	FCBTRK, X TRACK=1		INITIALIZE REST OF DISK (FREE-SPACE)	X=FCB ADDRESS	A=TRACK NUMBER	B≕SECTOR NUMBER	B MAKE SECTOR LINKAGE B #TRKSIZ+1 END OF TRACK?	INITR7 NO

ADD A #\$7 YES RTS	LOGICAL/PHYSICAL SECTOR TABLE FCB 00	M. FCB \$1 FCB \$A FCB \$13	J. J. J. J.			FCB #10 FCB #10 FCB #10 FCB #11 FCB #11 FCB #12 FCB #12	BOOT EQU * . BOOT PROGRAM STARTS HERE * END	
* *	* * *	* 17				* * *	*	
0225 8B 07 0227 39	0228 00		022C 02 022D 0B 022E 14		0235 05 0236 05 0237 17 0238 06 0239 0F	0.234 18 0.23C 10 0.23C 10 0.23E 08 0.23F 11 0.241 09 0.242 12	0243 0243	
0305 0306 0307 0308	0309 0310 0311 0312	0313 0314 0315 0316	0317	0321 0322 0323 0324	0326 0328 0329 0329	0332 0333 0334 0335 0337 0337 0339	0343 0344 0345 0346	
X CLEAR ERROR FLAG ISSUE I/O REQUEST	A, X ERROR? YES	RESTORE 'A'	CONVERT LEFT DIGIT	CONVERT RIGHT DIGIT E+1 SAVE X	T, X MAKE SECTOR NO. HEX	*	CTOR /	CONVERT BINARY TO HEX-ASCII HERE LSR A LS
CLR A CLR FCBSTA, X IOHUR SWI	FCB 19 STA A FCBSTA, X TST A BNE WRTERR	PUL A RTS	TAB BSR OUTHL STA A ERTYPE TPA	BSR OUTHR STA A ERTYPE+1 PSHX SWI	•	STA A SECT+1 LDA A FCBTRK, X BSR OUTH LDA A FCBTRK, X BSR OUTHR STA A TRACK+1 LDX **DERROR PRTMSG SWI	SW1 CHL RTS 1 "WA RTS CDISK ERROR: / RMB 2 CC / AT SECTOR / RMB 2 FCC /, TRACK / RMB 2 FCC /, TRACK / RMB 2 FCC /, TRACK / RMB 2 FCC / RMB 2	LER A LSR A LSR A LSR A LSR A AND A ##OF ADD A ##30 CMP A ##39 BLS #+4
		* *	WRTERR				* DERROR ERTYPE SECT TRACK	
00	000	01C0 32 01C1 39	01C2 16 01C3 8D 54 01C5 B7 01FF R	0109 0108 0108	0100 A6 08 0102 80 45 0104 B7 020C 0107 A6 08 0109 80 42	010B 01DE 01E2 01E7 01E7 01E	01F1 3F 01F2 1F 01F2 1F 01F4 44 01FF 0002 020C 0002 020E 2C 0216 0002 0218 0D	0219 44 0218 44 0216 44 0210 44 0210 84 0F 0217 88 39 0223 23 02
0244 0245 0246 0246	0248 + 0249 0250 0251	0252 0253 0254	0256 0257 0257 0258	0259 0260 0261 0262 0263 +		0270 0271 0273 0274 0275 0275 0276 0277 0278	0.281 0.283 0.284 0.285 0.286 0.287 0.289 0.290	0.293 0.294 0.295 0.296 0.296 0.298 0.299 0.300 0.300 0.300 0.300 0.300

```
BACK LINK TRACK/SECTOR
FILE NAME (8.3+E0T=13))
FILE TYPE
FILE ACCES CODE
FIRST TRACK/SECTOR
LAST TRACK/SECTOR
NUMBER OF SECTORS
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       NEXT FCB IN ACTIVE CHAIN
                                                                                                                                                                                                                                                                                       EQUIPMENT TABLE ADDRESS
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   FILE NAME (8.3 + E0T=13)
                                                                                                                                                                                                                                                                                                                                                                                                                                                     DRIVE NUMBER
TRACK NUMBER
SECTOR NUMBER
FWD LINK TRACK/SECTOR
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      INDEX INTO DATA BUFFER
SPACE COMPRESSION FLAG
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    128 BYTES/SECTOR
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  DEFAULT DRIVE=0
                                                                                                                                                                                                                                                                                                                      GENERIC DEVICE TYPE
                                                                                                                                                                                                                                                                                                                                                                                          DATA TRANSFER TYPE
DATA BUFFER ADDRESS
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  FILE TYPE
FILE ACCESS CODE
FIRST TRACK/SECTOR
LAST TRACK/SECTOR
NUMBER OF SECTORS
                                                       TRANSIENT COMMAND 'LINK' PROCESSOR
SYNTAX: LINK [DRIVE: ] FILENAME. EXT
MAKE SYSTEM LINKAGE TO FILENAME. EXT
                                                                                                                                                                                             * BLOCK ADDRESSING DEFINITIONS
                                                                                                                                                                                                                                                                                                                                                              STATUS
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          LDX #SYSFCB
CLR FCBDRV, X
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   BASE-PAGE EQUATES
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               * FCB FOR TRANSIENT
   NAM LINKER
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    #20
#22
#23
#25
#25
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                33
SECSIZ
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   * DISK ATTRIBUTES
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             RMB 2
FCC 'DSK'
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 BUFFER
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               FCBFWD EQU 12
FCBRAK EQU 14
FCBRVAM EQU 16
FCBRACS EQU 30
FCBRTS EQU 31
FCBRTS EQU 33
FCBRTS EQU 33
FCBRTS EQU 35
FCBRTS EQU 35
FCBRTS EQU 37
FCBRTS EQU 37
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                FIBNAM EQU O
FIBTYP EQU 13
FIBACS EQU 14
                                                                                                                                                                                                                                                                                    FCBEQT EQU O
FCBGDT EQU 2
FCBSTA EQU 5
FCBDTT EQU 6
FCBDBA EQU 7
FCBDRV EQU 7
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                FIBETS EQU 15
FIBLTS EQU 17
FIBNMS EQU 19
                                                                                                                                                                                                                                                           FCBDEF
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       FIBDEF
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               SECSIZ EQU
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    FCBTRK
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      FCBSCT
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              DESCRA
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               CUCHAR
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 SYSFCB
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            BUFFER
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   DESCRC
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   CLASS
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  JAI_UE
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          CE 0000 R LINK
6F 09
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 œ
                                                                                                                                                                                                                                                                                                                                                     0000 0005
0000 0006
0000 0007
0000 0009
0000 000E
0000 0010
0000 0010
0000 0011
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  0023
0025
0027
0027
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   0000
0000
000E
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   0021
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      000F
0000 0000
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 0000 0020
0000 0022
0000 0023
0000 0025
0000 0026
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           002A
0021
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         0800
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 0000 0000
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               0002
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   44
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 00000
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               0000
                                                                                                                                                                                                                                                                                                                         0000
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     0000
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               0000
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      0000
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             0000
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        0005
0007
0009
002A
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         OOAD
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             OOPP
                                                                                                                     0005
0006
0007
0008
0009
0010 +
                                                                                                                                                                                                                                                                                                                                                     0012 + 0013 + 0015 + 0015 + 0015 + 0015 + 0017 + 0017 + 0018 + 0018 + 0018 + 0018 + 0018 + 0018 + 0018 + 0018 + 0018 + 0018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 00018 + 
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 0019 + (0021 + (0022 + (0022 + (0022 + (0022 + (0022 + (0022 + (0022 + (0022 + (0022 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 + (0032 +
                                                                                          9000
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 0045
0046
0047
0048
0050
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             0052
0053
0054
0055
0055
0057
0057
0059
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  Σ
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 œ
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    WRTHE 2302 M
WRTHLK 0185 F
WRTERR 0102 F
XABX 2185 P
                  PSHALL 2151
PSHX 21CE
PULLAL 216A
PULX 21E7
PULX 21E7
PULN 2406
RGMSG 0296
KCBDEF 258C
KCBDEF 258C
KEAD 2388
                                                                                                                                                                                                                                                                              HEWIND 2384 PSECS17 0080 SECS17 0080 SECS17 0080 SUBARX 2227 PSUBARX 2299 PSUBARX 2283 PSUBARA 2265 PSUBARA 2
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 0216 F
001A
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               2183
0027
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   RKSIZ
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            J BL
TRACK
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  JULIE VALUE
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  XAB
```

FCBSCT 0008
FCBSLK 0000
FCBSTC 0000 R
FCBSTA 0005
FCBSTK 000C
FCBSTK 0000
FCBSTK 0000
FCBSTK 0000
FCBSTK 0000

2558

SIME

GETSC 0148 R
GETSC 0148 R
GREW 23EC M
GREW 2450 M
INITEX 258E M
INITER 0000 R
INITEX 0001 R
INITEX 0010 R
INITEX 0127 R
INITEX 0127 R
INITEX 0151 F

22CD 24D6 234F 239E 0219 021D

MOVS MOVS MUL16 NXTOK OPEN OUTHR PROMPT

24A2 22E7

œ

EKTYPE 01FF FCBDBA 0007

DSKS1Z 004C

231B M

243A 2369

CHOIN

CMWC 2572 M DELETE 2420 M DERROR 01F4 R DESCRA 0020 DIV16 2524 M DRVNO 00BE R

INITR 0003 R
6MBTRL 013D R
AUDAX 2232 M
AUDAX 2232 A
AUDX 224B M
AUDX 224B M
AUDX 224B M
AUDX AB 2000 M
BASEQU 2A2A M
BASEQU 2A2A M
BUOT 0243 R

FUBDEF 2650 M

FUBBRY 0009

Part	CHECK RC PERIOD? IF NOT, EKROR	COUNT PEKIOD GET TOKEN FROM CLI	CHECK RC UNAMBIG, NAME? IF NOT, ERROR	GET LENGTH OF EXT TOTAL LENGTH	POINTER TO FCBNAM	POINTER TO CLI NAME	FUNDAME INTO TOB	CLEAN STACK	ERRORS? YES	OPEN THE DIRECTORY	CHECK STATUS GOOD?	END OF DIRECTORY? NO	FILE NOT FOUND ON DISK		PRINT ERROR MESSAGE	POINT TO DIRECTORY NAME
Note of the color of the colo	FCB 47 LDA B RC CMP B #/ BNE LNK3	INC BUFFER+2 NXTOK			PSHX SWI FCB 5	PSHX SWI FCB 5		INS INS INS	INS TST B BNE LNK3	LDX #SYSFCB OPEND	SWI FCB 23 LDA A FCBSTA, X BEG LNK6	CMP A #1 BNE LNKSA	LDX #FNFND PRTMSG SWI FCB 49 RTS	FCC / FILE NOT FOUND/ FCB \$0D	<u>α</u> α	LDX FCBIND, X PSHX SWI FCB 5
100 1 1 1 1 1 1 1 1		œ) 3F : 2F D6 25 C1 01 26 CF	D6 22 FB 002C R CE 0010 R	05 05 FF 0024		0134 3F 0135 34			œ	3 3F 17 A6 05 27 1D	81 01 26 16	014A CE 0150 R 014D 3F 014E 31 014F 39		0160 3F 0161 1E 0162 39	0163 EE 27 0165 3F 0166 05
Court Fe Col Ce Ce Ce Ce Ce Ce Ce C		0126 0127 0128		0134 0135 0137				0148 0150	0151 0152 0153	0154 0155 0156		0161 0162 0163		0170 0171 0172 0173		
CLR FCBDTT, X CLR FCBDTT, X		GET USER RESPONSE	CHECK RC NUMBER? NO	•	VALID DRIVE NO.? (4 DRIVES) NO	SET DRIVE NO.	NUMBER ERROR				ERROR		CHECK RC UNAMBIG. NAME? YES FORMAT ERROR		POINT TO NAME	GET LENGTH OF NAME GET TOKEN FROM CLI
0.001. GE 0.05. R 0.001. GE 0.15. E R 0.001. GE 0.15. E R 0.001. GE 0.15. E R 0.001. GE 0.00. R 0.001. TD 0.02. T 0.002. SE 0.00. R 0.002. SE 0.002. SE 0.002. R 0.002. SE 0.002. R 0.002. SE 0.002. R 0.003. SE 0.002. R 0.004. GE 0.02. R 0.004. SE 0.004. R 0.004. SE 0.004. R 0.005. SE 0.006. R 0.006. SE 0.006. R 0	CLR FCBDTT, X LDX #PRMPT PRTMSG SWI	FCB 49 GTCMD SWI	448 #SYSFCB B RC B #3 LNK2	TST VALUE BNE LNK1	A VALUE+1 A #3 LNK1	STA A FCBDRV.X BRA LNK1A	LDX #NUMBER PRTMSG SWI	FCB 49 RTS	ER FCC / NUMBER ERROR/ FCB #OD	Z		NXTOK SWI FCB 47	LDA B RC CMP B #1 BEQ LNK4 LDX #FORMAT PRIMSG	SWI FCB 49 RTS	FCC FCB	
0061 0061 0062 0062 0065 0067 0077 0077 0077 0077 0077 0077	00AF 6F 06 00B1 CE 01CE + 00B4 3F	+ 00B5 + 00B6	+ 0087 30 0088 CE 0000 0088 D6 25 008D C1 03 008F 26 2F	00C1 7D 0027 00C4 26 0A	00C6 96 28 00C8 81 03 00CA 22 04	00CC A7 09 00CE 20 14	+	+ 00D4 31 00D5 39	0006 20 0053 00	+ 00E4 3F + 00E5 2E	00EA 26 E4	+ +	00FE D6 25 00FO C1 01 00F2 27 14 00F4 CE 00FA R	+ 00F7 3F + 00F8 31 00F9 39	00FA 20 0107 0D 0108 DE 20	010A FF 002A 010D 96 22 010F B7 002C + 0112 3F

0183 0167 CE 0010 R	LDX #SYSFCB+FCBNAM	POINT TO FCB NAME	0244	01CA 7E 0160 R	UMP LNKSA	K5A YES
+			0245	* oicd 39	RTS	
0186 + 016B 05 0187 016C C6 0C	FCB 5 LDA B #12	COMPARE 12 CHARACTERS	0247 0248			
0188 0189 + 016E 3F 0180 + 014E 12	CMPC SWI FIRE 18		0249	01CE 20 PF 01E1 04 *	PRMPT FCC / : FCB \$0.	FCC / SYSTEM FILE NAME? / FCB \$04
0170		CLEAN STACK	0252	k	ENŪ	
0173 31 0174 27 07	INS BEG LNK7	FOUND ENTRY IN DIRECTORY?				
0176 CE 0000 R + 0179 3F + 017A 1A 017B 20 C5	LDX #SYSFCB GETUR SWI FCB 26 BRA LNK5	GET NEW ENTRY				
017D CE 0000 0180 EE 27 0182 A6 OF 0184 E6 10	PPEE	POINT TO DIRECTORY ENTRY GET FIRST T/S				
0207 0186 CE 0000 R 0208 0189 A7 1F 0209 018B E7 20	LDX #SYSFCB STA A FCBFTS, X STA B FCBFTS+1, X	SAVE IN FCB				
018D EE 27 018F A6 11 0191 E6 12		GET LAST 1/S				
0213 0193 CE 0000 R 0214 0196 A7 21 0215 0198 E7 22	LDX #SYSFCB STA A FCBLTS, X STA B FCBLTS+1, Y	SAVE IN FCB				
019A 86 019C C6	0 C M	TRACK=0 SECTOR=3				
0218 019E A7 0A 0219 01A0 E7 0B 0220	STA A FCBIRK, X STA B FCBSCT, X TOURS	GET LINK SECTOR				
+ 01A2 3F + 01A3 13 01A4 6D 05 01A6 27 03	SMI SMI FCB 19 TST FCBSTA, X BEQ *+5	ERROR?				
01A8 7E 0160 R	JMP LNK5A	ERROR MESSAGE				
01AB CE 01AE 63 01B0 A6	# IL CI 1	MAKE 'OUTPUT' GET LINKAGE INFO.				
0231 01BZ E6 20 0232 01B4 B7 00A4 R 0233 01B7 F7 00A5 R	LUA B FUBFIS+1,X STA A BUFFER+122 STA B BUFFER+123	PUT IN LINKAGE SECTOR				
0234 01BA A6 21 0235 01BC E6 22 0236 01BE B7 00A6 R 0237 01C1 F7 00A7 R	LDA A FCBLTS, X LDA B FCBLTS+1, X STA A BUFFEK+124 STA B BUFFEK+125					
0238 0239 + 01C4 3F 0240 + 01C5 13 0241 01C6 6D 05 0242 01C8 27 03 **	10HDK SWI FCB 19 TST FCBSTA, X BEQ *+5	WRITE LINKAGE SECTOR ERROR? NO				

. B	TKANSIENT PEKIPHERAL-INTERCHANGE "PIP"	SET UP ADDRESSING EQUATES		DESCRIPTOR ADDRESS(2) DESCRIPTOR COUNT	CURRENT CHAR (2)	TOKEN RETURN CODE. TOKEN CLASS	BIN VALUE/1RANSFER ADDRESS (2)	TOP OF FCB CHAIN (2)	DISK FREE SPACE POINTER (8)	START OF TRANSIEN! AREA(2)	NEXT AVAIL TRANSIENT AREA (2)	BACKSPACE CHAR	DELETE LINE CHAR	DEPTH; LINES/PAGE	DEFINITE CHARGAINE	NULL COUNT	TAB CHAR	DUPLEX; FF=H, 00=F	EJECT COUNT	PAUSE; OU=YES ESCADE CHAP	DEPTH LINES/PAGE	DEPTH TEMP	WIDTH CHARS/LINE	SSECT TABLE ADDRESS	GENERIC DEVICE TYPE	STATUS	DATA TRANSFER TYPE	DAIN BUFFER HUDRESS	EQUIPMENT TABLE ADDRESS	GENERIC DEVICE TYPE	SIAIUS DATA TRANSFER TVRF	DATA BUFFER ADDRESS	DRIVE NUMBER	TRACK NUMBER	SECTOR NUMBER	BACK LINK TRACK/SECTOR	FILE NAME (8. 3+E0T=13.)	FILE TYPE	FILE ACCESS CODE	FIRST TRACK/SECTOR	LAST TRACK/SECTUR	NEXT FOR IN ACTION CHAIN	INDEX INTO DATA BUFFER	SPACE COMPRESSION FLAG	FILE NAME (8 3 + E0T=13)	
N NAM PIPPER	* TKANSIENT PEK		* BASEQU	DESCRA EQU \$20	HAR EQU	RC EQU \$25 CLASS EQU \$26	EQU	EQU	AB EQU	BMEM EQU #33		Egg	EQU	DP EQU #3B	3 2	E 19	TB EQU \$3F	EQU	10G	PS EGU \$42		CNT EQU	LWD EQU #46	ACBUER	RCBGDT EQU 2	EQU	RCBDTT EQU 6	RUBUBA EGO /	<u>B</u>	8	FCBSTA EQU 5				FCBSCT EQU 11	3 6	EBC	EQU	EGU	E00	2	FUBNIS EGO 33		EGU	FIBDEF	·
0000 0000				+ 0000 0020		+ 0000 0025 + 0000 0026	0000			+ 0000 0033	0000	0000			+ 0000 003r		+ 0000 003F			+ 0000 0042			+ 0000 0046	0000			0000	7000 0000 +	0000 0000 +		+ 0000 0002	0000			+ 0000 000B			0000		0000	0000	+ 0000 0023 + 0000 0025			0000	
0000	2000	5000	9000	+ 8000	0000	0011 +				0016 +			0020	0021 +	+ 7700 + 6600	0024	0025 +	+ 9200	4 0027	0028 +	+ 6200 + 0000	0031 +	0032 +		0035 +			+ 8EOO		0041 +	•	0043	0045	0046 +		1 0400			0052 +			+ SS00	+ 2500 + 2000	+ 8500	0020	

| | ~ ~ | . <u>.</u> . | 2. U | | 2 . | 2. 4 | L 2. | . 2 | 2 | | | - | | |
 | | - | | | | - | 4- | -
 | | _ | | | | | |
 | | | | | | |
 | | | | | |
 |
|--------------|------------------------------------|--------------------------------------|--|---|---|---|---|--|--|---|--|--|--|---
--|--|--|--|---|---|--|--
--|---|---|--|--|--
--|--|--|--|---|--|---
---|---|--|--|--
---|--|---|---|
| 017D
246E | 2301 | 22E7 | 22CD | 2406 | 234F | | | | 2151 | 21CE | 2157 | 2406 | | |
 | | | | | | | | 22027
 | 2185 | 64150 | | | | | |
 | | | | | | |
 | | | | | |
 |
| LUK7 | MOVC | MUL.16 | MUL8 | NX TOK | OPEN | CPEND | PRTERR | PRTMSG | PSHOLL | TSH
TSH
TSH
TSH | ;
;
;
;
;
;
;
;
; | FULDR | 3 | RCBDEF | KENIND
 | SECS17 | SUBABX | SUBAX | SUBBA | SYSECB | TABX | TXAB | VALUE
 | WAC I I | ٧٩١٠ | | | | | |
 | | | | | | |
 | | | | | |
 |
 | | | | | | | |
 | | | | | | | |
 | | | | | | |
 | | | | | |
 |
| ΣΣ | Σû | £Σ | Σ | Ξ | Σ | , | E | | Σ | | | E | | |
 | | | | | | | |
 | | | | Σ | | | | Σ
 | Σί | ×α | Σ | Σ | Σ. | ΕΣ | : œ
 | ž | oc o | ĸ Œ | œ | œα | _
 |
| 224B
2200 | 2A2A | 243A | 0026 | 231B | 2572 | 0023 | 2420 | 0022 | 2524 | 001E | 1000 | 2650 | 6000 | 9000 | 000
000
000
000
 | 2000 | 0000 | 0027 | 0021 | 0025 | 0023 | 0029 | 000B
 | 000 | 000 | OOOE | 2940 | 000 | 0000 | 0013 | 2488
 | 2558 | 00100 | 23EC | 24F0 | 24BC | 203E | OOPA
 | 0000 | 0000 | 00F0 | 00F4 | 0108 | 1
 |
| ADDEX | BASEQU | CHAIN | CLASS | CMPC | CMMC | CUCHAR | DESCRA | DESCRO | DIV16 | FCBACS | TCBBAK
TCBCBC | FCBDEF | FCBBRV | FCBDTT | FUBERI
 | FCBFWD | FUBGDT | FCBIND | FCBLTS | FCBNFB | FCBNMS | FCBSCF | FCBSCT
 | FCBSIA | FCRIVE | FIBACS | FIBDEF | FIBLTS | FIBNAM | FIBNMS | FMIFCB
 | FMTS | FORMOT | GETDR | GTCMD | INDEX | INTER | LINK
 | LINKER | ZX. | LNK2 | CNK3 | LNK
NK3 |
 |
| | 224B M LONZ 017D 2200 M LOADB 246E | 224B M LNK7 2200 M LOGDB 2200 M MOVC | 224B M LNK7 017D
3 2200 M L0ADB 246E
J 2A2A M MOUS 2301
202A R MUL16 22E7 | 2248 M LNK7 017D 2200 M LOADB 246E 2200 M LOADB 246E 2002 M MUVS 24A2 243A M MUL16 22E7 2006 M MURFR 0006 | 2248 M LNK7 017D 2200 M LOADB 246E 2200 M LOADB 246E 2301 MOVC 2301 2002 A MUL16 22E7 0026 MUL8 22E7 0026 MUL8 22E7 231B M NX10K 2406 | 2248 M LNK7 017D 2200 M LOADB 246E 2200 M LOADB 246E 2301 MOVC 2301 MOVC 2402 2403 MUL16 22E7 0026 2318 M NX10K 2406 2572 M OPEN 2315 M NX10K 2406 2572 M OPEN 2315 | 2248 M LNK7 017D 2200 M L0ADB 246E 2200 M L0ADB 246E 0024 R MUVS 2301 0026 MUL16 22E7 0026 MUL18 22E7 2318 M NUMBER 0016 2572 M OPEN 234F | 2248 M LNK7 017D 2200 M LOADB 246E 2200 M MOVS 2301 0024 R MOVS 2402 2434 M MUL16 22E7 0026 MULB 22E7 2318 M NUMBER 0016 2572 M NUMBER 230F | 2248 M LNK7 017D 2200 M LOADB 246E 2200 M MOVS 2301 0026 R MUL16 2267 0026 MULB 22CD 2318 M NUMBER 0016 2572 M NUMBER 0016 0020 PEND 239E | 2248 M LNK7 017D 224B M LOADB 246E 2200 M COADB 246E 2301 MOVS 24AC 2301 MOVS 24AC 2301 MOVS 24AC 231B M COADB 22CD MOVBER 00D6 231B M COADB 234E 00C3 PEND 234E 00C3 PEND 234E 234E 00C3 PEND 234E 234E 00C3 PEND 234E 234E 00C3 PEND 234E 00C3 | 2248 M LNK7 017D 2200 M LOADB 246E 2200 M MOVS 2301 0026 R MUL16 2201 0026 MUL8 22CD 2318 M NUMBER 0016 2572 M NUMBER 0016 2524 M PRTPG 2504 0022 PRTPG 2504 0022 PRTPG 2504 0022 PRTPG 2504 0002 PRTPG 2504 0002 PRTPG 2504 | 2248 M LNK7 017D 2200 M LOADB 246E 2200 M COADB 246E 2028 M MOVS 24A2 243A M MOVS 24A2 231B M NUMBER 00D6 2572 M NUMBER 00D6 2574 M NUMBER 00D6 2575 M NUMBER 00D6 25 | 2248 M LNK7 017D 2200 M LOADB 246E 2200 M MOVS 240E 243A M MOVS 24A2 243A M MUL16 22E7 0026 MUMBER 00D6 2552 M NUMBER 00D6 2572 M OPEND 234E 0023 PEMPT 01CE 0020 PEMPT 01CE 0020 PEMPT 01CE 0021 PEMPT 01CE 0022 PEMPT 01CE 0 | 2248 M LNK7 017D 2200 M LOADB 246E 2200 M MOVS 240E 243A M MOVS 240Z 231B M NUMBER 00D6 2552 M NUMBER 00D6 2572 M OPEN 234E 0020 PEND 239E 0020 PETTER 2454 0002 PETTER 2454 | 2248 M LNK7 017D 2200 M LOADB 246E 2200 M MOVS 2301 0026 MUL16 22E7 0026 MUL16 22E7 0026 MUL16 22E7 0020 MUL16 23F 0030 MUL16 23F 0 | 2248 M LNK7 017D 2248 M LOADB 246E 2200 M MOVE 2301 MOVE 2301 MOVE 246Z 2301 MUL16 22E7 MUL16 22E7 MUL16 22E7 2362 MUL16 22E7 2362 MUL16 22E7 2362 MUL16 22E7 2362 MUL16 2363 MU | 2248 M LNK7 017D 2200 M LOADB 246E 2200 M LOADB 246E 0024 MUL16 2267 0025 MUL16 2267 0026 MUL16 2267 0027 MUL16 2267 0023 MUL16 2406 0023 COEN 234F 0023 COEN 234F 0020 M COEN 237F 0000 M COEN 2167 0000 M COEN 2167 0000 M COEN 2167 0000 M COEN 2388 0000 M COEN 2388 | 224B M LNK7 017D 2200 M LOADB 246E 2200 M MOVE 2301 0026 R MUL16 22E7 0026 MUL16 22E7 24A2 0026 MUL16 22E7 24D 233B M NX10K 24D 2572 M OPEN 234F 0023 PERMEN 234F 234F 0020 PERMEN 234F 234F 0020 PERMEN 234F 255A 0021 PERMEN 245A 25CB 0022 PERMEN 245A 25CB 001E PERMEN 245A 25CB 0001E PERMEN 245A 25CB 0002 PERMEN 245A 21CF 0004 PERMEN 21CF 21CF 0005 PERMEN 21CF 21CF 0006 PERMEN 24CB 24CB 0007 PERMEN 24CB 24CB 0008 PERMEN 25B 20CB 0007 PERMEN 25B 20CB 0007 PERMEN 25 | 2248 M LNK7 017D 2200 M LOADB 246E 2200 M LOADB 246E 0024 R MUL16 2267 0026 MUL16 2267 0027 M NY10K 24D6 2572 M NY10K 24D6 2572 M OPEN 234F 0023 OPEN 234F 0020 PRTERR 2454 0021 PRTERR 2454 0020 PRTERR 2454 0020 PRTERR 2454 0020 PRTERR 2454 0021 PRTERR 2454 0021 PRTERR 2454 0021 PRTERR 2454 0021 PRTERR 2454 0022 PRTERR 2454 0022 PRTERR 2454 0022 PRTERR 2454 0022 PRTERR 2454 0000 PRTERR 2454 | 2248 M LNK7 017D 2200 M LOADB 246E 2200 M LOADB 246E 0024 R MUL16 2267 0025 MUL16 2267 0025 MUL16 2267 0023 DEEN 234F 0023 DEEN 234F 0020 PERFOR 237F 0020 PERFOR 237F 0020 PERFOR 237F 0020 PERFOR 237F 0020 PERFOR 2454 0021 PERFOR 2454 0022 PERFOR 2454 0022 PERFOR 2454 0020 PERFOR 2454 0020 PERFOR 2454 0020 PERFOR 2456 0000 PERFOR 2454 0000 PERFOR 2454 0000 PERFOR 2454 0000 PERFOR 2456 | 2248 M LNK7 017D 2200 M LOADB 246E 2200 M LOADB 246E 0024 R MUL16 2267 0025 M MUL16 2267 0026 M NX10K 2405 2318 M NX10K 2406 2572 M NX10K 2406 0023 DEEN 234F 0020 PKTERR 2454 0020 PKTERR 2454 0020 PKTERR 2454 0000 PKTERR 2454 | 224B M LNK7 017D 2200 M LOADB 246E 2200 M MOVC 2301 002A R MOVC 2301 002A MUL16 22E7 231B M NX10K 240E 231B M NX10K 240E 2572 M NX10K 240E 2572 M NX10K 240E 252 M NX10K 240E 252 M PKTERN 234F 252 M PKTERN 25A 250 M PULA 21E 250 M PULA 21E 250 M PULA 21E 250 M PKEND 23B 250 M P | 2248 M LNK7 017D 2200 M LOADB 246E 2200 M MUCL6 2301 0026 R MULL6 2267 0026 M NY 10K 246E 2318 M NX 10K 2406 2318 M NX 10K 2406 2318 M NX 10K 2406 2328 M NX 10K 2406 2328 M NX 10K 2406 2329 M NX 10K 2406 2320 M NX 10K 2406 2448 M NX 10K 2406 2449 M NX 10K 2406 2440 M NX 10K 2406 | 2248 M LNK7 017D 2200 M LOADB 246E 2200 M MUVS 2301 0026 R MUL16 2267 0026 M MUMBER 0267 2318 M NX 10K 2406 2572 M NX 10K 2406 2524 M NX 10K 2167 2650 M NX 10K 2167 | 2248 M LNK7 017D 2200 M LOADB 246E 2200 M MUVS 2301 0026 M MUL16 2267 0026 MUL16 2267 0027 MUL16 2267 0020 MUL16 2406 0020 MUL16 2406 0020 MUL16 2406 0020 MUL16 2406 0020 MUL16 2151 0000 MUL16 2267 0000 MUL16 0000 | 2248 M LINK7 017D 2200 M LONDB 246E 2200 M MUUS 2301 0026 MUUNS 246Z 2303 M MUL16 22CD 0026 MUL16 22CD 2305 M MUL16 22CD 0027 MUL16 22CD 0023 MUL16 24D6 2318 M NY10K 24D6 2318 M NY10K 24D6 2302 M NY10K 24D6 0023 CONDB COND | 2248 M LINK7 017D 224B M LOADB 246E 2200 M MUVS 2301 MOVS 24AE MUVS 2301 MUVS 24A2 MUVS 23436 M MUL16 22E7 MUVS 2345 MUL16 22E7 MUL16 22E7 MUL16 22E7 MUL16 22E7 MUL16 22E7 MUL2 21E MU | 2248 M LNK7 017D 2200 M LOADB 246E 2200 M MUUS 2301 0026 M MUL16 2267 0026 MUL16 2267 0027 MUL16 2267 0023 MUL16 2267 0023 MUL16 2406 2318 M NY10K 2406 2318 M NY10K 2406 0023 MUL16 2367 0020 M NY10K 2406 0020 MY10K 2289 0021 MY10K 2289 0021 SUBABX 2289 0022 MW10E 0027 0023 MW10E 0027 0026 MW10E 0027 0006 MW10E 2382 | 2248 M LNK7 017D 2200 M LOADB 246E 2200 M MUL16 2267 0026 M MUL16 2267 0026 MUL16 2267 0027 MUL16 2267 0023 MUL16 2267 0023 MUL16 2267 0020 MUL16 2287 0021 MUL16 2287 0021 MUL16 2287 0022 MUL16 2287 0027 MUL16 2287 0027 MUL16 2287 0027 MUL16 2287 0027 MUL16 2287 | 224B M LNK7 017D 2200 M LOADB 246E 2200 M MUVC 2301 0024 R MUVS 246E 2436 M MULI6 22CD 0026 M MURER 02CD 2345 M NY10K 24DE 252 M PWHT 21E 252 M PWHT 21E 252 M PULA 21E 252 M PULA 21E 252 M PULA 240c 2500 M PULA 240c 252 M PULA 240c 252 M PULA< | 2248 M LNK7 017D 2220 M LOADB 246E 2200 M MUL16 0024 R MUL16 2267 0025 M MUL16 2267 0026 MUL16 2267 0027 S11B M S11DK 24D6 231B M S11DK 24D6 231B M S11DK 24D6 231B M S11DK 24D6 231B M S11DK 24D6 2325 M NY10K 24D6 2250 M NY10K 22B3 2250 M NY10K 22 | 2248 M LNK7 017D 2220 M LOADB 246E 2200 M MUL16 2267 0026 MUL16 2267 0026 MUL16 2267 0027 MUL16 2267 0023 MUL16 2267 0023 MUL16 2267 0020 MUL1 | 2248 M LNK7 017D 2200 M LOADB 246E 2200 M MUL16 0024 R MUL16 2267 0025 M MUL16 2267 0026 MUL16 2267 0023 MUL16 2267 0023 COEN COEN 234F 0020 M NY10K 24D6 2318 M NY10K 24D6 2318 M NY10K 24D6 2318 M NY10K 24D6 0023 COEN COEN 234F 0020 M NY10K 24D6 0020 COEN COEN 234F 0000 COEN COEN COEN 238F 0000 COEN COEN COEN COEN COEN COEN COEN COEN | 2248 M LNK7 017D 22200 M LOADB 246E 2200 M MUL16 0024 R MUL16 2267 0025 M MUL16 2267 0026 MUL16 2267 0027 S11B M N710K 24D6 231B M N710K 24D6 231B M N710K 24D6 231B M N710K 24D6 2320 M N710K 24D6 0023 COENT COE | 2248 M LNK7 017D 2248 M LOADB 246E 2200 M MUVS 2301 0024 R MUL16 2267 0025 MUL16 2267 0026 MUL16 2267 0027 S11B M N710K 24D6 221B M N710K 24D6 221B M N710K 24D6 0023 COAD COAD COAD COAD COAD COAD COAD COAD | 2248 M LNK7 017D 2248 M LOADB 246E 2200 M MUVS 2301 0024 R MUVS 246E 0025 MUVS 246Z 2301B M WURER 0023 MURIER 0025 0023 COENT 01CE 0002 COENT | 2248 M LNK7 017D 22248 M LOADB 246E 2200 M MULL 6 2267 0026 MULL 6 2267 0026 MULL 6 2267 0027 MULL 6 2267 0023 MULL 6 2267 0023 MULL 6 2267 0023 MULL 6 2267 0020 MULL 7 2167 0020 MULL 7 2167 0020 MULL 1 2164 0021 MULL 1 2164 0020 MULL 1 2167 0020 MULL 1 2167 0020 MULL 1 2167 0021 MULL 1 2167 0022 MULL 1 2167 0020 MULL 1 2167 0020 MULL 1 2167 0021 MULL 1 2167 0022 MULL 1 2167 0020 MULL 2 2287 0000 MULL 2 2287 | 2248 M LNK7 017D 2248 M LOADB 246E 2200 M MULL 6 2267 0026 MULL 6 2267 0026 MULL 6 2267 0027 MULL 6 2267 0023 OFEN 234F 0023 OFEN 237F 0020 M NY10K 24D6 0023 OFEN 237F 0020 M NY10K 24D6 0020 MY10K 24D6 0020 | 2248 M LNK7 017D 2248 M LOADB 246E 2200 M MULL6 2203 M MULL6 2301 0026 M MULL6 2301 0026 M MULL6 2301 0027 M MULL6 2301 0023 M MULL8 2301 0023 M MULL8 2301 0023 M MULL8 2301 0020 M MULL9 2301 0020 M MULL8 2301 0020 M MULL8 2301 0020 M MULL8 2301 0020 M MULL8 2302 0020 M MULL8 0 | 2248 M LNK7 017D 2200 M LOADB 246E 2200 M MULL6 226T 0026 M MULL6 226T 0026 M MULL6 226T 0026 M MULL 8 226T 0020 M NT 10K 24D6 2218 M OFEN 234F 0023 OFEN 234F 0020 M NT 10K 24D6 0023 PRTERN 245F 0020 PRTERN 245F 0021 PSHALL 215F 0020 PRTERN 245F 0021 PSHALL 215F 0022 PRTERN 245F 0025 PRTERN 245F 0025 PRTERN 245F 0026 PRTERN 245F 0027 PRTERN 245F 0028 PRTERN 245 | 2248 M LNK7 017D 22248 M LOADB 246E 2200 M MULL6 2203 M MULL6 2301 0026 M MULL6 2301 0027 M MULL6 2301 0023 M MULL6 2301 0023 M MULL6 2301 0023 M MULL8 2301 0023 M MULL8 2301 0023 M MULL8 2301 0023 M MULL8 2301 0024 M MULL8 2301 0025 M MULL8 2301 0020 M M M M M M M M M M M M M M M M M M | 2248 M LUKY 017D 2248 M LOADB 246E 2200 M MULL 6 226T 2301 MULC 6 2318 M MULL 6 226T 2301 M | 2248 M LNK7 017D 22248 M LOADB 245E 2200 M MULL 6 2200 M MULL 6 2218 M MULL 6 2201 0026 M MULL 6 2201 0027 M MULL 6 2202 0023 M MULL 8 2202 0024 M MULL 9 2202 0025 M MULL 8 2202 0020 M MULL 9 0020 M MULL 9 0020 M MULL 9 0020 M MULL 9 0020 M M M MULL 9 0020 M M MULL 9 0020 M M M MULL 9 0020 M M M M M M M M M M M M M M M M M M | 2248 M LNK7 017D 2200 M LOADB 246E 2200 M MULL6 2200 M MULL6 2200 M MULL6 2201 0026 M MULL6 2201 0026 M MULL6 2201 0027 M MULL6 2202 0020 M MULL8 2202 0020 M MULL8 2202 0020 M MULL9 2202 0020 |

-	1	,
- 1	n	١

	169	0 I VO	
ILLEGAL INPUT DEVICE	ILLEGAL INPUT DEVICE END OF TABLE D I/O HANDLERS	S THAT THE DEVICE SUPPORTS CHARACTER S THAT THE DEVICE SUPPORTS BLOCK I/O POINT TO FCB CHECK STATUS IF BAD, ERROR MESSAGE	PRINT ERROR MESSAGE POINT TO FCB
R FDB LWRITE ** FDB DOPEN R FDB DOPEN R FDB DWRITE ** FCC 'LPT' R FDB SOPEN R FDB SOPEN R FDB LOPEN R FDB LOPEN R FDB LWRITE ** FCC 'MTA' R FDB LOPEN R FDB LWRITE ** FCC 'TTY' R FDB LOPEN R FDB LWRITE ** FCC 'TTY' R FDB LOPEN R FDB LWRITE ** FCC 'TTY' R FDB LCLOSE R FDB LWRITE ** FCC 'TTY' R FDB LWRITE	* FCC 'NUL' R FDB LOPEN R FDB LCLOSE FDB 0 R FDB LWKITE * FCB 0 * CHARACTER-ORIENTED *	* D-PREFIX INDICATES * L-PREFIX INDICATES * DOPEN TABX SWI FCB 3 OPEN SWI FCB 3 FCB 2 TT FCBSTA, X BNE IOERR *	RTS * IOERR PRTERR SWI FCB 30 * * * DCLOSE TABX SWI FCB 3 CLOSE SWI SWI
0279 035B 027E 0283 027E 0283 0282 02C4 0284 02D4 0289 0389 0289 0389 0280 0000 0291 4D 0294 02DE 0294 02DE 0298 0330 0298 0330 0297 0358	0287 4E 0284 02DE 028C 0321 028E 0000 02BO 035B 02BZ 00		02BB 39 02BC 3F 02BD 1E 02BE 39 02BF 3F 02C0 03
		++ ++	0170 0 0171 0 0173 + 0173 + 0175 0 0176 0 0177 0 0178 0 0187 + 0180 +
FILE TYPE FILE ACCESS CODE FIRST TRACK/SECTOR LAST TRACK/SECTOR NUMBER OF SECTORS BLOCKS FOR "PIP" ADDRESS OF INPUT DEVICE IN DEVIAB INPUT BUFFER ADDRESS OF OUTPUT DEVICE IN DEVTAR OUTPUT BUFFER	NUMBER OF FIRST TRACK ON DISK NUMBER OF FIRST SECTOR ON DISK NUMBER OF SECTORS/TRACK NUMBER OF TRACKS ON DISK NUMBER OF TRACKS ON DISK	BINAKY-FUKMAT FLAG OPEN HANDLER CLOSE HANDLER READ A CHARACTER WRITE A CHARACTER	ILLEGAL OUTPUT DEVICE
113 115 115 117 1180 1180 125 6 6 0 1180 125 25 25 25 25 25 256	141801ES	7 13 (FUB LCLCASE FUB LWRITE FUC 1978 FUB LOPEN FUB LCLOSE FUB CC 1979 FUB COPEN
FIBHTS FIBHTS FIBHTS FIBHTS FIBHTS FIBHTS INHND ** INHND ** OUTFCB OUTFCB ** OUTFCB ** OUTFCB **	X XONNN	** ** ** ** ** ** ** ** ** ** ** ** **	* *
	* DISK FSTIRK FSTSEC SECSIZ TRKSIZ DSKSIZ * HFLAG	0001 43 020E 8 DE 933	

B FCBIND+1, X FCBSTA, X RETURN GOOD STATUS	CHECK FOR VALID OUTPUT DEVICE OK?	NO, ERROR MESSAGE		RETURN ERROR STATUS X			POINT TO FCB	INPUT OR OUTPUT?	INPUT DOES NOTHING	OUTPUT CNTL-D (EOF)	OUTPUT CARRIAGE RETURN		POINT TO BUFFER	GET CHARACTER CNTL-D (EOF)? NO	RETURN EOF STATUS	×	CARRIAGE-RETURN? YES	MOVE POINTER SAVE 'A'	POINT TO FCB	ri, X RECOVER 'A' RETURN GOOD STATUS
STA B FCBIND+ CLR FCBSTA, X RTS		LDX #ERR4 PRIMSG	SWI FCB 49	LDX #OUTFCB LDA A #18 STA A FCBSTA, X	RTS LDX	BKH LUTZB	* LCLOSE TABX SWI	FCB 3 TST FCBDTT, X BNE LCLOS2	RTS	* .clos2			SAD LDX #INFCB LDX FCBIND, X		# 12	STA B FCBSTA, X RTS	LREAD2 CMP A #\$OD BEQ LREAD3 *	INX PSH A TXAB		STA B FCBIND+1, X PUL A REI CLR FCBSTA, X RE
0303 E7 28 0305 6F 05 0307 39	0308 FE 0156 R LOP 0308 EE 09 030D 26 0D		0312 3F 0313 31	0314 CE 012C R 0317 86 12 0319 A7 05	36 38	Z0 nC	0321 3F	0322 0323 0325	, * (0327 39	86 04		i	* 0330 CE 0000 R LREAD 0333 EE 27	84 84 87	දුපු ද	0340 E7 05 0342 39	0343 81 0D LRE 0345 27 0F *	0347 08 0348 36	+ 0349 3F + 034A 02 034B CE 0000 R 034E A7 27	0350 E7 28 0352 32 0353 6F 05
0245 0246 0247	0248 0249 0250 0251	0252	0255 + 0256 +	0257 0258 0259	0260 0261 0262	0263 0264 0265	0266 0267 0268 +		0272 0273	0274	0276	0279	0281 0282 0283	0284	0288 0288 0289	0290	0293 0294 0294	0296		0303 0304 0305
			X CHECK STAIUS EOF? NO	YES	ERROR? IF BAD, ERROR MESSAGE			CHECK SI					INPUT OR OUTPUT?	CHECK FOR VALID INPUT DEVICE	UK? NO, ERROR MESSAGE		RETURN ERROR STATUS		ISSUE 1/0 REQUEST	, X +1, X RE-INIT. FCBINDEX , X
FCB 21 RTS	LDX #INFCB	SWI FCB 24	LDA B FCBSTA, X CMP B #8 BNE *+3		TST B BNE IOERR	RTS	LDX #OUTFCB	WKITE SWI FCB 25 TST FCBSTA, X	BNE 10ERR	RTS		SWI	TST FCBDTT, X BNE LOPOUT		BNE LOPZ LDX #ERR3 DDIMEG	SWI FCB 49	LDX #INFCB LDA A #18 STA A FCBSTA, X	RTS LDX #INFCB	IOHDR SWI FCB 19	LDA A FCBDBA, X LDA B FCBDBA+1, X STA A FCBIND, X
*	* * ? DREAD			* *	* *	* *	* R DWRITE		*	*	* *	LOPEN		* ```	*		œ		* LOP2A	* LOP2B
+ 02C2 15 02C3 39	* * 02C4 CE 0000 R DREAD	+ 02C7 3F + 02C8 18	02C9 E6 05 02CB C1 08 02CD 26 01	02CF 39	02D0 5D 02D1 26 E9	0 2D3 39	* 02D4 CE 012C R DWRITE LDX #OUTFCB	+ 02D7 3F + 02D8 19	OZDB 26 DF	02DD 39		+ 02DE 3F	+ 02DF 03 02E0 6D 06 02E2 26 24	FE 002A EE 07	02E9 26 0D 02EB CE 03A7 F		CE 0000 86 12 A7 05	02F7 39 02F8 CE 0000 R	+ 02FB 3F + 02FC 13	02FD A6 07 02FF E6 08 0301 A7 27
								+ +					τ-			T T				

/ER-RUN′	ERROR <	ERROR '	WITCH'	'DEVTAB'	REGISTER RETURNS ADDRESS OF DEVICE BLOCK BIT CLEAR IF FOIND, SET IF NOT FOIND		EN IN CLI	POINT TO CLI NAME		POINT TO DEVICE TABLE				DEVICE NAMES ARE 3 CHARS		COMPARE NAMES			GET NEXT TABLE ENTRY	11 12 12 12 14 14 14 14 14 14 14 14 14 14 14 14 14			CHECK FOR END OF TABLE		CLEAN STACK			RE-SET FOR COMPARE						
ERRS FCC ' BUFFER OVER-RUN'	UMBER FCC / NUMBER FCB \$0D	* FORMAT FCC < FORMAT EF FCB \$OD	* SWITCH FCC / ILLEGAL SWITCH/	LOOK UP DEVICE IN	* * INDEX REGISTER RETURNS ADI * CARRY-BIT CLEAR IF FOLIND.	*	* WORKS ON PRESENT TOKEN IN	DLKUP LDX DESCRA	IMS	LDX #DEVTAB	PSHX	FCB 5		CMP B #3 BNE NOTFND		DLKUPZ CMPC SWI	FCB 18 BEG FOUND	*	LDX SAVEX	ADDBX	SWI FCB 10		BEG NOTFND	*	SNI	SNI	INS LDX DESCRA	PSHX	1.35 1.35 1.35 1.35 1.35 1.35 1.35 1.35		I ASS	FCB 5 LDA B #3	BRA DLKUP2	NOTFND INS
03D4 20	03E5 03F2	03F3 20 0400 0D	0401 20					0411 DE 20		+ 0415 CE 025A R		7 05 71 0488	041D D6 22	041F C1 03 0421 26 23		+ 0423 3F	+ 0424 12 0425 27 25	1	0427 FE 0455 K	000	+ 0420 3F + 0420 0A	H 2	0433 27 11	10.80%			0438 31 0439 DE 20		+ 043B3F + 043C05		+ 0440 3F	0441	0444 20 DD	0446 31
0367	0369 0370 0371	03/2 03/3 03/4	0375	0379	0380 0381 0382	0383	0384	0386	0388	0380	0391	0393	0395	9680	8660	0399	0401	0403	04040	0406	0407	0409	0410	0412	0414	0415	0416	0418	0419	0421	0423	0424	0426	0428
												INDEX																						
	Input a block and reset index reload cr.		GET BUFFER POINTER		yes, ERROR	SAVE 'A'		POINT TO FCB	× -	RECOVER 'A'	RIAGE RETURN?	IF SO, OUTPUT BLOCK AND RE-INIT.		BUFFER OVER-RUN			RETURN ERROR STATUS	×			SPECIAL OPEN FOR LINE-PRINTER	ISSUE A LINEFEED			SPECIAL CLOSE FOR LINEPRINTER	7.			PUT		INPUT DEVICE		/ ILLEGAL OUTPUT DEVICE/	
RTS	BSR LOP2 LDA A #\$0D RTS		LDX #OUTFCB LDX FCBIND, X STA A A.X		CFX #UUIBUF+236 BEG LWRIT2	PSH A		FCB 2 LDX #0U1FCB		4 €		BEQ LOP2A	RTS	LDX #ERR5	PRIMSG	FCB 49		STA B FCBSTA, X	<u>0</u>		JSR LOPEN	LDA A #\$OC		* 61.7	RTS		FCC / BAD INPUT		FCC / BAD OUTPUT	FCB #OD		FCB #OD	FCC / ILLEGAL	
*	READ3		L.WRITE		*	'	. –					*		R LWRITZ L	-			v) 0	*	* *	SOPEN	-1 O		* *	CLOSE	* *	KR1		* ERR2 F	ш *	3R3		ERR4 F	*
0355 39	0356 8D A0 0358 86 0D 035A 39		035B CE 012C R 035E EE 27 0360 A7 00) (0 (0 (0 (0 (0 (0 (0 (0 (0 (0 (0 (0 (0	0366 27 13	98 8980	+ 0369 3F	+ 036A 02 036B CE 012C R	67 27	32 72	2 g	0378 27 81	037A 39	037B CE 03D4 R	36 3200		88	0385 E7 05			0388 BD OZDE R	038B 86 0C	Ì		038F 39		0390 20		039B 20	03A6 OB	0347 20		03BD 20 03D3 0D	
0306	0308	0312 0312 0313	0314	0317	0319 0320	0321			0326	0328	0330	0831 0831	0333	0335	0336	0338 +	0339 0340	0341	0343	0344	0346	0347	0349	0351	0352	0353	0355	0356	0358	0359	0361	0363	0364	0366

STORE DRIVE NO. NUMBER ERROR	GET NEW CLI	GET A TOKEN	CHECK RC COLON? NO. FRROR			CHECK RC EQUALS?	ON	SET "COPY" MODE FLAG PROCESS INPUT	NAME? NO, ERROR	YES, (AND NOT DEVICE)	NAME? YES	NO, FORMAT EKROR	GET NEW CL.I	DEVICE NAME?	NO, TRY AS FILE	SAVE ADDRESS					PUT DEVICE NAME INTO PLB		CI FON STOCK	
LDX #OUTFCB STA A FCBDRV, X BRA PIP1B LDX #NUMBER PRTMSG	FKIMSU SWI FCB 49 UMP PIPNXT	NXTOK SWI FCB 47	LDA B RC CMP B #7: BMC DID10		SWI FCB 47			COM PIPFLG JMP PIP5	CMP B #1 BNE PIP2A	BRA PIP3	CMP B #1 BEG PIP2B	LDX #FORMAT	SWI SWI FCB 49	SR DIKIE	BCS FIP3	STX OUTHND LDX #OUTFCB+FCBGDT		FCB 5 LDX OUTHND	PSHX		LDA B #3 MOVC	SWI FCB 17		SNI
04A8 CE 012C R 04AB A7 09 04AD 20 08 A4AF CE 03ES R PIPIA	04B2 3F 04B3 31 04B4 7E 06F3 R	PIP1B 04B7 3F 04B8 2F	04B9 D6 25 04BB C1 3A	3		27	26	04CA 75 0798 R 04CA 7E 0597 R	* 04CD C1 01 PIPIC 04CF 26 06	04D1 20 28	04D3 C1 01 PIP2 04D5 27 08	* 04D7 CE 03F3 R PIP2A	04DA 3F 04DB 31	5	25 17	04E4 FF 0156 R 04E7 CE 012E R		04EB 05 04EC FE 0156 R		04F0 05	04F1 C6 03	04F3 3F 04F4 11	04F5 31	
0491 0493 0494 0495	0496 0497 0498 + 0499	0501 0502 + 0503 +	0505	0507	0509	0511	0513	0515 0516	0517 0518 0519	0520	0523 0523 0524	0525	0528	0531	0533	0535 0536	0537	0539	0541	0543	0544	0546	0548	0550
CLEAN STACK 0491 0492 0494 0494			OSOA TEMP. STORAGE OSOS OSOS	0507 0508 0508	INPUT DEVICE=DSK	DEVICE=DSK	NO COMPRESSION ON INPUT 0513	OUTPUT			0523 0523 0524 0524	0522 0522 0532	NO FILE NAME (BLANK FIRST) 052 DEFAULT FILETYPE=BINARY (00) 055	GET A TOKEN FROM CLI 0550	2000 2000	CHECK FOR ESCAPE 053	053/ 0536 0538	ESCAPEDONE 053	CHECK RC 034		OSA VALID DRIVE NO.? 054		VALID DRIVE NO. ? 054	ON ON
	0496 0497 CLEAN STACK 0498 0499		2 TEMP. STORAGE	AND PARSING ROUTINES	#DSKDEV INHND DEFAULT INPUT DEVICE=DSK	OUTHND DEFAULT OUTPUT DEVICE=DSK *INFCB	FCBSCF, X NO COMPRESSION ON INPUT #OUTFCB	FCBDRV, X DEFAULT DRIVE=0 FCBSCF, X NO COMPRESSION ON OUTPUT	PIPFLG INIT. PARSE FLAG NO BINARY REFORMAT HFLAG NO HEX REFORMAT	FCBSTA, X NO EKRORS A * D A * D	A FCBGDT, X A.#\S A FCBGDT+1, X	# /K FCBGDT+2, X	A #520 A FCBNAM, X NO FILE NAME (BLANK FIRST) A #0 DEFAULT FILETYPE=BINARY (00)	OK GET A TOKEN FROM CLI	521 CD8 47 - 57 PERSON - 57 PE	DESCRIPE O A O A CHECK FOR ESCAPE A ES				PIP2 NO	۲-	PIP1A NO	A VALUE+1 VALID DRIVE NO. ?	

	173	
"C"? NO FILETYPE=TEXT (03) SET SPACE—COMPRESSION ON "H"? NO	A T T T S S S S S S S S S S S S S S S S	GET ADDRESS OF HANDLER OPEN DEVICE OR FILE CHECK STATUS GOOD BAD OUTPUT ERROR GET NEW CLI RECOVER TOKEN EQUALS? NO, ERROR OF CLI DEFAULT DRIVE=0 NO ERRORS
CMF A # C BNE PIP4B LDA A #\$03 STA A F\$TYP, X INC FCBSCF, X BRA PIP4 CMP A # 'H BNE PIP4C	- α μα	, x , x , x , x , x , x , x , x , x , x
054E 26 08 ** 055E 26 08 ** 0550 86 03 ** 0554 6C 29 0556 20 08 ** 0558 81 48 PIP4B	055C 7C 0258 R 055F 86 03 0563 20 CB ### Comparison of the compar	
0613 0614 0615 0615 0615 0619 0620 0621	+ +	+ +
NOW OPEN DEVICE SAVE POINTER TO NAME SAVE LENGTH GET A TOKEN CHECK RC		FORMAT NAME INTO FCB CLEAN STACK ERRORS? YES GET A TOKEN CHECK RC SWITCH INDICATOR? NO GET SWITCH FROM CLI "B"? NO SET BINARY FLAG FILETYPE=BINARY (00)
BRA PIP4 LDX DESCRA STX SAVEX LDA A DESCRC STA A SAVEA NXTOK SWI FCB 47 LDA B RC	BNE PIP2A INC SAVEA NXTOK SWI FCB 17 LDA B RC CMP B #1 BNE PIP2A LDA B DESCKC ADD B SAVEA LDX #OUTFCB+FCBNAM SWI FCB 5	FMTS SWI FCB 52 INS
04F9 20 35 ** 04FB DE 20 PIP3 040E FF 0455 R 0502 B7 0457 R 0505 3F 0506 2F 0507 D6 25	050B 26 CA ** 050D 7C 0457 R ** 0510 3F ** 0511 D6 25 0514 C1 01 0516 26 BF ** 0518 D6 22 0518 D6 22 0518 FB 0457 R 0510 CE 013C R 0520 3F 0521 05 0522 FE 0455 R	0527 3F 0528 34 0528 31 0528 31 0528 31 0520 50 0520 50 0521 50 0532 2F 0531 2F 0532 16 25 0534 C1 2F 0534 C1 2F 0538 3F 0539 2F 0539 2F 0539 2F 0539 2F 0539 2F 0539 2F 0538 3F 0538 8F 0548 6F 0548 8F 05548 6F 05548 6F
0552 0553 0555 0555 0556 0557 0559 0559 0560 +	0563 0564 0564 0566 0566 0567 0571 0572 0573 0574 0575 0577 0578 0578 0578 0578 0578 0578	0583 0584 0584 0586 0588 0589 0590 0592 0594 0597 0598 0598 0598 0500 0600 0603 0603 0603 0604 0603 0604 0607

1	7	1

			IT FILE													1'	74																				
	NAME?		FORMAT ERROR FINISH AND CLOSE OUTPUT FILE	DEVICE NAME? NO, TRY AS FILE NAME	SAVE ADDRESS					CHARL PASSES	COLUMNIA INTO TOO			CLEAN STACK		NOW OPEN DEVICE		SAVE POINTER TO NAME	SAVE FNGTH	GET A TOKEN		CHECK RC	PERIOD?	NO, ERROR	COUNT PERIOD	GET A TOKEN		CHECK RC			MILLI-CHRU MHRE:		GET LENGTH OF EXT TOTAL LENGTH			MOVE NAME INTO FCB	
BRA PIP7	CMP B #1	3	LDX #FORMAT BRA PIPSB	JSR DLKUP BCS PIP7	STX INHND	LDX #INFCB+FCBGDT	SWI	FCB 5 LDX INHND	PSHX		LUM B #3		FCB 17	SNI	I NO	BRA PIP8	LDX DESCRA		LDA A DESCRC	: ¥	SWI	LDA B RC	CMP B #7.	BNE PIP6A	INC SAVEA	NXTOK	FCB 47	LDA B RC	۵ و		CMF B #Z	ğ	LDA B DESCRC ADD B SAVEA	#	POTTX OWI FICE S		IMS
0604 20 25	C1 01		060A CE 03F3 R PIP6A 060D 20 C9	060F BD 0411 R PIP6B 0612 25 17	FF 002A			061B 05 061C FE 002A R	1		0621 C6 U3	0623	0624 11 0625 31			0629 20 46	* 0428 DF 20 PIP7	FF 0455 R	0630 96 22	(Ct.) (a		_	0639 C1 2E	063B 26 CD	063D 7C 0457 R		0641 ZF	8	0644 C1 01	ì			22 0457	빙	0.654 3F	0656 F	0659 3F
0735	0737	0738	0740	0742	0745	0747		0750 +		0754 +	0756		0758 +	0760	0761	0762	0764	0766	0767	02/0		0771 +	0773	0774	07.70		0779 +	0780	0/81	0783	0784	07870	0787	6870	0790		0795 +
BMON 7117 ON	GET A TOKEN		CHECK RC NUMBER?	IN "COPY" MODE?	ND, O. K.	IN "COPY" MODE, MUST HAVE NUMBER	VALID DRIVE NO. ?	ON	VALID DRIVE NO. ?	ON		SAVE DRIVE NO.		NUMBER ERROR			Pass FCB ADDRESS		03 JUNE 200 5 130		CLOSE OUTPUT FILE	NEW C. T		GET A TOKEN		CHECK RC	NO, ERROR		OF A LOKEN		CHECK RC	END-OF-LINE:	VES. DISK-TO-DISK COPY		UNAMBIG. NAME? YES	WILD-CARD NAME? IF NOT, ERROR	
LDA A #\$20 cto a Econom.y	NXTOK	FCB 47	LDA B RC CMP B #3	BEW FIFSU	BEG PIP6	BRA PIP6A		BNE PIP5A	Œ	CMP H #3 BHI PIP5A	I DX #1NFCB	STA A FCBDRV, X	BRA PIP5D	LDX #NUMBER	PRIMSG	FCB 49	LDX #OUTFCB	IMS	FCB 2	LDX 5, X	JSR 0, X	TANGLE OM		NXTOK	SWI FCB 47	LDA B RC	BNE PIPSA		SMI	FCB 47		CMP 8 ##OU BNE PIPSE	VACATTA CINE		CMP B #1 BEQ PIP7	CMP B #2 BNE PIP6A	
74 05B0 86 20	(1352 H)	77 + 05B4 3F 78 + 05B5 2F	05B6 D6 05B8 C1	OSBC OSBC	05BF 27 45	05C1 20 47	0503	39 05C6 26 0D **	0508 96 28	05CC 22 07	9050	05D1 A7 09	77 05D3 20 14 **	OSDS CE OSES R	4	01 + 0508 3F 02 + 0509 31	03 OSDA CE O12C R	+	+ 05DE 02	7/ 05DF FE 0156 K 08 05E2 EE 05	05E4 AD 00	11 0564 75 0453 B 0105C	USES /E VOLS IN		14 + 05E9 3F 15 + 05EA 2F	OSEB D6	17 USED CI 3A 18 OSEF 26 E4		+ 05F1	+ +	05F3 D6	24 OSFS C1 OD 25 OSF7 26 OS	0550 75		29 05FC C1 01 P1P5E 30 05FE 27 2B	0600 C1 02 0602 26 06	*
0674	0676	0678	0890	0682	0684	9890	0687	06890	0691	0693	0694	9690	0690	6690	0700	0707	0703	0705	0706	0708	070	0710	0712	0713	0714	0716	0718	0719	0720	0722	0723	0724	0726	0728	0729	0732	0734

ERROR?	DDINI EUDOD MEGGAGES	TAIN ENDON TEOORGEO	SEND NULL CHARACTER			CHECK STATUS	MCACOOM GOODD TATOO	TAINI ERRON MESSANGE	BAD OUTPUT MESSAGE	CLOSE OUTFILE AND RE-PROMPT	POINT TO INPUT FCB		GET INPLIT CLOSE ROLLTINE	CLOSE DEVICE FILE	NEW TOKEN			NO N		CLOSE OUTFILE AND RE-PROMPT	DELIMITER?	ON	SOTUBOL FIGHT HEM SOCIOOD	TOTAL OF THE PROPERTY OF THE P	FURMAT EKKOK	POINT TO INPUT FCB			CLOSE INPUT DEVICE FILE	Issue Fir Front		GET NEW CLI LINE
TST B	DE 4 1 00	SWI		ěě				SWI			TXAB	FCB 2	LDX INHND	JSR 0, X	NXTOK SET	FCB 47			L.DX #PRMPT2	JMP PIPSB	LDA A	<u>د و</u>			MP PIP6A			LDX INHND	X SS	PRT	SWI FCB 49	GTCMD SWI
50	Ì	06B7	06B9 4	FE 0156 R	AD 00 CE 012C	60 05 27 Pr	ì	8090	06CY 1E 06CA CE 039B	OSD8 R	200	0601	06D2 FE 002A R	06D7 AD 00		06DA 2F	200	26 06	0713 R	7E 05D8 R	96 26	81 26	. 100 kg	/E (132)	7E 060A	06F3 CE 0000 R PIPNX	06F6		AD 00	CE O/OD	0702	
0980	0862			8930 8930	0870	0872	0874			0879			0884	9880			0880	0892	0894	3680 7000	0890	0898	0060	0902	0903	9060 9060			0910 0911	0912 0913	0914	
	CLEAN STACK		EKKUK? YES	IN "COPY" MODE?	PROCESS FILE-COPY	CHACK GOOD ALL	YES, ERROR				i	5						CLOSE FILES	P								SPACE-COMPRESS	COPY AS-IS	YES, REFORMAT		_	CHECK STATUS
SWI FCB 52	SNI					9	BEG PI	LDX #INFCB	TXAB	FCB 2	LDX 3, X			IMS					LDA	0 G		WARY FILE HERE	TST HFLAG	BEU PIPBB	UMP HEXFRM	(T FILE HERE	INC		UMP BINFRM	XC	ğ	LDX #INFCB
0658 3F 065C 34	31 31			98 R	0894	: :		* CE 0000 R PIP8	0674 3F	02	03		067D CE 0000 R PIP8A	0680 3F	0681 1E	0682 6D 05 0684 27 06	0	7E 05D8 R		188	* * *		58 R	27	0697 7E 0B51 R		60 29		* 7E OD43 R	ũ	200	
	SWI (0859 * TST B (0860 0684 5D TST B (0860 06	SWI SWI FCB 52 ** (86.0 0.684.50 ** TST B ERROR INS	SWI CLEAN STACK CR67 SWI CR67 SWI CW67 SW	SMI (852 (864 (864 50 464 50 TST B ERROR 1085) ** TST B ERROR 1086 (865 27 03 BEG PIPSC NO 1086 1 NO 1086	FCB 52	FCB 52	FCB 52	SMI FCB 52 SMI SEG PIPBC O684 5D SEG PIPBC O684 5D O684 5D	SMI	SHATE SHAT	FOB 52 F	FCB FCB	FCB 52 FCB 70 F	SM	FUB 52 FUB 54 F	FEB 52 FEB 04 FEB 52 FEB 04 FEB 52 FEB 04 FEB 52 FEB 04 F	Fight	SHADE SHAD	SH SH SH SH SH SH SH SH	SM1	Mail	High State Hig	Figs 52 Figs 64 Figs	1168 522 126 10	Figh 32 Figh	FIRST FIRS	188 18 18 18 18 18 18 1	158 158	Figs 22 Figs 32 Figs	Fig. 52 Fig.	156 157 158	State Stat

0700 70	0780 0780 0781 0781 0782	0 0760 CE 002C R 11 0763 3F	LDX #INBUF PSHX SWI	POINT TO INPUT SECTOR BUFFER
0712 04	AGAIN	+ +	581 FCB 5 FDA B #SFCS17	MOVE DATA FROM INPUT TO DUTHUT
# DISK-TO-DISK (NON-PACKING) # SYN1AX: FEB #0D # DISK-TO-DISK (NON-PACKING) # SYN1AX: PIP TODRV:=FRMDR # SYN1AX: PIP TODRV: # SYN1AX: PIP TODRV:=FRMDR # SYN1AX: PIP TODRV: # SYN1AX: PIP TODRV: # SYN1AX: PIP TODRV: # SYN1AX: PIP TODRV:=FRMDR # SYN1AX: PIP TODRV: # SYN	5860 9860	+	MOVC	
# DISK-TO-DISK (NON-PACKING) # SYN16X: PIP TODRV:=FRMDR # SYN16X: PIP TODRV: # SYN16X: PIP	0987	+	FCB 17	
# DISK-10-DISK (NON-PACKING) # SYNTAX: PIP TODRV:=FRMDF 0710 A6 09 0710 A6 09 0710 B8 30 0720 B7 07AB R 0730 3F 0730 3F 0730 B7 0730 3F 0730 B7 0730		076A	INS	CLEAN STACK
** SYNIAX: PIP TODRV:=FRMIDR ** 0715 CE 0000 R DTUCPY LDX #INFCB	0990	0 076B 31	INS	
0719 CE 0000 R DTUCPY LDX #INFCB 071C A6 09 0720 B 30 0720 CE 012C R STA A FRMDRV 0723 CE 012C R LDX #0UTFCB 0724 A5 09 0724 B 7 07AB R STADDL 1 0729 B 30 0729 CE 0729 R STADDL 1 0729 B 7 07AB R STADDL 1 0730 3F STADDL 1 0730 3F STADDL 1 0730 B 7 07AB R STADDL 1 0734 B 7 0AB R STADDL 1 0735 B 1 59 0736 CE 012C R DTDCP1 LDX #0UTFCB 0740 CE 012C R DTDCP1 LDX #0UTFCB 0740 CE 012C R DTDCP1 LDX #0UTFCB 0745 E 7 0B 0745 E 7 0B 0745 E 7 0B 0745 E 7 0B 0746 3F STADDL 1 0750 CB 05 0746 3F STADDL 1 0750 CB 05 0746 3F STADDL 1 0750 CB 05 0750 CB 0750 CB 0750 CB 0750 0750	Z660	07/0	LDX #OUTFCB	POINT TO "TO" FCB
071C A6 09 LDA A FCBDRV, X 0720 B2 07AB R STA A FRMDRV 0723 CE 012C R LDX #0UJFCB 0724 A5 09 LDA A FCBDRV, X 0728 B8 30 ADD A #\$30 0724 B7 07B6 R STA A TODRV 0729 CE 0729 R STA A TODRV 0729 CE 0729 R STA A TODRV 0729 CE 0729 R STA B TORR 4 0731 31 FCB 49 0734 A5 00 LDA A 0.7 SEC 0740 CE 012C R DTDCP1 LDX #0UJFCB 0735 B1 59 BNE DTDCP4 0736 B1 59 BNE DTDCP4 0736 B1 59 BNE DTDCP4 0737 B 00 LDA A #\$7 0736 CE 012C R DTDCP1 LDX #0UJFCB 0740 CE 012C R DTDCP1 LDX #0UJFCB 0740 CE 012C R DTDCP1 LDX #0UJFCB 0740 CE 012C R DTDCP1 LDX #0UJFCB 0745 E7 0B STA B FCBSCT, X 0745 E7 0B STA B FCBSTR, X 0745 E7 0B STA B FCBSTR, X 0746 B7 B8 STA B FCBSTR, X 0746 B7 B8 STA B FCBSTR, X 0747 CE 000 R STA B FCBSTR, X 0746 B7 B8 STA B FCBSTR, X 0746 B7 B8 STA B FCBSTR, X 0746 B7 B8 STA B FCBSTR, X 0747 B7 B8 BFTBRR 0752 27 07 B BEQ DTDCP2 0758 CE 0758 R DTDCP2 LDX #0UTBUF 0758 B1 59 BNE	0993 TO "FROM" FCB 0994	3 4 + 0770 3F	IOHDR SHI	WRITE "TO" SECTOR
071E 8B 30 ADD A #\$30 0720 B7 07AB R STA A FRMDRV 0726 A5 09 LDA A FRDRV, X 0728 8B 30 ADD A #\$30 0728 8B 30 ADD A #\$30 0728 B7 0786 R STA A TODRV 0720 CE 0799 R LDX #DTDL1 PRTPSG + 0731 31 GTCMD + 0732 3F SWI + 0733 30 FCB 49 0734 A5 00 LDA A 0.7 0736 CE 012C R DTDCP1 LDX BOUTFCB 0736 CE 012C R DTDCP1 LDX #DUFCB 0737 B E 20 CMP A # * Y 0736 CE 012C R DTDCP1 LDX #DUFCB 0740 CE 012C R DTDCP1 LDX #DUFCB 0741 A7 0A STA A FCBTRK, X 0745 E7 0B STA A FCBTRK, X 0745 E7 0B STA A FCBTRK, X 0746 A7 0A STA B FCBSCT, X 0746 A7 0A STA B FCBSTR, X 0746 A7 0A STA B FCBSTR, X 0746 A7 0A STA B FCBSTR, X 0750 6D 05 BEQ DTDCP2 + 0755 1E FCB 30 0756 CE 07CB R LDX #DRFR PRINSG + 0759 3F SWI + 0759 3F SWI + 0750 31 # FCB 49		. +	FCB 19	
0.720 87 0.748 K SIGN H FYRUNKV O7726 A6 09 LDA A FCBDRV, X O7726 A8 30 ADD A #\$30 ADD A #\$30 O720 CE 0.799 R LDA #DTDL1 PRTMSG SM1 A TODRV O721 CE 0.799 R LDA #DTDL1 PRTMSG SM1 A TODRV O731 31 FCB 49 GTCMD SW1 A TOTSG SW1			TST FCBSTA, X	CHECK STATUS
0726 46 09 LDA A FCBDRV, X 0728 8B 30 ADD A #\$30 0729 E 0729 R STA A TODRV 0721 CE 0729 R LDA #DTDL1 + 0731 31 FCB 49 + 0732 3F SWI + 0733 30 FCB 49 0734 AE 00 LDA BESCRA 0734 AE 00 LDA A 0.00 R LDA A FSTTRK 0736 CE 012C R DTDCP1 LDX #DUFP4 0736 E 00 R LDA A FSTTRK 0737 CE 000 R LDA B FCBTRX, X 0745 E 7 0B STA A FCBTRX, X 0745 E 7 0B STA A FCBTRX, X 0745 E 7 0B STA A FCBTRX, X 0746 E 7 0B STA A FCBTRX, X 0746 E 7 0B STA A FCBTRX, X 0747 CE 012C R DTDCP1 LDX #DUFP 0748 AT 0A STA A FCBTRX, X 0745 E 7 0B STA A FCBTRX, X 0746 E 7 0B STA A FCBTRX, X 0746 A 7 0A STA A FCBTRX, X 0746 A 7 0A STA A FCBTRX, X 0747 C 6 0 D STA A FCBTRX, X 0746 B FCBSCT, X 0746 A 7 0A STA A FCBTRX, X 0747 C 6 0 D STA A FCBTRX, X 0746 A 7 0A STA A FCBTRX, X 0746 A 7 0A STA A FCBTRX, X 0747 C 6 0 D STA A FCBTRX, X 0750 C 0 D STA A FCBTRX 0750 C 0 D STA A FCBTRX 0750 A 7 0A STA A FCBTRX 075		7 0//4 Z/ 0/ 8	BEG DIDCP3	. r.
0728 88 30 ADD A #\$30 0720 CE 0799 R LDX #DTDL1 + 0730 3F SW1 + 0731 31 FCB 49 10730 3F SW1 + 0732 3F SW1 + 0733 30 FCB 49 0734 A6 00 LDX DESCRA 0734 A6 00 LDX DESCRA 0736 81 59 BNE DTDCP4 0737 80 CMF A # 'Y 0738 81 59 CMF A # 'Y 0738 81 59 BNE DTDCP4 0738 81 59 BNE DTDCP2 0740 CE 012C R DTDCP1 LDX #0UTFCB 0740 CE 012C R DTDCP1 LDX #0UTFCB 0740 CE 012C R DTDCP1 LDX #0UTFCB 0750 CE 07CB R LDX #DRERR 0750 SE 015B R DTDCP2 LDX #0UTBUF 0750 CE 015B R DTDCP2 LDX #0UTBUF	ON.		PRTERR	PRINT ERROR MESSAGE
10	ASCII 1000	0 + 0776 3F	IMS	
PRTMSG + 0730 3F SW1 + 0731 31		0778 0	LDX #DWERR	"WRITE"
+ 0730 3F SWI + 0731 31 FCB 49 + 0732 3F SWI + 0733 30 FCB 48 0734 A6 00 LDA A 0.7 CMP A # 'Y 0738 81 59 CMP A # 'Y 0740 CMP A B # CBBTR, X 0750	1003		PRTMSG	
O'752 3F	1004	4 + 077B 3F	SWI	
+ 0732 3F SWI + 0732 30 FCB 48 0734 A6 00 LDA P 60.CRA 0738 81 59 CMF A # Y 0738 81 59 BNE DTDCP4 0736 26 59 ** LDA A #FSTYRK 073E C6 01 ** LDA B #FSTSEC 0740 CE 012C R DTDCP1 LDX #0UTFCB 0743 A7 0A STA A FCBTRK, X 0745 E7 0B STA A FCBTRK, X 0746 A7 0A STA A FCBTRK, X 0752 A7 0A STA A FCBTR, X 0752 A7 0A STA A FCBTRR 0753 BF CB 0TDCP2 0754 CF 0758 B DTDCP2 LDX #0UTBUF 0758 CF 0758 R DTDCP2 LDX #0UTBUF		1/0	100 47	
+ 0733 30 FCB 48 0734 AE 20 LDX DESCRA 0736 AE 00 LDX DESCRA 0736 26 59 BNE DTDCP4 073C 86 00 LDA A #FSTFKK 073C 66 01 ** LDA A #FSTFKK 073C 66 01 ** LDA B #FSTFCC 0740 CE 012C R DTDCP1 LDX #0UTFCB 0743 A7 0A STA A FCBTRK, X 0745 E7 0B STA A FCBTRK, X 0745 E7 0B STA A FCBTRK, X 0745 E7 0B STA A FCBTRK, X 0746 57 0A STA A FCBTRK, X 0746 57 0A STA A FCBTR, X 0750 6D 05 FCB 19 0750 CE 0758 R SWI + 0754 3F SWI + 0755 3F SWI + 0759 3F SWI + 0759 3F SWI + 0758 CE 0758 R DTDCP2 LDX #0UTBUF PSHX		077D CE	ΥΩΊ	
0736 A6 00	1008	8 0780 A6 0A 9 0787 E4 0B	LDA A FCBTRK, X	RECOVER 1/S
0736 81 59	1010	0784 50		NEXT SECTOR
0736 26 59 ** BNE DTDCP4 0736 26 00		0785 C1		END OF TRACK?
073C 86 00 LDA A #FSTTKK 073E C6 01 LDA B #FSTSEC 0740 CE 012C R DTDCP1 LDX #OUTFCB 0743 A7 0A STA A FCBTRK, X 0745 E7 0B LDX #INFCB 0744 A7 0A STA A FCBTRK, X 074C E7 0B LDX #INFCB 0744 A7 0A STA A FCBTRK, X 074C E7 0B LDX #INFCB 074A A7 0A STA A FCBTRK, X 074C E7 0B LDX #INFCB 074F 13 FCB 19 0750 6D 05 FCB 19 0750 6D 05 FCB 19 0752 27 07 BEQ DTDCP2 1 FCB 30 0754 GF 0758 R LDX #BRERR PRITMSG 1 0759 3F SWI 1 0759 3F SWI 2 075B C 075B R DTDCP2 LDX #BUBUF PSHX	KIP OVER	0787 26 B7	BNE DTDCP1	IF NOT, LOOP
073E C6 01	(A, B) TO FIRST T/S	4 0789 C6 01	LDA B #FSTSEC	IF SO, FIRST SECTOR
A	1015	078B 4C		NEXT TRACK
0745 E7 0B STA A FCBTRK, X 0745 E7 0B STA B FCBSCT, X 0747 CE 0000 R LDX #INFCB 0744 A7 0A STA A FCBTRK, X 0746 E7 0B STA B FCBSTK, X 0746 E7 0B STA B FCBSTK, X 0746 E3 FSWI 0747 E1 3 FCBSTA, X 0746 E3 FSWI 0752 E7 07 BEQ DTDCP2 0755 1E FCB 30 0755 CE 07CB R LDX #DRERR 0759 3F SWI 0758 CE 07CB R DTDCP2 LDX #DUTBUF 0758 CE 07CB R DTDCP2 LDX #DUTBUF 0758 CE 07CB R DTDCP2 LDX #DUTBUF 0758 CE 07CB R PRINSG 0759 SF SWI 0758 CE 0158 R DTDCP2 LDX #DUTBUF 0758 CE 0158 R DTDCP2 LDX #DUTBUF		078C		END OF DISK?
0745 E7 08 STA B FCBSCT, X (0745 E7 08 STA B FCBSCT, X (0744 A7 0A STA B FCBSCT, X (0746 E7 08 STA B FCBSCT, X (0746 E3 F SWI	10 "10" FCB T/S	/ U/8E Z6 BU *	BNE DIDOPI	AF NOI, LOOP
0747 CE 0000 R LDX #INFCB 0746 A7 0A STA A FCBTRK, X 074C E7 0B STA A FCBTRK, X 074F 33 IOHDR 0752 27 07 BEQ DTDCP2 0754 3F SWI 0755 CE 07CB R LDX #DRERR 0755 GE 07CB R DTDCP2 0756 CE 07CB R SWI 0757 3F SWI 0758 7F CB 49 0758 CE 0158 R DTDCP2 LDX #OUTBUF PSHX		0790 CE 07BB R	LDX #DTDL2	ISSUE "DONE"
0746 A7 0A STA A FCBTRK, X O74C E7 0B STA B FCBSCT, X I OHDBR SWI O74F 13 FCB 19 FCB 19 O752 27 07 BEQ DTDCP2 SWI O755 GE 07CB R LDX #DRERR O755 GE 07CB R LDX #DRERR PRTMSG O759 3F SWI O758 GE 0158 R DTDCP2 LDX #OUTBUF PSHX	TO "FROM" FCB		PRTMSG	
074E 3F SWI B FUBBLIX O74E 3F SWI		+ 0793	SWI	
074E 3F SWI 074F 13 FCB 19 0750 &D 05 FCB 19 0752 27 07 * PEQ DDCP2 * PRTERR 0754 3F SWI 0755 GE 07CB R LDX *BRERR 0759 3F SWI 0759 3F SWI 0758 GE 0158 R DDCP2 LDX *QUTBUF PSHX		Z + 0/94 31 3 0795 7E 06FF R DIDCP4	FUB 49	GET NEW CLI LINE
074F 13 FCB 19 0750 &D 05 TST FCBSTA, X 0752 27 07 * BEQ DTDCP2 * PRTERR 0754 3F SWI 0755 GE 07CB R LDX *BRERR 0759 3F SWI 0758 3F SWI 0758 CE 0158 R DTDCP2 LDX *QUTBUF PSHX				
0750 6D 05 TST FCBSTA, X 0752 27 07 * BEG DTDCP2 * PRTERR 0754 3F SWI 0755 CE 07CB R LDX *BDRERR 0759 3F SWI 0758 3F SWI 0758 CE 0158 R DTDCP2 LDX *BUTBUF PSHX		5 0798 0001 PIPFLG	RMB 1	PARSING FLAG
0754 3F SWI O755 1E PRTERR SWI O755 1E FCB 30 O755 3F SWI O756 CE 07CB R LDX #DRERR PRTMSG SWI O758 3F FCB 49 FCB 49 O758 CE 0158 R DTDCP2 LDX #0UTBUF PSHX	STATUS	4 * * * * * * * * * * * * * * * * * * *	CO # 0.00	
0754 3F SWI 0755 1E FCB 30 0755 CE 07CB R LDX #BRERR 0759 3F SWI 0758 31 # FCB 49 0758 CE 0158 R DTDCP2 LDX #0UTBUF PSHX	1028	079A 20	FCC COPY FROM DRIVE	· u
0754 3F SWI 0755 1E FCB 30 0756 CE 07CB R LDX #DRERR 0759 3F SWI 0758 CE 0158 R DTDCP2 LDX #0UTBUF		07AB	RMB 1	
0755 1E FCB 30 0756 CE 07CB R LDX #DRERR 0759 3F SWI 0758 31 * FCB 49 0758 CE 0158 R DTDCP2 LDX #0UTBUF	1030	07AC 20	FCC TO DRIVE /	
0759 3F PRINSG 0759 3F SWI CO75B CE 0158 R DTDCP2 LDX #0UTBUF		1 0786 0001 100KV	KAB 1	
0759 3F SWI 075A 31 FCB 49 075B CE 0158 R DTDCP2 LDX #0UTBUF PSHX		07BA	¥	
075A 31	1034			
075B CE 0158 R DTDCP2 LDX #OUTBUF	1035	07BB 20 07C9 0A0D	FDB \$0A0D	
		00		
	1038	8 0/CB 20 DREAK 9 0706 00	FCC / KEAD EKKOK/ FCB #OD	
	1040	0707		

FILE NOT FOUND'	FILE ERROR	CONTINUE PARSE OF CLI	JRY EKROR^	POINT TO DIR. NAME FIELD	FIRST	YES, SKIP FILE						-COMPARE DIR. NAME TO CLI NAME (WC)			CLEAN STACK		FOUND FILE?		SET NEW DINECTORY ENTRY				MARK FILE FOUND PRINT / COPY-/			>	MAKE DRIVE NO. ASCII		PRINT 'DRIVE: '		POINT TO FILE NAME IN DIRECTORY	PUT IN TERMINATOR PRINT 'FILE. EXT'	
FCC / FCB \$C	CP2 LDX #FEKROR PRTMSG SWI FCB 49	OMP PIP9	OR FCC / DIRECTORY EKROR- FCB \$0D	PR LDX FCBIND, X	CMP A #\$20	Œ	PSHX	SAT FCB 5	LDX #TMPBUF	IMS	FCB 5	CMWC #12	IMS	INS	SNI	INS	BEQ FILCP5	P4 LDX #CPYFCB	SWI	FCB 26	BRH FILLFI	ř	LDX #CPRMPT	PRTMSG		LDX #CPYFCB		LDX #DRIVE	PRTMSG	FCB 49	LDX #CFTFLE LDX FCBIND, X LDA A #\$04	STA A 12, X PRTMSG	SWI FCB 49
08D3 20 FNFND 08E2 0D *	08E3 CE 08EB R FILCP2 + 08E6 3F + 08E7 31	7E 06D0 R	OSEB 20 FERROR OSFB OD	OBFC EE 27 FILCP3	οä	27	2000	. 10	0906 CE OBBE R	6060	+ 090A 05	UNION CO OC	+ 090D 3F	_	0910 31 0911 31	8 1	0913 27 07	0915 CE 07E4 R FILCP4	+ 0918 3F	+ 0919 1A	*	091C CE 07E4 R FILCPS				0926 CE 07E4 R 0929 AA 09	8B 30	S E		0934	EE 27 86 04	A7	+ 093E 3F + 093F 31
1103 1104 1105	1106 1108 1108 1109	1110	1112	1115	1117	1118	1120	1122	1123	1125	1126	1128	1129	1131	1132	1134	1135	1137	1139	1140	1142	1143	1145	1146	1148	1149	1151	1153	1154 1155	1156	1158 1158 1159	1160	1162 1163
FCB \$0D FILE-COPY (PACKING) WITH WILD-CARD CAPABILITY	PIP TODRV:=FRMDRV:FILE.EXT "FILE" AND "EXT" MAY USE WILD-CARDS)					STORAGE FOR FILENAME (WC)								TO TEMP.					:	COPY FCB DRIVE		OUND' MARK				<i>ځ</i> ې				PARSE OF CLI	O ERROR		ISE OF CLI
3 6	ORV:=F "EXT"	E COPY				STORAGE F					BNAM			MOVE WC NAME TO TEMP.			CLEAN STACK		!	INIT	MAKE INPUT	CLEAR 'FILE-FOUND' MARK		A CUECK CTATES	G00D?	END-OF-DIRECTORY?	NO, ERROR	FOUND A FILE?	NO, ERROR	YES, CONTINUE PARSE OF	FILE-NOT-FOUND ERROR		CONTINUE PARSE OF CLI
FCB \$OD FILE-COPY (PACKING) W	SYNTAX: PIP TODRV:=F (WHERE "FILE" AND "EXT"	EXTRA FCB FOR FILE COPY	OYFCB RMB 2 FCC 'DSK' RMB 2		Α Έ	RMB 12		ILCPY LDX #TMPBUF	I ASS	FCB 5	LUX #INFOB+FOBNAM PSHX		FCB 5	i i	SWI FCB 17	SNI		SVI SCHAIT XII —	A FCBDRV, X		FCBDTT, X	CLR FCBSCF, X CLEAR 'FILE-F	:	FCB 23	BEG FILCP3 GOOD?	CMF A #1 END-OF-DIRECTOR	BNE FILCP2 NO, ERROR	FCBSCF, X) E	YES,	LDX #FNFND FILE-NOT-FOUND PRIMSG		JMP PIP9 CONTINUE PAR
O7E3 OD * FCB \$OD * * FILE-COPY (PACKING) W	* SYNTAX: PIP TODRV:=F * (WHERE "FILE" AND "EXT" *	* EXTRA FCB FOR FILE COPY *	PYFCB RMB FCC RMB	080E R FUB	0080 CPYBUF RMB	MPBUF RMB 12	* *	089A CE 088E R FILCPY LDX #TMPBUF	YUGL YUGHO +	OBSE OS FCE	COMPLIE COLO R LUX #INFCB+FCBNAM PSHX	08A2 3F SWI	+ 0843 05 FCB 5 0844 C4 06 IDA B #12	MOVC	7 11 FCB	08A8 31 INS	31 INS		A6 09 LDA A FCBDRV, X	OZE4 R LDX #CPYFCB INIT. 09 STA A FCBDRV, X	6F 06 CLR FCBDTT, X		OSBA 3F	FCB 23	BEG FILCP3 GOOD?	#1	FILCP2	TST FCBSCF, X	Z/ 03 BEW *+5 NO.	YES,	LDX #FNFND PRTMSG	SWI FCB 49	

1 , X +1, X GET NEXT FILE . X GET TNPHT FILE TYPE	A MAKE OUTPUT FILE OPEN OUTPUT FILE CHECK STATUS	PRINT ERROR MESSAGE OUTPUT ERROR	FORCE FILE CLUSED 1, X +1, X GET NEXT FILE BFAD CHAR FROM INPUT	EOF? YES ERROR? NO	BAD INPUT	WRITE CHAR. TO OUTPUT CHECK STATUS GOOD, LOOP FOR ANOTHER CHAR. PRINT ERROR MESSAGE
LDX #CPYFCB LDA A SAVEX LDA B SAVEX+1 STA A FCBIND, X STA B FCBIND+1, X OMP FILCP4 GE	STANGE ST	PRTERR SWI SWI FCB 30 LDX #ERR2 PRTMS0 SWI FCB 49 LDX #OUTFCB	CLOSE SWI FCB 21 FCB 21 LDX #CPYFCB LDA A SAVEX LDA A SAVEX STA A FCBIND, X STA B FCBIND, X STA B FCBIND+ UMP FILCP4 DX #INFCB	FCB 24 LDA B FCBSTA, X CMP B #8 BEQ FILCP9 TST B BEQ FILCP8		FILCPS LDX #UUIFUE WAITE SWI FCB 25 TST FCBSTA, X BEG FILCP7 *
07E4 R 0455 R 0456 R 27 28 0915 R	CE 012C R A7 1D : 3F : 14 6 05 27 1C	* 0948 3F 0949 1E 0940 CE 0398 R 0940 3F 094E 31	09B2 3F 09B3 15 09B3 06 07E4 R 09B7 B6 0455 R 09B A7 27 09B E7 28 09C1 7E 0915 R **	09C7 3F 09C8 18 09C9 E6 05 09CB C1 08 09CD 27 18 *	09D2 3F 09D3 1E 09D4 CE 0390 R 09D7 3F 09D8 31	09D9 CE 01ZC R 09DC 3F 09DD 19 09DE 6D 05 09E0 27 E2
1225 1226 1227 1228 1228 1230	1232 1233 1234 1235 1236 +	1240 1241 1243 + 1244 + 1245 1245 + 1247 + 1248 +	1249 1250 1251 1253 1254 1255 1256 1258	1261 + 1262 + 1264 1265 1266 1266 1266 1266 1266 1266 1266 1269 12	1270 1271 1272 + 1273 + 1275 + 1276 +	1278 1278 1280 + 1281 + 1282 1283 1284 1284 1285
PRINT < ? < GET USER RESPONSE	"YES"? IF NOT, GET NEW ENTRY SBNAM	SAVE DIRECTORY POINTER MOVE NAME TO INPUT FCB	CLEAN STACK	MOVE NAME TO OUTPUT FCB	OPEN INPUT FILE CHECK STATUS GOOD	INPUT ERROR FORCE FILE CLOSED
LDX #QMRK PRIMSG PRINT '?' SMI FCB 49 GTCMU GET USER RESPONSE SWI FCB 48	"YES"? IF NOT, +FCBNAM	SFYFCB FCBIND, X SAVE DIRECTORY POIN 5 5 8 #12 MOVE NAME TO INPUT	17. #OUTFCB+FC 5 SAVEX	T MOVE NAME TO OUTPUT I MOVE NAME TO CLEAN STACK	88	LDX #ERR1 INPUT ERROR PRTMSG SWI FCB 49 LDX #INFCB CLOSED SWI FCB 21
0940 CE 060F R LDX #GMRK PRTMSG + 0943 3F SMI + 0944 31 FCB 49 + 0945 3F SWI + 0945 3P SWI + 0945 3P SWI + 0945 3P SWI + 0946 3P PECPA	PESCAP 4 0,X 4 * 7 "YES"? FILCP4 IF NOT, *INFCB+FCBNAM	+ 0953 05 FCB 5 0954 CE 07E4 R LDX #CPYFCB 0957 EE 27 LDX FCBIND, X 0959 F+ 0455 R STX SAVEX SAVE DIRECTORY POIN + 095C 3F SWI + 095D 05 FCB 5 095E C6 0C LDA B #12 MOVE NAME TO INPUT	+ 0960 3F SWI + 0961 11 FCB 17. 0962 31 INS 0964 31 INS 0965 31 INS 0966 CE 013C R LDX #DUTFCB+FC PSWI + 0969 3F SWI + 0964 05 FCB 5 0968 FE 0455 R LDX SAVEX	+ 096E 3F SHA + 096F 05 FCB 5 0970 C6 0C LDA B #12 + 0972 3F SWI + 0973 11 FCB 17 0975 31 INS 0975 31 INS 0976 31 INS 0976 31 INS	R LDX #INFCB OPEN SWI FCB ZO TST FCBSTA, X BEG FLCP6	0981 CE 0390 R LDX #ERR1 PRTMSG + 0984 3F SMI + 0985 31 FCB 49 0986 CE 0000 R LDX #INFCB CLOSE + 0989 3F SMI + 098A 15 FCB 21

8D 07 BSR 01 8D E7 BSR PU 32 PUL A 8D 06 BSR 01 20 E2 * BRA PU 44 OUTHL LSR A 44 LSR A 44 LSR A	0846 84 0F 0UTHK AND A #\$0F 0848 88 30 A #\$30 OB4C 23 02 A #\$30 OB4C 23 02 A #\$0F 0B5C 39 A A #\$0F 0B5C 39 A A A A A A A A A A	8D C4 HEXFRM BSR 5D 27 03 BEQ 7E 0CA4 R HEX1 JMP 81 16 BNE 26 3E RPOCESS T 85 53 LDA 89 C3 BSR	0864 5D TST B 0862 26 F2	26 DC
			1381 1382 1383 1384 1386 1387 1389 1391 1391 1393 1393 1393	
BAD OUTFUT FINISH UP CLOSE INPUT	CLOSE OUTPUT X X 11, X GET NEXT FILE	Œ	WHB 1 CHECKSUM WHB 2 ADDRESS FIELD WHB 256 TEMP. BUFFER CHARACTER FROM BINARY FORMAT DX INHND DX INHND DX 7, X SR 0, X CALL INPUT HANDLER DA B FCBSTA, X RETURN STATUS TTS CHARACTER IN BINARY FORMAT	THND X X X CALL OUTPUT HANDLER UTFCB FCBSTA, X RETURN STATUS IN HEXADECIMAL FORMAT SAVE 'A' CHKSUM ADD CHAR. TO CHECKSUM CHKSUM
SWI FCB 30 LDX #ERR2 JMP PIP5B * FILCP9 LDX #INFCB CLOSE SWI FCB 21 FCB 21 LDX #OU1FCB	CLOSE SWI FOB 21 FOB 21 LDX #CPYFCB LDA A SAVEX LDA B SAVEX 1 STA A FCBIND, X STA B FCBIND+1, X UMP FILCP4 GE	FCB FCB FCC FCC FCB FCB FCB FCB	UM RMB 1 ES RMB 2 RMB 226 RMB 256 T A CHARACTER IN LDX INNND LDX 7, X LDX #INFCB LDA B FCBSI RTS T A CHARACTER	FUTBIN LDX OUTHND LDX 9, X JSR 0, X LDX #OUTFCB LDA B FCBSTA, X RTS * FUT A BYTE IN HEXAD * PUTHEX PSH A TAB ADD B CHKSUM STA B CHKSUM STA B CHKSUM
+ 09E2 3F 09E3 1E 09E4 CE 039B R 09E7 7E 05UB R 09EA CE 0000 R + 09ED 3F + 09EE 15 09EF CE 012C R	+ 09F2 3F + 09F3 15 09F4 CE 07E4 R 09F7 B6 0455 R 09F1 FC 0456 R 09F1 A7 Z7 09FF E7 28 0A01 7E 0915 R	** 0A04 20 0A08 04 ** 0A06 0A01 3A 0A06 0A 0A07 20 0A13 0A01 **	1320 0014 0001 CHKS 1321 0015 0002 ADDR 1322 0017 0100 ** 1324 ** 6E 1325 0817 FE 002A R GFTB 1327 081A EE 07 1328 081C AD 00 1339 0821 E6 05 1331 0823 39 ** 1333 1333 ** FU	0824 FE 0156 R 0827 EE 09 0828 AD 00 0828 CE 012C R 0830 39 0830 39 0831 36 0832 16 0833 FB 0A14 R 0834 F7 0A14 R

SCII										
CONVERT LEFT NIBBLE 'TO ASCI OUTFUT CHARACTER RECOVER BYTE CONVERT RIGHT NIBBLE TO ASC OUTFUT CHARACTER	CONVERT LEFT NIBBLE CONVERT RIGHT NIBBLE 0-97 YES	A-F GET BVTE FROM FILE CHECK STATUS	STATUS NONZERO XFER-ADDRESS MARK? NO	OUTPUT 'SO'		INIT. CHECKSUM ON RECORD BYTE-COUNT=3	GET ADDRESS-HIGH	OUTPUT ADDRESS-HIGH	GET ADDRESS-LOW	OUTPUT ADDRESS-LOW
OUTHL PUTBIN A OUTHR PUTBIN	6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	A #\$07 GETBIN B *+5	STATCK A #\$16 HEX2	TRANSFER-ADDRESS A #/S OUTPU PUTBIN B HEX1	A # 0 PUTBIN B HEX1	CHKSUM A #3 PUTHEX B HEX1	GETBIN B HEX1	PUTHEX B HEX1	GETBIN B HEX1	PUTHEX B
BSR BSR PUL BSR BRA	LSR LSR LSR LSR AND AND CMP CMP BLS	ADD RTS RTS BSR TST BEG	CM P	$\sigma \simeq \omega$	LDA BSR TST BNE	CLR LDA BSR TST BNE	BSR TST BNE	BSR TST BNE	BSR TST BNE	BSR 1
,	OUTHE.	: HEXFRM	* * HEX1	* PROCESS * PROCESS * BSI BSI	* *					
			œ			œ				•
07 E7 06 E2	33 OF	07 C4	0CA4 16 3E	53 53 F2	30 EB	0A14 03 · BF E1	OF DO	85 07	9.6 D2	A:B
80 33 80 80 80 80	4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	88 39 50 27	7E 81 26	88 80 80 80 80	86 80 50 26	F 28 8 5 7	8D 5D 26	8D 5D 26	8D 20 26	35
0B39 0B3B 0B3D 0B3E 0B3E	0843 0843 0844 0845 0845 0846 0846	084E 0850 0851 0853 0853	0856 0859 0858	085D 085F 0861 0862	0864 0866 0868 0869	0868 0866 0870 0872 0873	0875 0877 0878	0B7A 0B7C 0B7D	087F 0881 0882	0B84 0B86
00 0 C + N C	**************************************	N W Z IO & N W A A A A	+ N M + 10 .	0 N M A C - A 0	n + 10 -0 ≥ 0	`	+10.05.0			~ ~ ~

OUTPUT PARTIAL FRAME HERE (<30 BYTES)	INIT. CHECKSUM OUTPUT 'S1'	OUTPUT FRAME COUNT ADD OVERHEAD (ADDRESS+CHECKSUM) OUTPUT ADDRESS	NONZERO STATUS +1 GET DATA BVTE FKOM TEMP.	COUNT DOWN FINISH UP (CHECKSUM+C.R.)	
UT PARTIAL FR	CLR CHKSUM LDA A #'S USR PUTBIN 151 B 151 B 151 B 164 A #'1 UDA A #'1 UDA A #'1 USR PUTBIN 151 B 151 B	LDA A FCNT ADD A #3 USR PUTHEX TST B BNE HEX21 LDA A ADDRES USR PUTHEX TST B	UMP USR TST TST BNE BNE LDA	INX STY SAVEX JST SAVEX JST B TST B BNE HEX22 BNE HEX22 * UMP HEX1A	CLR CHKSUM CLR CHKSUM LDA A # 'S TST B BNE HEX22 LDA A # '1 USR PUTBIN TST B BNE HEX22 LDA A #30
TUC *	08EA 7F 0A14 R 08EU 86 53 08EF 8D 0824 R 08F2 5D 08F3 26 83 ** 08F7 86 31 08F7 8D 0824 R 08F 5D 0824 R	08FD B6 0413 R 0C00 8B 03 0C02 BD 0831 R 0C05 5D 0C06 26 A0 0C08 B6 0A15 R 0C08 B1 0831 R	27 7E	0C28 08 0C29 FF 0455 R 0C2C BD 0B31 R 0C3F 5D 0C30 26 DF 0C37 7E 0B89 R 0C37 7E 0B89 R	0C3A 7F 0A14 R HEX2D 0C3A 7F 0A14 R HEX2D 0C3 B6 53 0C42 50 0C43 26 CC
1470	1472 1472 1474 1474 1476 1478 1478 1480	1483 1483 1486 1486 1487 1489 1489 1489	1492 1493 1495 1496 1497 1500 1500 1501 1501 1503	1506 1506 1506 1507 1510 1511 1512 1513 1515 1516	151/ 1518 1518 1520 1521 1524 1524 1526 1526 1530
	OUTPUT INVERTED CHECKSUM OUTPUT CARRIAGE-RETURN		HERE. GET ADDRESS-HIGH NONZEKO STATUS GET ADDRESS-LOW	+1 GET FRAME COUNT INIT. BUFFER POINTER GET DATA BYTE:	STORE BYTE IN TEMP. COUNT DOWN INIT. BUFFER POINTER DATA FOR FULL FRAME (30 BYTES)? YES
BNE HEX1	LDA A CHKSUM COM A BSR PUTHEX TST B BNE HEX1 LDA A #\$OD BSR PUTBIN BSR PUTBIN BSR PUTBIN	BRA HEXFRM CMP A #\$02 BEQ *+5 UMP EOFST	HANDLE DATA-FRAME JSR GETBIN TST B BEG *+5 XZ1 JMP STATCK STA A ADDRES JSR GETBIN TST B BNE HEXZ1	STA A ADDRES+1 JSR GETBIN TST B BNE HEX21 STA A FCNT STA A SAVEA LDX #TBUF STX SAVEX JSR GETBIN TST B BNE HEX21	LDX SAVEX STA A O.X INX STX SAVEX BNE HEXZA LDX #TBUF STX SAVEX STX SAVEX CMP A #30 BHI HEXZD
OB87 26 CD	0B99 B6 0A14 R HEX1A OB8C 43 OB8D 8D A2 OB9D 26 C4 ** OB92 86 OD ** OB94 8D 8E OB94 8D 8E OB97 26 BD	20 B6 ** 81 02 HE 27 03 ** 7E OCBC R **	* HAND OBAZ BD OB17 R * ** OBA5 5D OBA6 27 03 * OBA8 7E OCA4 R HEX21 OBAB B7 OA15 R OB4E BD OB17 R OB81 5D OB82 26 F4 *	0884 B7 0A16 R 0887 BD 0817 R 0888 26 E8 ** 0880 B7 0A13 R 08C0 B7 0457 R 08C3 CF 0A17 R 08C4 F7 0A55 R 08C5 E7 0A55 R 08C5 E7 0A55 R 08C5 E7 0A55 R 08C5 E7 0A55 R 08C7 SD	A OBCF FE 0455 R OBDZ A7 00 OBD4 08 OBD5 FF 0455 R OBD8 Z6 EC OBD CE 0A17 R OBC 0A17 R OBE S F 0455 R

7	\circ T	

PRINT ERROR MESSAGE OUTPUT BAD		CLOSE INPUT FILE GET NEXT CLI TOKEN CHECK RC	OUTPUT EOF RECORD (7897)	OUTPUT CARRIAGE RETURN	"DONE" FINISH UP (MIKBUG) TO BINARY FORMAT	GET A BYTE ERROR? YES REMOVE ASCII BIAS VALID HEX?	0-9? YES A-F? ND	A-F? NO REMOVE ASCII BIAS
	CMP FIPSB T LDX #INFCB TXAB SWI FCB 2	LDX INHND LDX 5, X USR 0, X NXTOK SWI FCB 47 LDA B RC			LDX #PRMPT2 JMP PIP5B ** COF2 JMP PIP9B * * ** REFORMAT FROM HEX *	C JSR GETBIN TST B BNE INH2 SUB A ##30 BMI HEXBAD	CMP A #\$9 BLE INH2 CMF A #\$11 BMI HEXBAD	CMP A #\$16 BGT HEXBAD SUB A #\$7
* 00TERR 0CB4 3F 0CB5 1E 0CB6 CE 0338 R		00C1 FE 002A R 00C4 EE 05 00C6 AD 00 00C8 3F 00C9 2F 00CC C1 0D	8 25 E S 25	25 25 25 25 25 25 25 25 25 25 25 25 25 2	OCE8 CE 0713 R OCEB 7E 05D8 R OCEE 7E 06E7 R E0F2 * * * REF	0CF1 BD 0B17 R INHEX 0CF5 26 12 ** 0CF7 80 30 ** 0CF9 2B 0F	* OCFB 81 09 OCFD 2F 0A * OCFP 81 11 OD01 2B 07	* ODOS 2E 03 * ODO7 80 07 *
1592 1593 1594 + 1595 +	1598 1598 1600 1601 1602 +	+ +				1637 1638 1639 1640 1641		1649 1650 1651 1652 1653
АБВ ОVЕКНЕАВ	REMOVE 30 BYTES	**1	GET DATA BYTE. OUTPUT IT	MOVE ADDRESS COUNT DOWN OUTPUT INVERTED CHECKSUM	OUTPUT CARRIAGE RETURN	CONTINUE UNTIL BUFFER EMPTY R HANDLED HERE		FKIN+ EKKUK MESSAGE. INFUT BAD
STA A SAVEA ADD A #3 JSR PUTHEX TST B BNE HEX22		TST B BNE HEX22 LDA A ADDRES+1 USR PUTHEX TST B RNE HEX22	CE LDX SAVEX LDA A 0, X IDA STX SAVEX JSR PUTHEX TST B BNE STATCK		TST B BNE STATCK LDA A #\$0D JSR PUTBIN TST B BNE STATCK	FILE Strine	STATICK CMP B #8 * TST FCBDTT, X BNE OUTERR	FKIEKK SKIEKK FCB 30 LDX #ERR1 JMP PIP5B
0C4F B7 0457 R 0C52 8B 03 0C54 BD 0B31 R 0C57 5D 0C58 26 B7	B6 0A13 R 80 1E B7 0A13 R R6 0A15 R RD 0B31 R	0C68 5D 0C69 26 A6 ** 0C6B B6 0A16 R 0C6E BD 0B31 R 0C71 5D **	0C74 FE 0455 R HEXZE 0C77 A6 00 0C79 08 0C7 FF 0455 R 0C70 BD 0831 R 0C80 5D 0C81 26 21	FE 0415 R 08 FF 0415 R 26 E5 26 E5 B6 0414 R 43 RD 0831 R	0C96 5D 0C97 26 0R * 0C99 86 0D 0C9R BD 0B24 R 0C9E 5D 0C9F 26 03 *	7E OBEG R	0CA6 27 14 ** 0CA8 6D 06 ** 0CAA 26 08 **	OCAD 3F OCAD 1E OCAE DE 0390 R OCB1 7E 05U8 R
1531 1532 1533 1534 1535	1537 1538 1539 1540	1542 1543 1544 1545 1546 1548 1548	1550 1551 1552 1553 1554 1555 1555 1555	1559 1559 1560 1561 1562 1563 1563 1564	1567 1568 1569 1570 1571 1572 1573	1575 1576 1577 1578 1579 1580	1582 1583 1584 1584 1586 1586	1588 + 1589 + 1590 1591

1 Q つ		-	
	•		

	725 726 727 728 0 728 0 739 0 733 0 733 0 733 0	735 736 737 738 739 740 741	745 745 745 747 748 748 750 750 751 753 753 753	756 0 757 758 758 759 0 760 761 762	1765 1766 1767 1769 1770 1771 1772 1773 1773 1773
	:				, , , , , , , , , , , , , , , , , , ,
			×	ć. 2	
8	3BLE		<u> </u>	5 (SC	T
INVALID CHARACTER ERROR STATUS CHARACTER	GET A HEX DIGIT ERROR? MOVE TO LEFT NIBBLE	DIGIT	ADD INTO CHECKSUM GET A BYTE ERROR? YES	HEADER MARK? NO, KEEP LOOKING GET A BYTE TRANSFER ADDRESS ('SO')? NO	RESS HERE INIT. CHECKSUM GET A HEX BYTE DT USED GET ADDRESS-HIGH
INVALID CHAR ERROR STATUS CHARACTER	GET A HEX DIGIT ERROR? MOVE TO LEFT NI	T HEX DIG]	ADD INTO CI GET A BYTE ERROR? YES	HEADER MARK? NO, KEEP LOO GET A BYTE TRANSFER ADD NO	s HERE A HEX SED SED
INVAI	GET A ERROR? MOVE T	SAVE IT GET A H ERROR? MERGE D	ADD IN GET A ERROR?	HEADI NO, 1 GET (INIT. GET A GET A GET A
#NOTHEX INVALID CI SG . 49 B #\$FF ERROR STA' ' BAD HEX CHARACTER	X 2	TBUF HEX :T2 TBUF	CHKSUM CHKSUM TBIN 5	S E FRM 1 C C C C C C C C C C C C C C C C C C	HANDLE TRANSFER ADDRESS HERE CLR CHKSUM INIT. CHE BSR GETHEX GET A HEX TST B BNE BIN1 NOTE: FRAME COUNT NOT USED BSR GETHEX GET ADDRE
₹ ⊷ m		(4 4 ± m m m m	ж # 20 ж # X	E BINFRM R GETBIN E BIN1 E BIN1	E TRANSFER CLR CHKSUM BSR GETHEX TST B BNE BIN1 FRAME COUN BSR GETHEX TST B
		STA STA BSR TST TST BNE LDA ABA	ADD STA CLR RTS RTS TST BEQ	CMP BNE BNE BNE BNE BNE BNE BNE	CLR BSR TST BNE E: FRP BSR
INH2 * HEXBAD * NOTHEX	** 6ETHEX *		GET2 ** BINFRM BIN1 **		
œ	: * * © *	œ œ	«« « «	* *	œ
0D12 FF	18	0A17 BE 0C 0C	0A14 0A14 0B17 03 0E33	53 F3 0817 F3 30 5F	0A14 C6 E7 C1
39 CE OD 3.37 1:31 1:31 1:56 39	00 00 00 00 00 00 00 00 00 00 00 00 00		77.77 27.77 76. 39. 39. 77. 76. 77. 76. 77. 76. 77. 76. 77. 76. 77. 76. 77. 76. 77. 76. 77. 76. 76	81 26 35 26 26 26 26 26	788 % 8B
	0025 0027 0027 0028 0028	002E 002E 0031 0033 0034 0036	0038 0038 0038 00041 00042 00048 00046	004C 0050 0053 0054 0054 0056	0054 0050 0051 0060 0062 0064
655 655 655 655 655 655 655 653 653	6664 6665 6668 6668 6668 667 667 673	674 676 676 677 678 678 680	1682 1683 1683 1685 1688 1689 1691 1693	1695 1696 1698 1698 1699 1700 1701 1702	1704 1705 1706 1708 1708 1710 1711 1711 1712

	GET ADDRESS-LOW	1 TEST CHECKSUM GET CHECKSUM	GOOD? YES NO, ERROR M ERROR		BAD STATUS OUTPUT ADDRESS-HIGH	1 OUTPUT ADDRESS-LOW	DATA HEADER (NO HERE	INIT. CHECKSUM GET FRAME COUNT	REMOVE OVERHEAD BYTES GET ÁDDRESS-HIGH
BINI	A ADDRES GETHEX B BIN1	A ADDRES+1 A CHKSUM A SAVEA GETHEX B	A SAVEA BIN1B #CHKERR PIP5B	* CHECKSUM *OD A #*16 (PUTRIN B *+5	STAT2 A ADDRES PUTBIN B BINIC	A ADDRES+1 PUTBIN B BINIC BINFRM	NZ CMP A # / 1 BNE BIN3 HANDLE DATA-RECORD	CHKSUM GETHEX B BINIC	A #3 A FCNT GETHEX B BINIC A ADDRES
BNE	STA BSR TST BNE	STA LDA COM STA BSR TST BNE	CAN DE DE	FCB LDA JSR 1ST BEG	LDA USR TST BNE	LDA USR TST BNE		CLR USR TST	SUB STA JSR TST BNE
*	: ,	BIN1A	* * * * 0	CHKERR * BINIB	BINIC *	* *	₩.	*	* *
	Œ	αα α	מב מבמב	Œ	a a a	0c 0c 0c	1	Ͼ	αα α
E2	0A15 B9 DA	0A16 0A14 0A57 AA CB	0457 16 0089 0508	16 0B24 03	0E33 0A15 0B24 F4	0A16 0B24 EB	31 6F	0A14 0D25 DB	03 0A13 0D25 D0 0A15
26	80 80 50 26	87 80 80 26 26	B1 27 27 7E CE 7E 39	72BB 003	7E BB SD SD SS SS SS SS SS SS SS SS SS SS SS	86 80 50 26	26 25	7F 8D 5D 5D 26	80 80 80 26 80 87
00,65	0D67 0D6A 0D6C 0D6C 0D6D	006F 0072 0075 0076 0079 0078	0D7E 0D81 0D83 0D86	0.0089 0.0098 0.0098 0.009E 0.009F	ODA1 ODA7 ODAA ODAA	ODAD ODBO ODB3 ODB4	ODB9 ODBB	ODBD ODCO ODC3 ODC4	ODC6 ODC8 ODCE ODCF ODCF
1715	1717 1718 1719 1720	1721 1722 1723 1724 1725 1727 1727	1729 1730 1731 1732 1733 1734 1735	1736 1737 1738 1739 1740 1741	1744 1745 1746 1747 1748	1751 1752 1753 1754 1755	1757 1758 1759 1760 1761	1763 1764 1765 1765	1768 1769 1770 1771 1772 1773 1774

HANDILE ERROR STATUS				
END STATCK				
0E37 7E 0CA4 R				
1837 1838 1839				
GET ADDRESS-LOW OUTPUT DATA-HEADER MARK OUTPUT ADDRESS-HIGH 1 OUTPUT ADDRESS-LOW	OUTFUT FRAME COUNT GET A DATA BYTE (HEX FORMAT) OUTPUT DATA BYTE (BINARY FORMAT)	COUNT DOWN TEST AGAINST CHECKSUM GET CHECKSUM	OCCUP CARECUSOTTS NO GET NEW FRAME	YS9.7 NO, LOOK FOR IT YES, CLOSE FILE EOF ON INPUT? YES, CLOSE FILE
	LDA A FCNT USR PUTBIN 1ST B BNE STAT2 USR GETHEX 1ST B BNE STAT2 USR PUTBIN 1ST B BNE STAT2	DEC FCNT BNE BINZA LLDA A CHKSUM COM A STA A SAVEA JSR GETHEX TST B BNE STATZ		* HANDLE EOF ('S9') * * BIN3 CMP A #'9 * BNE BIN2B * BIN3A JMP PIP9 * STAT2 CMP B #8 * STAT2 CMP B #8
* * *	* * BINZA	* * * *		
333 7338 753884 788	00F5 26 3C 00F7 R6 0A13 R 00F0 50 00F0 25 33 0E00 B0 0D25 R 0E04 26 2D 0E04 B0 0R24 R 0E09 5D 0E04 26 27			0E2C 81 39 0E2E 26 F9 0E30 7E 06D0 R 0E33 C1 08 0E35 27 F9
1776 1777 1778 1778 1780 1781 1782 1784 1785 1786 1786 1787 1790 1791 1791	1795 1796 1797 1799 1799 1801 1802 1803 1803 1805 1805	1808 1809 1810 1811 1812 1813 1816 1816 1816	1820 1821 1822 1823 1824 1824 1825	1827 1828 1829 1830 1831 1833 1833 1835 1836

1 SECURITY NI COMMAND SECURITY PROCESSOR SECURITY IDRIVE: 1 FILENAME. EXT. VALUE E ACCESS CODE TO "VALUE" DDRESSING DEFINITIONS SDEF SU O GENERIC DEVICE TYPE SU 2 GENERIC DEVICE TYPE SU 2 STATUS	DATA TRANSFER TYPE DATA BUFFER ADDRESS DRIVE NUMBER SECTOR NUMBER SECTOR NUMBER SECTOR NUMBER FUL INK TRACK/SECTOR FILE TYPE FILE TYPE FILE ACCESS CODE FILE NACK/SECTOR NUMBER OF SECTORS	FILE TYPE FILE ACCESS CODE FIRST TRACK/SECTOR LAST TRACK/SECTOR NUMBER OF SECTORS 128 BYTES/SECTOR	DEFAULT DRIVE=0
* TRANSIENT COMMAND SECURITY* * SYNTAX: SECURITY IDRIVE: 1 FI * SET FILE ACCESS CODE TO "VAL * BLOCK ADDRESSING DEFINITIONS * FCBDEF FCBGT EQU O GENERIC FCBGT EQU 2 GENERIC	Free Contractions of the Contraction of the Contrac	FIBTYP EQU 13 FIBACS EQU 14 FIBLTS EQU 15 FIBNTS EQU 17 FIBNMS EQU 19 * * BASE-PAGE EQUATES * DESCRA EQU \$22 CUCHAR EQU \$22 CUCHAR EQU \$23 RC EQU \$25 CLASS EQU \$26 VALUE EQU \$27 * * DISK ATTRIBUTES * SECSIZ EQU 128	* FCB FOR TRANSIENT * SYSFCB RMB 2 FCC 'DSK' RMB 2 RMB 2 RMB 33 BUFFER RMB SECSI7 * CLR FCBDRV, X
0001 0000 0000 0003 0004 0005 0005 0007 0008 0009 0010 + 0000 0002 0017 + 0000 0002		+++++	0049 0050 0051 0051 0052 0053 0002 44 0054 00050 0007 00050 00050 00050 00050 00050 00050
SCLOSE 0455 R 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.03F 0.03F 0.0417 R 0.0417 R 0.		0000000000
PIP1B 0487 R PIP1C 040B R PIP2 040B R PIP2B 0407 R PIP2B 040F R PIP4B 0530 R PIP4B 0530 R PIP4B 0530 R PIP4C 0555 R PIP4C 0555 R	PIP4E 0574 PIP5 0587 PIP50 0503 PIP50 0503 PIP50 0503 PIP50 0508 PIP50 0569 PIP60 0569 PIP60 0606 PIP70 0606 PIP70 0607	0684 0684 0660 0660 0667 0667 0070 0700 0700 0713 2504 0042 2151 21167 0824	PUIDR 2406 M PUTHEX 0831 R GMMK 0A0F R KC 0025 KCBDBA 0007 KCBDF 258C M KCBDT 0000 KCBCT 0000 KCBCT 0000 KCBSTA 0005 KCBSTA 0005 KCBSTA 0005 KCBSTA 0005 KCBSTA 0005
GTCMD 24F0 M HEXIA 0856 R HEXIA 0898 R HEXZ 0898 R HEXZI 0808 R HEXZA 0809 R HEXZA 0809 R HEXZA 0809 R HEXZA 0809 R HEXZE 0023 R HEXZE 0023 R HEXZE 0023 R		0.2FD 0.31C 0.31C 0.320 0.330 0.330 0.356 0.378	NXTOK 2406 M OPEN 234F M OPEN 234F M OPEN 234F M OUTER 0158 R OUTER 0150 R OUTHL 0150 R OUTHL 0150 R OUTHL 0407 R PIP1 0457 R PIP1 0457 R
E0F2 0CEE R E0F3 0CBC R EMR1 0G90 R EMR2 039B R EMR3 03A7 R ERR4 03BD R ERR5 03D4 R EVBACS 001E FCBACS 001E FCBBAK 000E	FUBBER 0000 FUBBER 0009 FUBBER 0000 FUBBER 00000 FUBBER 00000 FUBBER 00000 FUBBER 00000 FUBBER 00000 FUBBER 00000 FUBBER 00000 FUBBER 00000 FUBBER 00000 FUBBER 00000	FERROR FIBNES FIBLTS FIBNAM FIBNAM FILCP2 FILCP3 FILCP3 FILCP3 FILCP4 FILCP6 FILCP6 FILCP6	FMTFCB 2488 FMFD 08D3 FURMAT 03F3 FURMAT 03F3 FURMD 044C FRETAB 002B FRHURV 07AB FSTSEC 0001 FSTSEC 0001 GETZ 06FTB 08L7 GETBR 23EC GETHEX 0D25
ADDABX 2219 M ADDAX 2232 M ADDKES OA15 R ADDKES OA15 R ADDXAB 2200 M BASEQU 2626 M BFLAG 0259 R BINI 0009 R BINIB 0099 R BINIC 0041 R	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0023 028F 028F 0022 0022 0025A 0035A 00411 00423 0036 0037 0037 0007 007CB 007CB	DIDCP1 0740 R DIDCP2 0758 R DIDCP2 0758 R DIDCP3 0770 R DIDCP4 0795 R DIDCP4 0795 R DIDCP4 0797 R DIDCP5 0707 R DWERR 0707 R DW 0040 EJ EMEM 0035

COUNT PERIOD GET TOKEN FROM CLI	CHECK RC UNAMBIG, NAME? IF NOT, ERROR	GET LENGTH OF EXT TOTAL LENGTH	POINTER TO FCBNAM	POINTER TO CLI NAME	FORMAT NAME INTO FCB	CLEAN STACK	ERRORS? YES	OPEN THE DIRECTORY		CHECK STATUS 600D?	END OF DIRECTORY? NO	FILE NOT FOUND ON DISK			PRINT ERROR MESSAGE	POINT TO DIRECTORY NAME	POINT TO FCB NAME COMPARE 12 CHARACTERS
	FCB 47 LDA B RC CMP B #1 BNE SEC3	LDA B DESCRC ADD B BUFFER+2 LDX #SYSFCB+FCBNAM		PSHX SWI SWI FCB 5	മലയ	I NS S	INS TST B BNE SEC3	LDX #SYSFCB OPEND	SWI FCB 23	LDA A FCBSTA,X BEG SEC6	CMP A #1 BNE SEC5A	LDX #FNFND PRIMSG	SWI FCB 49 RTS	FCC / FILE NOT FOUND/ FCB \$OD	PRTEHR SWI FCB 30 RTS		FCB 5 LDX #SYSFCB+FCBNAM PSHX SWI FCB 5 LDA B #12
0112 7C 002C R + 0115 3F	+ 0116 2F 0117 D6 25 0119 C1 01 011B 26 CF *	011D D6 22 011F FB 002C R 0122 CE 0010 R	0125 3F 0126 05 0127 FF 0024	012A 3F	•	012E 31 0130 31	0137 31 0132 5D 0133 26 B7	띥	0138 3F 0139 17	013A A6 05 SECS 013C 27 1D *	013E 81 01 0140 26 16	0142 CE 0148 R	0145 3F 0146 31 0147 39	0148 20 FNFND 0157 0D **	0158 3F 0159 1E 015A 39	015B EE 27 015D 3F	•
	0125 + 0126 0126 0127 0127 0127 0128 0128 0128	0130 0131 0132	0133 0134 + 0135 +	0137		0144	0147 0147 0148	0149	0152 +	0154 0155 0155	0158		0162 + 0163 + 0164 0164	0166 0167 0168	0168 0170 + 0170 + 0171 +		0177 + 0178
INPUT GET TOKEN FROM CLI	CHECK RC NUMBER? NO	VALID DRIVE NO. ? NO. ERROR	VALID DRIVE NO.? (4 DRIVES) NO	SET DRIVE NO.	NUMBEK ERROR			GET TOKEN FROM CLI	CHECK RC COLON?	IF NUI, EKROR	TO TOWN THE	UNAMBIG. NAME? YES	FORMAT ERROR		POINT TO NAME	GET LENGTH OF NAME GET TOKEN FROM CLI	CHECK RC PERIOD? IF NOT, EKROR
CLR FCBDT1, X NXTOK SWI	FCB 47 LDA B RC CMP B #3 BNE SEC2	TST VALUE BNE SEC1	LDA A VALUE+1 CMP A #3 BHI SEC1	STA A FCBDRV, X BRA SECIA	LDX #NUMBER PRIMSG SWI FCD AD	RTS * NIMBER FCC / NIMBER FDDOD	FCB #OD	A NXTOK SWI FCB 47	LDA B RC CMP B #/:	BNE SECT NXTOK	SWI FCB 47 FDA B RC	CMP B #1 BEG SEC4	LDX #FORMAT PRTMSG SWI		TFCC / FORMAT ERROR/ FCB #OD LDX DESCRA		SW1 FCB 47 LDA B RC CMP B #1. BNE SEC3
006F (+ 0062 2F 0083 D6 25 0085 C1 03 0087 26 2F *	00 bc 26 0A *	00BE 96 28 00C0 81 03 00C2 22 04	00C4 A7 09 00C6 20 14	OOCS CE OOCE R	00CB 39		SEC1A SEC1A OODC 3F OODD 2F	00DE D6 25 00E0 C1 36		00E4 3F 00E5 2F 00E6 D6 25		0	00F0 31 00F1 39	00F2 20 FORMAT 00FF 0D ** 0100 DE 20 SEC4 0107 FF 0076 R	0105 96 22 0107 B7 002C	0.008 3F 0.108 0.5 0.100 0.6 25 0.100 0.6 0.6 *
	0064 0065 0065 0067 0068	0069	0072 0073 0074 0075	0076	0079		0086	+ 0600 0088 + 0000	0091	009 4 300 5 4 300	+ 9600 + 0097 + 0098	00000	0102 0102 0103 0104 +	0105 + 0106 0107	0108 0109 0110	0113	0115 + 0118 + 0118 0119 0120 0121

OPEND 239E M PKTERR 2454 M PKTMSG 250A M PSHALL 2151 M PSHX 21CE M PULLAL 216A M	215, 216, 0025 EF 258C 2388 ND 2384 0008		SEC9 018B R SECSIT 0080 SECURE 0080 R SECURI 0000 RN SUBAN 227F M SUBAN 2297 M SUBAN 2289 M SUBAN 2283 M SUBAN 2283 M VALUE 0000 R VALUE 0027 WALUE 0027		
~ ~~~	-	DESCRC 0022 DIVA		2940 2940 2005 2005 2005 2488 2558 2558 2558	· ~
CLEAN STACK	FOUND ENTRY IN DIRECTORY?	GET TOKEN FROM CLI DELIMITER? YES NO, ERROR	GET TOKEN FROM CLI CHECK RC NUMBER? YES NO, ERROR SECURITY-VALUE TOO BIG? YES, ERROR	POINT TO DIRECTORY ENTRY GET NEW ACCESS CODE POINT TO AC. FIELD IN ENTRY STORE IT MAKE 'OUTPUT' WRITE DIRECTORY	RESTORE / INPUT/ ERROR? NO YES
CMPC SWI FCB 18 INS INS	INS BEG SEC7 LDX #SYSFCB GETDR SWI FCB 26 BPA SFC5	NXTOK SWI FCB 47 LDA A CLASS CMP A #4 BEG SEC8 UMP SEC3		LDX #SYSFCB LDX FCBIND, X LDA A VALUE+1 LDA B #FIBACS ADDBX SWI FCB 10 STA A O, X COM FCBDTT, X IOHDR	CLR 19 CLR FCBDT1, X TST A BEQ #+5 JMP SEC5A RTS END
+ 0166 + 0167 0168 0169 0169	0.189 0.16B 31 0.190 0.16C 27 07 * 0.191 0.16E CE 0.000 R 0.193 0.15E CE 0.000 R 0.194 + 0.171 3F 0.195 + 0.172 1A	+ +	+ 0180 3F + 0181 2F 0182 106 25 0184 C1 03 0186 27 03 0188 7E 00C8 R 0188 7D 0027 018E 26 F8	++ +	0231 + 0143 13 0232 0164 6F 06 0233 0164 6F 06 0234 0167 27 03 0235 0169 7E 0158 R 0237 016C 39

"="? EKROR	31 27A F30	OE I VHILDE			YES		NAME:		DUPLEX?	QN	YES	GET RESPONSE	HALF?			SET TO HALF DUPLEX			FULL?	EKROR			DOTE LEAN - TOLL		PAUSE?	<u>Q</u>			GET RESPONSE	∕ΟN	Ž.		PAUSE OFF		0=01/2=	1 C 2 C 2 C 2 C 2 C 2 C 2 C 2 C 2 C 2 C	1		PAUSE ON				2000	100001 80881	No.	ARM)	100 A	A VALUE+I SE! VALUE A O, X	
CMP B #/= BNE SET2	701	SWI	FCB 47	CHO B RC		6	BNF SFT9			BNE SETS	LDX DESCRA			BNE. SE 14	DA A #SFF	STA A DX			<u>G</u>	BNE SET2	4 44	CTA A TV	CMP SETNXT			BNE SET8					BNE SETO	I DA A ##FF		JMP SETNXT	>* \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \			LDA A #00		UMP SETNXT		PARM=VALUE FOUND	0000000	BCS SETS	0.10	FOUND ENTRY X=A(PARM)	, U	STA A O, X	
:	*					*		*						*	ŧ		~	*	SET4	:	*		œ	*	SETS		*				*	ķ.		: د	\$ 0F1'4	7	*			œ	*		1 1 0 *	1	*	* FOL	*		* ~
0021 C1 3D 0023 26 EC		+ 0025 3F	0026 21	0029 C4 020	27	ä	002F 26 49	ì	0031 8C 4458	76	띰	ક	8	0030, 26 07	003F 86 FF	67	7 E		0045 81 46	78	70 000	0 0	1	l	9C	0053 26 25		품 :	9 ;	0059 81 4E	C V	005D 86 FF	6	0061 7E 007F	0044 81 50			98	97 42	006C 7E 007F				0071 25 07	ì		0023 04 30	0075 A7 00	007F
0061	0063	0065 +	+ 9900	/900	6900	0020	00.72	0073	0074	0075	9200	0077	0078	× 000	0081	0082	0083	0084	0082	9800	2000	0000	0800	000	0092	6600	0094	000	9600	\ 000 000	0000	0100	0101	0102	010	0105	0106	0107	0108	0109	0110	0111	0112	0114	0115	0116	0117	0119	0120
97/ QJ QU	-UR CP/68	SET DP≔XX		nescripting Annaess(2)	DESCRIPTOR	3 CURRENT CHAR (2)	TOKEN CLASS			_			NEX! AVAIL IKANSIEN! AKEA (2)		DEPTH, LINES	DEPTH) WIDTH; CHARS/LINE			DOPLEX; FF=H, OO=F				DEPTH TEMP	WIDTH		GET TOKEN (PARM NAME)		MONOT OF THIS OF ACT	O INTOL	SCAPE?			DONE RETURN TO CLI	GET RETURN CODE				iA ERROR			T-\1	_	RA GET PARM NAME		# = # LDO			
1	* SET COMMAND FOR CP/68	SYNTAX:	*	DESCRO FOLL #20	DESCRC EQU \$22	CUCHAR EQU \$23	388	EGU		AB EQU	EGU	<u></u>	בים המו	Di FOH #34	EGU	CNT EQU	EQU	EQU	EQ	DX EGU #40	200	3 6	EQU	LDPCNT EQU \$45	I.WD EQU \$46		SETO NXTOK	SEI SEI	FUB 4/	LDA DESCRI	1			RTS *	SET1 LDA B RC	5	ш	*	R SET2 LDX #MSGA	PRTMS6	SWI	FUB 49	*	E13	LDX 0, X	*	SEI	FCB 47	LDA B RC
0000 0000				0000 0000	0000 +	+ 0000 0023	0000		+ 0000 0029				+ 0000 0037	+ 0000 0034						+ 0000 0040				+ 0000 0045	+ 0000 0046			# 0000 # -	1000	0002 DE 20	5 5	56		000A 39	000B D6 25	5	000F 27 08		0011 CE 008D		0014	+ 0015 31 0014 7F 007F			001B EE 00		+ 001D 3F	001E 2F	001F D6 25
0001	0003	0002	9000	/000 0000	6000	0010	0012	0013	0014	0015	0016	0017	0018	0000	0021	0022	0023	0024	0025	9700	7700	0029	0030	0031	0032	0033	0034	0030	9000	200	6E00	0040	0041	0042	0044	0045	0046	0047	0048			0001	0053	0054	0055	0056			0900

* FCC \WD\ FUB WD	* FCC 'NL' FDB NL	FCC /TB/ FDB TB	* FCC /EJ/ FDB EJ	* FCC 'ES'	* * * LINE PRINTER SET PARMS	* FCC 'LD' FDB LDP	FCC /LW/ FDB LWD	END												
00D8 57 00DA 003D	OODC 4E OODE OO3E	00E0 54 00E2 003F	00E4 45 00E6 0041	00E8 45	COEM COORS	OOEC 40 OOEE 0044	00F0 4C 00F2 0046													
0183 0184 0185	0186 0187 0188	0189 0190	0192 0193 0194	0195	0197 0198 0199	0200 0201 0202	0204 0204 0205	0207												
					EKROK ') SET PARM		FOR AN ENTRY	PUT PARM NAME INTO A, B	POINT TO TABLE	ND MATCH	NO MATCH		GET BP ADDRESS	POINT TO NEXT ENTRY	END OF TABLE? NO	NOT IN TABLE			
LDX #MSGB PRTMSG SWI	FCB 49 LDX #MSGC PRTMSG	SWI FCB 49 GTCMD	SWI FCB 48 LDX DESCRA	STX CUCHAR UMP SETO	FCC 'SYNTAX ERROR' FCB \$0D	FCC /INVALID : FCB \$OD	FCC /SET- / FCB \$04	* * SEARCH SETAB FOR	ТХАВ	SWI FCB 2 LDX #SETAB	CMP A 0, X BNE SETSR2	CMP B 1, X BNE SETSR2		LDX 2.X CLC RTS	XXXX	INX TST 0, X BNE SETSR1	SEC RTS	FCC /BS/ FUB BS	FCC /DL/ FDB DL	FCC /DP/ FUB DP
R SET8	E 31 FCB 49 CE 00AB R SETNXT LDX #MSGC PRTMSG			œ	* MSGA	SGB	200	* SEARC	* SEISRC TXAB	œ	ETSR1		* MATCH	:	* SETSR2			SETAB		
007A CE 009A 007D 3F	007E 31 007F CE 00AF	0082 3F 0083 31	0084 3F 0085 30 0086 DE 20	0088 DF 23 008A 7E 0000	008D 53	009A 49 00AA 0D	00AB 53 00B0 04			00B1 3F 00B2 02 00B3 CE 00CC	00B6 A1 00 00B8 26 08	00BA E1 01 00BC 26 04		00BE EE 02 00C0 0C 00C1 39	00C2 08 00C3 08 00C4 08	00C5 08 00C6 6D 00 00C8 26 EC	OOCA OD OOCB 39	00CC 42 00CE 0039	00D0 44 00D2 003A	00D4 44 00D6 003B
+	+	0128 + 0129 + 0130	0131 + 0132 + 0133					0145	0147 0148	+ +			0158 0159 0140				0171 0172 0173			

```
POINT TO NEXT ENTRY
           TRANSIENT TO LIST DEVICE ASSIGNMENTS
                                                     END OF TABLE?
                                                                                                                                                                                                                                                                                                                              SAVE POINTER
                        X:=A(PDTAB)
                                                                                                                                                   IF ORIGINAL ASSIGNMENT
                                                                                                                                                                                 윷
                                                                                                                                                                                                   웊
                                                                                                                                                                                                              * YES PRINT ASSIGNMENT
                                                                                   * MOVE DEV1, 2 TO MSG
                                                                                             STA A MSG
STA A MSG+6
LDA A 1, X
STA A MSG+1
STA A MSG+2
STA A MSG+2
STA A MSG+2
STA A MSG+2
                       TABX
SWI
FCB 3
STX PDTAB
                                                                                                                                                              LDA A 3, X
LDA B 4, X
CMP A 5, X
BNE STAT1
                                                     LDA A O, X
BNE *+3
                                                                                                                                                                                                                                                                                                       FCB 10
BRA STATO
                                                                                                                                                                                            CMP B 6, X
BNE STAT1
                                                                                                                                                                                                                                                                                                                                                           STATIA CMP A 5, X
                                                                                                                                                                                                                                                                                                                                                LDX PDTAB
                                                                                                                                                                                                                                SWI
FCB 5
FCB 5
PRTMSG
SWI
FCB 49
PULX
SWI
FCB 6
LLM B #7
ADDBX
NAM STAT
                                                                                                                                                                                                                                                                                                                               PSHX
SWI
FCB 5
                                                                                                                                                                                                                          STATOA PSHX
                                                                                                                                                                                                                                                                                                 ij
                                                                       RTS
                                                                                                                                                    SEE
                                                     STATO
                                                                                                                                                                                                                                                                                                                               STAT1
                                                                                                                0065 R
006B R
                                          œ
                                                                                               œœ
                                                                                                                                  œ œ
                                                                                                                                                                                                                                                                                                                                                 œ
                                                                                                                                            Œ
                                                                                              0064 F
                                          0002 FF 006E
                                                                                                                                 0066
006C
                                                     0005 A6 00
0007 26 01
0001 0000 0000 0000 0000 0003 0004 0005 + 0000 3F 0007 + 0001 03 0008 0002 FF 00 0009 0000 0005 A6 00
 0000 0000
                                                                                              0000 B7
0000 B7
0010 A6
0012 B7
0015 B7
0018 A6
001A B7
                                                                       68 6000
                                                          0011
0012
0013
0014
0015
0017
0017
0018
0020
0020
0021
0022
0023
0023
```

CMMC 2572 M CUCHAR 0023 DELETE 2420 M DESCRA 0020 DESCRA 0020 DIV16 2524 M DL 0036 DP 0038 DP 0038 DP 0040 EJ 0041 FCBCHN 0029
FUBBEF 2460 M
FUBBEF 2400 M
FWIFCB 2488 M
FWIS 2558 M
FWIS 2558 M
FWIS 2558 M
FWIS 2558 M
INDEX 248C M
INDEX 2 231B M 2572 M 0023 243A M 0026 2369 M 0037 2151 M 21CE M 216A M 21E7 M 2406 M 0025 ADDABX 2219 P ADDAX 2232 P ADDBX 2248 P ADDXAB 2200 P BASEQU 2A2A P EMEM 0033 0035 0043 0029 0033 DECNT PSHALL PSHX PULLAL OPEND PRTERR PRTMSG BS CHAIN CLASS CLUSE NY.TOK MUL.16 MUL.8 EUEMEM OPEN 150A 150A 156B FULX 2 2 2 15GC

2277 3 2229 3 2228 3 2265 3 2190 3 2183 3 20027 20027 23D2 3 2185 3

IVBX

TXAB VALUE WD WRITE

αααααααααεεεε 006F 007A 000C 007F 00B6 00C2 00C2

SETR SETAB SETNXT SETSR1 SETSR2 SETSR2 SUBABX SUBABX SUBABX

0064 00200

258C M 2388 M 2384 M 0000 R 0000 R 0011 R 0019 R

SET SE10 SE11 SE13

REMIND 2384

		1, 5		7	Z,	7.7	8	23	20	24	23	78	8	24	2	8	24	7	S	23	24	8	2	င်	22	2	7	83	8	င်	24	2	2	č
VOVIT	V 2 4 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	AUDHA	V000	HUUXHB	SEGU	ZI Ç	CLOSE	<u>န</u>	_ ₹	LETE	1/16	BDEF	BDEF	FMTFCB	47S	FIDE	CMD	XECE	XITE XITE	E E	OADB	2	SAC	99	JL.16	81	χį	OPEN	END	UTAB	RIERR	RIMSG	SHALL	> 1
3	Ě	₹ ₹	₫ 3	đ	à	٥	ಶ	ਹੈ	ਹੈ	ä	Ξ	ĭ	Ξ	ī	Ξ	3	5	Ħ	÷.	≍	=	Ĭ	ž	ž	Ĩ	Σ̈́	Ź	ð	ð	ī	<u>ā</u>	ā	ĭ	Ž
																			ITRY															
																			POINT TO NEXT ENTRY															
						ш													¥															
	Ü			Ę		Σ													욘							TRY AGAIN								
	NO MATCH			NO MATCH		Z													IN							₹								
!	Ž			2		10VE													5							ĭ		`						
						* FOUND ASSIGNMENT MOVE IN NAME			9+		+		œ+				~									Œ		н						
	BNE STAT2		CMP B 6, X	IAT2		NY NE		Ċ,	MSG	, X	MSG	×	MSG	PULX		FCB 6	PATO							INX		IAT1			\$0D					
	S	1	m m	S		SSIC		A A	4	4	: Œ	Q.	d	×	3	83	φ.		×	×	×	· ×	×	×	×	S.		FCC	FCB &		RMB 2		⊆	1
	W.	i	Ü	Z Z		Q Q		9	S		S		O:	. ⊒	Ø	L	8		Z	Z	Z	Z	Z	Z	Z	88		G.	E.				C Z	ì
						FOUN													STATZ									g			PDTAB			
		*			*	*	*		œ		œ		œ					*	S								*	MSG		*	4	*		
									44		006B F	!	0000																					
	0042 26 17		90	133				0	8	0	8	0	8	;	يپا	. ×	, E	i		. ~						8	1	_	_		202			
	2 26		Ē	5 26				4	4	4	E 27	4	2	i	75	. 00	7	i	Ö	Ö	Ö	Č	Č	000	č	2	i 1	20	6		006F 0002	1		
			0044 E1 (00				0048	004	00	00AF	005	0054		Ö	0058 06	00							0900					0067					
	_	N	m	4	ıc	4	: ~	. o	. 0.		· -	٠.	: 0) (+	+		. 00	. 0		; - -	٠,	ŧ (r	• •	·	. 40		۰α	0	٠. ح	; -		ιc	,
	1900	0062	900	7900	900	7400	900	900	Š	200	9	00	700	0	007	9	00	007	00	000	000	8	000	800	008	000	ď	8	à	Š	Š	00	0	3

ADDABX 2219 M
ADDAX 2232 M
ADDAX 2232 M
ADDXB 2220 M
BASEQU 2A2A M
CLOSE 2234 M
CLOSE 2318 M
CLOSE 2318 M
CMPC 2572 M
DIVI6 2524 M
FCBDEF 2650 M
FCBDEF 2650 M
FCBDEF 2650 M
GCMUC 2572 M
DIVI6 2524 M
FCBDEF 2650 M

_	-	
7	വ	п
	7	ч

+ 0015 000B FCBFWD EQU 11 + 0015 000C FCBFWD EQU 14 + 0015 0010 FCBNAM EQU 14 + 0015 0010 FCBNAM EQU 16 + 0015 0010 FCBNAM EQU 29	+ 0015 001F FCBFTS EQU 31 + 0015 0021 FCBFTS EQU 33 + 0015 0023 FCBNMS EQU 35 + 0015 0025 FCBNMS EQU 35 + 0015 0027 FCBNM EQU 37 + 0015 0027 FCBNM EQU 37 + 0015 0027 FCBSFF EQU 41	* DISK ATTRIBUTE SECTION	0017 0002 0019 44 0015 0002 001E 0041 R 0020 0021 0041 0080	*** *** *** *** *** *** *** *** *** *** *** *** ** *** *** *** *** *** *** *** *** *** *** *** *** ** *** *** *** *** *** *** *** *** *** *** *** *** ** *** *** *** *** *** *** *** *** *** *** *** *** ** *** *** *** *** *** *** *** *** *** *** *** *** ** *** *** **
0000 0000 N NAM RANDOM * CP/68 RANDOM-ACCESS FILES PACKAGE * COPYKIGHT 1979 BY HEMENWAY ASSOCIATES * COPYKIGHT 1979 BY NEMENWAY ASSOCIATES	0000 00CB N ENT CREATE BUILD A NEW RANDOM FILE 0000 02E6 N ENT ROPEN OPEN A RANDOM FILE 0000 0387 N ENT RCLOSE CLOSE A RANDOM FILE 0000 03C2 N ENT RREAD READ A BYTE FROM RANDOM FILE 0000 0404 N ENT RWITH WRITE A BYTE TO RANDOM FILE 0000 0404 N ENT EXPAND POSITION RANDOM FILE 0000 05E5 N ENT EXPAND ADD RECORDS TO RANDOM FILE	* VECTORS TO INDIVIDUAL ROUTINES * COCO 7E COCB R RNIVEC JMP CREATE COCO 7E COZE R JMP RCLOSE COCO 7E COZE R JMP RCLOSE COCO 7E COZE R JMP RMRITE COCO 7E COZE R JMP RWRITE JMP RWRI	0022 DESCRC EQU \$22 0023 CUCHAR EQU \$23 0025 RC EQU \$25 0026 CLAS EQU \$25 0027 VALUE EQU \$27 0027 FCBCHN EQU \$28 0028 FNETAB EQU \$28 0033 BMEM EQU \$28	0.035 EMEM EQU #35 0.037 CMEM EQU #37 0.038 DL EQU #39 0.03B DP EQU #38 0.03C DPCNT EQU #36 0.03C ML EQU #36 0.03F TB EQU #37 0.041 EQU #41 0.041 EQU #41 0.042 ES EQU #42 0.043 ES EQU #45 0.044 LDPCNT EQU #45 0.045 LDPCNT EQU #45 0.045 ES EQU #45 0.045 ESBEF 0.005 FCBEQT EQU 5 0.005 FCBEQT EQU 5 0.007 FCBEQT EQU 5 0.007 FCBEQT EQU 5

0111 26 CE BNE CRERR YES *	0113 17 TBA	WRITE	0114 37 5W1	16 05 LDA A FCBSTA, X		*	2C L.DA A FCBRSZ, X	LDA B FCBRSZ+1,	3	011E 3F SW1)2 	BNE CRERR	*	0124 17 TBA	3	3F	, 19 FCB 25	A6 05 L.BA A FCBSTA, X	0129 26 B6 BNE CRERR YES	* !	O128 CE O017 K LDX #KNUFCB	AF 30 CLR FCRACE, Y NO COR	84 FF 100 to #8FF	A7 06 STA	86 20 LDA A #\$20	A7 10	30 TSX	EE OO LDX	OTST AS 09 LDA A FUBBRY, A DEL DAIVE NOMBER	07 09 STA	# CONTRACT OF TAXABLE PROPERTY AND A PARTY OF TAXABLE PROPERTY OF		BUILD INDEX OF RECORD POINTERS	* DATA SPACE BUILT USING RNDFCB	* POCCOD BOTHIED WAS THORED &		. N	m		0	IMS	FCB 20	A6 05 LDA A FCBSTA, X	0148 27 03 BEG CR6 NU	OLAG 75 ONE 1 R CREA . IMP CRERR YES	*	CE 0017 R CR6 LDX #RNDFCB	0150 Ho IF LDH H FCBF13/A OE1 FINST INDEX 0152 E6 20 LDA B FCBF13+1, X FIRST SECTOR	30 TSX	0155 EE 00 LDX 0, X POINT TO FCB		
0184	0186		2000		0191	0192	0193	0194		0196 +		0199	0200	0201	0202		0204 +	0205	9020	0207	0208	0203	0210	0212	0213	0214	0215	0216	0217	0210	0220	0221	0222	0223	0224	0220	0227	0228	0229	0230		0232 +	0233	0234	0520	0237	0238	0240	0241	0242	0244 +	
SAVE FCB ADDRESS			COMPRESSION UPF		2, X TYPE=02 (RANDOM)	(CHECK RECORD SIZE>0			× -1		FREDR NO 11	į			BE SURE LOCAL FCB IS CLOSED			RECOVER FCB ADDRESS			STA A FCBSTA, X PUT IN ERROR STATUS		STREET, COURTS ACTUAL A		S = 12 12 12 12 12 12 12 12				MUST BE CMXRNUM					ERROR NO. 12			POTNT TO ECB		, x output	OPEN FILE				YES	SCOODS OF BECODES	10. Of 11.00		UP FILE=NU. UP RECURDS, SIZE	WRITE NO. OF RECORDS		h, x error?	
CREATE PSHX SWI	FCB 55	CLR FCBSTA, X	CLK FUBSCE, X			TST FCBRSZ, X	BNE CR2	*	TST FCBRSZ+1, X	BNE CRZ	* - DA A #11	TARREST A TARREST A PART A PAR	*	R CRERR LDX #RNEFCB		IMS	FCB 21	PULX		FCB 6	STA A FCBST	, N	× WNGGUD AU - COU		*	TXAB	SWI	FCB 2	LDX #MXRNUM	SUBABX	CMI COD	PUD 12	į	CR3 LDA A #12	BRA	> C		LDA A #\$FF	◂	OPEN	IMS	FCB 20	LDA A FCBSTA, X	BNE CRERR	**	LDA B FCBRN#+1, X	*	* FIRST FUUR BYTES	WRITE	IMS	LDA A FCBSTA, X ERROR?	
+ 000B 3F		Ψ,	000F 6F 29	5 60	Α7	<u>(19</u>	0009 26 10		6D	00DD 26 0C	80 78 JUOU	3		OOE1 CE OO17 R		+ 00E4 3F	+ 00E5 15		00E6			OOEA 39		OOF 1 27 00	ì			+ 00F0 02	OOF1 CE 09B0		+ 00F4 3F	00F6	; ;	30 98	OOFA 20 E5	0000			A7		+ 0103 3F	+ 0104 14		0107 26 DB	70 000	1 P				+ 010D 3F	+ 010E 19 010F A6 05	
0123		0126	0127	0129	0130	0131	0132	0133	0134	0135	0136	013	0130	0140	0141	0142		0144			0147	0148	V 410	0100	0152	0153			0156		010		0161	0162	0163	0164	016	0167	0168	0169			0172	0173	0174	0176	0177	01/8	0180		0183	

BEQ CR7C NO 3 JMP CREHR YES	F 3	FCB 25 LDA A FCBSTA, X ERROR? BNE CR7B YES	LDX *RNDFCB LDA A FCBIND+1, X SUB A FCBDBA+1, X FIND POINTER TSX	LDX 0, X POINT TO FCB WRITE WRITE POINTER SENT	FCB 25 LDA A FCBSTA, X ERROR? BNE CR7R YES		NOW HAVE INDEX FILE AND DATA FILE APPEND DATA FILE TO INDEX FILE	LDX #RNDFCB	CLOSE DATA FILE NOW DUPLICATE MOST OF 'CLOSE' ROUTINE IN CP/68 DOES NOT UPDATE DIRECTORY	TXAB FCB ADDRESS IN (A, B)		LDX FCBCHN GET HEAD OF FCB-CHAIN PSHX SAVE X	SWI FCB 5	SUBABX AT THIS FCB?	FCB 12 PULX RESTORE X	SWI FOB FOR BNF CRRA	A FCBNFB, X	8 4	B FCBCHN+1	CMP A FCBNFB, X	BNE CR88 NO CAUCHE CR88 NO CAUCHE DECEMBER COD	BNE CR8B NO DESINED TODE	X FCB-CHAIN TO GO AROUND THIS FCB
01AA 27 03 01AC 7E 00E1 R	01AF 1 + 01B0	0313 + 01B1 19 0314 01B2 A6 05 0315 01B4 26 F6	01B6 CE 0017 R 01B9 A6 28 01BB A0 08 01BD 30	01BE	++	01C6 7E 0176 R	0330 * NC	01C9 CE 0017 R	0335 * CL 0336 * DL 0337 * DC		+ 0100 +	0342 OICE DE 29 0343	0344 + 01D0 3F 0345 + 01D1 05	+ 01D2	+	0350 + 01D4 3F 0351 + 01D5 06 0352 01D6 26 0A	0108 86		01DE D7	01E2 A1	56	0168 26 10	0365 * FIX
EFROR? YES	WRITE SECTOR	ERROR? YES	FIRST POINTER =4 WRITE POINTER	ERROR? YES	X INIT. TEMP. RECNUM	OR EACH RECORD	×	INIT. TEMP. RECSIZ	OF NULLS			ERROR?	YES	COUNT DOWN RECORD	LOOP UNTIL RECORD DONE	COUNT DOWN NO. OF RECORDS		FR FF		GET TRACK GET SECTOR	POINT TO FCB		ERROR?
FCB 25 I.DA A FCBSTA, X BNE CR5A		FCB 25 LDA A FCBSTA.X ERROR? BNE CR5A YES	LDA A #4 FIRST POINTER WRITE POINTER SWI FCB 25	CBSTA, X A	LDA A FCBRNM, X LDA B FCBRNM+1, X STA A RNMTMP INIT. TEMP. STA B RNMTMP+1	. <u>9</u>	* CR7 LDA A FCBRS7, X IDA B FCBRS7+1, X	A RSZTMP INIT. TEMP. B RSZTMP+1	ITE OU	0180 CE 0017 R CR7A LDX #RNDFCB 0183 4F CR7A LDX #RNDFCB	SMI	8 25 A FCBSTA, X	BNE CR5A YES	RSZTMP	STX RSZTMP BNE CR7A LOOP UNTIL RECORD	Ģ	RNMTMP DONE?	* OUTPUT INDEX BLOCK HERE		A FCBTRK, X GET B FCBSCT, X GET			FCBSTA, X

Y SECTOR	FFER UFFER FREE-SPACE SECTO		194		INKS INDEX FILE TO FIRST T/S
MAKE "OUTPUT" GET DRIVE NO. LIMIT RANGE (0-3) 2 BYTES/TABLE ENTRY ACCESS FREE-SPACE SECTOR	GET FREE TRACK GET FREE SECTOR POINT TO SECTOR BUFFER X PUT NEW T/S INTO BUFFER X POINT TO DATA FCB WRITE OUT UPDATED FREE-SPACE SECTO	EKROR? VES POINT TO FCB	ERROR? YES ERROR?	YES ERROR? YES GET FIRST T/S	POINT TO FCB POINT TO BUFFER UPDATE FORWARD LINKS POINT TO FCB GET LAST T/S OF INDEX FILE POINT LOCAL FCB TO FIRST T/S
COM FCBDTT, X LDA A FCBDRV, X AND A #\$03 ASL A LDX #FRETAB ADDAX	SWI FCB 9 LDA A 0, X LDA B 1, X LDX #RNDBUF STA A SECSIZ-2, X STA B SECSIZ-1, X LDX #RNDFCB	19 FCBSTA, X R9A , X BLOCK=0,	WRITE SWI FCB 25 LDA A FCBSTA, X BNE CR9A WRITE SWI FCB 25 LDA A FCBSTA, X	BNE CR9A WRITE SWI FCB 25 LDA A FCBSTA, X B BNE CR9A LDX #RNDFCB LDA A FCBFTS, X (LDA B FCBFTS, X (LDA B FCBFTS, X (LDA B FCBFTS, X (LDA B FCBFTS+1, X (LDA B FCBFTS+	LDX O, X LDX FCBDBA, X STA A O, X STA B 1, X TSX LDX O, X LDA A FCBTRK, X LDA B FCBSCT, X LDA B FCBSCT, X LDA A FCBTSY, X STA A FCBTS, X
023D 6 023F 6 0241 8 0243 4	0247 3F 0248 09 0248 64 00 0248 E6 01 0250 A7 7E 0252 E7 7F 0254 CE 0017 R	0257 3F 0258 13 0259 A6 05 0258 26 42 0250 30 025E EE 00 0260 4F	0261 3F 0262 19 0263 A6 05 0265 26 38 * 0267 3F 0268 19 0269 A6 05		027A 30 027B EE 00 027B EE 07 027F A7 00 0281 E7 01 0284 EE 00 0284 EE 00 0288 E6 08 0288 E6 01 0289 C 017 R 0290 A7 04
	0435 + 0438 + 0438 + 04439 04442 04442 04442	0445 0448 0448 0448 0448 0453 0453	0453 + 6464 + 64	0465 0466 0466 0468 + 0470 0472 0473	0476 0477 0478 0478 0480 0481 0483 0484 0485 0486
SAVE X S POINT TO THIS FCB	LDA A FCBNFB+1, X LDA B FCBNFB+1, X PULX PULX FCB 6 STA A FCBNFB, X UPDATE LINKAGE AROUND FCB STA B FCBNFB+1, X BRA CRBC	<u> </u>	FCBSCT, X AT END OF DISK? CRSE NO A FCBBAK, X FIXUP FOR END-OF-DISK B FCBTKK, X B FCBTKK, X CRSCT, X CRSF OR WRITE LAST DATA SECTOR	CBSTA, X ERROR? B YES CBNMS, X ONE MORE SE CBNMS+1, X 0 0 0 CBNMS, X CBNMS, X CBNMS, X CBNMS+1, X	B FCBSCT. X A FCBLTS, X A FCBLTS, X B FCBLTS+1. X FCBDTT, X A #0 B #FRESEC FREE-SPACE SECTOR B FCBRX. X C B FREE-SPACE SECTOR C C C C C C C C C C C C C C C C C C C
	LDA LDA SWI STA STA STA STA	CRSB LDX	R8D LDA STA STA BRA SWE IOH	LDA ADD STA STA	LDA W FI STA B FI STA B FI CLR FCB CLR FCB STA B FI STA B
*	•	œ	00 00 00 00 00 00 00 00 00 00 00 00 00	* 0	
+ 01EA 3F + 01EB 05 + 01EC 3F + 01ED 03	01EE A6 25 01F0 E6 26 + 01F2 3F + 01F3 06 01F4 A7 25 01F6 E7 26 01F8 20 04	EE CE CE 27	g 94770 g	00 0000000	0227 E6 0B 0229 A7 21 022B E7 22 022D 6F 06 022F 86 00 0231 C6 03 0233 E7 0B 0235 E7 0B + 0237 3F + 0238 13 0239 A6 05

```
READ NUMBER OF RECORDS FROM FILE
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 STA A FCBRNM+1, X
READ RECORD SIZE FROM FILE
OPEN A RANDOM-ACCESS FILE
CALL WITH ADDRESS OF FCB IN INDEX REGISTER
MUST HAVE EXTENDED FCB (170 BYTES)
SET DRIVE, FILENAME
                                                                                                                                                                                                                                                  RECOVER FCB ADDRESS
                                                                                                                                                                                                                                                                                                                                      LDA A FCBTYP, X CHECK TYPE OF FILE CMP A #2 RANDOM? BEQ ROP3 YES
                                                                         SAVE FCB ADDRESS
                                                                                                                                                                                                                                                                                                                                                                                                ERROR NUMBER 14
CLOSE FILE
                                                                                                                                 NO COMPRESSION
                                                                                                                    INIT. STATUS
                                                                                                                                                                                                                                                                                             STA A FCBSTA, X SET STATUS RTS
                                                                                                                                                              OPEN FILE
                                                                                                                                                                                                     LDA A FCBSTA, X ERROR?
BEQ ROP2 NO
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                LDA B FCBSTA, X ERROR?
BEQ ROP3B NO
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               LDA B FCBSTA, X EKROR?
BNE ROP3A YES
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    FCB 24
LDA B FCBSTA, X ERROR?
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   FCB 24
LDA B FCBSTA, X EKROR?
                                                                                                                                                INPUT
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      STA A FCBRNM, X
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              STA A FCBRSZ, X
                                                                                                                   CLR FCBSTA, X
CLR FCBSCF, X
CLR FCBDTT, X
OPEN
                                                                                                                                                                                                                                                                                                                                                                                                                                          FCB 21
BRA ROPERR
                                                                                                                                                                                                                                                                                                                                                                                                LDA A #14
CLOSE
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         BRA ROPZA
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                SNE ROP3A
                                                                                                                                                                                         FCB 20
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  FCB 24
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 FCB 24
                                                                                                      FCB 5
                                                                                                                                                                                                                                                                                FCB 6
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     READ
                                                                                                                                                                              IMS
                                                                                                                                                                                                                                                  ROPERR PULX
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       SE
                                                                                                                                                                                                                                                                                                                                                                                                                               3
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            TBA
                                                                         ROPEN
                                                                                                                                                                                                                                                                                                                                                                                                               R0P2A
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      ROP3B
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           40P3A
                                                                                                                                                                                                                                                                                                                                       R<sub>O</sub>P<sub>2</sub>
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      ROP3
                                                                                                                 02E8 6F 05
02EA 6F 29
02EC 6F 06
                                                                                                                                                                                                                                                                                                                                      02F9 A6 1D
02FB 81 02
02FD 27 06
                                                                                                                                                                                                     02F0 A6 05
02F2 27 05
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       0320 3F
0321 18
0322 E6 05
                                                                                                                                                                                                                                                                                                                                                                                                                            0301 3F
0302 15
0303 20 EF
                                                                                                                                                                                                                                                                                            02F6 A7 05
02F8 39
                                                                                                                                                                                                                                                                                                                                                                                                02FF 86 0E
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                0307 E6 05
0309 27 03
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         030C 20 F3
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    030E A7 2A
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               0312 E6 05
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           0314 26 F5
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        0316 A7 2B
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 031A E6 05
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              031C 26 ED
                                                                                                                                                                                                                                                                                                                                                                                             0578 02FF 86 0E
0579 0580 + 0301 3F
0581 + 0302 15
0582 0303 20 EF
0584 0305 3F
0586 + 0305 3F
0586 + 0307 E6 05
0589 0307 E6 05
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   0.602 + 0.318 3F
0.603 + 0.319 18
0.604 0.31A E.6 0.5
0.605 0.31C 26 EU
0.607 0.31E A7 2C
0.608
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           031E A7 2C
                                                                                       02E6 3F
02E7 05
                                                                                                                                                                          02EE 3F
02EF 14
                                                                                                                                                                                                                                                                02F4 3F
02F5 06
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 0310 3F
0311 18
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           030B 17
                                                                                                                                 0560 0
0551 0
0553 +
0563 +
0564 +
0565 0
0566 0
0568 +
0568 +
0568 +
0570 +
0571 0
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         0590 (0591 (0592 (0594 (0594 (0594 (0596 (0596 (0596 (0596 (0596 (0599 (0599 (0599 (0590 (0599 (0590 (0599 (0590 (0599 (0590 (0599 (0590 (0599 (0590 (0599 (0590 (0599 (0590 (0599 (0590 (0599 (0590 (0599 (0590 (0599 (0590 (0599 (0590 (0599 (0590 (0599 (0590 (0599 (0590 (0599 (0590 (0599 (0590 (0599 (0590 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (0599 (
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    0610 +
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        + 6090
                             0553
0554
0555
0557
0558
                                                                                                                                                                                                                                                                                                                         0573
                                                                                                                                                                                                                                                                                                                                                    0575
0576
0577
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        0601
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          LDX #RNDFCB
LDA A FCBLTS, X
LDA B FCBLTS+1, X GET LAST T/S OF DATA FILE
                                                                                                                                                                                                                                                                                                                                                                                                                                                                     WRITE LAST INDEX SECTOR
                                                                                                                                                                                                                                                                                                                                                     OF SECTORS
                                                                                                                                                                                                                     UPDATE BACK LINKS
                                                                                                                                                                                                      POINT TO BUFFER
                                                                                                                                                                                                                                                                WRITE SECTOR
                                                                                                                                                                                                                                                                                                                                                 LDA A FCBNMS, X GET NO. OF S.
LDA B FCBNMS+1, X
TSX
LDX 0, X
ADD B FCBNMS+1, X
ADC A FCBNMS, X
STA A FCBNMS, X
STA B FCBNMS, X
STA B FCBNMS+1, X
                                                                                                                                                                                                                                                                                                                                                                                             POINT TO FCB
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   POINT TO FCB INPUT
                                            READ SECTOR
                                                                                                                                 CLEAN STACK
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               READ SECTOR
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           CLEAN STACK
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              CLOSE FILE
                                                                                                                                                                                                                                                                                                         ERROR?
YES
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        ERROR?
Yes
                                                                                       ERROR?
                                                                                                                                                                            OUTPUT
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  ERROR?
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   OUTFUT
                               IN THE
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    DONE
 LDA A FCBF1S+1, X
STA A FCBSCT, X
CLR FCBDT1, X
                                                                                                                                                                                                                                                                                         FCB 19
LDA A FCBSTA, X
BNE CR9A
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    TSX
LDX 0, X
CLR FCBDT1, X
STA A FCBTKK, X
STA B FCBSCT, X
IOHDR
                                                                                      LDA A FCBSTA, X
BEQ CR9B
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               LDA A FCBSTA, X
BNE CR9A
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       LDA A FCBSTA, X
BNE CR9A
                                                                                                                                                                          COM FCBDTr, X
PUL, A
                                                                                                                                                                                                     LDX FCBDBA, X
                                                                                                                                                                                                                 STA A 2, X
STA B 3, X
LDX #RNDFCB
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  COM FCBDTT, X
                                                                                                                                 INS
JMP CRERR
                                                                        FCB 19
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   FCB 19
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          FCB 19
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            FCB 21
                                                                                                                                                                                                                                                                                                                                                                                                                                                                    IOHDR
Swi
                                            COHDR
                                                                                                                                                                                                                                                                COHDR
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 CLOSE
                                                           IMS
                                                                                                                                                                                                                                                                             IMS
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           SWI
                                                                                                                                                                            CR9B
                                                                                                                                                CR9A
                                                                                                                                                                                                                 02A7 A7 02
02A9 E7 03
02AB CE 0017 R
                                                                                                                                                œ
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           œ
                                                                                                                                 029E 31
029F 7E 00E1
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          02C9 CE 0017
02CC A6 21
 0292 A6 20
0294 A7 0B
0296 6F 06
                                                                                                                                                                          02A2 63 06
02A4 32
02A5 EE 07
                                                                                                                                                                                                                                                                                                       02B0 A6 05
02B2 26 EB
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       02DB A6 05
02DD 26 C0
                                                                                                                                                                                                                                                                                                                                                                                            EE 00
EB 24
A9 23
A7 23
E7 24
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       A6 21
30
30
EE 00
EE 00
A7 0A
E7 0B
                                                                                      029A A6 05
029C 27 04
                                                                                                                                                                                                                                                                                                                                                    A6 23
E6 24
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  02C5 A6 05
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                02C7 26 D6
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 02DF 63 06
                                                                                                                                                                       0501 02A2 63 06
0502 02A4 32
0503 02A5 EE 07
0504 02A7 A7 02
0505 02A9 E7 03
0506 02AB CE 00
0507 + 02AE 3F
0509 + 02AF 13
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   02C3 3F
02C4 13
                                                                        0299 13
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          02D9 3F
02DA 13
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              02E1 3F
02E2 15
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      02E3 3F
02E4 06
02E5 39
                                                          0298 3F
                                                                                                                                                                                                                                                                                                                                                                               ဓ္ဗ
                                                                                                                                                                                                                                                                                                                                                                                                          02BB
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 02D3 (02D5 (02D7 1
                                                                                                                                                                                                                                                                                                                                                                                                                            OZBD
                                                                                                                                                                                                                                                                                                                                                                                                                                        02BF
02C1
                                                                                                                                                                                                                                                                                                                                                     02B4
                                                                                                                                                                                                                                                                                                                                                                               02B8
                                                                                                                                                                                                                                                                                                                                                                                               02B9
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     oggo
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    0201
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      0546 +
0547 +
0548 (
                                                                                                                                                                                                                                                                                                       0543 +
0544 +
0545
                                           0492 0493 0494
 0489
                                                                                    0495
0496
0497
0499
0500
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      0538
0539
0540
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  0541
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 0542
```

CMP * CMP * BRE * BRA * BRA * BRA * BRA * CLR	** CLOSE A R ** CALL. WITH TXAB SWI	BEG RCLOS4 YES, GOOD
26 CA 26 CA 26 C5 26 C5 20 D8 6F 2E	0.382 A7 0.384 7E 0.388 0.5 0.388 0.5 0.388 0.5 0.388 0.5 0.388 0.5 0.390 0.391 3F 0.392 3F 0.393 3F 0.393 3F 0.395 27 0.397 EE 0.397 EE 0.398 86	
0673 0674 0675 0675 0677 0677 0680 0681 0683	0.685 0.686 0.688 0.688 0.690 0.691 0.692 0.692 0.695 0.695 0.696 0.696 0.696 0.700 0.	0732 0733 0734
YES 1. X TABLE IN FCB -FCBRTB TOTAL LENGTH OF TABLE CLEAR A BYTE LOOP UNTIL DONE	TEX	A FCBTRK, X NEW SECTOR? B FCBSCT, X
BNE ROP3A YES STA A FCBRS7+1, X CLEAR INDEXING TABLE IN FCB LDA B #FCBRTE-FCBRTB TOTAL CLR FCBRTB, X INX INX BNC B BNE ROP4 LOOP I	TEX CLD X 0, X CLD X 0, X SWI SWI SWI SWI SWI SEAD Y STX RINTMP STX RINTMP STA B FCBSCT, X CLD X STA B TWPSCT INX SWI	LDA A FCBTRK, LDA B FCBSCT,
* * * * ROP 4	0P5A	*
0324 26 E5 0326 A7 2D 0328 C6 78 0326 6F 32 0320 08 0320 5A	0331 EE 00 0333 86 32 0335 8F 02 0336 09 0337 FF 0005 0338 EE 00 0344 E7 01 0348 E7 01 0348 E7 01 0348 E7 01 0355 FF 0005 0357 18 0355 3F 0355 7E 0301 0356 3F 0357 18 0358 E6 05	036E A6 0A 0370 E6 0B
	++ ++ ++	

CHECK FILETYPE=02 (RANDOM) CHECK FILETYPE=02 (RANDOM) LDA A FCBTYF, X CMP A #2 BEQ RRED4 YES LDA A #14 NO, ERROR 14 BRA RREDER LED4 TST FCBDTT, X READ OR WRITE? BEQ RRED5 READ	HING FROM WR DR B 19 B 19 A FCBSTA, X A FCBSTA, X A FCBDTT, X M FCBDTT, X M I B 24 B FCBSTA, X RRED6	RED6 STA A SAVEA SAVE DATA BYTE JMP RWRIT6 UPDATE RANDOM FILE POINTERS WRITE A BYTE INTO A RANDOM-ACCESS FILE CALL WITH ADDRESS OF FCB IN INDEX REGISTER BYTE TO BE WRITTEN IN "A" REGISTER RETURN STATUS=15 WHEN LAST BYTE OF FILE WRITTEN RRITE STA A SAVE DATA BYTE PSHX SAVE FCB 5 TXAB SWI FCB 5 CHECK THAT FILE IS OPEN (LOOK AT FCB-CHAIN) LDX FCBCHN BEQ RWRIT2 NO ACTIVE FCBS-ERROR 13 FCB 5 SUBABX AT THIS FCB? SUBABX SWI FCB 5 SUBABX FCB 5 SUBABX SWI FCB 5 SUBABX FCB 5 SUBABX FCB 5 SUBABX FCB 12
* * * * * * * * * * * * * * * * * * * *	*** * * * * * * * * * * * * * * * * * *	R RRED6 * WRITE * BYTE * RETUH * CHECK * CHECK * CHECK * CHECK
03DF EE 00 03DF EE 00 03E1 A6 1D 03E3 81 02 03E5 27 04 03E7 86 0E 03E9 20 EU	03EF 3F 03F0 13 03F1 A6 03F3 26 03F5 6F 03F8 18 03F9 E6 03FB 27 03FU 17	0400 B7 00CA 0403 7E 040A 0406 B7 00CA 0408 3F 040C 02 040D DE 29 040F 27 0C 0411 3F 0412 05
0.797 0.798 0.799 0.801 0.803 0.805 0.805 0.805 0.806 0.806 0.806 0.807 0.808		0831 0832 0834 0835 0836 0837 0840 0840 0841 0841 0841 0841 0842 0843 0843 0843 0844 0845 0845 0855 0855 0855 0855 0855
LDA A #14 NO, ERROR 14 BRA RCLERR * RCLOS4 TST FCBDT1, X READ OR WRITE? BEG RCLOS5 IF READING, FINISH CLOSE * IOHDR WRITE SECTOR SWI FCB 19 LDA A FCBSTA, X EKROR? BNE RCLEKR YES * CLR FCBDT1, X MAKE INPUT RCLOSS PULX RECOVER FCR ADDRESS	SMI FCB 6 CLOSE SWI FCB 21 FCB 21 FTS A BYTE FROM A INTH ADDRESS O WITH ADDRESS O WITH ADDRESS OF SWI FCB 5 FT FCB 5 FCB 5 FCB 5 FCB 5	* CHECK THAT FILE IS OPEN (LOOK AT ACTIVE FCB-CHAIN) * LDX FCBCHN BEG RRED2 NO ACTIVE FCB=ERROR 13 * RRED1 PSHX * SWI FCB 5 SUBABX SWI FCB 6 BEG RRED3 YES, GOOD * LDX FCBNFB, X NO, TRY NEXT FCB IF THERE IS ONE BRE RRED1 * RREDER PULX SWI FCB 6 BEG RRED3 YES, GOOD * TRY NEXT FCB IF THERE IS ONE BRE RRED1 * SWI FCB 6 STA A FCBSTA, X RETURN ERROR STATUS CLR A RETURN NULL BYTE
	то тю то	95 0 0 125 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
03AD 86 0E 03AF 20 EC 03B1 6D 06 03B3 27 08 03B5 3F 03B6 13 03B7 A6 05 03B9 26 E2	03BD 3F 03BE 3F 03C0 15 03C1 39 03C2 3F 03C3 05 03C3 05	03C6 DE 29 03C8 27 0C 03C8 3F 03C8 3F 03C8 3F 03C6 3F 03C6 3F 03C8 3F 03D4 26 F1 03D8 3F 03D8 3F 03D8 3F 03D8 4F 03D8 4F 03D8 4F

LDA A FCBSTA, X EKROR? BNE RWTERR YES	* * BRING IN FORWARD-LINKED SECTOR AND UPDATE LINKS *	LDA A FCBTRK, X CMP A FCBLTS, X AT END OF FILE? BNE RWRITS NO	* LDA B FCBSCT, X AT END OF FILE? BNE RWRITS NO	* LDA A #15 YES, ERROR 15 BRA RWIERR *	RWRITS CLR FCBDT1, X SET TO "READ" LDA A FCBFWD, X GET FORMARD 17NK 1/8	B FCBSCT, X B FCBSCT, X READ NEW SECTOR	19 (19)	LUH H FLES H, A FLANOR; BNE RWIERR YES *	LDX FCBDBA, X POINT TO SECTOR BUFFER LDA A O, X LDA B 1, X GET NEW FORWARD LINKS	0, X A FCBFWD, X UPDATE FORWARD L	B FCBFWD+1, X FCBDBA, X A 2, X	LDA B 3, X GET NEW BACKWARD LINKS TSX LDX 0, X		1 m m 4 4	. n Z	NOW UPDAT	LDA B	A FCBPOS, X B FCBPOS+1,	CMP A FCBRSZ, X BEYOND RECORD LENGTH? BHI RWRIT8 YES, MUST INC. RECORD NO.	* BCS RWRIT7 NO
0920 0460 A6 05 0921 0462 26 BB	0922 0923 0924	0925 0464 A6 0A 0926 0466 A1 21 0927 0468 26 08		0931 0932 046E 86 0F 0933 0470 20 AD 0934	0472 6F 0474 A6	0937 0476 E6 00 0938 0478 A7 0A 0939 047A E7 0B	+ 047C 3F + 047D 13	0943 04/E A6 05 0944 0480 26 9D 0945	0946 0482 EE 07 0947 0484 A6 00 0948 0486 E6 01	0488 30 0489 EE 048B A7	048D E7 048F EE 0491 A6	0493 0495 0496	0498 A7 049A E7	0960 049C H6 07 0961 049E E6 08 0962 04A0 CB 04 0963 04A0 CB 04	04A6	40 040	04AC E6 04AE CB	04B0 89 04B2 A7 04B4 E7	04B6 04B8	0979 04BA 25 04 0980
	FOUND FCB?	TRY NEXT FCB IF THERE IS ONE	OR 13	i 6 A FCBSTA, X RETURN ERROR STATUS	POINT TO FCB	(RANDOM)	0	ERROR 14	MAKE "OUTPUT"	CHECK FOR END OF BUFFER	END OF BUFFER?	END OF BUFFER?	ECTOR BUFFER	POINT TO SECTOR BUFFER GET DATA BYTE STORE IT	MOVE BUFFER POINTER	LIDRATE BOTHITED		ND GET NEW SECTOR	SET TO "WRITE"	WATER SECTOR
	FUB 6 BEG RWRIT3	* LDX FCBNFB, X NO, BNE RWRIT1 *		FCB 6 STA A FCBSTA, X RE RTS *	WRIT3 TSX LDX 0, X	AT FILETYP	LUA A FUBIYE, X CMP A #2 BEG RWRIT4 GOOD	* LDA A #14 NO, BRA BUTERR	• <		SBC	*	STORE DAT	* RWRT4A LDX FCBIND, X LDA A SAVEA STA A O, X	TAAB TXAB SWD 0		rωş	* * WRITE OLD SECTOR OUT AND *	WRT4B	IONDK SWI FCB 19
0415	. 0416 06 0417 27 0B	0419 EE 25 041B 26 F4	C	. 0420 06 0421 A7 05 0423 39	0424 30 0425 EE 00	;	0427 66 1D 0429 81 02 042B 27 04	042D 86 0E	86	54 3 C	8 B 5	F 17	ì		044E 3E	0451 30 0452 EE	0456 E7 28 0458 20 50		045A 86 FF 045C A7 06	045E 3F 045F 13
	0861 + 0862 0863	0863 0865 0865		0870 + 0871 0872 0872	0874	0876 0877 0878	0879 0880 0881	0882 0883 0884	0885 0886 0887	6880 0888 0889	0891 0892 0893	0894	0897	0899 0900 0901 0902	0904 0905 +		0910 0911	0912 0913 0914	0915	0917 0918 + 0919 +

CLIP B FCBRS7+1.X BEYOND RECORD IN.O. 1044 04F9 20 EB	LDA A #14 IF NOT, ERROR 14 BRA POSIA	LDX FCBRCD, X GET RECORD NUMBER	W CHECK THAT RECORD NUMBER IS VALID			BNE POS3B MUST BE >0	Ę.	BRA POSIA		A C AUT TO FOR	A FCBRNM, X	œ.	a	SECT FORCE, A BUST BE CHRIM		FCB WAS IN "WRITE", FINISH LAST SECTOR		TST FCBDTY, X	CONTRIBUTE TO LEGGE	IOHDR WRITE LAST SECTOR		L.DA A FCBSTA, X EKROR?	POS1A	C. R. ECBRIT, X. MAKE INPLI		J FIND PROPER FCB-TABLE ENTRY FOR RECORD		"KINIMP" (U PUSILIUN WILHIN HABLE "BUMTMB" TO BECORD NO	TITE TO RECORD NO. TO POSITION WITHIN INDEX	"B" TO POSITION WITHIN INDEX BLOCK			NS		STX RINIMP POINT TO START OF TABLE IN FCB	I DX 0, X POINT TO FCB	FCBRCD, X		STX RNMTMF RECORD NUMBER DESIRED		FIRST 4	4 BYTES OF FIRST INDEX SECTOR=RNM, RSZ	**************************************		LDX RNMTMP AT DES	BEQ POSS IF SO, BRANCH
CHAP B FCBNS74.1 X BEYONG RECORD LENGTH? 1943 04F8 84 0FE B.	*	P0S3	MON *	_	>	:	* P0S3		* 0	5001					*	* IF	*		*					*	*	MON *			* *	*	* 1	FUS/4			œ							*	*		R POS4A	
### CHAP B FCBRS7+1.X BEVOND RECORD LENGTH? 1044 ### IT PULX NO. NORMAL RETURN 1046 SHI RMRITB	88 20	ĒE				56	98	20		9 1	1 &	E6	G (7 E	Ì			65	· ·		0515	0517 A6	26	7	5				•		è	8	051F	0250		S H	Ш	6	ï				ď	88	FE 0001	27
## CMP B FCBRS7+1. X BEYOND RECORD LEN ## LDA A SAVEA ## RTS ## LDA A FCBRCD, X LPDATE RECORD NUM # ADD B # 1 ADD A # 0 # ADD B # 1 ADD A # 0 # ADD B # 1 ADD A # 0 # ADD B # 1 ADD A FCBRCD+1. X # ADD B # 1 ADD A FCBRCD+1. X # ADD B # 1 ADD A FCBRCD+1. X # ADD B # 1 ADD A # 0 # ADD B # 1 ADD A # 0 # ADD B # 1 ADD A FCBRCD+1. X # CHECK THAT FILE IS OPEN (LOOK AT FCB-CH FCB SW) # CHECK THAT FILE IS OPEN (LOOK AT FCB-CH FCB SW) # ADD A FCB SW # ADD A # ADD A FCB SW # ADD A # ADD A FCB SW # ADD A # ADD A # ADD A FCB SW # ADD A FCB SW # ADD A # ADD A # ADD A FCB SW # ADD A	1043 1044 1045	1046	1048	1049	1051	1052	1054	1055	1056	7050	1059	1060	1061	1063	1064	1065	1066	1067	1069				1074	1075	1077	1078	1079	000	1087	1083	1084	1080			1086	1091	1092	1093	1094	1095	1097	1098)))))	1101	1102	1103
	CMP B FCBRSZ+1, X BHI RWRIT8	PULX NO.	FCB 6	R LDA A SAVEA	2	LDA A FCBRCD, X	20 20			DOC3 POSITION FIRE TO	POSITION RANDOM-ACCESS FILE TO DESIRED	OF FCB IN INDEX REGIS		SAVE FCB			TXAB		-	CHECK THAT FILE IS	YU I	NO ACTIVE FCB=ERROR				AT	TAN COLUMN	P.C.B. 1.2	130		BEG POS2 YES,	I DX FORMER, X TRY NEXT FOR IF THERE IS	POSI		L.U.A. A. #13			₫			È	TSX C YOUR TO	A FCBTYP, X	CAP A #2	BEQ POS3	*
048E 040C	0981 0982 0983		+ 9860	0987	6860	0660	0991	6660	0994	0000	8660	6660	0001	1002	1003 +			1006 +		1009	1010		1013	1014			1013		1021 +	+	1023			1027	1028	1030 +				1034		1037				1042

ERROR? No Ye.S	GET SECTOR OF RECORD ERROR? YES	GET POINTER OF RECORD ERROR? YES	K, TMPSCT, TMPPNT SET TRACK SET SECTOR MAKE "INPUT" READ IN DATA SECTOR ERROR?	GET FORWARD LINKS UPDATE LINKAGE POINT TO SECTOR BUFFER GET BACKWARD LINKS UPDATE LINKAGE	RECOVER FCB ADDRESS
SWI FCB 24 LDA B FCBSTA, X BEQ POSSD TBA UMF POSIA	STA A TMPTRK READ SWI FCB 24 LDA B FCBSTA, X BNE POSSC	A TMPSCT 24 B FCBSTA, X POSSC		LDX FCBDBA, X LDA A 0, X LDA B 1, X TSX LDX 0, X STA A FCBFWD, X STA A FCBFWD, X STA A FCBFWD, X STA A FCBBA, X LDX FCBDBA, X LDA A 2, X LDA A 2, X LDX 0, X STA A FCBBAK+1, X STA B FCBBAK+1, X STA B FCBBAK+1, X STA A FCBBAK+1, X	TXAB SWI FCB 2 PULX SWI
+ 0581 3F + 0582 18 0583 E6 05 0585 27 04 0587 17 P0S5C 0588 7E 04EA R	058B B7 00C7 + 058E 3F + 058F 18 0590 E6 05 0592 26 F3	0594 B7 00C8 + 0597 3F + 0598 18 0599 E6 05 0598 26 EA	059D B7 00C9 R 05A0 B6 00C7 R 05A3 F6 00C8 R 05A6 A7 0A 05A6 A7 0A 05A6 AF 06 + 05AC 3F + 05AC 3F + 05AC 13 054E A6 05	0.582 EE. 07 0.584 A6 00 0.588 30 0.588 A7 00 0.588 EE 00 0.588 EE 00 0.558 EE 00 0.558 EE 00 0.556 EE 00	+ 05D3 + 05D4 + 05D5
1165 1166 1167 1168 1169 1170	1172 1173 1174 1175 1177 1177	1179 1180 1181 1183 1184 1185		1202 1203 1203 1203 1204 1210 1211 1211 1211 1211 1211 1211	1221 1222 1223 1223 1224 1224
AT END OF AN INDEX SECTOR? IF SO, BRANCH COUNT A BYTE COUNT A BYTE FOR INDEX BLOCK AT FND OF TAMPY REOCK?	LOOP UNTIL DONE		R ENTRY IN FCB-TABLE D DATA START OM TABLE INTO FCB	OF INDEX FFER CORD DATA	YECORD
		RESET "A"	RESET "B" IS TO PROPER ENTRY II IS TO RECORD DATA ST SAVE A GET T/S FROM TABLE X PUT T/S INTO FCB	X ERROR? NO YES POINT TO BUFFER POINT TO RECORD	X INIT. FCBIND +1, X FE INDEX BLOCK GET TRACK OF RECORD
CMP A #SECSIZ A BEQ POS4D I INC A DEC B RED POS4F	BRA POS4A TABLE POINTER LDX RINTMP INX	INX STX RINTMP LDA A #4 BRA FOS4C IT DOWN RECORD NUMBER	DEX STA RNMTMP LLDA B #3 BRA POS4A "RINTMP" POINTS "A"	STA B FCBSCT, X IOHDR SWI SWI FCB 19 LDA A FCBSTA, X ERROR? BEQ POSSB NO WIP POSIA YES LDX FCBDBA, X POINT TO BU LDA A TMPPNT ADDAX SWI FCB 9 TXAB SWI FCB 2 TXAB TXAB SWI LDX O, X	A FCBIND, X B FCBIND+1 THE 3-BYTE
A #SECSIZ POS4D A B B POS4F	20 F1 * BRA POS4A * * * * * * * * * * * * * * * * * * *	08 INX FF 00C5 R STX RINTMP 86 04 LDA A #4 20 EF. * BRA POS4C * * COUNT DOWN RECORD N * FE 00C1 R POS4E LDX RNM1MP	PF 00C1 R STX RNMTMP C6 03 LDA B #3 20 DA ** NOW "RINTMP" POINTS ** NOW "RINTMP" POINTS ** "A" "A" "A" "A" "A" "A" "A" "A" "A" "	E7 0B STA B FCBSCT, X 10HDR 8 3F SW1	A7 27 STA A FCBIND, X E7 28 STA B FCBIND+1 * NOW READ THE 3-BYTE * READ

OLD RECNUM+FCBRCD < MXRNUM	NO, ERROR 12 UPDATE RNUM	POINT FILE TO FIRST T/S MAKE 'INPUT' NO SPACE COMPRESSION SAVE LAST T/S	READ FIRST T/S OF FILE ERROR? NO YES GET NEW RNUM POINT TO SECTOR BUFFER PUT NEW RNUM IN BUFFER	GET FORWARD LINKS UPDATE FORWARD LINKAGE POINT TO SECTOR BUFFER GET BACKWARD LINKS	UPDATE BACKWARD LINKAGE SKIP FIRST 8 BYTES OF BUFFEK NEW BUFFER POINTER
BRA EXPERR TSX LDX 0, X CK THAT NEW RECNUM= LDA B FCBRNM, X LDA B FCBRNM+1, X ADD B FCBRCD+1, X ADC A FCBRCD, X LDA MXRNUM	gel″g ç∢ a∢	: w < w f f c w < w	, K		LDX 0, X STA A FCBBAK, X STA B FCBBAK+1, X LDX FCBDBA, X LDA A #8 ADDAX SWI FCB 9
0614 20 E1 * 0616 30 EXP4B 0617 EE 00 * CHE(0619 A6 2A CHE(0610 EB 2F 0610 E9 2F 0617 A9 2E 0617 A9 2E	7 3 0C 2B 3C	65 67 67 67 67 67 67 67	5 3F 6 13 7 27 03 7 E 05F7 7 E 05F7 8 E 6 2B E 6 2B E 7 04	966 HE 77 HE 30 HE	0668 EE 00 066E E7 0F 0670 EE 07 0672 86 08 0674 3F 0675 09
**************************************		1308 1309 1310 1311 1313 1314 1315 1315 1315		1330 1331 1332 1333 1335 1336 1336 1338	1340 1341 1342 1343 1344 1345 1345 1347 1348
FCB 6 STA A FCBIND, X SET BUFFER POINTER STA B FCBPOS, X CLR FCBPOS, X LDA A #1 STA A FCBPOS+1, X LDA A SAVEA RTS EXPAND A RANDOM-ACCESS FILE EXPAND A RANDOM-ACCESS FILE CALLED AUTH ADDRESS OF FCR IN INDEX PEGISTED	INALS	IS OPEN (LOOK AT FCB-CHAIN) NO ACTIVE FCBS=ERROR 13 AT THIS FCB?	YES NO, EKROR 13 FORCE FILES CLOSED	SWI FCB 6 STA A FCBSTA, X RETURN ERROR STATUS RTS * EXP3 TSX LDX 0, X * CHECK THAT FILE IS RANDOM-ACCESS (TYPE=02) CMP A #2 RANDOM-RESS (TYPE=02) RPA A #2 PADOM?	NO, E GET N NONZE
FCB 6 STA A FCBIND, X STA B FCBINL+1, X CLR FCBPOS, X LDA A *1 STA A FCBPOS+1, X LDA A SAVEA RTS * EXPAND A RANDOM-ACCES * CALLED WITH ADDRESS DIS	×	T FILE FCBCHN EXP2 5 5	SWI FCB 12 PULX SWI FCB 6 BEG EXP3 * EXP2 LDA A #13 EXPER LDX #RNUFCB CLOSE CLOSE SWI FCB 21 PULX	SWI FCB 6 STA A FCBSTA, X RTS * EXP3 TSX * CHECK THAT FILE IS CMP A FCBTVP, X CMP A FCBTVP, X BFG FYA	
05D6 06 05D7 A7 27 05D8 E7 28 05DB 6F 30 05D 86 01 05E1 B6 00CA R 05E4 39		□ (3	oc 0017 R	05FC 3F 05FD 06 05FE A7 05 0600 39 0601 30 0604 A6 1D 0606 81 02	36 26 EE 20 20 20 20 20 20 20 20 20 20 20 20 20
1226 1227 1228 1229 1230 1231 1232 1233 1234 1234		1247 + 1248 1249 1250 1251 1253 + 1253 + 1255 1255 + 1255 1255 + 1255 12		1269 + 1270 + 1271	1280 1281 1283 1283 1284 1285 1286

X EFROR? NO	YES	X GET FIRST T/S OF DATA	+1, X		WRITE INDEX IMPOR		, X EFROR? VFS		WRITE INDEX SECTOR			VES YES		FIRST INDEX PUINIER=4				YES	**	r1, x	INIT, TEMP. RECNUM		FOR EACH NEW DATA RECORD	2	×. × +	INIT. TEMP. RECSIZ	+1 BOINT TO DATA FOR		WRITE NULL TO DATA			YES		COUNT DOWN RECSIZ	100 INTI RECSIZED			COUNT DOWN RECNUM	DONE?		OUTPUT INDEX BLOCK FOR NEW RECORD		
LDA A FCBSTA, X BEQ EXPSB	UMP EXPERR	LDA A FCBFTS, X	LDA B FCBFTS+1, X	LDX 0, X	WRITE	FCB 25	LDA A FCBSTA, X	DIAL LAT OF	TBA HD175	SWI	FCB 25	ENE EXP8A		LDA A #4	1715	FCB 25		BNE EXP8A	I DA A FORROD, X	<u> </u>	Œ	SIA B KNMIMP+1	THROUGH HERE FOR	•	LDA A FCBRS/, X	Œ	STA B RSZIMP		WRITE	SWI	LDA A FCBSTA, X	BNE EXP8A	LDX RSZTMP			BNE EXPYR	LDX RNMTMP		BEG EXP10		UT INDEX BLOC	LDX #RNDFCB	
06C7 A6 05 06C9 27 03	06CB 7E 05F7 R EXP8A	* 060E A6 1F EXP8B	E6 20		1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	06D6 19	06D7 A6 05	(101) 20 FU *	06DB 17	06DC 3F	0600 19	06DE A6 05 06E0 26 E9		06E2 86 04	36 7370	OAES 19				2 W	B7	00C2	* 1.00P	!	06F4 A6 2C EXP9	B7	F7 00C4 R	06FE CE 0017 K EXP9H		0702	0703 17 0704 A6 05		OZOB FE OOC3 R	60	# (070F 26 ED *	FE OOC1 R	60	0715 FF 00C1 R	*	* 0UT	071A CE 0017 R	
1410	1412 1413	1414	1416	141/		1420 +		1423	1425	1426		1429	1431	1432		1435 +	1436	1437	1438	1457	1441	1442	1444	1445	1446	144/	1449	1450	1452		1454 +	1456	1458	1459	1460	1461	1463	1464	1465	1467	1468	1469	
		UPDATE BUFFER POINTER	READ INDEX FROM FILE		ERROR?	ON	YES		INDEX TRACK=0? (END OF INDEX)	YES			cauaaa	YES				ERROR?	YES	LOOP HINTH FAND OF INDEX FOUND		NOW START ADDING NEW STUFF	MAKE 'OHTPUT'		!	BACK UP ONE BYTE IN BUFFER			GET FURWARD LINK	SAVE T/S OF OLD DATA START		OF OLD DATA RECORDS	NOW INITIALIZE DATA FCB AND MAKE NEW RECORDS	GET DRIVE NO. OF FILE		INIT. DATA STATUS	NU SPACE CUMPRESSION	"OUTPUT"	/HV 14-17-17-17	NAME SELANK INIT DOINE NO	OPEN DATA FILE		
SWI FCB 2	TSX LDX 0, X	STA PEBIND, X	SIA B FUBIND+1. A READ	SWI CCD 34	LDA B FCBSTA, X	BEG EXP7B	TBA	JMP EXPERR	TST A	BEQ EXP8	BEAD		FCB 24	BNE EXP7A		READ	UNI UNI UNI UNI UNI UNI UNI UNI UNI UNI	LDA B FCBSTA, X	BNE EXP7A	COVE AND	LAT HAS	FOUND END OF OLD INDEX, NOW START ADDING	Y TIBULE		B FCBIND+			æ	LDA A FCBFWD, X	Œ	Ø	RINTMP IS FIRST SECTOR OF OLD DATA RECORDS	INITIALIZE DATA FCB	LIDA B FCBDRV, X	LDX #RNDFCB		CLR FCBSCF, X	ĮΦ	Œ	STA A FCBNAM, X	۹ 🕳	IMS	
0676 3F 0677 02	0678 30	A7	067D E7 28 EXP7	067F 3F	0680 18 0481 F4 05	0683 27 04		7E 05F7 R		068A 27 0E	*	068C 3F	0680 18	0690 26 F3	*	!	0692 3F	0673 18 0494 FA 05	0696 26 ED		0678 ZU ED		Ç	069F 64 27	19 19 19	88	0682 82 00 0684 87 27	E7		<u>ن</u>	F7 00C6 R	* * *		*	i H	6F 05	₽ (06BB 86 FF	8		06U3 E7 09	0605	
1349 +			1354 1355		1357 + 1358 +	1359	1360 1361	1362	1363 1364	1365	1366	1368 +	1369 +	13/0	1372	1373	1374 +	13/5 +	1377	1378	13/9	1381	1382	1383	1385	1386	1387	1389	1390	1392	1393	1394 1395	1396	1397	1399	1400	1401	1402	1404	1405	1406	1408 +	

OUND THIS FCB								GET NEXT FCB IN CHAIN	L00P		SECTOR TO FILE		AT END OF DISK? YES	AT END OF DISK?	!!	FIX-UP FOR END OF DISK			autos 186 i attam			EKKUK'? NO	· !	YES	ONE MORE SECTOR IN COUNT				GET LAST 1/S		UPDATE LT, LS	MOXF / TNPLT:		FREE-SPACE SECTOR	READ FREE-SPACE SECTOR
* * FIX FCB-CHAIN TO GO AROUND THIS FCB *	PSHX SWI FCB 4		FCB 3 LDA A FCBNFB, X	١.,	SWI FCB 6		BRA EXPIOC	* EXPIOB LDX FCBNFB, X	BRA	œ	* * WRITE OUT 1 AST DATA SEC		TST FCBTRK, X BEQ EXP10D	* TST FCBSCT, X		EAFIOU LUM M FUBBHK, X LDA B FCBBAK+1, X	STA A FOBTRK, X STA B FOBSOT, X	i iii	* FXP10F TOWNS	4	19	LUM A FUBSIA, X BEG EXP106	* 1	K EXPIOR OMF EXPERK	Œ	ADD B #1	Œ	STA A FCBNMS, X	D Q	LDA B	4	STA B FCBLTS+1, X	ړ∢	LDA B #FRESEC STA A FORTRY, X	STA B FCBSCT, X IOHDR
	0764 3F	0766 3F	0767 03 0768 A6 25 0768 E4 25	27 CT W0 /2	0/65 3 7 0765 06		07.72 20 04		0776 20 E4	0778 CE 0017		!	077B 6D 0A 077D 27 04	077F 6D 0B	3 ;		0787 A7 0A 0789 E7 0B	ŝ			# ? !!!	0781 46 03 0791 27 03	i	0/93 /E 05F/		0798 E6 24 0790 CB 01	6	079E A7 23	. &	E6	A:	07A8 E7 22 07A8 6F 06	8	07AE C6 03 07B0 A7 0A	
1532 1533 1534	1535 1536 +		1540 +		1545 +	1546	1548	1549 1550	1551	1552 1553	1554 1554			1559	1562		1565 1566		1568		1571 +	1573		15/5		15/8		1581				1586			
GET T/S OF DATA RECORD	WRITE INDEX TRACK	ERROR?	YES	WRITE INDEX SECTOR		ERROR?	011		FIND POINTER		WRITE INDEX POINTER		FRUK: VES	LOOP UNTIL ALL RECORDS DONE		WILL NO DIRECTORY OF PHILE	DATA FCB ADDRESS IN (A, B)		GET HEAD OF ECB-CHAIN	; ; ;			AT THIS FCB?				ON			MAKE NEW CHAIN HEAD			AT DESIRED FCB?	ON	AT DESIRED FCB? NO
∢	O,X TE WRITE INDEX TR T	STA, X	* BNE EXPSA YES * TDA	TE WRITE INDEX SE	3 25	LDA A FCBSTA, X ERROR?		LDA PA	A FCBDBA+1, X	×	WRITE WRITE INDEX POINTER SMI	25	LUM A FCBSIA, X BNE EXPRA		NOW PATCH LINKAGES FROM NEW STUFF TO OLD	* TINS() CLUSE DHIM TILE *IID NO DIRECTORT *	EXP10 LDX #RNDFCB DATA FCB ADDRESS IN		Ü	ACTIVE FCB-CHAIN	PSHX	FCB 53	×	FCB 12	PULX	SWI FICE A	P10A	* * * * * * * * * * * * * * * * * * *		A FCBCHN MAKE NEW CHAIN	+1		XPIOA CMP A FCBNFB, X	BNE EXP10B NO	CMP B FCBNFB+1, X AT DESIRED FCB? BNE EXP10B NO
A FCBTRK, X GET T/S OF DATA B FCBSCT, X	EF 00 LDX 0, X WRITE WRITE INDEX TR	0725 19 FCB 25 0726 A6 05 LDA A FCBSTA, X	BNE EXP8A	WRITE WRITE INDEX SE	SW1 FCB 25	U.DA A FCBSTA, X	4 BNE EAFOR	17 R LDX	AO OS SUB A FCBDBA+1, X	SC LSX C. X		073C 19 FCB 25	LUM A FCBSIA, X BNE EXPRA	BRA EXP9 LOOP UNTIL ALL	NOW PATCH LINKAGES FROM NEW STUFF TO OLD	* TINS() CLUSE DHIM TILE *IID NO DIRECTORT *	LDX #RNDFCB DATA FCB ADDRESS IN	IMS	GET HEAD OF EC	* DELETE FCB FROM ACTIVE FCB-CHAIN	PSHX	OF FCB	SUBABX	+ 0740 3F SM1 + 0740 0C FCB 12		+ 0/45 37 5M1 + 0745 06 FCR 6	0750 26 0A BNE EXP10A	*	E6 26 LDA B FCBNFB+1,	97 29 STA A FCBCHN MAKE NEW CHAIN	D7 2A STA B FCBCHN+1	20 1C BRA	A1 25 EXPIOA CMP A FCBNFB, X	26 14 BNE EXP10B *	B FCBNFB+1, X EXP10B

ERROR? YES	GET T/S OF INDEX SECTOR	REWIND DATA FCB	MAKE 'INPUT' READ FIRST DATA SECTOR	ERROR? NO CLEAN STACK	MAKE COUTPUT'	UPDATE BACKWARD LINKS WRITE SECTOR	EKROR? YES 3 OLD DATA	GET LAST T/S OF DATA MAKE 'INPUT' PEAR SECTOR	ERROR? YES GET FIRST T/S OF OLD DATA	POINT TO SECTOR BUFFER UPDATE FORWARD LINKS	MAKE 'OUTPUT' WRITE SECTOR
SWI FCB 19 LDA A FCBSTA, X BNE EXP11	LDA A FCBTRK, X LDA B FCBSCT, X			FCB 19 LDA A FCBSTA, X BEG EXP11A INS	1 JMP EXPERR 1A PUL A COM FCBDTT, X		SWI FCB 19 LDA A FCBSTA, X EKROR? BNE EXPII YES * LINK NEW DATA SECTORS TO OLD DATA	LDA A FCBLTS, X LDA B FCBLTS+1, X STA A FCBTRK, X STA B FCBSCT, X CLR FCBDTT, X	SWI SWI FCB 19 LDA A FCBSTA, X BNE EXPII LDA A RINTMP+1 LDA B RINTMP+1		FCB 6 COM FCBDTT, X IOHDR SWI
+ 0803 3F + 0804 13 0805 A6 05 0807 26 16	# 1658 0809 A6 0A # 1659 0809 E6 0B	080E A6 080E A6 0810 A7 0812 A6	+	+ 0819 081A 081C 081C	081F 0822 0823	16/8 0823 EE 0/ 16/9 0827 A7 02 1680 0829 E7 03 1681 0828 CE 0017 R	+ 082E 3F + 082F 13 0830 A6 05 0832 26 EB	1687 1690 0834 A6 21 1691 0836 E6 22 1692 0838 A7 0A 1693 0836 E7 0B 1694 083C 6F 06	1695 + 083E 3F 1697 + 083F 13 1698 0840 A6 05 1699 0842 26 DB * 1700 0844 B6 00C5 R 1702 0847 F6 00C6 R	+ 084A + 084B 084C 084E 0850 + 0852	1711 + 0853 06 1712 0854 63 06 1713 1714 + 0856 3F
ERROR? YES	MAKE COUTPUT GET DRIVE NO.	LINII MANUE (U-3) 2 BYTES/TABLE ENTRY ACCESS FREE-SPACE TABLE	GET FREE T/S	POINT TO DATA SECTOR BUFFER PUT NEW T/S INTO BUFFER LIBRATE IPPOTED FREE-SPACE SECTOR	ERROR? YES	0 '0 '	WRITE INDEX TRACK=0 ERROR? YES	WRITE INDEX SECTOR=0 ERROR? YES	WRITE INDEX POINTER=O ERROR? YES	GET FIRST T/S OF DATA POINT TO INDEX FCB POINT TO SECTOR BUFFER UPDATE FORMARD LINKS	WRITE LAST INDEX SECTOR
SWI FCB 19 LDA A FCBSTA, X BNE EXPIOF	COM	AND H #\$U3 ASL A LDX #FRETAB ADDAX	SWI FCB 9 LDA A 0, X LDA B 1, X	LDX #RNDBUF STA A SECSIZ-2, X STA B SECSIZ-1, X LDX #RNDFCB		* WRITE LAST INDEX BLOCK=0, * TSX LDX 0.X		WRITE SWI FCB 25 LDA A FCBSTA, X BNE EXPIOF	* WRITE SWI FCB 25 LDA A FCBSTA, X BNE EXPIOF * RE-LINK INDEX TO DATA		STA B 1, X TSX LDX 0, X IOHDR
1593 + 07B4 3F 1594 + 07B5 13 1595 07B6 A6 05 1596 07B8 26 D9	07BA 63 07BC A6	566	+ 07C4 3F + 07C5 09 07C6 A6 00 07C8 E6 01	1608 07CA CE 0041 R 1609 07CD A7 7E 1610 07CF E7 7F 1611 07D1 CE 0017 R	+ 07b4 3F + 07b5 13 07b6 A6 05 07b8 26 B9		0700 4F + 070E 3F + 070F 19 07E0 A6 05 07E2 26 AF	+ 07E4 3F + 07E5 19 07E6 A6 05 07E8 26 A9	+ 07EA 3F + 07EB 19 07EC A6 05 07EE 26 A3	07F0 CE 0017 R 07F3 A6 1F 07F5 E6 20 07F7 EE 00 07F6 EE 07 07FC A7 00	1650 O7FE E7 01 1651 O8OO 30 1652 O8O1 EE OO 1653

```
SWI
FCB 6
RTS
                                                      EN D
1776 + 08B0 3F
1777 + 08B1 06
1778 08B2 39
1779
1780
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   DECREMENT (CLOSE WILL ADD BACK)
                                                         MAKE 'INPUT'
GET FIRST T/S OF OLD DATA
                                                                                                                                                                                                                 GET LAST T/S OF NEW DATA
                                                                                                                                                                                                                                                                                       POINT TO SECTOR BUFFER UPDATE BACKWARD LINKS
                                                                                                                                                                                                                                                                                                                                                                                                                                                                              OF SECTORS
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          POINT FILE TO END
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    TOTAL BOTH FILES
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              READ LAST SECTOR
                                                                                                                                                                                                                                                                                                                                                                          MAKE 'OUTPUT'
WRITE SECTOR
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               MAKE 'OUTPUT'
CLOSE FILE
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     MAKE 'INPUT'
                                                                                                                               READ SECTOR
                                                                                                                                                                                                                                                                                                                                                                                                                                                                              GET NO.
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          ERROR?
NO
               ERROR?
YES
                                                                                                                                                                        ERROR?
Yes
                                                                                                                                                                                                                                                                                                                                                                                                                                    ERROR?
YES
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     YES
                                                                                                                                                                                                                                                                                                                                                                                                                                                                          LDA A FCBNMS, X
LDA B FCBNMS+1, X
TSX
ADX A
ADD B FCBNMS, X
ADC A FCBNMS, X
SUB B #1
STA A #0
STA B FCBNMS, X
STA B FCBNMS, X
STA B FCBNMS+1, X
LDA A TMPTKK
LDA A TMPTKK
LDA A TMPTKK
LDA A TMPTKK
CLA FCBNTX, X
STA A FCBNMS+1, X
LDA A TMPTKK
STA B FCBNMS+1, X
LDA A TMPTKK
STA B FCBNMS+1, X
STA A FCBTTK, X
STA B FCBTTK, X
STA B FCBTTK, X
                                                                                                                                                                                                                 LDA A FCBLTS, X
LDA B FCBLTS+1, X
PSHX
                                                      CLR FCBDTT, X
LDA A RINIMP
LDA B RINIMP+1
STA A FCBTRK, X
STA B FCBSCT, X
IOHDR
            LDA A FCBSTA, X
BNE EXP11
                                                                                                                                                                        LDA A FCBSTA, X
BNE EXP11
                                                                                                                                                                                                                                                                                                                                                                                                                                    LDA A FCBSTA, X
BNE EXP11
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         LDA A FCBSTA, X
BEQ EXP12
                                                                                                                                                                                                                                                          SWI
FCB 5
LLDX FCBDBA, X
STA B 2, X
STA B 3, X
PULX
SWI
FCB 6
COM FCBDTT, X
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               COM FCBDTT, X
CLOSE
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   JMP EXPERR
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           SWI
FCB 21
PULX
                                                                                                                                                          FCB 19
  FCB 19
                                                                                                                                                                                                                                                                                                                                                                                                                      FCB 19
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              FCB 19
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                IOHDR
                                                                                                                                              SWI
                                                                                                                                                                                                                                                                                                                                                                                                        SWI
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             EXP12
                                                      085C 6F 06
085E B6 00C5 R
0861 F6 00C6 R
0864 A7 0A
0866 E7 0B
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            23
24
0007 R
0008 R
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     08A9 7E 05F7 R
1715 + 0857 13

1716 0858 A6 05

1718 0856 26 C3

1719 0856 EF 06

1720 0864 F6 00C

1721 0864 E7 0B

1723 0864 E7 0B

1724 0864 E7 0B

1725 1868 E

1725 1868 E

1726 1868 E

1726 1868 E

1727 1868 A6 05

1728 0866 A6 05

1729 0866 A6 05

1731 0866 A6 05

1732 0866 A6 05

1734 0876 E 07

1734 0878 E 07

1734 0878 E 03

1735 0874 E 07

1736 0874 E 07

1737 0878 E 03

1738 0876 A7 02

1738 0876 A7 02

1738 0876 A7 02

1738 0876 A7 02

1738 0876 E 07

1738 0876 A7 02

1738 0876 A7 03

1744 0876 A876 A7
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       089D 6F 06
089F A7 0A
08A1 E7 0B
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    EB 24
A9 23
C0 01
                                                                                                                                                                                                                                                                                                                                                                                                                                    0880 A6 05
0882 26 9B
                                                                                                                                                                                                                                                                                                                                                                                                                                                                              A6 23
E6 24
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          08A5 A6 05
08A7 27 03
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         င်
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 င်
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               08AC 63 06
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         1749 0886 E6 24
1750 0888 30
1751 0888 EE 00
1752 0889 EE 00
1753 0889 EE 00
1754 0889 EE 00
1755 0891 B2 00
1756 0893 A7 24
1756 0893 B6 00
1759 0897 B6 00
1759 0897 B6 00
1759 0897 B6 00
1750 0897 B6 00
1760 0897 B7 08
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             08AE 3F
08AF 15
                                                                                                                                                                                                                                                                                                                                                                                                                                                                              0884 A6
0886 E6
0889 E0
0889 EE
0881 A9
0881 A7
0893 A7
0895 E7
0897 E6
0897 F6
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              1772
1773 +
1774 +
```

\$0030100F6 \$12100Z00424040A2803020100104D230500000000000FFFFFFFFFFFFFFFFFFFFFFFFF	SIL101 JIEPTO/IEPT
οc	HAFEDS 03F7 R HAFEDER 0400 R HAFEDER 03DB R HAKITI 0411 R HAWRIT2 041D R HAWRIT2 041D R HAWRIT5 0472 R HAWRIT5 0472 R HAWRIT6 040A R HAWRITE 0406 R HAWRITE 0406 R HAWRITE 0406 R HAWRITE 0406 R HAWRITE 0405 R HAWRITE 0407 R HAWRITE 0407 R HAWRITE 0003F HAWRITE 0003F HAWRITE 2183 M VALUE 0003T HAWRITE 23D2 M XABX 2185 M XABX 2183 M VALUE 0003D HAWRITE 23D2 M XABX 2185 M
2462 2267 2260 0036 0036 N 0468 K 2406 Z347 D 2396	PUS1 04DC R PUS1A 04DC R PUS1A 04EA R PUS3A 0500 R PUS3B 0504 R PUS3B 0504 R PUS4B 0536 R PUS4B 0536 R PUS4C 053B R PUS5C 053B R PUS5C 053B R PUS5C 056E R PUS5C 056E R PUS5C 056E R PUS5C 056E R PUS5C 056C R PUC5C R PUC5C 056C R PUC5C R PUC5C R PUC5C R
	EXP7 067F R EXP7A 0685 R EXP8A 0668 R EXP8A 0668 R EXP8A 0666 R EXP8AND 0565 R EXPAND 0565 R EXPERS 0006 R EXBRACH 0007 R
× 65 6	CLASS 0026 CLGSE 2369 M CMEM 0037 CMEM 2318 M CR2 006B R CR3 006B R CR3 006B R CR3 006B R CR5 0140 R CR5 0140 R CR6 0140 R CR7 0176 R CR7 0176 R CR7 0176 R CR8 0162 R CR8 0176 R CR8 0162 R CR8 0163 R CR9 016 C78 R CR9 076 R C

\$12106B08613BD06ED3F06393F0530EL04EE00EL04A6004724FBA6013F06393F69 \$12106CE05EE07A60008BD06EU810P26F6860ABD06EDC6648600BD06ED5A26F83F060B \$12106EC399F0530EE04EL00EL0A37F46005724FAA701333F0639F033F05E07U6DA \$112070A46A60008BD07A85A260C860DBD07A8B1

\$12.10E15783F05CA033F12272C3F0A3F0A08083F023F0ADE203F055F03bD0026E43F37 \$12.10E310A5F03AD0020993CA5EA6LF0A5AFDEZ727ABAD0020A73F0AEF0A0F023F25 \$12.10E510A5F03AD0020993CA65EA0LF0A59A707E7088A80A70A513394A4F52ADCD \$12.10E510A5F03AD0020993CAF55ADA707E552524F520D200000000455525AE \$12.10EA74F523A2000000A0D30EL05AA652722BD15FAB70B91AA65BD15FEB70B92AA89 \$12.10EA70A5AA7038AA6A03B7A6AA70A5C3CA6853F3139CE0A59F06.8A43A7028453 \$12.10EC9AFA7038A4EA70A3F02C6AA093F0AA707F0887H0EC827ABCE0A09FF0EC9CE48 \$12.10EC9AFA7038A4EA70A3F02C6AA093F0AA707F0887H0EC827ABCE0A09FF0EC82025 \$12.20C0589F0EC9A7003F0ECE987D7A90F5E

\$1210C140h093F31DF2320168126266F0A592006810427U320BB3F13CE0A09DF234B \$1210C327F00227CD022BF20A600BBF23812027F32207810B26A162032815F97 \$1210C5023022031BD0D11858027ODC60185012702CA02BD0CB2201885A02705BD0C4D \$1210C4F8C700F85042710DF20E600C1242603BD0CD49725B725597F00267F00259E6 \$1210C8CDF23A6007C002208DF23BD0D11854026FF146225AD72C1042D037E0D07BDB1 \$1210CAA0D7CDF27C603704DDF23A6007C002208BF23BD0D1185012702C6028B92 \$1210C6E8985026E596224A97722C1052DDF23BF20BE23A6007C002208BF23BD9E \$1210CFA0D1850226EFD6725AD72C1052D007EDDD25DF27C603DE2395 \$1210CAA0D785026E7087331317E0285912048

\$1210F13250C815F2208CE09A93F09A600394F390100DE207F0D237F0D24D6220908ED \$1210F315A26FCD627RF0D6CB70D245A272909BF0D6C484848A8BA0D2457OD245A27B1

\$12.10D4F1809BDD6CB70D235A270E09BD0B6C484848BA0D23B70D23FE0D2339A678
\$12.10D4F1809BD0B6CB70D235A270E09BD0B6C484848BA0D23B70D23FE0D2339A678
\$12.10D48DB19270C380073990000000000004FF0D7AFF0D7AFF103F02FB0D78B9AC
\$12.10D48DB17B70D77F70D78085A26FC3F0E6000748FF0D7ASF03F0D78B9AC
\$12.10D490D77B70D77F70D784FC40AFE0D7ASF06097A0D7924D3FE0D7AB
\$12.10D673893F2F1655G1032704cE0R68B5F31398131DE274E00CF0F272021CE0E342041
\$12.10D651CE053E2017CE06AD2012CE0E5200DCE0E442008CE0F712003CE0E7C3F55
\$13.120D651CE0533F035F03BF23RD11175F04F43

 \$12.11.10F00000000000000CE10&D3F2C3F1E&D052&043F253F1E397F00277F0028C8 \$12.11.12B30EE0526F0&E79&F003F14&D05270139A&1B2728101271ECE11503F313034 \$12.11.14REE053F153920494AC454AC44AC20A&AE7545950F3559S050F303034 \$12.11.14REE053F153920494AC454AC450AB1840227033F15398D22B7139EBD10B7139FB4 \$12.11.1878D18B713A530E6058D10F139FA70008FF13957A13A52AE120C23F18E405FB \$12.11.11.C37CDA&ACA62AF053A13F133A5AC16AB187AAA51F5D052&28BD4&RECEFB \$12.11.11.10AD6D052716CE11F13F339205245A447494E4720445352A4F520BF2B5 \$12.11.1F1962&08052716CE11F13F3392C6245AE4144P494E4720445352A4F520BF2FB5

\$12.11.206.256.10.126F2DE20FF.139F96.22B7.13453F2FD6.25C.2F26F07C.13453F2FD6.25EF \$12.11.226C.10.124D5D6.22FB.1345F1.02FF.1345F3F55F53F3.13.13.13.15.10.28D6.E.1051 \$12.11.2446.DBD1.68FC.E106.DA6.675.66.726.70.13F055.66.63F1.15.70F0.70.64.FEF.051 \$12.11.284FFA706.8F134F06.39CE1.7947E513F0.2F055.64C4943415445204E414D450D74 \$12.11.284FFA706.3F134F06.39CE1.7947E5139204455504C4943415445204E414D450D74 \$12.11.284FFA706.3F134F06.8F3552495455504C4943415445204E414D450D74 \$12.11.284FFA706.3F134F06.8F3735737495726455504C4943415445204E414D450D74 \$12.11.2746.2545.8B104.2774767786.E006.2766.271438F51128D5D5 \$12.11.2746.7546.7440D4449524543544F5259C5

S11213F00625C1002752C13A26EE3F2FD625C162

\$1211424F87C139F3F2FD625C1012706C102Z70220D7D622FB139ECE136E3F05FE1308 \$12114A213&DCE137FFF139FCE13947F13A5E001A20025057C13A520F5EB01A900362B \$12114C0FF13A0FE139EB613A58B30A7003208FF139EFE13A008088C139E26D1CE1319 S1211448A03F053F3431313131C10227BF7F136C7F136D7D13A4270FCE0A59864CA7AE \$12114843F02CE07F73F04A707E708B613A5A7093F17A605274B81012644B6136CF6B1 \$121140F7E3F31393F1E39EE27A600812026037E15F23F05CE136E3F05C60C3F3531D7 8121140001270AC1022706CE0B6B3F3139DE20FF13A09622B7139E3F2FD625C12E2661 S1211466028650A7038654A7048613058R30B71317CE13043F31CE131A3F31CE07CDF1 \$11214FC31313127037E15F2CE136AC6208627BD

S121150BE700094426FA3F05CE07CDEF273F05C60C3F113F063F0608083F05CE0748

\$1211529CDEE27860D16BD15FA3F06A7000817BD15FEA700080883F05CE07CDEE2794 \$1211547860E16BD15FA3F06A7000817BD15FEA7000808083F05CE07CDEF27A60F168C \$1211565BD15FA3F06A7000817BD15FEA70008083F05CE07CDEE27A61016BD15FA3F34 \$121158306A7000817BD15FEA70008087F05CE07CDEE27A611168D5E3F06A7000844

\$12.115A1178D5AA70008083F05CE07CDEE27A&12168D4A5F06A70008178D42A700084E \$12.115DF08083F05CE07CDEE27EE138C3F03F08FF136C3F063&8D21A700083271 \$12.115DD8D1FA70008178D15A70008178D13A700CE13433F31CE07CD3F1A7E149644D0 \$11215F844444840F8B30813923028B0739CE4B

\$121160A106D3F0530EE073F05C61F3F1131313131CE106D3F02CE10973F04A707E7C3

\$12114820000040DCE07CDA7093F02CE07F73F04A707E7086F053F17A60527078101CB \$1211628086F053F256D05260FDE27270B3131313131313131314E000E16783F05CE4D S12116A027137E14E2EE27A6008120260139CE07CD3F1A20E5A60B811A2601396F059E \$1211646107D3F05C60C3F1131313131CE16663F313131313131313131313 81211664153920554E41424C4520544F20434841494E3A200000000000000000000000000000000 S12116BE393F05A609CE07CDA7093F02CE07F73F04A707E7086F053F17A605270B8177

\$12116DC01273B3F06A7057E14E2EE27A60081202607CE07CD3F1A20E23F0530EE023A \$11216FA8A103F093F05G6A07273131312AAE \$12117709FACE07CDEE273F023F06A7272728AF05393F06A7053930EE053B9B30EE057B \$1211777A6A05370139EE27&D02272AEE173F3F3F3130EE058A12A7053920A449C45205E \$121177A6A4454C455A452D50524F5A45440DDE2927466D06270DA6093F0530AD

\$1211763F677410927083F05E2526F87237374708703F31861230EE0567053920E7 \$121178144454C455445204552526F8723131CE17805731861230EE0567053920E7 \$121178164056E27660F641030EE05671F6720E976411E61230EE0567053920E7 \$121178DE622786206700E07CD86F657136F066D05270139609CE106D67098606 \$121178D06056036706F670866058F1366060676676E6718978709E055290 \$12117806106D0603670866058F136062646E07667E67139F500EE0552 \$1211808106D060948CE002R3F0932A700E701CE106D63063F136F06A60526A630EE34 \$12118240E106D650363F136F083F136405286EE07764139E7103252970027260646 \$1211840E106D63063F136F063F13660593PE056510326297D0027260646 \$12118628628810322046708E058615670539DE20FF139F96228713653FE76 \$121188226E63F2F0625C101270830EE058615A70539DE20FF139F96228713853FE76 \$1211888026E63F2F0625C101270830EE058615A70539DE20FF139F96228713853FE76 \$12118BC05FE139E3F053F343131315D26BF397E07CD30EE056F046F048604A70BC1 \$12118DA4F05A407E408A727E7283F134D240CEE274D00270930EE056F0539A70539C8 \$11218F830EE058401A7053930EE05A427E62850

\$1211907CB208900A727E728EE073686803F09323F0B270830EE054FA70520C630EE27 \$121192505A60B4C811B27CB7E18D830EE05EE273F0530EE0786103F093F05C6153FC5

\$1211A4227E62830EE05A727E7286F236F246F216F226F0E6F0FA609840348CE002895 \$1211A603F093F05A600E6014D27035D262730EE076F068600C603A70AE70B3F1363E9 \$1211A7E064D2705A705313139EE07A67EE67F3F063F05A700E70130EE074D27035DF1

S1211ABA63064D2705A705313139EE07A600E6013F064D260B5D2608860730EE05A7BA \$1211AD80539A700F70130EF05FF07C&7C&F04085A2&FA30FE057E19D930A&05E&067& \$121149C26078607A705313139A71FE720A70AE70B3F1B6D0527033131398F063F135F S1121AF6DE29272E3F053F0C3F062A0AA625E6CC

\$1211B230BEE2524E43F03840DA7053930EE054D042401394D0A27014bDB240AA60ECA S1211B05269729U72A2023A1752614E12626103F053F03A625E6263F06A725E72620BC

\$1211B7DE70B3F1363064D2701396609840348CE002B3F09A60660130EE05EE07A7E0 \$1211BB907R119552605F119562736E2746002A1E30EE056D292715EE274C260486F3 \$1211BD720200EA700862030A704EF056F0539EE27083F0530A706323330EE05A72742 \$1121BF5E7286F0539A60AA121260BA60BA1220A \$1211B41E60FA70AE70B200E3F13A623E624CB018900A723E724&F04BD195A30EE05A5 S1211R5F63066D05270139A60AE60BA721E7223F1B6D052701396F068600C603A70A14 \$1211B9B7EE77F30EE053F13A7053930EF056D0627058612A70539A627E628E008A246

\$1211022E728EE07A600E60130EE05A700E70DEE07A602E60330EE05A70EE70F7E1B58 S1211C40C430EE056D0426058612A70539A627E628E008A207B119552605F119562743 \$1211C5E4530A604EE056D292727847F81202621EE27E6002A0B5A2A0CE70030EE05B9 \$1211C7C6F0539C6F120F4083F0230EE05A727E72820C2EE276D0026FEA700083F020F \$1211C9A30EE05A727E7286F0539A6098403CE002B483F09A6002718E601271430EE97 \$1211CB805EE07A700E70130EE053F134D2706A705397E1D3AA623E624CB018900A709 \$1211CD623E724A60AE60BA70EF70FA6098403CE002B483F09A600E6013F0530EE07BD S1211C0426058608A70539A60CE60DA70AE70B3F134D26F0A607£608CB048900A72762

\$1211D03313139EE07A600E6013F06A700E70130EE05A70CE70DA607E608CB04890010 \$1211D3FA7053930EE056D0627058612A705396F053F156D052701393F14397FED01CB 81211D21A727E728A60FE60FE607A702E703C67C6F04085A26FA7E1C5F860730EE05B2 S1121CF4A70AE70B6F063F1363064D2705A705E5

\$1211DR705202D8580270486092025C680863CB7EC038640B7EC02B6EC0036842CB75A \$1211DD5EC038640B7EC027+EC0232A700085A26DE4F30A706EE07AA05A705393F03EF \$1211D5D7FEC037FEC077FEC0086FFR7EC02B7EC068604B7EC01B7EC07862CB7EC0316 \$1211D788680B7EC02860CBD1E4D393F03A6090C464646AAA0BE60AEE07BD1E41BD1E48 \$1211D99742402204917BD1E66C6058602BD1E4DB6EC008508270ABD1E5E5A26EE86C5

\$1211E0280460008B7EC068630R7EC025626F23332BD1E41BD1E74240220CA17BD1E48 \$1211E2066C6058604BD1E4D8606BD1E4DB6EC008508270ABD1E5E5A26E9860520A9B3 S1211E3E4F20A6BD1E5EB7EC068620B7EC023934B6EC0032B7EC02B4EC012AFBB4ECEE \$1211E5C003936860AB7EC023239B7EC068610B7EC028608BD1E4D39B6EC0084202616 SLORIEZA034F0C390D860A39EF

S1121DF3A6090C4646AAAAAABE60AEL073637C683

PERTER JOAN LIMITS: 0100 1E81

1686 16BF 1900 1930 1955 1960 1860 1866 6091 84E 2081 18CF OD7C D42 050 0025 1127 641 CVDB CLOADB 1 CCHAIN 1 SEMPTY BOTOMD YOLXNE **COPENU PUTDR** SCLOSE PWRITE RGETDR PDELET **GEMTEC** DIRECT PINIDE COMUR SOPEN PREAD PREMI SFIO CKHB CKHB UNRESOLVED UNRESOLVED UNRESOLVED UNRESOLVED UNRESOLVED UNRESOLVED UNKESOLVED UNRESOLVED UNRESOLVED UNRESOLVED UNRESOLVED UNKESOLVED UNRESOLVED 0765 0768 078F 0792 07B0 07B3 0795 0798 07B6 OZEC OMEO RUSRIO 079B 07A1 0789 0100 07C1 0704 07C1 YOM MOD RI_OADR PCL.DST **HOWYMS** eusk6 eusk7 eusri eusrz EQTAR PDTAB NILW MTOT euska RUSRB **PUSR9 @USR4** SUSRS BIOS

RDSEC WISEC

PPRTMS

209

\$12120000B7211BF7211C3F2FDE20A6009143260939CE20F13F317E207ED625C10126FB \$121201EF2D622C10326ECCE211D3F05DE203F05C6033F113131313F2FD625C13D0A \$121203C26D33F2FD625C10126CBD622C10326C5CE21203F05DE203F05C6033F1131E7 S121205A313131CE211D3F05CE21203F05C6033F123131312625CE211DBD20BB243D \$121207813CE20FE3F31CE21123F313F30DE20DF237E2006A602A700A603A70120E8AB S1212096GE2120BD20BB240220D9EE023F05CE211DBD20BB240220CB3F023F043F04B0 S12120B4A700E7017E207E3F05FE211B3F05C6033F1227233F063F0A080808083F0547 \$12120D230EE028603103F0D3F0Z30A702E7033F066D0026D931310D393F0631310CD7 S11220F03953594E544158204552524F520D49BD

S12120FF4ES6414C49442044A556494345204E414D450D41535349474E2D20040000CD S106211D000000BB

\$12120008E400F7FEC017FEC037FEC077FEC008&FFB7EC02B7EC068604B7EC01B7EC8& \$121201F07862CB7EC038680B7EC02860CBD21478603C600CE4010BD20DDCE4010A6F4 S121203C7AF67BB74090F74091A67CE67UB74092F74093CE4014FF4096B64091F640CC \$121205A90R74095F74094CE4010BD20DD8D3A8116260C8D34B7409A8D2FB7409B2055 S1212078EE810226218D24B740988D1FB740998D1AB7409C8D15FE4098A70008FF4072 S1212096987A409C26F020C9FE409A6E00FL40968C40902707Ä60008FF409639F640A5 \$121208494B64095B140932607F14092260220DACE4010E600A601F74094B740958DC6 S12120D20ACE4014A60008FF40963936860AB7EC0232B7EC068620B7EC02B6EC00844D 811220F02027037EE11317B7EC068610B7EC0226

\$121211107037E113C680843CB7EF038640B7LC02B6EC0036862CB7EC038640B7LC52 \$121213R027FEC0232A700085A26DE3936B6EC0032B7EC02B6EC012AFBB6EC0039CE7B \$12120FF86088D44C60586028D3EB6EC008508270B860AB7EC025A26ED7EE1138580C8 S10A2159215E3F3139200A29

\$12120B40000000CE20006F096F063F2FDE20A6009143260139D625C10326347D0053

\$12120622726009628810322076E200047092016CE20E43F317E2221204E554D4245B3 \$11220F052204552524F52003F2FB625C134264 \$12120FF23F2FB625C1012714C1022716CE21143F317E222120464F524D41542045E \$1217111D52524F5200DE20FF202A9622B7202C3F2FD625C12E26D87C202C3F2FB625C5 \$1212134KC1012704C10226C9D622FB202CCE20AC3F05FE202A3F053F31313131C142

S10421EF3FAC

\$12121D13F31CE223C3F313F30DE20A600815926C6CE20103F05CE2000EE27FF20AAF9 \$12121590227B0CE20006F293F17A60527288101261F6D2927037E2221CE217A3F3194 \$12121B329CE222F3F31CE2000A6098B30B72239CE22393F31CE2000EE278604A70C0F 812121777E22212046494C45204E4F5420464F554E440D3F1E7E2221EE27A600812011 \$12/2/9527/13F05CE20AC3F05C60C3F353131312707CE20003F1A20B4CE20006C21 S11221EF3F05C60C3F1131313131CE20003F1C6A

\$12121FEB620AAF620ABA727E7286D05269CCE22133F3120952046494C452044454C10 S121221C455445440DCE222F3F313F30DE20DF237E20B82044454C455A452D200400F9 S10C223A3A04203F2004111A09A2

\$121201 From the resolution of the resolution \$12120B42049AE2044524956452000203F200496288403EE2000A7098B30B720BECE15 \$12120B220AA3F313F30BE20A6008159270139CE20006F0A8601A70B3F02CE22443F10 \$11220F004A707E7088B466D05270220136C0B24 \$12120FF3F02CE22C43F04A707E7088D316D052702207A6C0B3F02CE202A3F04A70736

\$1212159ED8601A70bA70A1650C11B2609C6014C814D26024F5FB7202A378D33F7204A \$12121772B338D3A4D26045U260139A70A378U21E70B3320D50E21923F3139494E4933 \$12121955449414C49564154494F4E204641494C45440B3F050E222A3F0A09E6003F39 \$12121B30639364F6F053F13A7054D26023239168D55B72200178D53B722013F05A668 \$121211UE7083F05GE202AC67E4FA700085A26FA8601A700A7013F068B7E6D052704D6 \$12121D10B8D46B7220D460B8D43B7220E460A8D38B72217A60A8D35B72218CE21F514 S1212133B204F20766C0B7F20A87F20A98D6C6D052702203DA60B4C811A2704A70B20F1 S11221EF3F313F1F39394449534B204552524F1A

S12121FE523A000020415420534543544F522000002C20545241434B200000D4444F8 \$121221C4444840F8B30813923028B073900010A13020B14030C15040D16050E170666 S10F223A0F1807101908111A0912464366

\$12120B43F313F30CF2000D625C103262F7D0027260A96288103220A7092014CE201B \$12120B2D63F313920AE55AD424552204555254F520D3F2FD625C13A26E43F2FD62556 S121213B26B7CE20003F17A605271D81012616CE21503F31392046494C45204E4F547B \$121217720003F1A20C5CE2000EE27A60FE610CE2000A71FE720EE27A611E612CE26CD \$121219500A721E7228600C603A70AE70B3F136B0527037E2160CE20006306A61FE676 \$12120Fh4154204552524F520DDE20FF202A9622B7202C3F2FD625C12E26DA7C202C51 \$121211D3F2FU625C10126CFU622FB202CCE20103F05FE202A3F053F34313131315UDF S121215920464F554E440B3F1E39EE273F05CE20103F05C60C3F1231313131312707CF47 \$12121B320B720A4F720A5A621F627B720A6F720A73F136B0527037E216039205359B7 S12121D15S54454D2046494C45204E414D453F2004B72217A40A8D3SB72218CE21F598 S11220F0C1012714CE20FA3F313920464F524DFB

\$12122580000434F4E22DE73212330235B50545222DE23212330000050545022DE23CB \$1212276210000235B44534B22R322BF22C422D44C50542388238F0000235B4D544186 \$121229422DE23212330235B54545922DE23212330235B4E554C22DE23210000235B4C \$12122B2003F038F146D052601393F1E393F03SF1539CE20003F18E605C108260139E5 \$12122D05D26E939CE212C3F196D0526DF393F036D062624FE202AEE07260DCE23A71D

\$1212357A08&0D39CE212CEE27A700088C225827133&3F02CE212CA727E72832CE2149 \$12123752C810D278139CE23D43F31C&12CE212CE70539BD22DE8&0C20CC3920424147 \$1212393A420494E50555A0D204241A4204F555A5055540D20494C4C4547414C20499F \$121231B39CE212CZ0DC3F036D0626013986048D2F860D202BCE2000EE27A6008104E9 \$12123B14E505554204445564943450D20494C4C4547414C204F555450555420444512 \$11222EE5F31CE20008612A70539CE20003F13C2 \$12122FUA607E608A727E7286F0539FE2156FE09260DCE23BD3F31CE212C8612A7057E \$12123392608CE2000C608E70539810D270F08363F02CE2000A727E728326F05398DF9 \$12123CF564943450D20425546464552204F5645522D52554E0D204E554D424552204A S11223ED4552524F520D20464F524D41542045F8-

\$121241AFF24551622C1032623F127725FE2455C60B3F0AFF2455A60027113131310C \$121243831DE203F05FE24553F05C60320DD3131310D39313131FE24550C390004 \$12124560000CE227BFF202AFF2156CE20006F29CE212C6F096F297F27987F22597FD2 \$121247422586F058644A7028653A703864BA7048620A7108600A71D3F2FDE20A6002D \$12124929143260139D625C10326367D0027260D962881032207CE212CA7092008CED6 \$12)240E0126062028C1012708CE23F33F317E26F3BD24112517FF2156CE212E3F0596 \$11224E0FE21563F05C6033F11313131312035F2 \$12123FC52524F520D20494C4C4547414C205357495443480DDE203F05CE225A3F0584 \$12124B023E53F317E26F33F2FD625C13A26F03F2FD625C13D26067327987E2597C1C1

\$12124FBBE20FF24559&22B724573F2FB625C12E2&GA7C24573F2FB625C1012&BFB63A \$121251922FB2457CE213C3F05FE24553F055F343131315D2&A73F2FB625C12F2&FF \$12125373C3F2FBE20A&00CE212C81422&077C2259&F1D20E481432&088&03A71D&CF7 \$1212552920D8814826097C22588603A71D20CB815426068603A71D20C1CE240120D8 \$12125736ACE212C3F02FE2156EE03AD00CE712C6D052708CE239B3F317E26F3DE2026 \$1212591E600C13D2673CE227BFF202ACE20006F096F05844AA7028653A703864BA7AF \$12125AF048620A7103F2FD625C10327077D2798274520477D0027260D962881032204 S12125CD07CE2000A7092014CE23E53F31CE212C3F02FE2156EE05AD007E26F33F2F57

\$12126543F05F24553F053F34313131315102274570279827037E2894C1012799CE49 \$121267220003F02FE2024EE03AD00CE20003F1E6D052706CE23907E25D8A61D8103D2 \$12126362FD625C12E26CD7C24573F2FD625C1012704C10226BED622FB2457CE201016 \$121261820023F05FE202A3F05C6033F11313131312046DE20FF24559622B724573FCC \$12125FA2719C101272BC10226062025C1012705CE23F320C9BD24112517FF202ACE37 \$11225ERD625C13A26E43F2FD625C10D26037EFF

812126AFE605C108271C5D27033F1E4FF2156EE09AD00CE212C6D0527LC3F1ECE23E9 \$1212400987E25D83F02FE202A6E05AP003F2FU425010D26060E27137E25D89626818A \$121269027087D2258270D7E2B516C297D225927037E2D43FE2O2AEE07AD00CE200057 S11226EA0426037E25977E260ACE20003F02FE9B

\$1212753073F1ECE27CR3F31CE21583F05CE202C3F05C6803F1131313131CE212C3F33 \$1212771136D0527073F1ECE27D73F31CE20000A60A660B5CD11A26B7C6014C814D2650 \$121278FB0CE27BB3F317E26FF000A20434F50592046524F4D20445249564520002022 812126F9202AEL05AD000E270D3F313F30DE20DF237E2458205049502D0420444F4EBF \$1212717450DCE2000A6098B30R727ABCE212CA6098B30B727B6CE2799SF313F30DE09 \$121273520A600815926598600C601CF212CA70AF70BCF2006A70AF70B3F136D0527UC \$12127GB205245A144204552524F520D205752495445204552524F520D000044534BB5 \$11227E90000280E0000000000000000000000000 \$12127AD544F2044524956452000203F200420434F505920434F4D504C4554450AOD3F

\$12128CADOCE28D33F317E26D02046494C45204E4F5420464F557E440DCE28EB3F3124 S11228E87E26D0204449524543544F525920452F

812128F752524F520DEL27A600812027113F05CE288E3F05C60C3F35313131370795 \$12129150E27E43F1AZ0A00E27E46C290E2A043F310E27E4A6098B30B7ZA00EZA00CA0 \$12129333F31CE27E4EE278604A70C3F31CE2A0F3F313F30DE20A600815926C6CE2039

\$12120F31750261280302B0F81092F0A81112B0781162E038007390E2D123F3106FF08 \$1212D11392042414420484558204348415241435445520D8DCA5D26184848488FF8

\$1212D4D5326F3BD2B175D26F38130265F7F2A148DC65D26E78DC15D26E2B72A158D9D \$1212D6Ek95D26DAB72A16R62A1443B724578DAA5D26CBB124572716CE2D897E25D8E3 81212D2F2A178DBESD260CF62A171B16FB2A14F72A145F39BD2B175D27037E2E338113 \$1212D8920434845434B53554D204552524F520D8616BD2B245D27037E2E33B62A15FB \$1212DA7BD2B245U26F4B62A16BD2B245D26EB7E2D438131266F7F2A14BD2D255D2693 \$1212DC5DB8003B72A13BD2D255D2AD0B72A15BD2D255D26C7B72A168602BD2B245D01 S1122DE326BCR62A15BD2R245U2645B62A16BD7F

\$1212DFZZR245UZ&3CR&2A13RD2BZ45UZ&33BDZDZ55UZ&2DBDZBZ45UZ&277AZA132&AA \$1212E10EFB62A1443B72457b02D255D2615B124572706CE2D897E25D87E2D438139A1 S1152L2F26F97E26D0C10827F97L2CA4535A412041205B

\$121203 Control of the state

\$12120D2424552204552524F520D3F2F14625C13A24E43F2FD625C1012714CE20F23F59 \$121208425C103262F7D0027260A962881032204A7092014CE20CE3F3139204E554D37 \$11220F0313920464F524D415420455224F52E0

\$12120FH0U4E20FH202A9622B7202C3F2FU625C12E26DA7C202C3F2FD625C10126CF70 \$121213B05271D81012616GE21483F31392046494C4520464F55420464F554E440D3F62 \$12121591639E6273F05CF20103F05C60C3F1231313131707CE20003F1A20C53F2FC3 \$12121779626810427037E20EC3F2F14625C10327037E20C87D002724F8CF2006E27CF \$12121100622FB202CCF20103F05FE202A3F05F34313131315D26B7CE20003F17A638

\$11E21959628060E3F0AA7000E200063063F136F064D27037E21583920006361

S0032000IIC

\$12170003F2FDE20A6009143260139D625C1012708CE208D3F317E207FDE20EE003F59 S1212030260786FF97407E207F81462608860097407E207F8050532625UE204600810E \$12120564E260786FF97427E207F815926A9860097427E207F8D4025079628A7007E6D S1712078207FCE209A3F31CE20AB3F313F30DF20DF237E200053594E544158204552FB \$121701E2FD625C13D26ED3F2FD625C103Z742C10126498C4458261ADE20A6008148CA

\$1717096524F520D494E56414C494420534554205041524D0D5345542D20043F02CE6C \$12120B420CCA1002608E1012604FF020C390808086D0026FC0B3942530039444CCB \$12120D200364450003B5744003D4E4C003E5442003F454A0041455300434C440044DF

811220F0405700464F524D4154204552524F5207 STOTZOFFORCE

S0032000EC

STATUS

\$121201F206CA603E604A1052613E106260F3F05CE20643F313F06C6073F0A20CA3FFC \$12120003F03FF206EA600260139B72064B7206AA601B72065B7206BA602B72066B7AC \$171203C05FE206EA1052&17E10&2&13A600B720&AA601B720&BA602B720&C3F0&20CE

\$119205AD108080808080820DC202020203D202020200D00001D

NAM SWIPDK	* DISK DRIVERS FOR SOUTHWEST TECHNICAL PRODUCTS		* COMMANDS	400	EQU \$1B SEEK	FDRDC EQU \$8C READ A SECTOR FILLING FOLL \$0C LIBITE A SECTOR	211111	EQC	200	INTER ENU *8019 SECREG FOU *8014	EGU		* FCB DEFINITIONS:	:	FCBSTA EQUIS STATUS FCRDBA FOLL 7 DATA RUFFER ADDRESS	EQU 9 DRIVE#	EQU 10	FORSCI EGO 11 SECTOR#	* USER REGISTERS		EQU 6 RETURN		ENT		Z	CTRKO FOLLO	Ego	200		CURIV ENU 4		* INIT THE DISK SYSTEM	AINTOV LDA A MARC	STAR	Œ	Œ		CLR CDRIV	*	* READ A SECTOR	. RDSEC TABX GET FCBADR	IMS	LDA A FCBDRV, X GET DRIVE*	#CTRKO F	ADDAX ADD OFFSET SMI	
z																							Z	Z 2	Z																					
0000 0000				4000		0000 0080				0000 8019					0000 0005			8000 0000			9000 0000			0000 000E	0400 0000	0000 0000				0000 0004			32 70 0000	0 6	26	6	6	000A /F 0004				000E 3F	CT.	0012 CE 0000	0015 3F	
0001	0002	000	0000	9000	8000	0000	0011	0012	0013	500	9100	0017	0018	6100	0020	0022	0023	0024	0029	0027	0028	0030	0031	0032	2000	0035	9600	0037	800	0039	0041	0042	200	0045	0046	0047	0048	00400 V 00	0051	0052	0054	0055 +	0057	0028	0000	

~12	
-----	--

RY 17ED?	ACKO			FCBADR TRACK# SECTOR BUFFER ADDRESS	SECTOR	GAIN?		· Œ		GET DRIVE#	SET		'R'V	DRIVE INITIALIZED? YES						FCBADR	IRACK# SECTOR#	DATA BUFFER ADDRESS		IN?
GET ENTRY INITIALIZED? YES	SEEK TRACKO ERROR	ERROR		Ĕ <u>ê</u> êĔ	GET A S	RETRY AGAIN? YES		X=FCBADR		GET DR	ADD OFFSET		GET ENTRY	DRIVE I YES						GET FCE	GET TH	GET DAT	¥	TRY AGAIN? YES
9 B 0, X B #\$FF RDSEC1	RESTOR QUITIO	DRIVE QUIT10	A #5 A RCNT	UXH, X A FCBTRK, X B FCBSCT, X FCBDBA, X	eREAD QUIT10	RCNT RDSEC2 QUIT	A SECTOR		œ		×	0	B 0, X	B #\$FF WTSEC1	RESTOR	QUITIO	DRIVE QUITIO	A 4.00		UXH, X	A FCBTKK, X GET TRACK# B FCBSCT, X GET SECTOR	FCBDBA, X	ewrite Quitto	RCNT WTSEC2
LEDA CMP	JSR BCS	JSR BCS	LDA	E P P P P P P P P P P P P P P P P P P P	JSR BEQ	DEC BNE BRA	A S	TABX	SWI	4	•	SE SE	d l	S S	SR	BCS	JSR BCS	LDA STA	<u>.</u>	TSX	99	ĽĎ	JSR BEG	DEC
,	*	* RDSEC1	*	* RDSEC2	*	* *	* WRITE	. WTSEC							*	*	WTSEC1	*	*	WTSEC2		,	* :	*
	Œ	œ			œ										Œ		œ						Œ	
5 09 E6 00 C1 FF 26 05	0152 53	0127 4E	0 0 0 0	07 08 08 07	007F 3C	0005 ED 30			<u>ս</u> Ր	A6 09	}	டம		F 8		21	0127 1C	S 5				02	OOAF OA	0005 EU
26 26 26	BD 25	8D 25	98	8 H 8 H 8	BD 27	26 26 20			₩ 6	8 1	3	۳ S	8	C1		22	BD 25	86	`	8 3	8 3	Н	8D 27	7.6 26
0016 0017 B 0019 C	001D 0020	0022 0025	0027	002E 002E 002E 0030	0034 0037	0039 0035 003E			0040 3F	0042		0047		004B	004F	0052	0054	0059	9	005D 005E	0060 0062	0064	6900 9900	006B 006E
0061 + 0062 0063 0064	0065 0066 0067	0068	0071 0072 0073	0074 0075 0076 0077 0078	0080 0081 0082	0083 0085 0085	0088	0600 6800	0091 +	8600 8600	0095	0096 +		0099 0100	0101	0103	0105	0107 0108	0110	0111	0113 0114	0115	0116 0117 0118	0117 0120 0121

OK? YES ERROR STORE RC IN USER A-REG GET FCBADR X	RK, B=SECTOR) READ DELAY 128 BYTES/SECTOR DATA REG FULL? YES	A BYTE IN BUFFEK SAIN	EG BUSY? YES (A=TRK, B=SECTOR, X=DATA BUFFEK) EG 128 BYTES/SECTOR FG DREG EMPTY? YES
A QUITIO A #5 A UA, X UXH, X A FCBSTA, A FCBSTA,	SECTOR (A=TRK, R SEEK A A #FDRDC RE R DEL30U DE R B #128 12 A CMDREG A A CMDREG T A #\$02 DA E READ2 YE	A #1 READ1 READ3 A DATREG A O, X B READ1	A #*1C A #1 WBUSY WBUSY WBUSY SECTOR (A- SEEK A #FDWRC A #FDWRC A #TDWRC B #128 A CMUREG A #2 WRITE2
TST BEG LDA TSX STA CRA STA STA STA	A SE USR STA USR LDA LDA BIT		RT SE BRITA STATES CONTROLLED A A COUSTA BRITA B
* QUIT * * QUIT10		* * * * * * *	READS WBUSY WBUSY WRITE WRITE WRITEI
	8 8 8 8 8	8	81 86 81 8 87 81 8 8 8
02 05 05 05 05	0008 8018 8018 00F5 80 80		8018 9018 9018 90055 800 8018
27 27 30 30 A7 A7 A7 39	86 87 87 87 87 87 87 87 87		26 26 39 39 39 25 26 26 26 26 26 26 26 26 26 26 26 26 26
0070 0071 0073 0075 0076 0076	007F 0082 0084 0087 0086 0086 008F	0093 0095 0097 0096 0096 0096 0096	0044 0045 0045 0045 0046 0087 0087 0087 0087 0087
0122 0123 0124 0125 0127 0129 0132 0133	0136 0137 0138 0140 0141 0144 0145	0153 0157 0157 0157 0157 0158 0158	0163 0163 0164 0165 0165 0167 0170 0173 0173 0173 0174 0175 0177 0178

13

GET CURRENT TRACK SAVE IN TABLE INIT REGISTER NEW CURRENT DRIVE

DELAY A SECOND

JSR DEL1S

READY? YES

MOTOR LDA B CMDREG AND B #\$80 BEQ MOTOR1

* * TURN ON MOTORS (A=UNIT#)

SWI FCB 6 RTS EKROR CODE

LDA A #10 SEC RTS * DRIVE SELECT (A=DRIVE#)

* MOTOR1 CLC RTS AND A #3 PSH A BSR DSEL1 BSR MOTOR BCS DRIV1

DRIVE

READY? YES

LDA B CMDREG AND B #*80 BEQ MOTOR1 GET CURRENT TRK

LDA B TRKREG STA B 0,X STA A DRVREG STA A CDRIV BSR DSEL1 LDA A 0,X STA A TKKEG POINT TO TABLE

* DSEL1 LDX #CTRKO LDA B CDRIV BEQ DSEL3

DRIVI

INX DEC B BNE DSEL2

* DSEL2 RESTORE COMMAND

LDA A #FDRSC STA A CMDREG JSR DEL30U

ERROR

RESTOR PSH A JSR DRIVE BCS RESTR1

* RESTORE SEEK TRACKO (X=CTRKx)

* DSEL3 RTS

SERIOR CO245 + COTOL CO245 + CO2		* :	* *	ž		*	*				*			*	ž		*	* :	× 2	i			*							_	ä		*	S		*	, D			*	S	* :	* *			*		
COCTO 20 OCT 2				F6	27			F6	7	27	į	8							40	8	80	8	C	F 6	E7	B7	97	80	1 10 1 10	BD 00F5	32			IJ ;	10	4								36	BD 0127	C7	86	B7
NOTE SE S		0247	0248	0250	0251	0253	0254	0256	0257	0258	0259	0260	0262	0263	0264	0265	0266	/970	0269	0270	0271	0272	0273	0275	0276	0277	0278	0279	0280	0282	0283	0284	0285	0286	0787	0289	0230	0291	0292	0293	0294	0295	0220	0298	0299	0301	0302	0303
NOTE SE S																																																
NOTE SE S									တ																																							
0005 26 F5	BUSY? YES	ERROR	SET A BYTE			OO AGAIN	WAIT FOR BUSY						ON TRACK?	ÆS		JO, STORE TRACK#	2000	SEEK CUMMHNU		WAIT FOR BUSY	1	SET SECTOR																										
0005 26 53 0007 20 00 0008 B7 801B 0005 56 6A 0005 56 6A 0005 84 50 0005 84 50 0005 84 50 0005 84 50 0005 84 50 0005 84 50 0005 80 80 80 80 0006 80 80 80 80 0006 80 80 80 0006 80 0006 80 0006 80 80 0006 80 80 0006 80 80 0006 80 80 0006 80	H #1 WRITE1	WRITE3	ν, ο Υ		æ	WRITE1					4 14 1	CK IN A-KEU				A DATREG	DELSOU	A CMUDES	DELSOU	WBUSY		B SECREG	DELGA		NSECS										SECOND	- Second			ın O		A #2	0000#		DEL 1SA		DFI 15A	-	Œ
0005 26 53 0007 20 00 0008 B7 801B 0005 56 6A 0005 56 6A 0005 84 50 0005 84 50 0005 84 50 0005 84 50 0005 84 50 0005 84 50 0005 80 80 80 80 0006 80 80 80 80 0006 80 80 80 0006 80 0006 80 0006 80 80 0006 80 80 0006 80 80 0006 80 80 0006 80	BNE	BRA		STA	DEC	BNE	JSR			RTS		EEK - KA				STA	¥ .	T O	SS	SR			RTS		ELAY 30		NI DOE	TEX T	ž į	X X	DEX	INX	X I	RIS	INC VOIS				FC	PSH	HO!	Š			i i	B E	1	PUL 1
0005 25 51 61 60 60 60 60 60 60 60 60 60 60 60 60 60	*	×	¥ 3			,		*	ERI I	,	* 1	* *	SEE		*		r		æ	œ	* (*	*	* 1	DEI:							*	* *	5 : *	DEL					,	140		*		*	:
00000000000000000000000000000000000000	5 f2	ပွ	00	801B		ΕΑ			20				8019	11				9010			3	501A	7																		2	0000		Ü.		₫.	.	
	25 25 26 26 27	20	98	B7	14 54	30 26			84				B1	27		B7	<u>.</u>	0 0	BD	BD	1	74	4 39								60 H	80 g	60 0	88				FE 3F	FF 05	36	88	Ü		56		4 5 1 4	1	C 32
	3 g	000	00	o o	Š	100	100		io C	Too			000	100		ioc !	000	200	00	OOE		000	S			1	000	5 6	000	00	00F	00F	00 P	Ė						010	010	010	010	010	(0 0		010

```
K K E E E E K E
                                                                                                                                                                                                                                                                                                                                        ααΣαααααΣ
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             Σ
2454
2250A
22151
21151
22157
2406
00075
00075
00028
00028
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00088
00
PRTERR PRTERR PRTMSG PRTMSG PSHALL PSHALL PHULLAL PHUL
  ΣΣ
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               αααΣΣαα
                                                                                                                                                                                                                                                                                                                                                                                                                                    ΣΣ
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            ~~~
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              ETTTTTTTCCCTTTTTT
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         00108
00075
2524
01127
01127
01131
00009
00009
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
00008
0
0000E
0000E
0000F
0000F
0000F
0000E
0000E
0000E
RUSEC WITDK GENERAL ADDRAY ADDRAY ADDRAY ADDRAY ADDRAY ADDRAY ADDRAY ADDRAY ADDRAY CHAIN C
                                                                 CTRKx: =00
                                JSR WBUSY
CLR O.X
CLC
                                                                                                                                                               Φ
                                                                                                                                                      PUL
RTS
                                                                                                                                                                                                                                               S
                                                                                                                                  *
RESTR1
                                œ
```

BD 00A7 6F 00 0C

0160 0163 0165 38

0166

1	READ LINK SECTOR GET FIRST T/S	0/1 10/1 130	051 173	_	INIT BUEFER INDEX		INIT, PRESENT T/S		READ FIRST SECTOR	2 2 2	LE INTO MERIORY	GET A DATA BYTE FROM FILE	TRANSFER-ADDRESS?			GET TRANSFER ADDRESS		GET NEW DATA FRAME		NO NO			GET HUURESS		SELECTION SECTION		GET DATA BYTE	STORE BYTE		OUNT THIS			GET NEW DATA FRAME	GET TRANSFER ADDRESS	GO THERE	BYTE FROM SYSTEM FILE IN 'A' REGISTER		28 NEED NEW SECTOR? YES
Note	#22 # ∢	B 内 B O	t m <			4 CE CE	Œ			Maraya GAO	2131513								2	N C		Ö <	ιö															CPX BEQ
SALPE CP/68 BOOTSTRAP PROGRAM SALPE CP/68 BOOTSTRAP PROGRAM SALPE CP/68 BOOTSTRAP PROGRAM SALPE SYSTEM FILE LINNED AG FOLLOWS: TRACK O, SECTOR I, BYTE 27-FIRST TRACK 123-FIRST SECTOR 123-FI	CE 0010 BD 00CB CE 0010 A6 7A	E6 7B B7 0090 F7 0091 64 7C	E6 7D	F7 0093	FF 0096	B6 0091 F6 0090	B7 0095	F7 0094 CE 0010	BD OOCB	* *	* *	8D 3A	81	}	81) 34	B7 009C	B7 009D	20 EE	000	26 21		8D 24	8D 1F	B7 0099	8D 1A R7 009F	1	8D 15 FF 0098 C	A7 00	08	74 0095	26 F0	00	2 0 0 2	FE 009C C	9 E 00		*	FE 0096 C 8C 0090 C 27 07
NAME	0061 0062 0063 0064	0065 0066 0067 0068	0069	0071	0073	0075	9200	0077	6700	0080	0082	0083	\$ 100 000 100 100 100 100 100 100 100 100	9800	0087	8800	0600	0061	0092	0094	0095	9600	8600	6600	0100	0102	0103	0105	0106	0108	0109	0110	0112	0113	0114	0117	0119	0120 0121 0122
-6 666 66666 555555555	N NAM BOOT * SWIPC CP/68 BOOTSTRAP PROGRAM * ASSUMES SYSTEM FILE LINKED AS FOLI	TRACK O, SECTOR 1, BYTE 122-FIRST 123-FIRST 124-LAST T	125-LAST SECTOR 126.7 FREE-SPACE	BOOTS SYSTEM FROM DRIVE 0:		DEFINE DISK-DRIVE INTERFACE ADDRES	OOOB FURSC EQU \$OB	001B FDSKI EQU \$1B SEEK 008C FDRDC ERU \$8C READ A	* 6	8014 DRVREG EQU \$8014 DRIVE REGISTER 8018 CMDREG EQU \$8018 COMMAND REGISTE	8019 TRKREG EQU \$8019 TRACK REGISTER	801A SECREG EQU \$801A	4 ALCON 022 CONTROL *	NOTE: ALL VARIABLES IN COMMON, CODE IS	* 70000 * 70000	0010	O090 C CMN	0092 C CMN	0096 C CMN	O098 C CMN	COOP C CMN	COSE C CAN FONT. 1	OO9F C CMN		* *	7E E113 ERROR	* BEGIN BOOT HERE	* 0 0000	SE OVOT U STAKE LUS #STAUK+15 INIT. STAUK POI 86 01 SECTOR=1	C6 01 LDA B #1	BD 013F R USR DRIVE SETUP DRIVE 0	65 UB LIDH H #FINSC ISSUE KESTUKE B7 8018 STA A CMDREG	BD 0136 R JSR DEL30U 30 USEC	B6 8018 STARTZ LDA A CMDREG	26 F9 BNE START2	* NOW GET SYSTEM LINK INFORMATION	*	86 O1 LDA A #1 C6 OO LDA B #O TRACK O,

GET A BYTE SAVE IT COUNT DOWN DO AGAIN BUSY?	FF STATUS	FAILED 5 TIMES, ERROR CLEAN STACK GOOD READ , SECTOR IN "A"	YES NO 30 USEC WAIT SEEK COMMAND 30 USEC WAIT BUSY? WAIT UNTIL DONE	υ
2 LDA A DATREG STA A O.X INX DEC B BNE READ1 3 LDA A CMUREG BNT A #\$01 BNE READ3	4 AND A #\$1C BEG READS DEC RCNT BNE RDSEC2	JMP 5 PUL RTS EK TRAC	BEG SEEK2 STA B DATREG BSR DEL30U LDA B #FUSKI STA B CMDREG BSR DEL30U 11 LDA B CMDREG BSR DEL30U BSR DEL30U BIT B #\$01 BNE SEEK1	STA
**************************************	* 010C 84 1C	.9.*gr ***g .9.*gr ***g	27 13 F7 801B 8D 11 C6 1B F7 8018 F6 8018 C5 01 26 F9	26 F9 ** B7 801A SE 08 ** 08 09 09 08 09 08 10 08 1
0185 0186 0187 0188 0189 0191 0192	0195 0196 0197 0198 0199 0200	0202 0203 0204 0204 0205 0207 0208 0209	0211 0212 0213 0213 0214 0215 0217 0219 0219	0,220 0,221 0,222 0,224 0,225 0,227 0,228 0,231 0,231 0,232 0,233 0,233 0,235 0,235 0,235 0,235 0,237 0,237 0,237 0,237 0,237 0,237 0,237 0,237 0,237 0,237
GET BYTE MOVE POINTER CHECK FOR LAST SECTOR	NOT LAST EDF-GO TO TRANSFER ADDRESS	GET FORWARD T/S LINK UPDATE PRESENT T/S READ NEW SECTOR GET DATA BYTE RE-INIT. INDEX	CTOR	SAVE SECTOR SAVE TRACK SAVE TRACK SETUP DRIVE O RESTORE BUFFER POINTER FIVE RETRIES GET TRACK GET SECTOR GET BUFFER POINTER PUT BACK ONTO STACK READ COMMAND 30 USEC WAIT SECTOR SIZE=128 BYTES REG. FULL? YES BUSY? YES ERROR
LDA A O, X INX STX INDEX RTS C LDA B PTS CMA A PTS+1 CMP A CTS+1 DAR GETS+1	BRA E		SIS SIE-SE DRIV TRAC SECT BUFF PSH	PSH A PSH B SH B STAVEX2 JSR DRIVE LDX SAVEX2 LDA A #5 STA A RCNT STA A RCNT PSH B DSH A PSH B S3 JSR SEK LDA A #FDRDC STA A CMDREG JSR DEL30U LDA B #128 LDA A #502 BNE READ2 BNE READ2 BNE READ3
* 0099 A6 00 009B 08 009C FF 0096 C 009F 39 * 00A0 F6 0094 C GETSEC 00A3 B6 0095 C	20 DA	00B2 CE 0010 C GETSZ 00B3 E6 00 00B7 A6 01 00BC B7 0094 C 00BC B7 0095 C 00BC B0 04 00C1 CE 0014 C 00C4 A6 00	* * * * * * * * * * * * * * * * * * * *	00CB 36
0123 0125 0125 0127 0128 0130	0132 0133 0135 0138	0139 0140 0142 0143 0144 0145	0148 0150 0150 0153 0153 0154 0156	0159 0160 0160 0163 0164 0165 0167 0173 0173 0173 0173 0173 0173 0173 017

AUDRES 0.09C C
BOOTT 0.005 RN
BOOTT 0.005 R
BUILDOUT 0.005 R
BUILDOUT 0.006 C
CMDREG 8018
DATREG 8018
DRIVE 0.136 R
DRIVE 0.136 R
DRIVE 0.136 R
DRIVE 0.006 C
CHOOL R
CHORD 0.006 C
CHORD 0.006 C
CHOOL R
CHORD 0.006 C
CHORD 0.00

```
DRIVE=0
30 USEC WAIT
                                           NO, ERROR
                            READY?
YES
                       B CMDREG
B #$80
DRIVE1
                                                    DRVREG
DEL30U
    LDX #0000
INX
BNE DRV1
                                          ERROR
                                                    CLR BRA
                       LDA
AND
BEQ
                                       *
DRIVEO JMP
                                                    DRIVE
         DRV1
                                           œ
    0000
                       8018
80
03
                                                    8014
DS
                                          7E 0000
   014C CE
014F 08
0150 26 F
                       F6
C4
27
                                                   0152
0155
0157
                                          0159
                                                    015C
015F
0246
0247
0248
0249
0250
0253
0253
0254
0255
0256
0256
0256
```

 $\circ \overset{\mathbf{Z}}{\mathbf{Z}} \propto \mathbf{Z} \propto \mathbf{Z} \propto \mathbf{C}$

+ 000A0 0035 EMEM EQU \$35 END OF + 000A0 0037 CMEM EQU \$35 NEXT PA + 000A0 0039 BS EQU \$39 BACKSPN + 000A0 0030 DP EQU \$38 DEPTH; + 000A0 003C DPCNT EQU \$38 DEPTH; + 000A0 003C DPCNT EQU \$30 MIDTH; + 000A0 003C NL EQU \$35 NULL C + 000A0 003F TB EQU \$3F TAB CH + 000A0 0041 EJ EQU \$41 EJECT + 000A0 0041 EJ EQU \$41 EJECT + 000A0 0041 EJ EQU \$41 EJECT + 000A0 0042 PS EQU \$42 PAUSE;	+ 00AA 0043 + 00AA 0044 + 00AA 0045 + 00AA 0046 00AB 0001 D 00BE 0001 D 00C2 04 **	March Marc	0099 00DA A6 00 LDA A 0, X GET FIRST CHAR. OF RESPONSE 0100 00DC 81 59 CMP A #*Y WAS IT 'YES'? 0101 00DE 27 01 BEQ INITK2 IF SO, CONTINUE 0103 00E0 39 ** RTS IF NOT, QUIT 0104 00E1 CE 0000 R INITK2 LDX #FCBSPC POINT TO FCB 0105 00E4 6F 0A LDA A #1 0108 00E8 A7 0B ** INITIALIZE HEAD OF FREE-SPACE BLOCK 0110	0113 * TXAB 0114 + OOEA 3F SWI 0115 + OOEB 02 FCB 2 0116 + OOEC CE OO2A R LDX #BUFFER 0119 + OOEF 3F SWI 0120 + OOF0 04 FCB 4 0121 OOF1 A7 07 STA A FCBDBA, X
INITIALIZE A DISK FOR CP-68 OPERATING SYSTEM FOR SWIPC 5 INCH FLOPPY DISKS TRACK 0, SECTOR 1 BOOTSTRAP TRACK 0, SECTOR 1 HEADER OF FREE-SPACE LIST TRACK 0, SECTORS 2-18 DIRECTORY SPACE TRACKS 1-35 FREE-SPACE	### 6080 ### 8 128 BYTES PER SECTOR ### 8 128 BYTES PER SECTOR 18 18 18 18 18 18 18 1	0006 FCBDTT EQU 6 DATA TRANSFER TYPE 0007 FCBDTT EQU 6 DATA BUFFER ADDRESS 0009 FCBDRV EQU 7 DRIVE NUMBER 0000 FCBRK EQU 10 TRACK NUMBER 0000 FCBFWD EQU 11 SECTOR NUMBER 0010 FCBRME EQU 14 BACK LINK TRACK/SECTOR 0010 FCBRME EQU 14 BACK LINK TRACK/SECTOR 0010 FCBRME EQU 14 FILE TYPE 0010 FCBRAPE EQU 30 FILE TYPE 001 FCBRTYP EQU 29 FILE TYPE 002 FCBFTS EQU 30 FIRST TRACK/SECTOR 002 FCBFTS EQU 31 FIRST TRACK/SECTOR 002 FCBFTS EQU 33 LAST TRACK/SECTOR 002 FCBFTS EQU 33 LAST TRACK/SECTOR 002 FCBNMS EQU 35 NUMBER OF SECTORS 002 FCBNMS EQU 35 INDEX INTO DATA BUFFER	0.002 FCBSCF EQU 41 SPACE COMPRESSION FLAG *	0.022 DESCRIPTOR COUNT 0.023 CUCHAR EQU #22 DESCRIPTOR COUNT 0.025 RC EQU #22 TOKEN RETURN CODE 0.026 CLASS EQU #26 TOKEN CLASS 0.027 VALUE EQU #27 BIN VALUERTRANSFER ADDRESS (2) 0.027 FCBCHN EQU #27 TOP OF FCB CHAIN (2) 0.028 FKETAB EQU #28 DISK FREE SPACE POINTER (8) 0.038 BMEM EQU #33 START OF TRANSIENT AREA(2)
	++++	++++++++++++	+ +	0053 + 0064 0055 + 0064 0055 + 0064 0057 + 0064 0057 + 0064 0059 + 0064

013A	013C 013E	013F	1410	0143	0144	0145	0148	0149	014E	014F	0151	7070	0154	0155	0157		0158	0158	0150	015F				0162		0166	/910	0168	01710				01/E 017F	0180			0186		018h
0183 0184	0185	0187	0189	0190	0191	0193	0194	0195	0197	0198	0199	0201	0202	0203	0205	0206	0207	0209	0210	0211	0213	0214	0216	0217	0219 +	0220 +	0222	0223	0224	0226	0227		0230 +	0232	0234 +	0235 +	0237	0238 0239 + 0240 +	0241 0242 0243
STA B FCBDBA+1, X PSHX	SWI FCB 5	CLEAR ONT RUFFER EXCEPT FOR LAST 2 RYTES		LDX #BUFFER	B #SECSIA			DEC B BNE INITES)	Œ ¢	STA A (), X	:		ROB WATER K WRITE BLOCK 3	×	BEQ *+6 OK	BRA INITO FATAL DISK FRENR. DULT		WKTELK	INC FUBSCLY SECTOR#4 CLR BUFFER+SECS17-2		INITIALIZE DIRECTORY TO ZERO		INITR4 BSR WRIBLK - WRITE DIRECTORY BLOCK IST FCRSTA.X CHFCK FOR DISK FRROR	*+*	RRA INITO FATAL DISK ERROR. GLITT		A FCBSCT, X	CMP A #TKKSIZ DONE WITH TRACK?	INITRS YES	STA A FCBSCT, X	BRA INITK4 NO, CONTINUE WRITING		STA A FCBSCT, X SECTOR=1	Y CWIND Y	INITIALIZE REST OF DISK (FREE-SPACE)		AFTCB HUNRES AFTRACK NUMBER BFSECTOR NUMBER	* INITRS INC B MAKE SECTOR LINKAGE CMP B #TRKSIZ+1 END OF TRACK?
OOF3 E7 08	00F5 3F 00F6 05	* *		00F7 CE 002A R	و لا	A7 00		0100 SA 0101 26 FA	ì	88	0105 A7 00 0107 A7 01	ì	0109 3F	0106 06 0108 80 7F	010D 6D 05	010F 27 04	* ************************************	i	20 76	9	7F 00A9 R	* *		011D 8D 6C 1	27 02	* 0123 20 3D	Á	0125 A6 0B		27 04	**************************************		86 01	0132 A7 0B	£ 5	* *	*	* * * :	0137 5C 1 0138 C1 13
	0124 + 0125 +	0126		0129			0133				0138			0142 +		_	0146				_	0103		0156 (0159		0162			0166	0168	-	0171		0174 0175	0176	0178	0180 0181 0182

INITR7 NO		#1 YES, SECTOR=1	NEXT TRACK	SIZ+1 END	INITR7 NO	LAST SECTOR POINTS TO 0,0				DETSC GET PREC	FER+1	RESTOR		A DONE? (=0) INITES NO		B DONE? (=0)		YES, DONE!!!	ECBTRK, x	SAVE LSEC		FUBSUL, X			ERROR MESSAGE		MSG OUTPUT ERROR MESSAGE		49 DETRICK TO CLI		'INITIALIZATION FAILED'	3	C TO PSEC	F.G	SAVE X-REGISTER		BL POINT TO LOGICAL/PHYSICAL TABLE	ADD LOGICAL OFFSET		SECTOR STARTS AT 1		RESTURE X-REG			SECTOR WITH EKROR CHECKING
BNE IN					BNE IN		CLR B		T 0				_	ENT IN		TST BE		RTS	STA A	œ	8	<u> </u>			ERROR		LDX #QMSG PRTMSG	IMS	7 CB 4	2	FCC /IN			B-REG	PSHX	SWI	LDX #TBL	ш	SWI FOR 10	3	α	Y IN	FCB 6	'n	
æ	*		I	ပ	α *		O		o Alini	LOC	iΩ	ā.	Δ (1	<u>~</u> ac	*	₽ ₽	*	iz ,	NITES		ă i	νŒ	ă	* *	FATAL		INITQ P		à	*	QMSG FC	Ĭ *	CONVE	* LSEC IN	GETSC PS	U) [L	- 🖰	ы	ם נט	DEX	LDA	2 0,	ш.		* WRITE A
									r.		œ																۰. ۵ <u>د</u>			-	_	•	T	~ ~	_		œ							*	*
60		o I		23	07			000	H700	33	002B		30	04		5	•		θo		21	90	55				0168										OIFF				00				
26		S	ĄÇ	8	56	4F	i N	2	7 2	8 6	F7	8	8	4F) 	3 2	ì	33	67	37	8	3 6	20				Ä	<u></u> е		;	946	ą				^ω 8	GE	į	ž Q	8	E.6	æ	င် ဇ	<u>,</u>	
0136		0130	013E	013F	0141	0143	0144	4	0140					0152		0154		0157	0158			0150					0162	0165	0167		0168				į	017E	0180	,	0183	0185	0186 E	0188	0189		
0183	0184	0185	0186	0187	0188	0160	0191	0192	0194	0195	9610	0197	0198	0700	0201	0202	0204	0205	0202	0208	0209	0210	0212	0213	0215	0216	0217	0219 +	0220 +	0222	0223	0225	0226	0227	0229	0230 +			0235 +		0237	0239 +	+		0243

86888888888		: : : : : : : : : : : : : : : : : : : :	282882		
LAG	DIGIT . DIGIT	O. HEX). HEX		
A' ERROR FLAG I/O REQUEST	LEFT RIGHT	CTOR N	ACK NC RROR P	I HERE 10HT 110HT	1
SAVE 'A' CLEAR ER ISSUE I/ ERROR?	RESTORE 'A' CONVERT LEFT DIGIT	SAVE X MAKE SECTOR NO.	MAKE TRACK NO. HEX PRINT ERROR MESSAGE CALL CP/68 "WARMSTART"	EKTYPE RMB 2 EKTYPE RMB 2 FCC ' AT SECTOR ' SECT RMB 2 FCC ', TRACK ' TRACK RMB 2 FCB % D * CONVERT BINARY TO HEX-ASCII HERE * CONVERT BINARY TO HEX-ASCII HERE * LSR A	₹ %
×	- N - N - N - N - N - N - N - N - N - N	YPE+1 SCT, X T T+1 TRK, X	× + × ×	2 AT SECTOR (2 AT SECTOR (2 AT SECTOR (3 A AT SECTOR (4 AT	
PSH A CLR A CLR FCBSTA, X IOHUR FCB 19 STA A FCBSTA, X TST A		A ERTYPE+1 K S 3 5 OUTHL A SECT OUTHR OUTHR A SECT+1 A SECT+1 A SECT+1	SR OUTHL TA A TRACK DA A TRACK DA A CETRK, X DA A TRACK+1 TA A TRACK+1 DX #DERROR RTMSG RTMSG RTMSG WI WI TS TS	2 2 1 SK 2 2 1 SK 3 1 SK	## d
	PUL RTS TAB BSR STA TBA BSR BSR	STAN PSHA SWHX CLDA BSR C STA C BSR C STA C STA C STA C	BSR OU STA A LDA A LDA A BSR OU STA A SWI SWI SWI SWI SWI SWI FCB 4	FCC RMB	
* WRTBLK	* WRTEKR		*	DERROR ERTYPE SECT TRACK * CONVI OUTHL	
	oc Io		α αα 0 οσ		
8 5 8	54 01D5 52	0106 0106 0162 0162 0163 008	37 01EC 0A 34 01EB 01CA	9 9 8 E	36
36 4F 6F 6F 3F 13 13 10	32 33 33 14 17 17 80 80 80 80 80	2	8D 8D 8D 8D 8D 8D 8D 8D 8D 8D 8D 8D 8D 8	20002 20002 20002 20002 444 444 444 444	
0188 36 018C 4F 018D 6F 018F 3F 0191 A7 0193 4D	0196 0197 0198 0199 0198 0198		0186 8D 0188 B7 0188 B7 0188 B7 0186 B7 0167 GE 0167 3F 0167 3F	0105 0107 0107 0162 0162 0166 0167 0173	01F3
0244 0245 0246 0246 0247 0248 0250 0251 0252	0255 0255 0257 0258 0258 0250 0261	0263 0264 0265 + 0265 + 0267 0269 0270 0271	02/4 0275 0276 0277 0279 0281 0281 0283 0284 0283	0.287 0.289 0.290 0.291 0.293 0.294 0.295 0.297 0.297 0.300 0.301	0304

					LE																									PROGRAM STARTS HERE			
					TAB																									ĕ			
Q N	YES				SECTOR																									BOOT			
BLS *+4	ADD A #\$7	RTS			LOGICAL/PHYSICAL SECTOR TABLE		FCB 00		FCB \$1				-	FCB #8		FCB \$12				FCB #2						FCB SE				EQU *		!	END
,	k	*	*	*	* LOGI			*	TBL																		*	*		R BOOT	*	*	
02	07																													_			
23 (9B	39					င္ပ		5	90	9	10	ဗ္ဗ	æ	g	12	05	Ğ	Ŕ	02	07	ဗ	11	Š	60	Ä				0211			
01F9	O1FB	01FD					OIFE		OTEF	0200	0201	0202	_	0204	0205										020F	0210				0211			
305	305	800 000 000	310	311	312	313	314	315	316	317	318	319	320	321	322	323	324	325	326	327	328	329	330	331	332	933	334	335	936	337	338	336	040

TTER	ш-	FOR CP/68 SYSTEM35 TRACKS, 18 SECTORS/TRACK	COPYRIGHT: 1979 HEMENWAY ASSOCIATES, BOSTON MASS.	FSTOB 15H CP/69 BASEBAGE	DESCRIPTOR ADDRESS(2)		CURRENT CHAR (2)	TOKEN RETURN CODE		BIN VALUE/TRANSFER ADDRESS (2)	TUP UF FCB CHAIN (2)	DIOK TREE SPECE FUINIER (8)	STAKE OF EKANSTENT AKER(Z)		NEXT AVAIL TRANSIENT AREA (Z)	DETACHETH THER	DELETE LINE CHAR	DEPTH; LINES/PAGE	DEPTH TEMP	WIDTH; CHARS/LINE	NULL COUNT		DUPLEX; FF=H, OO=F	EJECT COUNT	PAUSE: OU=YES	ROCETH CHEK Plotte - tain coom	DEPT TOTAL SERVICE	MININ CLASS A INC	WILLIAM CHANGLINE	BUILDE DOD TOMODO 1.				GET USER RESPONSE			CHECK TOKEN		ON -			YES, ERROR	TATA NIMBER TOO STOOL		(4 INTIVES TOT SWITE)			MAKE	THE DATAGE NO.				GET RESPONSE
NAM FORMATTER				RASFOLI	DESCRA EQU \$20	EQU	CHAR EQU		EQC		3 6	PREIRE EGU #ZB	200	2 E	Ę	200	E INC	EGO	NI EGU	EGU	EGO	EQC	Eau			70 FEED #46	E PAGE) L		CTORT INY #DEMPT1		SMI	FCB 49	GTCMD	SWI	FCB 48	Ω		BNE NOTNUM			BNE BADNUM	1 100 0 001 1644		Įά		CHOMOG# VC		STA A DNUM	PRTMSG	IMS	FCB 49	GTCMD
* 0000 0000	* *	* *	: *	*	0000 0000							0000 0028	0033	0030	0000 0037				3500	0030	OOSE	003F	0040	0000 0041	2000	200	100	0046	*	* 4 5500 EE 0000		0003 3F	0004 31		0005 3F	0000	2	ಽ	000B 26 22	1	E 2	0010 26 28	, 26, 28	2 20	3	į	0018 CF 0043 B	SB 30	B7		0020 3F		
0001	0003	0005	0000	8000 8000	0010 +						0019	+ 4 000	+ 8100	+ 400	0000	+ 1700	0022 +	0023 +						+ + 6200 6200		0031	0033		0035	0036	0037	+ 8600	+ 6600	0040	0041 +	0042 +						0048									0058 +		0900
017E 24F0	INDEX 24BC M INITDK 253E M		00E1	0110	0130	0137	0145	3 0158	CSS7 M	FLUT COOK		0046	2501	2001		2200	2000	3000	2000	1000 1000 1000 1000	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	100	0110 T		7504	0042	IM. 1. 2151	21CE	AL 216A	21E7	œ	0168	0025	L.	23B8	2384	0800 71	0162	227F	SUBAY 2299 M	2265	TABX 219C M	003F					0027	0030	2302		RR 0198	XABX 2185 M
00C3 RN GETSC 017E 0113 R GTCMD 24F0	M INDEX 24BC	2246 M INITE 0000 3 2200 M INITE 0162	2A2A M INITR2 OOE1	OZ11 R INTR4 011D	0039 INITRS 0130	002A R INITE 0137	Z43A M INITRY 0145	INITRS 0158	2307 FI 10HUR 2330	7307 T.DP	2572 M CASS	2072 11 2.40E 38 0023 1 MD 0046	SACO MICHOLOGICA MICHOLOGICA	OTES D MOUS 2402	0020 Mill 14 22F7	1112 STEE	25.25 M NOTE 20.25	0030 NATO VOCADA	ODEN AND AND AND AND AND AND AND AND AND AN	COOC CAPTO	CASC OFFICE CASE	0000 N 1000 O 10	OCAC CENTRAL CATA	PROBLEM SACA	0035 PATAG 2500	PE 01115 R PS 0042	0043 PSHOLL 2151	PSHX 210E	OOOE FULLAL 216A	0029 PULX 21E7	0007 PUITIN 2406	M GMSG 0168	0009 RC 0025	. 0006 RCBDEF 258C	0000 KEAD 23B8	COIF KEWING Z384	000C SECS17 0080	OOOZ SECT OLEZ	SUBABX 227F	0021 SUBHY 2299	0025 SUBBA 2263	0023 TABX 219C	0029 TB 003F	OOOB TBL OIFF	OOOO R TRACK OFF	0005 TRKSIZ 0012	0000 TX6B 2183	OO1D VALUE 0027	2940 M WD 003D	CB 2488 M WRITE 23D2	2558 M WRIBLK 018B	02B WRTERR 0198	23EC M XABX 21B5

LDA A #1 STA A SECTOR START OF SECTOR LOOP	LOOP FOR SECTORS 1-18	1 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	UP LDA A #%FF		B #4	BSK PULBY! 4-BYLE SYNC LDA A #\$FE		٥	•		CLR O, X ZERO	A SECTOR	STA A 0, X SECTOR NO.	INX CID 0. Y I FNGTH=128	* S	A ##F7	INX H C X CAC	Œ	LDA B #11 peo pitovi 11-8/TF ID-60P		B #6	PULBY! 0-5715.		INX	B #128	BSR PUTBYT 128 BYTES OF DATA (0000)	€ 4	•	STA A O, X PAD		END OF SECTOR DATA		LDA A SECTOR	INC H STA A SECTOR BUMP SECTOR	A #19 DONE?	SECLOP	Č	FINISH OUT TRACK WITH LUNG GAP	Œ	LDA B #200		
009B 86 01 009D B7 008B R	* *	* (9 00A0 86 FF SECLUP 0 00A2 C6 07	0004 8D	0007 C6	00A8	OOAD A7	17 00AF 08	00B3 A7 00	0085 08	11 00B6 6F 00	00B9	OOBC		0051 of	00C2 86	00C4	0007 86	12 00C9 C6 0B	OOCD 4F	OOCE C6	0000	00D4 A7	59 00D6 08	0008	OODO	OODE A7	00E0 08	. 00E1 85 FF	OOES	* *	*	00E6	73 OOEY 4C	OOED 81 13	OOEF 26	*	* *	00F1 86 FF	00F3 C6	00F3	00F9 8D
0124	0126	0128	0129	0131	0133	0134	0136	0137	0139	0140	0141	0142	0144	0145	0148	0148	0148	0151	0152	0103	0155	0156	0158	0159	0161	0162	0163		0166	0168	0169	0171	0172	01/3	0175			0178 (3400 BYTES) 0179		0181	0182	0184
		WAS RESPONSE "YES"?	IF NOT, RETRY	IF SO, BEGIN FORMATTING	CHECK FOR "ESCAPE"		IF NOT, ERROR	0001100	DISK DRIV		IF SO, RETURN TO SYSTEM	EKROR MESSAGE			BETRY		/ BAD DRIVE NUMBER/									START AT TRACK O	BUILD TRACK IMAGE	WRITE TRACK IMAGE TO DISK		BUMP TRACK	DONE?	FOOL ON THE BOWE	BACK TO BEGINNING			DUMMY RCB (FOR "PRTERR")	FMT' ERROR	DUMMY "FOBSTA" IN MEMORY "IRKBIE" (340)		POINT TO BUFFER		8-BY1E GAP
SWI FCB 48	LDX DESCRA		BNE START	JMP FORM2		LDA A O, X			SHIDS	FCB 51	RTS	LDX #BADMSG		SWI	FCB 49		ည္မမ	2	FCB	FCC 'DRIVE NUMBER'		2 FCC /DRIVE /	' READY?	FCB #04	אין חופע חבעב	CLR TRACK	A JSR TRKBLD	RS	LDA A TRACK		CMP A #35		UMP START		KATA .		50	RMB 1 A TRACK IMAGE		Č	LDA A ##FF	٠ <u>٣</u>
					* NOTNUM		_				_	R BADNUM					BADMSG		PRMPT1			PRMPT2	=	Š	5	FORM2	* FORM2A							Ş	SECTOR	SAVEX	į	EKRCOD * FORM	5	TRKBLD		

5	2	2
5	۵)

	0157 0157 0158	0159 08 015A 09 015B B6 8016 015E 84 03	0160 88 0162 27	0166	016A 33 016B F7 801E 016E 20 EB	0170 BE 008C 0173 F6 8018 0176 26 A5	0178 39	0179		017E 017F 0180 0181	0182 0180		
0247 0248 0249 0250	0251 0252 0253 0253	0.254 0.255 0.254 0.258	0259 0260 0261 0261	0265 0264 0264 0265	0268 0267 0268 0269	0270 0271 0272 0273	0274 0275 0276 0277	0278	0280 0281 0282 0283	0284 0285 0284 0287 0289	0290 0291 0292 0292		
DONE! PUT BYTE INTO TRKBUF	DONE? LOOP ON COUNT IN "B"	TO DISK DRIVE , DRIVE NO. IN "VALUE+1"		TURN DRIVES ON READY? YES	NO, LONG DELAY	ANOTHÈK DELAY	READY NOW? YES	ID HERE	ISSUE ERROR MESSAGE	CLEAN STACK (JSR TKKWRT) RETRY	DISABLE INTERRUPTS GET DRIVE NUMBER 30 USEC DELAY SET TRACK NO. 30 USEC DELAY	"SEEK TKACK" COMMAND 30 USEC DELAY WAIT ON BUSY	SAVE STACK POINTER POINT TO TRACK IMAGE
RTS SYT STA A O, X INX	DEC B BNE PUTBYT	RTS TR	(EG EQU \$8014 (EG EQU \$8018 (EG EQU \$8018	KT LDA B CMDREG AND B #\$80 BEQ TKKW2	LDX #O DEX BNE T1	LDX #0 DEX BNE T2	LDA B CMDREG AND B #\$80 BEQ TKKW2	* * DISK EKRORS HANDLED HERE *	0.70	SWI FCB 30 INS INS JMP START	2 SEI LDA A VALUE+1 STA A DRVREG BSR DEL30U LDA A TRACK STA A DATREG BSR DEL30U IDA A #418	STA BSR BIT BNE	STS SAVEX LDS #TRKBUF-1 LDA A #\$F4
OOFB 39 ** * OOFC A7 OO PUTBYT		36	0103 8014 DRVKEG 0103 8018 CMDREG 0103 801B DATREG	0103 F6 8018 THKWRT 0106 C4 80 0108 27 20	010A CE 0000 7 010D 09 71 010E 26 FD *	0110 CE 0000 0113 09 T2 0114 26 FD *	0116 F6 8018 0119 C4 80 011B 27 0D	* * *	0110 F7 0091 R 0120 CE 008C R 0123 3F	000	*** 012A 0F	80 80 26 26	0148 BF 008C R 014B 8E 018B R 014E 86 F4
0185 0186 0187 0188 0188	0190 0191 0192	0193 0195 0195 0197	0198 0199 0200 0201	0202 0203 0204 0205	0206 0207 0208 0209	0210 0211 0212 0213	0214 0215 0216	0218	0220 0221 0222 0222		0229 0230 0231 0232 0234 0235 0235	0238 0239 0240 0241 0243	0244 0245 0246

"WRITE TRACK" COMMAND	CHECK STATUS BITS DRO AND BUSY BITS INVERT BUSY BIT DATA REQUEST? YES OTHERWISE, FDC DONE		START OF TRACK IMAGE
STA A CMDREG DEX INX DEX DEX INX DEX DEX	LDA A CMDREG AND A #\$03 EOR A #\$01 BEG TRKLOP BIT A #\$02 BNE TRKRGT BRA THKDON	B DATR TRKLOP SAVEX B CMDR DSKEKR	INX DEX INX DEX INX DEX INX SEX SEX EQU #+10
	* * * * * * * * * * * * * * * * * * *	* * * * * 30	EL30U
8018	8018 03 01 F7 02 02 06	801B EB 008C R 8018 A5	ć
00 00 00 00 00 00 00 00 00 00 00 00 00	88 88 27 27 26 26 33		08 08 08 08 09 018C
0150 0153 0154 0155 0156 0157 0158	015E 015E 0160 0162 0164 0168	016E 016E 0170 0173 0176	0179 0178 0178 0170 0170 0176 0180 0181
0247 0248 0249 0250 0251 0252 0253 0254 0255	257 258 258 260 262 263 264 265 265	268 268 270 272 272 274 275 275	275 280 281 282 283 284 285 286 287 288 289 290 291

X 2219		0001 0000 0000	z ~		NAM SMOKEDRV		
RUDHA 2232 M	6000	5000	*	DISK	IS FOR SMOKE-SIG	DRIVERS FOR SMOKE-SIGNALS BFD-68 AND CP/68 S	SYSTEM
B 2200	2454 M	0004	*	USES	INFO. FROM D. L. PUCKE	PUCKETT'S ARTICLE IN "68" J	JOURNAL
0041		5000	* *	VUL. I ISSUE	-	•	
BADNUM OOSH K BASEQU 2A2A M	101.L 2151 M	0007	*		BFD-68 ROM INTERFACE ADDRESSES	SSES	
Σ	21CE M					Alo populativi ner	
0039	ΣX		8026 8029	INITE CAP SE	\$8026 INITHE	d t	
CHAIN 243A M	7 OPFC R	0006 7E		o La		WRITE A DISK SECTOR	
	2406 M	0009 7E		g.		SEEK TRACK ZERO	
6 8018	0025	000C 7E	806C M	E S	#806C WRITE T	WRITE TRACK REG. OF FDC BEAD 18ACK BEG. OF EDC	
0037	RCBOEF 258C M			DIRKA GER			
CMPC 2018 M	7350 FI	0016	. 30	* BFD-68 RAM ADDRESSES	ADDRESSES		
23/2 4R 0023	008C R						
	P 00A0 R	0012			\$A07B		
	R COSB R	0019 0012 A07C			#A07C		
2420				30	#AO7E		
DESCRA COZO	7,77 H				!		
	22B3 A	0023	*	* CP/68 FCB DEFINITIONS	DEFINITIONS		
003A	3 2265			i	1		
ξ	OLOD R	0012			STATUS	() () () ()	
	0113 R	0012		E C	BOFFER ANDRED	BUTTEN AUUKESS Dotor nimber	
	8X 219C M					MBEN	
8014	003F	0028 0012 000R		200			2
USKERR OILD R	INACK OOST R	7100		3			24
	0180	0031	*	USER	REGISTER DEFINITIONS		
Σ	0170 R			i	1000	200 to 100 to 10	
3 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	· 015B R	0012		UA EGU 6	. A	"A" REGISTER ON STACK	
	T 016A R	0034 001Z 0007		XH EWO	: Y		
FCBCHN 0029	TEXECT OLICE R	0036	: *	* BASE-PAGE TRACK TABLE	TRACK TABLE		
2002	0103	0037	*				
	2183 M	0012		EGG			
2558	0027	0012		EGU			
2 0073	OOSD	0012		2			
	WRITE 23D2 M	0041 0012 0003		CTRK3 EQU 3			
FURMAT 0000 KN	21183	7100					
23EC		0044	*		ENTRY POINTS TO DRIVERS		
24F0		6		Ì	NATION.		
INDEX 24BC M		0046 0012 0012 0047 0012 001F	Z Z	EN I	RDSEC		
2335		0012			WTSEC		
0044		0049	* 1		3/3/10		
T 0045		0000	* *	* INI - IMI - IVE			
LOGDB 246E M		0012 86	. ש ב	INTEK	##FF		
2301		0014 97		STA		INIT. TRACK TABLE	
2402		0016 97	ت 5				
MU.16 22E7 M			2.2	STAR	CTRK3		
22CD		0010 20	E2			INIT. INTERFACE	
4 002F		0058	*				
ν 010		0059	* *	READ	A DISK SECTOR		
OPEN 234F M		non.			, ,		

		20 311 90	NO ON		YES		DRIVE 19	02		YES			DRIVE 22	⊋ `	0.27	753		O SULON OF FUNDAME		CET DOLLE IN DI ACE	SEL DAIVE IN LEACE					POINT TO FCB			Cir Linada HUO	GET DRIVE NO.	LIMII KANGE DE-EDOMAT DOTUE NO	POINT TO TRACK TABLE	ADD IN OFFSET			GET ENTRY	INITIALIZED?	YES		C		SAVE TABLE POINTER			SAME CHOPENT TOACK		_	NECOVER INDEE LOINIER			INII. IABLE ENIKY	SAVE TABLE ENTRY
DRIVE	* 1=10 * 2=20	* + + + + + + + + + + + + + + + + + + +	2 X		LDA B #\$08	SKH SELL	ET1 CMP				BRA SETD			BNE SEI3			BRH SEID	**************************************	ב ב	* CETH CIA D DDIME	e to	* * * *	*	* WRITE A DISK SECTOR	*	WISEC TABX					FIND R ##CG			IBS	FCB 9	LDA B O, X	CMP B #\$FF	BNE WTSEC1		* RESTORE DRIVE TO TRACK		XXX	IMS	FUR S	USK RDIKKK	ACHORD GOT	DIE X	Y INC	FCB 6	a 1	× '0 a u	WTSEC1 PSHX
		4000	26 04		0075 C6 08	71 07	0079 81 01	26 04		8	007F 20 0A		81 02	0083 26 04	ò	0083 C6 20	70	00 70 0000	000	0000 E7 0075	39 77	i	•	•	•		008F	+ 0000 03		ğ ç	0093 84 03	9 6	3	+ 009C 3F		009E E6 00		00A2 26 10	•	•	•	•	+ 00A4 3F	0000 000	0086 BU 000F R	<u> </u>	COVO OG	+ 000AF 3F		K [00BZ E/ 00	-
0122	0124	0126	0128	0129	0.730	0131	0133	0134	0135	0136	0137	0138	0139	0140	0141	0142	5410	244C	2440	0770	0140	0149	0150	0151	0152	0153	0154	0155	0156	015/	0100	0170	0161	0162	0163	0164	0165	0166	0167	0168	0169			01/2	01/3	1170	0176	0177		0179	0180	0182
GET FCB ADDRESS		SAVE FCB ADDRESS	LIMIT RANGE		POINT TO TRACK TABLE	HUD IN OFFICE		GET ENTRY	INITIALIZED?	YES		O	1	SAVE TABLE PUINIER		TOPIC NOVIE COLL GALL	MEND FOU INHER NEGLOTER	FUL DURKEN! INHUR IN FLHUR	DECOUSE TABLE DOINTED				INIT. TABLE ENTRY		SAVE TABLE POINTER			SET TRACK OF DRIVE	INTO FDC	PUINI IO PUB	GET FRACE		N L			RECOVER TABLE POINTER			IN TRACK	CALL "READ SECTOR"	ERRORS?	ON ON	00000	YES, EKKUK NU. 5		"A" NI OLITATO MOLITAGO	<u> </u>	And IN STATIS				BFD-68
* RDSEC TABX		STX SAVEX		BSR SEIDRV	LDX #CTRKO	HIJIHA	FCB 9	LDA B O, X	CMP B #\$FF	BNE RDSEC1	*	* RESTORE DRIVE TO TRACK O		PSHX	I MA	FUE 3		SIR BIRACK		rock State	ECB A				RDSEC1 PSHX		FCB 5	STA		LDX SAVEX	LUM M PUBLIKY, X	I DY ECBURA. X			ñ	PULX	SWI	FCB 6	STA	SS		BEG DONE2	0	C# 8 #M"	**************************************				STA B FCBSTA, X	RTS	* *	* REFORMAT DRIVE NO. FOR BFD-68
	001E 3	0020 DF 04	£ &	0026 8D 4A	0028 CE 0000	36 300			OOZF C1 FF	26					0033	0034 05	BU 000F	0038 F7 A07C		30 3000		0040	0041 E7 00				0044 0	F7 A07C	BD 000C	出 ;	00411 R6 0R	0 1	7 6	ì	Ł		0020		A7 00	BD 0003	20	0064 27 02	ò	0066 C6 05	00 0700		מ ע	i d	1			
	0063 +	0065	0067	8900	6900	0070		0073	0074	0075	9200	7.200	0078			0081 +	2800	5000	1000		0087 +		6800	0600	0091		+ 8600	\$600	0000	9600	\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \	0000	0100	0101	0102	0103	0104 +	0105 +	0106	0107	0108	0109	0110	0111	0112	2170	110	0115	0117	0118	0119	0121

```
wwww.Xeffer
                                                                                                                          Σ
                                                                                                                                                         EKKKE
SECTOR A07D
SET1 0079
SET2 0081
SET3 0083
SETD 0088
SETDRV 0072
SUBANKED 0007
SUBANKE 2229
MATTER 0007
                    0012
                SET TRACK OF DRIVE
INTO FDC
POINT TO FCB
GET TRACK
GET SECTOR
GET BUFFER ADDRESS
PUT INTO BFD-68 PLACES
                                                                                                                                    NEW ENTRY
CALL "WRITE SECTOR"
ERROR CHECK AND FINISH
                                                                                                        RECOVER TABLE POINTER
SWI
FCB 5
STA B TRACK
JSR WTRKR
LDX SAVEX
LDA A FCBTRK, X
LDA F CBDBA, X
STA B FCCT, X
STA B SECTOR
STA B SECTOR
STA B UFPNT
PULX
SWI
FCB 6
STA M O, X
                                                                                                    + 00CD 3F
+ 00CE 06
00CF 47 00
00D1 BD 0006 R
1D4 20 8D
                             œ
               0086 F7 A07C
0089 BD 000C R
008C DE 04
00C0 E6 08
00C1 EF 07
00C7 F7 A07D
00CA FF A07E
00B4 3F
00B5 05
0163
0163
0165
0165
0168
0198
0193
0194
0198
0198
0198
```

. RDSEC 001E RN
ADDARX 2219 M
ADDBAX 2219 M
ADDBAX 2219 M
ADDBAX 22219 M
ADDBAX 2222 M
ADDBAX 2222 M
ADDBAX 2222 M
ADDBAX 2222 M
ADDRAS 2222 M
CLOSE 2348 M
CHWC 2572 M
DOVIC 2573 M
DOVIC

GET FIRST T/S	3/1 187 1 1/S				INIT. BUFFER INDEX		INIT. PRESENT T/S		READ FIRST SECTOR		FILE INTO MEMORY		DET A DATA BYTE FROM FILE	NO NOTES HELDEREGOS.	}		GET FRANSFER ADDRESS	,	GET NEW DATA FRAME		DATA FRAME?			GET ADDRESS	•	4	GET FRAME COUNTER	GET DATA BYTE		STORE BYTE		COUNT DOWN		GET NEW DATA FRAME		GET TRANSFER ADDRESS	SO THERE	FROM SYSTEM FILE	/A/ REGISTER	100 NEED NEU CECTODO	YES YES	SET BYTE	Off.: 401:11.
LDA A 122, X LDA B 123, X STA A FTS	C 00 4	[W 4		LDX #BUFFER+4	LDA A FTS+1	Ω		STA B PTS			LOAD SYSTEM	1	BSR	BNF BOOT?				STA GETBYT			<u>a</u> i	BNE BUUL4			BSR GETBYT	r Č			ě	STA A O, X	STX SAVEX		BNE BOOTS	BRA BOOT1		Š	X 'O AWD	BYT	RETURN BYTE IN 'A	BYT LDX INDEX		1 DA A 0. X	C
0024 A6 7A 0026 E6 7B 0028 B7 0090 C	ĕ [2 4	E6 7D B7 0092	F7 0093	0038 CE 0014 C	B6 0091	F6 0090	B7 0095	0047 F7 0094 C	BD 00C6		MON *	6	0050 8U 3A BUUTI	3,4	Ì	80 34	87	005B 8U ZF	20 EE			0004 Z6 Z1 *	8D 24	B7	006B 8D 1F	80 16	B7 009E C	0075 8D 15 BOOT3	FE 0098 C		007D FF 0098 C		0083 26 F0	0085 20 C9		11 L	0089 6E 00 *		0C * *	008C FE 0096 C GETBYT	27 07	* 0094 46 00	8
0061 0062 0063	0064	0066	8900	6900	0071	0072	0073	0074 0075	0076	0077	0078	0079	0000	0082	0083	0084	0000 0000	0086	(0088	6800	0600	00041	6600	0094	0095	0097	8600	9099	0101	0102	0104	0105	0106	0108	0109	0110	0111	0114	0115	0117	0119	0120	0122
BOOTSTRAP PROGRA	LINKED AS FOLLOW	SECTOR 1, BYTE 122-FIRST TRACK 123-FIRST SECTOR	⊢ (125-LAST SECTOR 126.7 EREE-SPACE HEADED		DRIVE 0:	INTERFACE ADDRESSING	INIERFACE AUDRESSI					INIT INTEREDOR PIA	DISK SECTOR		READ FUC TRACK REG.	ALL VARIABLES IN COMMON. CODE IS ROM-ARLE	o T		Ø.													INIT. STACK POINTER DRIVE O IN BEN-FORMAT		INIT. INTERFACE	GET TRACK FROM FDC	SEEK TRACK 0		STSTEM LINK INFURMATION	TRACK O. SECTOR 1		READ LINK SECTOR	
	ASSUMES SYSTE	* TRACK 0, SECT	*	* *		* BOOTS SYSTEM FROM DRIVE	* * DEFINE DISK DDIVE		EQU	EQU		BUFFNI EUU *AU/E	INITE FOIL \$8024	EGU	EQU	RDTRKR EQU #8072	* NOTE: ALL VARIABLE				CAN FIS. Z	CMN PTS, 2		CMN SAVEX, 2	CMN ADDRES, 2	CMN FCNT, 1	CMN RCNT, 1	* ERROR JUMP VECTOR	*	ERROR JMP \$E113	* BEGIN BOOT HERE		START LDS #STACK+15			STA B TBACK	USR RESTOR	1014	* NOW UE! SYSTEM LIN	LDA A #1		USR RDSEC LDX #BUFFER	
z																			ပ	ပ	ے د	<u>ں</u> ر	Ç,	ه ن	ں د	Ü	ü						000년 08 08	A07B	8026	8072	8038				O	0006 R 0010 C	

>	っ	х	
•	4	v	

ADDRES 009C C BOOT 0000 RN BOOTI 0050 R		oc ⊱				UNDEX COSE N INDEX COSE C INITE 8026		1 009F		SAVEX 0098 C	A07D 0000	0003 A07C			
MOVE POINTER	CHECK FOR LAST SECTOR	NDT LAST	NOT LAST	EOF-GO TO TRANSFER ADDRESS	GET FORWARD T/S LINK	UPDATE PRESENT T/S	READ NEW SECTOR	GET DATA BYTE	RE-INIT. INDEX	TINE			PUT DATA INTO PLACE	READ DATA SECTOR EKROR? NO	YES
STX INDEX RTS	44	ENE GETS2	CMP B LTS BNE GETS2	BRA B0014		LDA A 1, X STA B PTS STA A PTS+1	BSR RDSEC		INX STX INDEX RTS	* SINGLE-SECTOR READ ROUTINE	DRIVE=0 TRACK='B'	SECTOR='A' BUFFER='X'	STA	STX BUFFNT JSR READS TST B BEQ DONE	JMP ERROR RTS END
FF 0096 C 39	0094 C GETSEC 1	3 8600 07	0092 C 02	•	* 0010 C GETS2 00	01 0094 C		00 to	08 FF 0096 C 39		* * *	* * :	* AO7C RDSEC	AO7E 8029 03	7E 0000 R * * * * * * * * * * * * * * * *
0097 FF 0 009A 39	F6	00A1 B1 0 00A4 26 0	00 A6 F1 009	00AB 20 D	E 6	8 7 E	6 8 6		00C1 08 00C2 FF 0 00C5 39				F7 B7	00CC FF 6 00CF BD 8 00D2 5U 00D3 27 0	00D5 7E C
0123 0124	0126	0128	0130	0133	0135 0136 0137	0138	0141	0142	0144	0147	0151	0153	0155 0156 0157	0158 0159 0160 0161	0162 0163 0164 0165 0165

Figure 1 Figure 1 Figure 2 Figure 2 Figure 3	0001 0003 0004	0000 0000	N NAM INITER * INITIALIZE A DISK FOR CP-68 OPERATING SYSTEM * FOR CHECK E TAKEL ELODDY DISK	0061 + 0062 + 0063 +		<u> </u>	\$35 \$37 \$39	END OF TRANSIENT AREA (2) NEXT AVAIL TRANSIENT AREA (2) BACKSPACE CHAR DELETE LINE CHAR	
THORAC O, SECTION 1 PROFESTIVE FREE-SPACE LIST COMPA	0 4 2 5		* FUK SWIPC 3 INCH FLUPPY DISKS				9 0 4 4	H; LINES/PAGE	
Fight Figh	200		TRACK O, SECTOR 1 BOOTSTRAP			<u> </u>	3 3 3 3	H CHARS/LINE	
The color of the	2 6 2 6 2 6		TRACK () SECTOR 1 HEADER OF FREE-SPACE TRACK () SECTORS 2-18 DIRECTORY SPACE				# 3E # 3E	COUNT	
	010		TKACKS 1-35				\$40 \$40		
Color Colo	011			0071 +			\$.41		
	717			0072 +			\$42		
Council Court Council Cour	513		FOLL 128 NOTES DED	0073 +			\$43	PE CHAR	
October Color Co	015		EQU 18 18 SECTORS PER	0075 +		Ž	4 4 4 4 4 10		
FILE-CANTROL BLOCK ADDRESSES	016		EQU 34 34 TRACKS ON DISK (LESS				\$46	H CHARS/LINE	
Color Colo	017		**************************************	0077					
Control Cont	010		* TILETUNINUL BEUCK HUNKEVURV				·INI.	K IN DRIVE /	
Color Colo	020						۰,		
Common concernment Common	0021 +		EQU 0			FOF	¥.		
CONTROL CONT	022 +		EQU 2	0082					
Color Colo	+ + 500		EQU 5	0083	8000			ENTRY POINT FROM CLI	
	1000		E01 7	1800	è				
Common Common Figher Equity Common Common Figher Equity Common Common Figher Equity Common Common Figher Equity Common	4 900		F01 9		0 0		A VALUE+1	GET DRIVE NUMBER	
+ 0000 0000 FGENDLE KBU 11 SECTOR NUMBER PROCESSION OCCOR FGENDLE KBU 12 SECTOR NUMBER PROCESSION OCCOR FGENDLE KBU 14 BACK LINK TRACK/SECTOR 0099 00000 FGENDLE KBU 14 BACK LINK TRACK/SECTOR 0099 00000 FGENDLE KBU 14 BACK LINK TRACK/SECTOR 0099 00000 FGENDLE KBU 14 BACK LINK TRACK/SECTOR 0099 0000 FGENDLE KBU 14 BACK LINK TRACK/SECTOR 0099 0000 FGENDLE KBU 14 BACK LINK TRACK/SECTOR 0099 0000 FGENDLE KBU 14 FILE NAMER (B 3-ECTOR SECTOR 0000 0001 FGENDLE KBU 14 FILE NAMER (B 3-ECTOR 14 FILE NAMER) FESTIVE	027 +		EQU 10		CF 0000		#FCRSPC	POINT TO FOR	LUNIVES!
Cougo Octoo CFEBAME EQUI 14 PAGK LINK TRACK/SECTOR OVOC 0000 OCCC FCEBAME EQUI 14 PAGK LINK TRACK/SECTOR OVOC 0000 OCCC FCEBAME EQUI 14 PAGK LINK TRACK/SECTOR OVOC 0010 CFEBAME EQUI 14 PAGK LINK TRACK/SECTOR OVOC 0010 CFEBAME EQUI 14 PAGK LINK TRACK/SECTOR OVOC 0010 CFEBAME EQUI 14 PAGK ENDI 13 FTIE TYPE TACK/SECTOR OVOC 0010 CFEBAME EQUI 14 FTIE TYPE OVOC 0010 CFEBAME EQUI 15 FTIE TYPE OVOC 0010 CFEBAME EQUI 15 CFEBAM	028 +		EQU 11		A7 09	STA	A FCBDRV, X		
COUNTY C	029 +		EQU 12		8B 30	ADI	Æ	MAKE DRIVE NUMBER ASCII	2
Construction Cons	+ + + + + + + + + + + + + + + + + + +		EGU 14 BACK LINK		B7 OOBE	STE	∢ :	PUT IN PROMPT LINE	29
	032 +		EQU 29 FILE TYPE		CE OORA	X G		TOMOGO TIGHTIO	
Composition	033 +		EQU 30			- d	2		
+ 0000 0021 FCBMF E0U 35	034 +		EQU 31			. E	B 49		
		0000	EQU 33	0092		910	Œ	GET USER RESPONSE	
		0000	EQU 35 NUMBER OF SECTORS			30	H		
Concologies Fuebrace Education Concologies			EGU 3/ NEX! FUB IN ACTIVE FOR 30	+	ŭ	<u>.</u>	B 48		
Common color Comm	+ 680		EQU 41		1 4 1 4	י המ	DESCRIP		
OCCOR OCCO	040				2 2			II YES'?	
OUNDE 44 FCC 1DSK DISK OUNDE 1 OUNDE 1 OUNDE 2 OUNDE	041		RMB 2		27	BEG	INITRZ	ģ	
NOW	242		, nsk						
NOW	243		1 1		00E0 39			IF NOT, QUIT	
## COMPAND LINE INTERPRETER BASE—PAGE LOCATIONS 0.0054 & 0.00 ## COMMAND—LINE INTERPRETER BASE—PAGE LOCATIONS 0.10 0.0056 & 0.1 ## COMMAND—LINE INTERPRETER BASE—PAGE LOCATIONS 0.10 0.10 0.0056 & 0.1 ## INITIAL PROPERTY 0.000 0.0050	045		32		0000		10000	and of TMICO	
NOCAH OOSH BUFFER RMB SECSIZ SECTOR BUFFER O107 OOE6 86 01	046		<u>}</u>		4		#FUBSFU FUBTEK: Y	TRACK O	
* COMMAND-LINE INTERPRETER BASE-PAGE LOCATIONS 0109 00E8 A7 0B * INIT	047				8				
* COMMAND—LINE INTERPRETER BASE—PAGE LOCATIONS 0109 * * INIT	048		*		A7	STA	₫		
# BASEQU	040		* COMMAND-LINE INTERPRETER BASE-PAGE LOCATIONS	0109					
+ 00AA 0020 DESCRA EQU \$22 DESCRIPTOR ADDRESS(2) 0113 * ALL. + 00AA 0022 DESCRA EQU \$22 DESCRIPTOR COUNT + 00AA 0022 CICHAR EQU \$23 CURRENT CHAR (2) 0113 * * ALL. + 00AA 0025 RC EQU \$25 TOKEN RETURN CODE 0114 + 00EA 3F 0116 + 00EC CE 002A R + 00AA 0025 CLASS EQU \$27 TOKEN CLASS 0117 00EC CE 002A R + 00AA 0027 VALUE EQU \$27 TOP OF FCB CHAIN (2) 0118 + 00EC CE 002A R + 00AA 0028 FCBCHN EQU \$28 DISK FREE SPACE POINTER (8) 0120 + 00F0 04 + 00AA 0038 BMEM EQU \$33 START OF TRANSIENT AREA(2) 0121 00F1 A7 07	000			0110			ZE HEAD OF	FREE-SPACE BLOCK	
+ 004A 0022 DESCRE EQU \$22 DESCRIPTOR COUNT + 00AA 0023 CUCHAR EQU \$23 CURRENT CHAR (2) + 00AA 0025 RC EQU \$25 TOKEN RETURN CODE + 00AA 0025 RC EQU \$25 TOKEN RETURN CODE + 00AA 0027 VALUE EQU \$25 TOKEN RETURN CODE + 00AA 0027 VALUE EQU \$25 TOP OF FCB CHAIN (2) + 00AA 0028 FKETAB EQU \$25 TOP OF FCB CHAIN (2) + 00AA 0028 FKETAB EQU \$33 START OF TRANSIENT AREA(2) + 00AA 0038 BMEM EQU \$33 START OF TRANSIENT AREA(2) + 00AA 0039 0027 VALUE COUNTER (8) + 00AA 0030 0030 0027 VALUE COUNTER (8) + 00AA 0030 0030 0030 0030 0030 0030 003	+ 020		FOIL #20	1110		4 4	4000		
+ 00AA 0023	053 +		EQU \$22	0112		* HLL /ERU	EXCEP! FUR		SECTOR 1
+ 00AA 0025 RC EQU \$25 TOKEN RETURN CODE	0054 +		EQU \$23	0114		•	œ		
+ 00AA 0026 CLASS EQU \$26 TOKEN CLASS + 00AA 0027 VALUE EQU \$27 BIN VALUE/TRANSFER ADDRESS (2) 0117 00EC CE 002A R + 00AA 0029 FCBCHN EQU \$27 TOP OF FCB CHAIN (2) 0118 + 00AA 002B FRETAB EQU \$2B DISK FREE SPACE POINTER (8) 0120 + 00E0 04 + 00AA 0033 BMEM EQU \$33 START OF TRANSIENT AREA(2) 0121 00F1 A7 07	055 +		EQU \$25		OOEA 3F	MS			
+ COCHA COZ7	056 +		EQU #26	+		5	8 2		
+ 00AA 0033 BMEM EQU \$38 START OF TRANSIENT AREA(2) 0120 + 00F0 04	+ + 0000		EWU #2/			CDX	#BUFFER		
+ 00AA 0033 BMEM EQU #33 START OF TRANSIENT AREA(2) 0120 + 00F0 04	+ + 620		EQU \$2B		00FF 3F	APA APA	× -		
00F1 A7 07	+ 090		EQU \$33		00F0 04	. L			
				_	00F1 A7 07	STA	A FCBDBA, X		

INITR7 NO B #1 YES, SECTOR=1 A #DSKSIZ+1 END OF DISK? INITR7 NO	LAST SECTOR POINTS TO 0,0 TRACK LINK SAVE LSEC GET PSEC 1 SECTOR LINK RESTORE LSEC WITTE SECTOR DONE? (=0) NO	DONE? (=0) NO YES, DONE!!! X SAVE LSEC GET PSEC X GET LSEC KEEP WRITING	OR MESSAGE #QMSG OUTPUT ERROR MESSAGE SG 49 RETURN TO CLI *NDTIALIZATION FAILED' \$CD -REG	SAVE X-REGISTER POINT TO LOGICAL/PHYSICAL TABLE ADD LOGICAL OFFSET SECTOR STARTS AT 1 GET PSEC RESTORE X-REG TH EKROR CHECKING
BNE INC PMP	* CLR A CLR B CLR B * CLR B * INITK7 STA A BUFFER STA B BUFFER+1 P'UL B P'UL B BY HTBLK TST A BUFFER+1 BY HTBLK TST A BUFFER+1 BY BY HTBLK TST A BUFFER+1 BY		* FATAL ERR INITG LDX INITG PRTM SWI FCB RTS AMSG FCC GMSG FCC ** CONVERT L	GETSC PSHX SAVE X-REGISTER SMI FCB 5 LDX #TBL POINT TO LOGICAL ADDBX ADD LOGICAL OFF SWI FCB 10 SECTOR STARTS A' LDA B 0, X GET PSEC PULA B 0, X RESTORE X-REG SWI FCB 6 RTS * WRITE A SECTOR WITH ERROR CHECKING
0183 013A 26 09 0184 013C C6 01 0187 013F 81 23 0188 0141 26 02	0189	0201 0202 0203 0155 26 01 0205 0205 0205 0157 39 0206 0158 47 0A 0208 0158 80 21 0210 0150 E7 0B 0211 0157 33 0211 0150 20 05 0213	0215 0216 0217 0162 CE 0168 R 0219 + 0165 3F 0220 + 0166 31 0221 0167 39 0222 0168 49 0223 0168 49 0224 017D 0D 0225 0226	0229 0230 + 017E 3F 0232 + 017F 05 0233 + 0180 CE 01FF R 0234 + 0183 3F 0235 + 0184 0A 0236 0185 09 0237 0186 E6 00 0239 + 0189 3F 0240 + 0189 36 0241 018A 39
000000				
STA B FCBDBA+1,X PSHX SWI FCB 5 CLEAR OUT BUFFER EXCEPT FOR LAST 2 BYTES	17-2 TRACK, SECTOR=1	WRIELK WRITE BLOCK 3 FCBSTA, X CHECK FOR DISK ERROR *+6 INITQ FATAL DISK ERROR, QUIT WRIBLK OUT OF RANGE "BSR WRIBLK" FCBSCT, X SECTOR=4 BUFFER+SECSI7-2 BUFFER+SECSI7-1	TORY TO ZERO WRITE DIRECTORY BLOCK X CHECK FOR DISK ERROR OK FATAL DISK ERROR, QUIT T, X NEXT SECTOR IZ DONE WITH TRACK? YES	BRA INITR4 NO, CONTINUE WRITING STA A FUBSCT, X SECTOR=1 STA A FUBSCT, X TRACK=1 TAB INITIALIZE REST OF DISK (FREE-SPACE) X=FUB ADDRESS A=TRACK NUMBER B=SECTOR NUMBER B=SECTOR NUMBER CMP B #TRKSIZ+1 END OF TRACK?
	* LDX #BUFFER LDA B #SECSIZ CLR A INITR3 STA A O.X INX BEC B BEC B BNE INITR3 * LDA A #1. STA A 0.X STA A 0.X	~ L &	* INITIALIZE DIRECTORY TO ZERO * INITRA BSR WRTBLK WRITE DIR TST FCBSTA, X CHECK FOR BEG *+4 OK * BRA INITA FATAL DIS * LDA A FCBSCT, X INC A NEXT SECT INC A ATTAKSIZ DONE WITH BEG INITRS YES * CTO A FCRSCT; X	** INITES LDA A *I INITES LDA A *I INITES LDA A FCBSCT, X STA A FCBTRK, X TAB ** INITIALIZE REST OF D ** X=FCB ADDRESS ** A=TRACK NUMBER ** B=SECTOR NUMBER ** INITES INC B ** TRKSIZ+1*
00F3 E7 08 . 00F5 3F . 00F6 05	00F7 CE 002A R 00FA C6 7E 00FD 4F 00FD 47 00 00FD 5A 0100 5A 0103 86 01 0103 86 01	9 3F 6 06 8B 7E 8B 05 27 04 20 4F 20 76 6C 0B 7F 00A9		012C H7 05 0130 86 01 0130 86 01 0134 A7 08 0136 16 0137 5C
0122 0123 0124 + 0125 + 0126	0128 0129 0130 0131 0133 0134 0136 0137		0155 0157 0157 0158 0160 0161 0165	0168 0168 0169 0170 0171 0173 0174 0174 0176 0177 0177 0181 0181

* LOGICAL/PHYSICAL SECTOR TABLE

FCB 00

표

FCB #1 FCB #6 FCB #8 FCB #3 FCB #3 FCB #3 FCB #3 FCB #5 FCB #7 FC

ş

BLS *+4

ADD A #\$7

BOOT PROGRAM STARTS HERE

EQU

```
R BOOT
         01F9 23 02
                              0
                                                                                                                                                                                                                                                                                                                                                                                                     0211 0211
                               O1FB 8B
                                                                                                                                        01FF 01
0200 06
0202 10
0203 03
0204 08
0204 08
0205 0D
0207 05
0208 0A
0209 0F
0209 0F
0209 0F
0209 0F
0209 0F
0200 0C
0200 0C
0200 0C
0200 0C
0200 0C
0200 0C
                                                                                                                   DIFE OO
                                                      OTED
        BSR OUTHR CONVERT RIGHT DIGIT
STA A EKTYPE+1
PSHX
SWI
SWI
FCB 5
LDA A FCBSCT, X
BSR OUTHL MAKE SECTOR NO. HEX
STA A SECT+1
LDA A FCBSCT, X
BSR OUTHR
STA A SECT+1
LDA A FCBTRK, X
BSR OUTHR
STA A TRACK
LDA A FCBTRK, X
BSR OUTHR
STA A TRACK
LDA A FCBTRK, X
BSR OUTHR
STA A TRACK
LDA A FCBTRK, X
BSR OUTHR
STA A TRACK
LDA A FCBTRK, X
BSR OUTHR
STA A TRACK+1
LDX #DERROR
PRIMSG PRINT ERROR MESSAGE
                                                                                                                                                                                                                       CONVERT RIGHT DIGIT
                                                                                                                                                                                                                                                                                               MAKE SECTOR NO. HEX
                                                                                                                                                                                                                                                                                                                                                                                                                                                PRINT ERROR MESSAGE
                                                                                                                                                                                     CONVERT LEFT DIGIT
                                       CLEAR ERROR FLAG
ISSUE I/O REQUEST
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 * CONVERT BINARY TO HEX-ASCII HERE
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  CALL CP/68 "WARMSTART"
                                                                                                                                       RESTORE 'A'
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     GET NIBBLE
MAKE ASCII
>9?
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          SHIFT RIGHT
                 SAVE 'A'
                                                                                                   ERROR?
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         FCC ' AT SECTOR '
RMB 2
FCC ', TRACK '
RMB 2
FCB #0D
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       *
DEKROR FCC 'DISK ERROR: '
EKTYPE RMB 2
--- AT SECTOR '
    *
WRTBLK PSH A
CLR A
CLR A
CLR A
CLR FCBSTA, X
                                                                                      STA A FCBSTA, X
TST A EF
BNE WRTERR YE
                                                                                                                                                                                   BSR OUTHL
STA A ERTYPE
TBA
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     AND A #$0F
ADD A #$30
CMP A #$39
                                                                                                                                                                                                                                                                                                                                                                                                                                                         SWI
FCB 49
                                                                          FCB 19
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   SWI
FCB 31
RTS
                                                                                                                                     PUL A
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         LSR A
LSR A
LSR A
                                                                                                                                                                        WRTERR TAB
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    OUTHR
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       OUTHE
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              TRACK
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     SECT

        0244
        **

        0245
        018B 36
        MF

        0246
        018C 4F
        05

        0248
        018D 6F
        05

        0249
        + 018F 3F
        05

        0250
        + 0190 13
        05

        0251
        - 0191 42
        05

        0252
        - 0194 26
        02

        0253
        - 0194 26
        02

        0253
        - 0194 26
        02

        0254
        - 0196 37
        02

        0255
        - 0197 39
        02

        0256
        - 0197 39
        01

        0257
        - 0198 17
        01

        0258
        - 0198 17
        01

        0259
        - 0197 39
        01

        0250
        - 0198 17
        01

        0260
        - 0197 18
        01

        0260
        - 0198 17
        01

        0260
        - 0141 187
        01

        0260
        - 0141 187
        01

        0260
        - 0141 187
        01

        0270
        - 0144 18
        01

        0270
        - 0141
```

TER	OBOGGOAM TO EODMAT SOFT-SECTORET MINIELOPPY DISKS	ASSUMES BED-68 HARDWARE WITH ROM AND PIA	M30 IRACKS, 18 SECTORS/IRACK	COPYRIGHT: 1979 HEMENWAY ASSOCIATES, BOSTON MASS.	JUVOJSVO 67/05 HOI IOVEST	DESCRIPTOR ADDRESS(2)	DESCRIPTOR COUNT	CURRENT CHAR (2)	TOKEN RETURN CODE	TOKEN CLASS	VALUE/TRANSFE	TOP OF FUB CHAIN (Z)	DISK FREE SPHCE FOINER (0) Stadt of toanstent ADEA(3)	STAND OF TRANSFER AREA (2)	NEXT AVAIL TRANSIENT AREA (2)	BACKSPACE CHAR	DELETE LINE CHAR	DEPTH; LINES/PAGE	DEPTH TEMP	WIDTH; CHARS/LINE	NULL COUNT		DOPTERS FFER, COST	FOECT COUNT PAISE: COMYES	ESCAPE CHAR	DEPTH LINES/PAGE	DEPTH TEMP	WIDIH CHARS/LINE	. PROMET FOR DRIVE				GET USER RESPONSE		CHECK TOKEN	NUMBER?			NUMBER 100 BIG?		E+1 NUMBER TOO BIG?		1 YES, ERROR	72 TSSIE SECOND PROMPT					GET RESPONSE	
NAM FORMATTER	MOOD OF MOODOGO	ASSUMES BFD-68	FOR CP/68 SYSTEM===35 TRHCKS,	COPYRIGHT: 1979	1000	DESCRIPTION #30	<u> </u>	Eau	EQU	E		ES	E B	BENERAL FOR #55		E	EQU	EQU	DPCNT EQU \$30				200	EU #41	EOC	EGU	NT ERU	LWD EQU \$46	* * * * * * * * * * * * * * * * * * *		SMI	FCB 49	GTCMD	SWI	100 400 -	CMP B #3	BNE NOTNUM		TST VALUE	DNE DHUNO	LDA A VALUE+1	CMP A #2	BHI BADNUM	CTGMSG# YU!		STA A DNUM	PRTMSG	I de la	GTCMD 43	!
N 0000 0000	* *	* *	* *	*	*	0000		0023	0025	0026	0027	0029	002B	0033		0039	003A	003B	0030	0030	003E	003F	0040	0000 0041	0043	0044	0045	9400 0000		0000 CE 0053 K 5	30 0000	0004 31			0006 30	0009 C1 03	26		2	87. 97. 0100	0012 96 28	8	0016 22 22	į	CE (VOS)	001D B7 0069 R		0020 3F	0021 31	
0001	0002	0004	0000	0007	8000		0010	0012 +	0013 +	0014 +	0015 +	0016 +			+ 6100			0023 +	0024 +	0025 +	0026 +		0028 +	0029 +			0033 +	0034 +	0035	0038	7500	0039 +			0042 +	0044	0045	0046	0047	0048	0049	0051	0052	0053	(AU34	0026	0027		+ 6500	8
												Ť.	₩ W			E					_			# T		-		_	5A M	_		_		₩ 88			77. X								E 68	2,7 3.D			98 R	
		INDEX 24BC INITDK 253E	~	INITE 016Z	INITES OF		INITRS 0130		CALC CATINI		LUP 0044	×	9				MULIO ZZE/	MULG ZZUJ	XOY.		_				O C		PSHALL 2151				PULIDR 240		SDEF			SECSIZ DOBO				m	TABX 219C			TRKSIZ 0012	TXAB 2183	WID OUSD			WATERR 0198	
	0113	ADDABX 2219 M ADDAX 2232 M	224B		0033	5	0039	~	1047 7000	CL755 0020	0037	2318		4R 0023	2420			DESCRI COZZ	4000		LN		2200	bx 0040		0030	EXITE OLDS R				FCBDBA 0007		FCBDRV 0009 FCBDIT 0006				FUBGDI OOGZ					FUBSCF 0029		0002	FCBTRK 000A	001D 2940	FMTFCB 2488 M	2558		GETDR ZBEC M

LDA A #1 STA A SECTOR START OF SECTOR LOOP	P FOR SECTORS 1-18	LDA A		ъ В #4	BSR PUTBY! 4-BYTE SYNC	Œ	A TRACK	STA A O, X TRACK NO.	CL.R O, X ZERO	INX LDA A SECTOR	Œ	INX CLR 0, X LENGTH≖128	1	LDA A #FF7 STA A O, X CRC	:	LDA A #\$FF	₹ .	C K	Œ.	LUA A #≸FB STA A O, X DATA-ADDRESS MARK	XNI	B #128	BSR PUTBY1 128 BYTES OF DATA (0000) LDA A #\$F7			STA A O, X PAD	į		LDA A SECTOR	A SECTOR	CMP A #19 DONE? BNE SECLOP LOOP THROUGH 18 SECTORS	UT TRACK WI		മെമ്	LDA B #200 BSR PUIBYT
009B 86 01 009D R7 008B R *	* LOOP	86 FF		# %	00A9 8D 51 00AB 86 FE	A 0	86 86	00B3 A7 00		0088 08 0089 B6 008B R			80	00C2 86 F7	8		8	OOCE C6 06	8		0006 08 0007 4F	රි	00DA 8D 20 00DC 86 F7	00DE A7 00				7N-1 * *	00E6 B6 008B R	37	00ED 81 13 00EF 26 AF	* * :	Ή. Έ	3 G	
0124 0125 0126	0127	0129	0130	0132	0134	0136	0138	0139	0141	0142	0144	0146	0147	0148	0150	0151	0153	0155	0156	0158	0159	0161	0162	0164	0166	0167	0169	0171	0172	0174	0175	0177	0179	0181	0183 0184
	WAS RESPONSE "YES"?	>	SO, BEGIN FORMATTING	CHECK FOR "ESCAPE"		NOT, ERROR	RE-INIT. DISK DRIVES		SO, RETURN TO SYSTEM	ERROR MESSAGE				BER/								o Year Id	ון יאונא י	TRACK IMAGE TRACK IMAGE TO DISK		ACK	DONE? LOOP UNTIL DONE		BACK TO BEGINNING		DUMMY RCB (FOR "PRTERR")	YFMTY ERROR DUMMY "FCBSTA NAMMODY "TEKRHE" (3400 BYTES)	ION I INNECT COLOR DITEST	POINT TO BUFFER	8-BYTE GAP
	3	H	Ħ	중		H	RE		IF S	ERROF			RETRY	VE NUM			MEEK?					CTABT	C E	BUILD TRACK WRITE TRACK		BUMP TRACK	DONE?		BACK TO		DUMMY F			POIN	8-BY
2 3F SWJ 3 30 FCE DE 20 LDX		BNE START	JMP FORM2	Œ	91 43 CMP A ES	0035 26 03 BNE BADNUM IF	AGLINI	003/ 3F SW1 0038 33 FCB 51	RTS IF	ADNUM LDX *BADMSG	PRIMSG 1.3F SWI	FCB 49	ZO BF BRA START RETRY	20 BADMSG FCC	0052 OD FCB \$0D	PRMPT1 FCB \$0A	44 FCC / DRIVE NUMBER? / O4 FCB \$04	*	44 PRMPT2 FCC /DRIVE /	FCC / READY?	U+ FORMAT DISK HERE	* 7F 0084 R FORMS CLR TRACK STABT	* * * * * * * * * * * * * * * * * * *	FORMZA JSR TRKBLD JSR TRKWRT	OOSA R LDA A TRACK	OOSA R STA A TRACK	81 23 CMP A #35 DONE? 26 EF BNE FORM2A LOOP UN	*	7E 0000 R JMP START BACK TO *	TRACK	02 SAVEX RMB 2	646 FCC /FMT/ /FMT/ 10001 ERRCOD FMB I DUMMY 1 TAGIK IN MEMY IN MEM A TRACK IN MEM FORM A TRACK IN MEM FOR	1000000 4 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	0154 R TRKBLD LDX #TRKBUF FF LDA A #*FF	C6 08 LDA B #8 8D 61 BSR PUTBYT 8-BY

	PSHALL 2151 M PSHX 21CE M FULLAL 216A M		PUTDR 2406 M	SDEF	2384	SAVEX OOSC R SECLOP OOAO R	JR 008B	SE11 0118 R SET2 0120 R	0128	SETD 0128 R	X 227F		SUBXAB 2265 M	1ABX 219C	TRACK 008A R	TRKBLD 0092	TRKWRT 0110	TXAB 2183	DEOO OM	WRITE 23D2 M XABX 21B5 M		z															
x 2219 2232	X 224B M AB 2200 M SG 0041 R	003A 2A2A	0033	N 243A M			2572	AR 0023	0020	0022	6 4224 F		003C		0040	0035	.0043 R	0029	2940	m	00/3	9000 0000	002B	IR 23EC M ID 24F0 M	24BC	E 2335 M	0044		0046	2301 m 24A2 m	\$ 22E7	22CD #		24D6 234F	ID 239E M	0063	1
АБВАВХ АВВАХ	ADDXAB ADDXAB RAHMSG	BADNUM	BMEN	CHAIN	CLOSE	CMEM		CUCHAR	DESCRA	DESCRC	101 V 18	DNG	UPCN1	DSKERR	χΞ	EMEM	ES	FCBCHN	FIBDEF	FMIFCB	FORMZ	FORMZA	FREIAB	GETDR	INDEX	IOHOR	da i	LOADB	3	MOVS MOVS	MUL16	S Z	MONTON	S TOK	CPEND	PRMPT2	4
DONE!	PUT BYTE INTO TRKBUF	DONE?		TO DISK DRIVE	", DRIVE NO. IN "VALUE+1"	ED HERE		1	ISSUE EKKUK MESSAGE		CLEAN STACK (USR TRKWRT)			+1 GET DRIVE NUMBER	NOT ZERO?	DRIVE NO. IN BFD-68 FORMAT		DRIVE 12	ON.	DRIVE 1 IN BFD-FORMAT		DRIVE 2? NO		DRIVE 2 IN BFD-FORMAT		NOT 0, 1, 2 IS ERROR	SET DRIVE IN PLACE	SET TRACK IN PLACE		SET SECTOR IN PLACE	SET BUFFER POINTER	SEEK TRACK	ERROR?	YES		START OF TRACK IMAGE	
RTS	STA A O, X	DEC B		\vdash	IN "IKKBUF",	ERRORS HANDLED HERE	STA B ERRCOD	LDX #SAVEX	SWI	FCB 30	INS	JMP START		LDA A VALUE+1	BNE SET1		BRA SETD	CMP A #1		LDA B #\$10		CMP A #2 BNF SFT3		LDA B #\$20		JMF BADNUM	20	a	Œ			JSR \$82AB	TST B	BNE DSKERR	RTS	EQU #+10	END
*	* PUTBYT		*	* WRITE		DISK	* DSKERR						* *	RKWKT	*	*	×¢	SET1	*			SET2	*		*	SET3									: :	TRKBUF	*
00FB 39	00FC A7 00	00FF 5A 00FF 5A 0100 24 FA	Ŷ	0102 39			0103 F7 0091	0106 CE	+	+ 010	010B 31	010D 7E 0000 R		0110 96 28	0112 26	0114 C6 08	0116 20	0118 81 01	011A 26	011C C6 10	OI IE ZO	0120 81 02	777	0124 C6 20	27 9710	0128 7E 003A R	012B F7 A07B	012E B6 008A R	0134 B6	0137 B7	013D FF A07E		0146 5D		0149 39	014A 0154 R	
0185	0187	0183	0192	0193 0195	0196	0198	0199	0201	0202	0204	0205	0207	0208	0210	0211	0213	0214	0216	0217	0219	0220	0222	0224	0225	0227	0228	0230	0231	0233	0234	0236	0237	0239	0240	0242	0243	0245 0246

Name	STX TW INIT DATA BUFFER ADDRESS TABX X: =FCBADR	SWI FCB 3 PCP GETTITE GET TOWN AND STATES	SON SELDIS SEL DATVE, INNER AND SECTOR PSHX SAVE FCBADR SWI	FCB 5 TO DECECY DEAR A CENTRAL		IMS FICE A		* * RE-SET ERROR CODES TO CP/68 VALUES		CMP A #0 O BECOMES 10 RNF #+4		LDA A #10			#+C	L.DA	BRA RDSECO	I.DA		** RDSEC1 CLR A NO EKRORS≕0		RDSECO STA A FCBSTA, X SET ERROR CODE RTS		* INIT BASE PAGE DRIVE, TRACK AND SECTOR	SETDIS LDA A ECRORU, X GET DRIVE #	INC A OFFSET +1	₫,	HSL H		ASL A	HSL A	(A TRK	LUM A FCBSCT, X GET SECTOR DFC A DFFSFT -1	A SCTR	2. T. S.		WRITE A SINGLE SECTOR		. WISEC 1ABX X:=FCBADR SWI
OND OND OND N		+ +	+	# +		+ +		0075 0075		0016 81	2	001A 86	77.0	001E 81	40 97 07W	0022 86 12	0024 20	0026 86	0028 20 01	002A 4F		00ZB A7 05 00ZD 39			002E A6 09	0030 4C	0031	0033	0034	0032	0037 97	0039 A6	0038 97	003F 4A	0040 97	0047.39				+ 0043 3F
O000 0000 N		n N	CIOR READ AND WRITE ROUTINES	ERCUM DISK DRIVE SYSTEM	1978 BY HEMENWAY ASSOCIATES INC	ISS. UZIII S RESEKVED			CONGCO A GOT FINDMOTHON	GENERIC DEVICE TYPE	STATUS	DATA TRANSFER TYPE DATA BUFFER ADDRESS		DRIVE NUMBER	SECTOR NUMBER	FWD LINK TRACK/SECTOR	BHCK LINK INHCK/SECTOR	TES	0	TRACK	DESIRED SECTOR	BHCKWHAD LINK FORWARD LINK	SECTOR BYTE COUNT	DATA ADDRESS VECTOR CONTINUATION ADDRESS	ALTERNATE TARGET ADDRESS					INITIALIZE DRIVES		VES		•		SECTOR		X:=FCBADR		×
0000 0000 0000 0003 0000 0003 0000 0005 0000 0005 0000 0005 0000 0006 0000 0006 0000 0006 0000 0006 0000 0007 0000 0007	*	ENT .	SINGLE				** *	FCB	- - - -	3 3 3 3 3	E3		į		8	8	2		FOR	EGU		EQU	Э С		EQU		VECTURS	EQU	E :	EgC	: *		A PINTINK IMP INITER	*	* *			. RDSEC TABX SWI	FCB 3	LDX FCBDB4
- 500000000百百百百百百百百万万万万万万万万万万万万万万万万万万万万万万	0000 0000	0000 0003	\$ \cdot 0	0 0	٥,	5.	m e	· 10 •	0000	0000	0000	0000		0000	0000	0000	O COLOR	ο	0000	0000	0000	0000	0000	2 0 0 0 0 0	0000	•	۰.	0000	0000	0000		m ·			~! **		ء ھر	+	+	

READ 23B8 H KEWIND 2384 M SCTR 0002 SUBABX 227F M SUBAX 2279 M SUBAX 2299 M SUBAX 2297 M SUBAX 2297 M SUBAX 2183 M WISECO 0000 R WISECO 0000 R WISECX COOF XABX 2185 M	
. KUSEC 0003 RN - WISEC 00043 RN - AUDARN 2219 M - AUDARN 2000 RN - BAKENIN 0003 BASEGU ACAD M - BAYLON 0007 CHAIN 213 B M - CLOSE 2369 M - CLOSE 2369 M - CHOC 2572 M - DELETE 2420 M - DIVIG 2572 M	4 ~ 16873
INIT DATA BUFFER ADDRESS X: =FCBADR GET DRIVE, TRACK AND SECTOR SET FOR FULL SECTOR X SAVE FCBADR GET FWD LINK WRITE A SECTOR X: =RCADR OK TO CP/68 VALUES O BECOMES 10 OTHERS BECOME 5 NO ERRORS=0	
FCB 3 LDX FCBDBA, X STX TA STX TA SWI FCB 3 BSR GETDTS CLR BYTCNT STA A ADDRES+1 STA A BAKLNK LDA A FCBBAK, X SWI STA A BAKLNK+1 SWI FCB SWI FCB SWI SWI STA FWDLNK SWI STA FWDLNK SWI SWI STA FWDLNK SWI) i
0122 + 00044 03 0123 + 0045 EE 07 0124	

LDA A O, X GET FIRST CHAR. OF RESPONSE CMP A # Y WAS IT YES?? BEQ INITR2 IF SO, CONTINUE * RTS IF NOT, QUIT * INITR2 LDX #FCBSPC POINT TO FCB CLR FCBTRK, X TRACK=0 I DA A # 11	STA A STA A INITIALIZE ALL ZERO EX	* TXAB SWI FCB 2			FCB 5 * CLEAR OUT BUFFER EXCEPT FOR LAST 2 BYTES * ON ** PROPERTY ** PROPERT		LDA A #1 TRACK, SECTC STA A O, X STA A 1, X PULX SWI FCB 6	BSR WRTBLK WRITE BLOCK 1 TST FCBSTA, X CHECK FOR DISK ERROR BEQ *+6 OK * BRA INITQ FATAL DISK ERROR, QUIT	##TBL BRA WRTBLK OUT OF RANGE "BSR WRTBLK" INC FCBSCT, X SECTOR=2 CLR BUFFER+SECSIZ-2 CLR BUFFER+SECSIZ-1 * INITIALIZE DIRECTORY TO ZERO INITRA BSR WRTBLK WRITE DIRECTORY BLOCK INITRA BSR WRTBLK WRITE DIRECTORY BENOCK BEQ *+4 OK
00 01 00 00 00 00 00 01	4	+ 016A 3F + 016B 02		0171 A7 07 0173 E7 08 + 0175 3F	0176	F 00 F	0183 86 01 0185 A7 00 0187 A7 01 + 0189 3F + 018A 06	018B 8D 7E 018D 6D 05 018F 27 04 0191 20 4F	0193 20 76 0195 6C 0B 0197 7F 0128 R 0196 7F 0129 R 0190 8D 6C 019F 6D 05 0161 27 02
0062 0063 0063 0064 0065 0065 0067	0070 0070 0071 0072 0073	0075 0076 0077 0077	0079			0093 0093 0093 0093 0093 0093	0009 0100 0101 0102 0103 0103	0105	01113 01113 01113 01114 01116 01117 01118 01118
NAM INITER INITIALIZE A DISK FOR CP-68 OPERATING SYSTEM FOR PERCOM FLOPPY DISKS TRACK O, SECTOR 1 HEADER OF FREE-SPACE LIST TRACK C, SECTORS 2-10 DIRECTORY SPACE TRACK C, SECTORS 2-10 DIRECTORY SPACE	ACK ACK		EKROR STATUS FLAG DATA BUFFER ADDRESS DRIVE NUMBER	IRACK NUMBER SECTOR NUMBER TRACK LINK POINTER SECTOR LINK POINTER	FILE-CONTROL BLOCK DISK OUTPUT	** ** ** ** COMMAND-LINE INTERPRETER BASE-PAGE LOCATIONS ** ** ** ** ** ** ** ** ** ** ** ** **	DISK IN DRIVE '	ENTRY POINT FROM CLI GET DRIVE NUMBER LIMIT RANGE (ICOM PERMITS 4 DRIVES)	MAKE DRIVE NUMBER ASCII PUT IN PROMPT OUTPUT PROMPT GET USER RESPONSE
* INITIALIZE A DISK FOR CI * FOR PERCOM FLOPPY DISKS * TRACK 0, SECTOR 1 * TRACK 0, SECTOR 2 * TRACK 0, SECTOR 2 * TRACK 0, SECTOR 3	шx	EGU		FUBIRK END 10 FUBSCT END 11 FUBTIR END 12 FUBSLK END 13		* BUFFEK RMB SECSIZ * COMMAND-LINE INTER * DESCRA EQU \$20	ESC #27 FCC / INIT. RMB 1 FCC / ? / FCB #04	* * * * * * * * * * * * *	H T CEDICY A ##30 A DRVNO #PROMPT MSG M 49 MD B 48 B 48 DESCRA
0000 0000	0100 0000	0022	0005	0000 0000 0000	0002 44 0001 FF	0100	002/ 00001 04	8	3.1 3.1 3.1 3.1 3.1 3.1 3.1 3.1 6.20
0000	0000	0000	0000	0000		002A		0143	0146 8 0146 8 0151 0 0151 0 + 0155 + 0155 + 0157 0158 1

RETURN TO CL.I	'INITIALIZATION FAILED' \$0D	CONVERT LSEC TO PSEC LSEC IN B-REG	SAVE X-REGISTER		ADD LOGICAL OFFSET	10 SECTOB STABLE AT 1	Ē	RESTORE X-REG			* WRITE A SECTOR WITH EKROR CHECKING		PSH A SAVE 'A' CLR FCBSTA, X CLEAR ERROR FLAG	ISSUE I/O REQUEST		TST FCBSTA, X ERROR? BNE WRIERR YES	,			OUTHL CONVERT LEFT DIGIT	A ERTYPE	OUTHR CONVERT RIGHT DIGIT	A ERTYPE+1 K SAVE X			A FCBSCT, X OUTHL MAKE SECTOR NO. HEX	A SECT	A FCBSCT, X	SECT+1	×	OUTH, MAKE TRACK NO. HEX	A TRACK	OUTHR	A TRACK+1	SERVOR PRINT ERROR MESSAGE	c	47 CALL CP/68
RTS	FCC 'IN FCB \$0D	RT LSE	PSHX	FCB 5 LDX #TBL	ADDBX SWI	FCB	LDA B	PULX ELX	FCB 6	RTS	A SEC	:	PSH A	TOHO!	FCB 19	TST FI		RTS H	TAB		STA A TBA		STA A	IMS		88. A		LDA A	•			S 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	ιō		PRTMSG	SHI	SWI
	MSG		E1.SC								WRITE	:	WRTBLK						RTERR																		
01E7 39	O1E8 49 0	* * *	01FE 3F	01FF 05 0200 CE 027D R		0204 0A	0206 E6 00	20 0000	0209	020A 39	* *		020B 36 W			0210 6D 05 0212 26 02		0215 39	16	8D 54	0219 B7 0253 R 021C 17		021F B7 0254 R	0222	0223 0	0224 A6 0B 0226 8D 45	B7	022B A6 0B	0 Z	A6 0A	80 37		£ 8	B7	0248	0243 3F	•
0183	0185	0188 0188	0190 0191 0192 +	0193 +	0195	0197 +	0199	0200		0203	0205	0206	0207 0208		0210 +	0212	0214	0216	0217	0219	0220	0222	0223	0225 +	0226 +	0227 0228	0229	0230	0231	0233	0234	0235	0237	0238	0239		0242 +
FATAL DISK ERROR, QUIT	A FCBSCT, X A NEXT SECTOR	SIZ-1 DUNE WITH TRACK?	CCT,X I NO, CONTINUE WRITING	SCT, X SECTOR=1	×		AEST OF DISK (TREE-STACE)	KESS	UMBER	TOOL OF CHILD	B #TRKSI7+1 END OF TRACK?	ON	YES, SECTOR=1	EXT TRA	A #DSKSIZ+1 END OF DISK; INITR7 NO	AST SECTOR POINTS TO 0,0			SAVE LSEC	ER+1 SECTOR LINK		DONE	ON .	DONE? (=0)	ON .	YES, DONE!!!		rk, x	SAVE LSEC	CT, X	GET LSEC	, KEEP WRITING		SAGE	OUTPUT ERROR MESSAGE		
INITO			A FCBSCT, X INITR4	4 4	Œ			CB ADD	B=SECTOR NUMBER				B #1				. m	A BUFFER	A B GFTSC		. B MRTRIK		INITR	6	INITER				ב המדינה			INITRO		* FATAL ERROR MESSAGE	LDX #GMSG	PRTMSG	SW1 FCB 49
BRA	INC	2 88 E E E E	STA	INITES LDA STA	STA		INITIALIZE	× <	- U)	JIN1 70.	S G	BNE	LDA	INC	N S	2	CLR.	R7 STA	PSH R SH	STA	PUL.	TST	BNE.	TST	BNE	RTS			E a	STA	5	BRA		TAL ER			FCB
				11			Z * *	* 1	* *	*	CHILNI		*			*		* INITR7		œ			*	•		*	*	INITRB				,	¢ *	* FA	* INITO		
	*	*	*	i		* :			•									œ																	α	:	
30	80	e t	E G	01 080	7 OA	•						60 9	5 01		1 23 6 02 8 02	,.	.,	002A	33	002B			5 04	_	5 01	^		40 Z	,			22			OTER R	: ! !	# 12
	A6 0B	81 08 27 04			A7					Ċ	01B8 C1 0B	26	01BC C6 01	40	01BF 81 23 01C1 26 02		01C4 5F	B7 002A	01C8 37	F7 002B	01CE 33	9 4	0102 26 04	0104 50	0105 26 01	01107 39		Ğ!	01UA 3/	E G	ဗ္ဗ	01E0 20 D5			OTE2 CE OTE8 R		01E5 3F 01E6 31

```
* CONVERT BINARY TO HEX-ASCII HERE
"WARMSTART"
QUIT
                                       SHIFT RIGHT
                                                      GET NIBBLE
                                                                                    * LOGICAL/PHYSICAL SECTOR TABLE
      *
DERROR FCC 'DISK ERROR: '
ERTYPE RMB 2
               ' AT SECTOR '
               FCC ' AT SECTOR
RMB 2
FCC ', TRACK '
RMB 2
FCB $0D
                                                      A ##0F
A ##30
A ##39
                                                                     A #$7
FCB 31
RTS
                                                                                          ٤
                                                                                                σσσσ
                                       LSR 128
                                                      ADD CMP
                                                                     ADD
                                                                                          FCB
                                                                                                OUTH,
                        TRACK
                                                      OUTHR
                  SECT
                                                                                             # 님
                                                      5883
5883
                                                                     07
         0002
20
20
20
20
20
0002
0002
                                                      98
88
23
                                                                     88
                                       4 4 4 4
                                                                          3
                                                                                          027C 00
                                                                                                026D
026E
026F
0270
0246
                                                                     0279
        0248
0253
0255
0260
0262
0264
0264
                                                                           027B
                                                      0271
0273
0275
0275
                                                                                                027D
027E
027F
0280
0281
0282
0283
0283
0284
0285
```

INITR 0143 RN

AUDDAN 2229 M

AUDDAN 2229 M

AUDDAN 2229 M

AUDDAN 2229 M

BUFFER 0224 M

CLOSE 2348 M

CLOSE 2348 M

CLOSE 2348 M

CLOSE 2348 M

DELETE 2420 M

DELETE 2420 M

DECRA 0020

DIV16 2524 M

DRWNO 013E R

DRWNO 013E R

DRWNO 0009

FCBDR 0000 R

FCBDR 0000 R

FCBDR 0000 R

FCBSCT 000B

FWITTRS 01BS R

INITRS 01BS R

I

CEEEEE CC

21CE 216A 22406 22406 01E8 2258C 2258C 227F 2227F 2227

Σ

#BUFFER+4 HIDEX A FIS+1 B FTS A PTS+1 INIT. PRESENT T/S B PTS B PTS #BUFFER #BUFFER SYSTEM FILE INTO MEMORY	GETBYT GET A DATA BYTE FROM FILE A #\$16 TRANSFER-ADDRESS? BOOT2 NO GETBYT A ADDRES GET TRANSFER ADDRESS GETBYT A ADDRES+1	H HDALLS GET NEW DATA FRAME A ##02 DATA FRAME? BOOTA NO GETBYT A SAVEX GET ADDRESS	GETBYT A FCNT GET FRAME COUNTER GETBYT GET DATA BYTE SAVEX A 0, X STORE BYTE	SAVEX FCN1 FCN1 BOOT3 BOOT1 GET NEW DATA FRAME ADDRES GOT TRANSFER ADDRESS O, X GO THERE	NTA BYTE FROM SYSTEM FILE THE IN 'A' REGISTER INDEX #BUFFEK+256 NEED NEW SECTOR? GETSEC YES	A 0, X GET BYTE INDEX MOVE POINTER B PTS A PTS+1 CHECK FOR LAST SECTOR A LTS+1 OFTS2 NOT LAST
LDX STA LDA STA STA STA LDX LDX LOAD	* BSR 116 BSR CMP CMP CMP CMP SA4 BSR CM11A C STA	20 EE * BRA 81 02 B00T2 CMP 26 21 * BSR 8D 24 BSR 8D 71 BC STA 8D 11 BC STA 8D 11 BC BSR	BY 0119 C STA 8D 1A BSR BY 011C C STA 8D 15 800T3 BSR FE 0118 C LDX AY 00 STA 08 INX	STX DEC BNE BRA 0014 LDX	RETURN BY ETBYT LDX CPX BEG	0083 A6 00
R E HEADER	0071 0072 0073 0074 0075 0075 0077	0077 0081 0082 0083 0083 0084 ROM-ABLE 0085	0089 0089 0090 0091 0092 0093 0093	0096 0097 0098 0100 0101 0102 0103	0106 0107 0108 0109 0110	0112 0115 0116 0116 0117 0117 0119 0119
NAM BOOT PERCOM CP/68 BOOTSTRAP PROGRAM ASSUMES SYSTEM FILE LINKED AS FOLLOWS: TRACK 0, SECTOR 1, BYTE 250-FIRST SECTOR 251-FIRST SECTOR 253-LAST SECTOR 253-LAST SECTOR 253-LAST SECTOR	KOM DRIVE O: IVE INTERFACE ADDRESSING	ES IN COMMON, CODE IS	BUFFER, SECS17 FTS, 2 LTS, 2 INDEX, 2 SAUSY, 2 SAUDRES, 2 FCNT, 1	OR : : : INIT. STACK POINTER INIT DELUES	INFOR	DSEC READ LINK SECTOR BUFFER SECSIZ-6, X GET FIRST T/S SECSIZ-5, X FTS FTS SECSIZ-4, X GET LAST T/S SECSIZ-2, X LTS LTS+1
* * * * * * * * * * * * * * * * * * *	* * BOOTS SYSTEM FROM DRIVE O: * DEFINE DISK-DRIVE INTERFACE * INTRK EQU \$CO27 * RDSECX EQU \$CO0C * DRV FOU! \$COOC	TRY EQU SCTK EQU TW EQU SECSIZ EQU * NOTE: ALL		* ENROR JUMP VECTOR * EKROR EQU #E113 * BEGIN BOOT HERE * BEGIN BOOT HERE * DOOF C START LDS #STACK+15	30N ***	0010 C
0001 0002 0003 0004 0006 0006 0007 0008 0009	0011 0012 0013 0015 0015 0017 0017 0000 C00C 0018	00000		0036 0037 0038 0039 0040 0041 0042 0043 0043 0043	9000 8000 9000 9000	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0

ADDRES 011A C BOOT 0000 RN BOOT1 003F R BOOT2 0051 R	0000 0000 0000 0110	GETBYT 0.11.0 C GETBYT 0.07.8 R GETSZ 0.09.C R GETSEC 0.08.A R INDEX 0.11.6 C INTRK C.02.7 LTS 0.11.4 C PTS 0.11.4 C	KDSEC 0085 R KDSECX COOC SAVEX 0118 C SCTR 0002 SECSIZ 0100 STACK 0000 C STACK 0000 R TWK 0001	
NOT LAST EOF-GO TO TRANSFER ADDRESS	GET FORWARD T/S LINK UPDATE PRESENT T/S BEAD NEW SECTOR	KEAD NEW SECTOR GET DATA BYTE RE-INIT. INDEX	OFFSET OF SECTOR=-1	SAVE BUFFER ADDRESS INIT. DRIVE EKROR? YES
. B LTS : GETS2 N BOOT4	#BUFFER 0, X A 1, X B PTS B PT	BSK KUSEC LDX #BUFFER+4 LDA A O.X INX INX STX INDEX RTS SINGLE-SECTOR READ ROUTINE	E=0 OR='A' OR='A' EH='X' B SCTR B TRK	TW A ##40 A DRV RDSECX *+5 ERROR
CMP BNE BRA * *	# GETS2	BSX LDX LDA INX STX STX RTS * SINGLE - SE	** DRIV ** TRAC ** SECT ** BUFF ** STA	STX LDA STA LUSR A STA BCC - A BCC - A
0095 F1 0112 C 0098 26 02 009A 20 DA	009C CE 0010 C 009F E6 00 00A1 A6 01 00A3 F7 0114 C 00A6 8P 0115 C	97 48 48	44 77 70	00BA DF 16 00BC 86 40 00BC 97 00 00C3 24 03 00C5 7E £113
0123 0124 0125 0126	0128 0129 0130 0131 0132	0134 0134 0135 0137 0138 0139	0142 0143 0144 0145 0147 0147 0147 0149	0151 0153 0153 0154 0155 0155 0158

A	E
ADABX 33 ADDAX 33 ADDBX 33 ADXAB 32 ASSIGN 3	ENLARGE 95, 96 EQTAB 54 EXIT 6 EXPAND 87
	_
В	F
BINFRM 77 BOOT 4 BUILD 95, 96	FCBPOS 84 FCBRCD 84 FCBRNM 83 FCBRTB 84 FCBRSZ 84 FILCPY 77
С	FILENAMES 2 FMTFCB 38, 67
CHAIN 43, 66 CHRTAB 58, 62 CLI 58	FMTS 39, 54
CLOSE 41, 69 CMDTAB 58	G
CMPC 34 CMWC 39 CONRCB 59 CREATE 85 CVDB 61, 62 CVHB 62	GCHRTB 62 GETBYT 72 GETDR 40 GETSC 74 GTCMD 38, 59
_	Н
D	HEXFRM 77
DELETE 4, 42, 67, 95 DESCRA 62 DESCRC 62 DEVNAM 60	HSCAN 62
DEVIAN 60 DEVIAB 75 DIRCMD 65	I
DIRECTORY 5, 40 DISPATCH 54 DIV16 35 DLKUP 75 DRIVER 80 DSCAN 61 DSKSIZ 100	INCON 56 INDEX 35 INICMD 65 INITDK 42, 59, 70 INITIALIZE 6, 73 INLIN 55 INRDR 56
DTDCPY 77	IOHDR 37, 55

Index

J	P	T
JMPCMD 63 JUMP 7	PDSRCH 55, 71 PDTAB 54 PIP 8, 75 PLACE 95, 96 POSITION 87	TABX 32 TRKBLD 80 TRKSIZ 100 TRKWLT 81 TXAB 32
L LINK 7, 74 LOAD 7 LOADB 43, 64 LOCATE 95, 96 LODCMD 59, 64	PRTERR 38, 60 PRTMSG 38, 59 PSHAL 32 PULAL 32 PULX 32 PUTDR 40	U USR1-USR11 43
М	R	V
MOVC 34 MOVS 35 MUL8 33	RCLOSE 86 RDRIN 57 READ 42, 69 RENAME 10	VALUE 61
MUL16 34	RENCMD 64 REWD 41, 70 RDSEC 70, 73 ROPEN 86	WARM3 59
N NSCAN 61 NULL 55	RWRITE 87	WARMST 42, 59 WILDCARD 3 WRITE 42, 70 WRTBLK 74 WTSEC 71
NXTOK 35, 60	SAVE 10 SAVCMD 64	·
0	SBABX 33 SBXAB 33 SECSIZ 99	X XABX 32
OPEN 41, 68 OPEND 40 OTLIN 56 OTPCH 57 OUTCON 56 OUTLPT 57 OUTPCH 57	SECURITY 11, 78 SEMPTY 11 SET 11, 78 SFILE 66 STATUS 13, 78 SUBAX 33 SUBBX 33 SUBCMD 63 SUBFCB 60 SUBFLG 60 SUBMIT 13	